# Public Document Pack Cyngor Bwrdeistref Sirol Pen-y-bont ar Ogwr **Bridgend County Borough Council**



Swyddfeydd Dinesig, Stryd yr Angel, Pen-y-bont, CF31 4WB / Civic Offices, Angel Street, Bridgend, CF31 4WB

Rydym yn croesawu gohebiaeth yn Gymraeg. Rhowch wybod i ni os mai Cymraeg yw eich dewis iaith.

We welcome correspondence in Welsh. Please let us know if your language choice is Welsh.



Annwyl Cynghorydd,

# Cyfarwyddiaeth y Prif Weithredwr / Chief **Executive's Directorate**

Deialu uniongyrchol / Direct line /: 01656 643148 /

643147 / 643694

Gofynnwch am / Ask for: Gwasanaethau

Democrataidd

Ein cyf / Our ref: Eich cyf / Your ref:

Dyddiad/Date: Dydd Gwener, 1 Gorffennaf 2022

# PWYLLGOR GWASANAETHAU DEMOCRATAIDD

Cynhelir Cyfarfod Pwyllgor Gwasanaethau Democrataidd O Bell trwy Timau Microsoft ar **Dydd Iau**, 7 Gorffennaf 2022 am 10:00.

#### **AGENDA**

2.

- 1. Ymddiheuriadau am absenoldeb
  - Derbyn ymddiheuriadau am absenoldeb gan Aelodau.
- Datganiadau o fuddiant

Derbyn datganiadau o ddiddordeb personol a rhagfarnol (os o gwbl) gan Aelodau / Swyddogion yn unol â darpariaethau'r Cod Ymddygiad Aelodau a fabwysiadwyd gan y Cyngor o 1 Medi 2008.

- 3. Cymeradwyaeth Cofnodion dderbyn am gymeradwyaeth y Cofnodion cyfarfod y 21/10/2021
- 4. Arweiniad Enghreifftiol i Gyfansoddiad a Chyfansoddiad Cymru Gyfan 9 - 294
- 5. Materion Brys

I ystyried unrhyw eitemau o fusnes y, oherwydd amgylchiadau arbennig y cadeirydd o'r farn y dylid eu hystyried yn y cyfarfod fel mater o frys yn unol â Rhan 4 (pharagraff 4) o'r Rheolau Trefn y Cyngor yn y Cyfansoddiad.

Nodyn: Sylwch: Oherwydd rhesymau iechyd a diogelwch ni fydd y cyfarfod hwn yn cael ei gynnal yn ei leoliad arferol. Bydd hwn yn gyfarfod rhithwir a bydd Aelodau a Swyddogion yn mynychu o bell. Bydd y cyfarfod cael ei recordio i'w drosglwyddo drwy wefan y Cyngor. Os oes gennych unrhyw gwestiwn am hyn, cysylltwch â cabinet\_committee@bridgend.gov.uk neu ffoniwch 01656 643147 / 643148.

3 - 8

# Yn ddiffuant

# K Watson

Prif Swyddog, Gwasanaethau Cyfreithiol a Rheoleiddio, AD a Pholisi Corfforaethol

# Dosbarthiad:

CynghowrwyrCynghorwyrCynghorwyrS AspeyM L HughesT ThomasP FordRM JamesG WalterRM GranvilleR J SmithE D Winstanley

D T Harrison I M Spiller

# Agenda Item 3

#### PWYLLGOR GWASANAETHAU DEMOCRATAIDD - DYDD IAU, 21 HYDREF 2021

COFNODION CYFARFOD Y PWYLLGOR GWASANAETHAU DEMOCRATAIDD A
GYNHALIWYD YN O BELL - TRWY DIMAU MICROSOFT DYDD IAU, 21 HYDREF 2021, AM
10:00

# **Presennol**

Y Cynghorydd E Venables - Cadeirydd

S Aspey RE Young SG Smith

G Thomas

A Williams

Ymddiheuriadau am Absenoldeb

DG Howells, RM James, B Sedgebeer, SR Vidal a/ac MC Voisey

Swyddogion:

Mark Galvin Rheolwr Gwasanaethau Democrataidd Dros dro
Michael Pitman Swyddog Gwasanaethau Democrataidd – Pwyllgorau

203. <u>DATGANIADAU O FUDDIANT</u>

Dim

204. CYMERADWYO COFNODION

PENDERFYNWYD: Bod cofnodion cyfarfod 24/06/21 yn cael eu

cymeradwyo fel cofnod gwir a chywir

# 205. RHAGLEN DATBLYGU AELODAU

Cyflwynodd y Pennaeth Gwasanaethau Democrataidd dros dro adroddiad, er mwyn rhoi diweddariad i'r Pwyllgor ar gyflawni Rhaglen Hyfforddi a Datblygu Aelodau'r Cyngor a gweithgareddau cysylltiedig. Gofynnwyd i'r Pwyllgor hefyd nodi pynciau i'w cynnwys ar y Rhaglen Datblygu Aelodau.

Dywedodd y Rheolwr Gwasanaethau Democrataidd Dros Dro fod sylw cynyddol wedi'i roi i Ddatblygu Aelodau Etholedig. Roedd Mesur Llywodraeth Leol (Cymru) 2011 yn gosod y dylai awdurdodau lleol roi mwy o bwyslais ar Ddatblygu Aelodau. Ychwanegodd fod aelodau yn cael eu hannog i nodi eu hanghenion datblygu eu hunain a chymryd rhan lawn mewn gweithgareddau dysgu a datblygu.

Tynnodd y Rheolwr Gwasanaethau Democrataidd Dros Dro sylw at y Sesiynau Hyfforddi a Datblygu Aelodau a gafodd eu cynnal ers cyfarfod y Pwyllgor Gwasanaethau Democrataidd ar 25 Mawrth 2021 a gafodd eu rhestru yn 4.1.1 o'r adroddiad. Rhoddodd fanylion hefyd am sesiynau briffio'r Aelodau yn ogystal â sesiynau hyfforddi'r Pwyllgor Rheoli Datblygu a gafodd eu cynnal ers y dyddiad hwn. Amlinellwyd y rhain yn 4.2 a 4.3 o'r adroddiad, yn y drefn honno.

Darparodd y Rheolwr Gwasanaethau Democrataidd Dros Dro restr o'r pynciau arfaethedig i'w hystyried ar gyfer Aelodau drwy gyfrwng sesiynau Briffio Aelodau a sesiynau hyfforddi a datblygu Aelodau a gafodd eu hamlygu yn adrannau 4 i 6 o'r adroddiad.

Dywedodd y Rheolwr Gwasanaethau Democrataidd Dros Dro fod y Cyngor wedi dyfeisio ystod o gyrsiau e-ddysgu i Aelodau i gefnogi eu hanghenion dysgu a datblygu. Mae

darparu cyrsiau e-ddysgu yn rhoi cyfle i Aelodau ymgymryd â'u dysgu a'u datblygiad o bell ar adeg gyfleus ar eu cyflymder eu hunain. Roedd y cyrsiau E Ddysgu canlynol ar gael.

- Sefydlu Corfforaethol
- o GDPR y DU
- o Offer Sgrin Arddangos
- Ymwybyddiaeth o Ddiogelwch Tân
- Cod Ymddygiad TG
- o Diogelu Plant ac Oedolion
- o Trais yn erbyn Menywod, Cam-drin Domestig a Thrais Rhywiol

Gofynnodd Aelod a ellid darparu sesiwn hyfforddi ar Strategaeth Carbon Niwtral 2030 gan ei fod yn credu na fyddai pob Aelod mor wybodus ag eraill ar y pwnc pwysig hwn. Eglurodd y Rheolwr Gwasanaethau Democrataidd Dros Dro fod hwn yn bwnc a oedd wedi'i gadarnhau ac y byddai'n cael ei drefnu yn y dyfodol agos. Esboniodd na fyddai hyn yn cael ei drefnu gan y Gwasanaethau Democrataidd ond yn hytrach drwy'r Gyfarwyddiaeth Gymunedau a darparwr allanol.

Gofynnodd Aelod beth y gellid ei wneud i hyrwyddo Aelodau i fynychu'r sesiynau hyfforddi. Dywedodd fod yr un Aelodau yn mynychu llawer o'r sesiynau.

Pwysleisiodd y Rheolwr Gwasanaethau Democrataidd Dros Dro mai drwy gyfarfodydd grwpiau gwleidyddol oedd y ffordd fwyaf buddiol o hyrwyddo'r sesiynau hyfforddi gan y byddai hyn yn dod yn uniongyrchol o arweinwyr grwpiau i'w Haelodau, Ychwanegodd y byddai hefyd yn cysylltu â'r Aelodau annibynnol i bwysleisio pwysigrwydd y sesiynau. Ychwanegodd hefyd nad oedd y sesiynau'n orfodol ac felly eich bod ond yn gallu annog Aelodau i gymryd rhan yn yr hyfforddiant a'u hatgoffa o fanteision y sesiynau hyn.

Ychwanegodd y Rheolwr Gwasanaethau Democrataidd Dros Dro y gallai gynnal y sesiynau cyn cyfarfod o'r Pwyllgor roi hwb i bresenoldeb. Fodd bynnag, roedd llawer o Aelodau wedi beirniadu'r dull hwn gan fod amser ar y sgrin wedi cynyddu'n sylweddol ar gyfer y diwrnod penodol hwnnw ac roedden nhw'n teimlo bod hyn yn anodd bryd hynny oherwydd materion fel gormod o straen ar y llygaid a'r gallu parhaus i ganolbwyntio ar sesiynau'r cyfarfod.

PENDERFYNWYD: Bod yr Aelodau wedi nodi cynnwys yr adroddiad.

#### 206. DIGWYDDIAD POSIBL I YMGEISWYR

Cyflwynodd y Rheolwr Gwasanaethau Democrataidd Dros Dro adroddiad a oedd yn gofyn am farn y pwyllgor ynghylch cynlluniau dros dro i ddarparu digwyddiad 'Darpar Ymgeisydd', ar gyfer unrhyw un sy'n ystyried sefyll fel Cynghorydd Cyngor Bwrdeistref Sirol Pen-y-bont ar Ogwr (BCBC) yn yr Etholiadau Llywodraeth Leol yn 2022.

Esboniodd fod Wythnos Democratiaeth Leol yn cael ei chynnal bob blwyddyn ym mis Hydref gyda'r diben o:

- gryfhau'r cysylltiadau rhwng cynrychiolwyr etholedig a'u cymunedau
- cynyddu cyfranogiad yn y broses ddemocrataidd
- cynnwys dinasyddion mewn materion cymunedol
- cynyddu eu gwybodaeth am sefydliadau a phrosesau democrataidd lleol

Ychwanegodd, gyda'r Etholiadau Llywodraeth Leol ym mis Mai 2022, y byddai Wythnos Democratiaeth Leol eleni (11-17 Hydref 2021) yn amser priodol i ddechrau paratoi ar gyfer yr etholiadau ac i gynnwys yr etholwyr yn weithredol yn y broses ddemocrataidd.

Yn dilyn hyn, cynigiwyd y dylid cynnal dwy sesiwn ymgeiswyr ar 18 a 25 Tachwedd, a'u diben oedd amlinellu gwybodaeth am rôl y cynghorydd a swyddogaethau'r Cyngor a darparu gwybodaeth am sut y daw unigolyn yn ymgeisydd yn yr etholiad a'r prosesau ar gyfer cael ei ethol. Byddai'r pynciau'n cael eu rhannu'n ddwy ran fel y dangosir yn 4.1 o'r adroddiad. Roedd rhagor o wybodaeth am y sesiynau yn 4.3 o'r adroddiad.

Ychwanegodd y Rheolwr Gwasanaethau Democrataidd Dros Dro y byddai fideo hefyd yn cael ei ddarparu i ddarpar ymgeiswyr a fyddai'n dilyn diwrnod ym mywyd cynghorydd. Byddai'r sesiynau'n cael eu hyrwyddo ar wefan y Cyngor yn ogystal ag mewn bwletinau newyddion a chan Aelodau a thrwy Gynghorau Tref a Chymuned.

Esboniodd fod 3 o bobl wedi bod â diddordeb mewn mynychu'r sesiwn gyntaf a 6 o bobl yn yr ail sesiwn hyd yn hyn.

Dywedodd Aelod fod llawer o awdurdodau lleol eraill wedi darparu fideos tebyg a'i fod yn amheus y dylai awdurdodau lleol fod yn darparu'r fideos hyn. Roedd yn credu bod hyn yn gost ddiangen ac y gallai'r fideos gael eu cefnogi'n ariannol gan CLILC yn lle hynny.

Roedd y Rheolwr Gwasanaethau Democrataidd Dros Dro yn credu mai tua £800 oedd y gost am ffotograffydd ar gyfer fideo yn cynnwys tri Aelod. Dywedodd mai diben y fideos oedd gwneud yr hyn a gafodd ei wneud mewn blynyddoedd blaenorol pan oedd staff ac Aelodau'r Cyngor yn rhoi'r briff i ddarpar ymgeiswyr yn Ystafell y Cyngor yn y Swyddfeydd Dinesig. Gyda Covid-19 yn effeithio ar y gallu i gyfarfod fel hyn, roedd yn rhaid archwilio llwybrau eraill. Ychwanegodd y gallai roi adborth ar bryderon yr Aelodau ar y gost i Reolwr y Grŵp - Gwasanaethau Cyfreithiol a Democrataidd.

Esboniodd Aelod y byddai wedi cymryd rhan yn y gwaith o hyrwyddo'r diwrnod ym mywyd cynghorydd pe bai'n sesiwn holi ac ateb, ond nad oedd yn cefnogi'r gost o £100 y pen ac yn credu nad oedd hyn yn anghenraid.

Gofynnodd y Cadeirydd am ba hyd y byddai'r sesiynau'n para.

Dywedodd y Rheolwr Gwasanaethau Democrataidd Dros Dro ei fod wedi'u trefnu am 2 awr, ond roedd yn rhagweld y byddai'r sesiynau'n para dim mwy na 90 munud.

Gofynnodd y Cadeirydd a oedd y sesiynau'n cael eu cofnodi a'u rhoi ar wefan y Cyngor i'r cyhoedd eu gweld. Roedd y Rheolwr Gwasanaethau Democrataidd Dros Dro yn meddwl nad oedden nhw'n cael eu cofnodi ond nad oedd yn gweld problem o ran gwneud hynny. Roedd y Cadeirydd yn meddwl y gallai ddenu mwy o bobl a oedd wedi meddwl am y syniad, ond nad oedd yn gallu mynychu'r sesiynau posibl i ymgeiswyr, drwy eu rhoi ar y wefan ac y byddai'n eu hannog i gysylltu â Chyngor Bwrdeistref Sirol Pen-y-bont ar Ogwr i gael rhagor o wybodaeth.

Gofynnodd y Cadeirydd faint o bobl a oedd yn y sesiynau a gafodd eu cynnal yn 2017, cyn yr etholiadau y flwyddyn honno. Dywedodd y Rheolwr Gwasanaethau Democrataidd Dros Dro fod tua 20 o bobl wedi mynychu'r sesiynau hynny.

PENDERFYNWYD:

Bod y Pwyllgor Gwasanaethau Democrataidd wedi ystyried y cynigion ac wedi rhoi mewnbwn i'r digwyddiad(au) ar gyfer darpar ymgeiswyr sy'n cael eu cynllunio i ddigwydd ym mis Tachwedd 2021.

#### 207. RHAGLEN SEFYDLU AELODAU 2022

Cyflwynodd y Rheolwr Gwasanaethau Democrataidd Dros Dro adroddiad a oedd yn amlinellu'r Rhaglen Sefydlu Aelodau arfaethedig ar gyfer 2022 i'w hystyried gan y Pwyllgor.

Esboniodd fod angen hyfforddiant a datblygiad Aelodau o dan Ddeddf Llywodraeth Leol 2000, Mesur Llywodraeth Leol (Cymru) 2011 a Strategaeth Dysgu a Datblygu Aelodau Etholedig y Cyngor a gafodd ei gymeradwyo gan y Cyngor ym mis Gorffennaf 2021. Ychwanegodd y bydd nifer o Aelodau Etholedig newydd a rhai sy'n dychwelyd yn dilyn yr Etholiadau Llywodraeth Leol yn 2022. Roedd Rhaglen Sefydlu yn gyfle datblygu pwysig gan ei fod yn galluogi Aelodau i ymgyfarwyddo'n gyflym â'r ffordd y mae'r Cyngor yn gweithio, gan gynnwys rheolau a gweithdrefnau, cymhlethdodau rôl yr Aelod Etholedig, yn ogystal â'u helpu i integreiddio'n gyflym i'r Cyngor ar ôl iddynt gael eu hethol.

Hysbysodd y Rheolwr Gwasanaethau Democrataidd Dros Dro yr Aelodau i'r 3 cham yn adran 4 o'r adroddiad a gafodd eu trafod yng nghyfarfod diwethaf y Pwyllgor.

Yn Atodiad 1 yr adroddiad roedd y Rhaglen Sefydlu Aelodau arfaethedig lawn ar gyfer 2022. Dylid nodi bod y rhaglen ddrafft, yn ôl ei natur a'i hamserlen, yn cynnwys rhai dyddiadau/amseroedd a chynnwys drafft a allai newid i raddau ac felly y gallai fod yn destun rhai addasiadau.

Ychwanegodd, yn dilyn y broses ethol, ac i gefnogi'r rhaglen sefydlu aelodau, y byddai pecyn gwybodaeth hefyd yn cael ei ddarparu i'r aelodau pan fyddan nhw'n llofnodi Derbyniad y Swydd. Bydd hyn yn cynnwys gwybodaeth ddefnyddiol fel polisïau a gweithdrefnau craidd, amserlen cyfarfodydd a rhifau cyswllt defnyddiol y Cyngor a'i bartneriaid. Cynigiwyd bod y Rhaglen hefyd yn cynnwys digwyddiadau lle bydd Aelodau'n gallu cwrdd â swyddogion i gael gwybod mwy am wasanaethau'r Cyngor yn ogystal ag Aelodau newydd eraill, ac Aelodau sy'n dychwelyd.

Gofynnodd Aelod a fyddai'n bosibl rhoi mwy o bwyslais ar y broses o bennu'r gyllideb, cyllid y Cyngor a pham mae gan y Cyngor gronfeydd wrth gefn. Roedd yn meddwl nad oedd digon o wybodaeth ymhlith yr Aelodau presennol am yr uchod ac y byddai Aelodau newydd sy'n ymuno hefyd yn elwa o gael gwell dealltwriaeth o hyn. Ychwanegodd y byddai sesiynau briffio Aelodau hefyd yn ffordd dda o fynd ar drywydd hyn.

Mae'r Cadeirydd yn sôn bod y pwnc cynaliadwyedd ariannol heb ei drefnu tan fis Mehefin 2022. Gofynnodd a oedd hyn yn amseru addas neu a ddylid ei gyflwyno'n gynharach na hyn.

Esboniodd y Rheolwr Gwasanaethau Democrataidd Dros Dro fod rhan gyntaf y rhaglen sefydlu yn hanfodol ac wedi'i rhwymo braidd gan gyfyngiadau amser. Esboniodd, unwaith y pennwyd aelodaeth y Pwyllgor yng nghyfarfod blynyddol y Cyngor, ei bod yn ofynnol sicrhau bod Aelodau'n cael eu hyfforddi'n gywir cyn eistedd ar Bwyllgorau, yn enwedig y pwyllgorau trwyddedu, rheoli datblygu, llywodraethu ac archwilio a'r Panel Apeliadau. Ychwanegodd fod y rhaglen sefydlu ar ffurf ddrafft ac yn agored i newid, felly efallai y bydd newidiadau'n cael eu gwneud i sicrhau bod pynciau'n cael eu trefnu mewn trefn fwy priodol, gan gynnwys cael eu hategu gan sesiynau hyfforddi a datblygu eraill, os a phryd y barnwyd bod hynny'n angenrheidiol.

**PENDERFYNWYD:** 

Bod y Pwyllgor wedi ystyried yr adroddiad ac wedi rhoi ei farn fel y'i mynegwyd uchod, ar y Rhaglen Sefydlu Aelodau ddrafft ar gyfer 2022.

# 208. <u>ADRODDIAD DRAFFT PANEL ANNIBYNNOL CYMRU AR GYDNABYDDIAETH</u> <u>ARIANNOL 2022/2023</u>

Cyflwynodd y Rheolwr Gwasanaethau Democrataidd Dros Dro adroddiad a oedd yn gofyn i'r Pwyllgor roi ei farn mewn perthynas ag adroddiad drafft Panel Annibynnol Cymru ar Gydnabyddiaeth Ariannol. Dywedodd y byddai unrhyw ymateb yn cael ei gyflwyno i IRPW a Chymdeithas Llywodraeth Leol Cymru erbyn y dyddiad cau, sef 26 Tachwedd 2021.

Esboniodd fod yr adroddiad drafft yn cynnig rhai newidiadau i'r gydnabyddiaeth gyfredol a ragnodir ar gyfer Aelodau Etholedig ar lefelau Prif Athrawon (Bwrdeistref Sirol) a Chynghorau Tref a Chymuned. Tynnodd sylw at elfennau allweddol yr adroddiad drafft fel y'u rhestrir yn 4.2 ymlaen o'r adroddiad.

Tynnodd y Rheolwr Gwasanaethau Democrataidd Dros Dro sylw at y ffactorau allweddol sy'n sail i benderfyniad y Panel a gafodd eu rhestru yn 3.21 o'r adroddiad drafft. Ychwanegodd fod yr adroddiad drafft hefyd yn ymdrin â newidiadau a/neu ofynion o ran Cefnogi gwaith aelodau etholedig awdurdodau lleol, uwch gyflogau penodol neu ychwanegol, Trefniadau Rhannu Swyddi, Cynorthwywyr i'r Weithrediaeth yn ogystal â darpariaeth cynllun pensiwn llywodraeth leol (CPLIL).

Tynnodd sylw at y pwyntiau a gafodd eu gwneud yn 3.2 o'r adroddiad drafft ynghylch diogelwch Aelodau yng ngoleuni marwolaeth drasig ddiweddar Syr David Amess AS. Roedd hyn yn rhywbeth a gafodd ei amlygu iddo ef ei hun a Rheolwr y Grŵp – Gwasanaethau Cyfreithiol a Democrataidd ac roedd yn cael ei ystyried yn agosach yn dilyn y drasiedi, h.y. er diogelwch parhaus yr Aelodau sy'n ymgymryd â'u rolau a'u cyfrifoldebau.

Roedd Aelod yn credu na ddylid gwneud sylwadau ar y swm o arian y mae'r Panel Annibynnol ar Gydnabyddiaeth Ariannol wedi'i gynnig i Aelodau, gan fod hwn yn sefydliad ar wahân i lywodraeth leol ac felly'n ddiduedd.

Dywedodd Aelod y dylai'r ystyriaethau gan y Panel Annibynnol ar Gydnabyddiaeth Ariannol fod wedi'u cysylltu â'r arolwg newid i'r ffiniau. Esboniodd fod gan lawer o ardaloedd mawr wardiau aml-Aelod a bod rhai Aelodau yn y Wardiau hyn yn fwy gweithgar nag eraill. Roedd hyn wedi achosi i lwyth gwaith llawer o'r Aelodau mwy gweithgar gynyddu'n sylweddol tra bod Aelodau llai gweithgar yn ymgymryd â llai o'r gwaith hwnnw am yr un tâl. Ychwanegodd pe bai'n cael ei ailethol yn yr etholiadau nesaf ac oherwydd y newid i'r ffiniau, gallai ei lwyth gwaith dreblu tra bod y gydnabyddiaeth yr un fath â wardiau llai.

Cytunodd y Cadeirydd ei bod yn sefyllfa anodd i Aelodau fod ynddi, h.y. wardiau amlaelod yn ogystal â wardiau o wahanol feintiau. Gwnaeth y Cadeirydd sylwadau ar benderfyniad 10 ac 11 o adroddiad drafft y Panel Annibynnol ar Gydnabyddiaeth Ariannol gan nodi bod Aelodau, yn y gorffennol wedi datgan nad oedd angen darparu'r rhain gan eu bod yn credu ei bod yn gyffredin sefydlu'r seilwaith hwn. Gofynnodd beth oedd barn yr Aelodau ar hyn yn awr, gan fod y galw am weithio o bell wedi cynyddu. Cododd hyn gan y sylwyd bod llawer o Gynghorwyr yn cael problemau cysylltu yn rheolaidd yn ystod neu cyn cyfarfod.

Roedd y Rheolwr Gwasanaethau Democrataidd Dros Dro o'r farn bod hyn yn rhywbeth y byddai CBSC yn ymchwilio iddo, yn ogystal â'r holl awdurdodau lleol wrth symud ymlaen gyda gwaith parhaus o bell a hybrid yn debygol ar gyfer y dyfodol hirdymor. Ychwanegodd fod gwaith yn cael ei wneud yn Siambr ac Ystafelloedd Pwyllgora'r Cyngor i hwyluso cyfarfodydd mewn modd hybrid. Roedd disgwyl i'r gwaith ar Siambr y Cyngor ddechrau rywbryd yn y flwyddyn newydd a'r ystafelloedd Pwyllgora ym mis

Rhagfyr eleni. Ychwanegodd fod ffurflenni asesu risg yn cael eu datblygu i'r Aelodau eu llenwi, lle'r oedd angen iddynt basio meini prawf iechyd a diogelwch penodol, er mwyn mynychu cyfarfod yn Swyddfeydd y Cyngor. Os nad oedden nhw'n gallu pasio'r meini prawf, bydden nhw'n parhau i fynychu'r cyfarfod o bell.

# PENDERFYNWYD: Bod y Pwyllgor yn:

- a. nodi cynnwys yr adroddiad;
- b. Darparu ymateb mewn perthynas ag Adroddiad Blynyddol Drafft IRPW 2022-23;
- c. Cymeradwyo y dylid cyflwyno unrhyw ymateb gan y Pwyllgor i IRPW a CLILC erbyn y dyddiad cau, sef 26 Tachwedd 2021.

### 209. <u>EITEMAU BRYS</u>

Dim

#### BRIDGEND COUNTY BOROUGH COUNCIL

#### REPORT TO DEMOCRATIC SERVICES COMMITTEE

#### 7 JULY 2022

#### REPORT OF THE MONITORING OFFICER

#### ALL WALES MODEL CONSTITUTION AND CONSTITUTION GUIDE

# 1. Purpose of report

- 1.1 The purpose of this report is for the Democratic Services Committee to consider the model constitution and constitution guide.
- 2. Connection to corporate well-being objectives / other corporate priorities
- 2.1 This report assists in the achievement of the following corporate well-being objectives under the **Well-being of Future Generations (Wales) Act 2015**:-
  - 1. **Supporting a successful sustainable economy** taking steps to make the county borough a great place to do business, for people to live, work, study and visit, and to ensure that our schools are focused on raising the skills, qualifications and ambitions for all people in the county borough.
  - 2. Helping people and communities to be more healthy and resilient taking steps to reduce or prevent people from becoming vulnerable or dependent on the Council and its services. Supporting individuals and communities to build resilience, and enable them to develop solutions to have active, healthy and independent lives.
  - 3. **Smarter use of resources** ensure that all resources (financial, physical, ecological, human and technological) are used as effectively and efficiently as possible and support the creation of resources throughout the community that can help to deliver the Council's well-being objectives.

# 3. Background

- 3.1 Section 37 of the Local Government Act 2000 requires Welsh County Councils to prepare and keep up to date a written constitution containing such information as Welsh Ministers may direct, a copy of the Authority's standing orders, a copy of the Authority's code of conduct for Members and such other information as the Authority considers appropriate.
- 3.2 Each County Council has its own individual constitution to reflect that Council's particular circumstances with differences as to the documents it believes appropriate to include in its constitution.
- 3.3 Section 45 of the Local Government and Elections (Wales) Act 2021 (the Act) now requires principal councils to publish a constitution guide which explains in ordinary language the content of the Council's constitution. The principal councils must also

publish their constitution and constitution guide electronically and provide on request either free of charge or at a charge (representing no more than the cost of providing the copy).

# 4. Current situation/proposal

- 4.1 The All Wales Monitoring Officers' Group, via funding from the Welsh Local Government Association commissioned Browne Jacobson to draft a new model constitution and a model constitution guide due to provisions coming into force under the Act. A report was presented to full Council on 20<sup>th</sup> January 2021 detailing the main provisions of the Act.
- 4.2 The new model constitution (attached as **Appendix 1**) has been considered by each local authority and has become common place. The style of the new constitution is more intelligible to the public and there is less duplication. Whilst the new model contains detailed provisions, adopting the model constitution does not require all of those detailed provisions to be adopted and therefore following a local review by officers, some of the Council's existing detailed provisions concerning procedure rules, codes of conduct and protocols have been retained in order to preserve continuity and retain the valuable work that has been done over the years in reviewing each part of the existing constitution.
- 4.3 Attached at **Appendix 2** to this report is a constitution guide based on the Browne Jacobson model guide and adapted to reflect Bridgend Council's constitutional arrangements. Members views are sought on any suggested changes to the draft guide. A beginner's guide summarising all the provisions of the constitution is also attached as **Appendix 3**.
- 4.4 The model constitution and guides will be presented to Cabinet in July for endorsement with the recommendation for approval and adoption by full Council.

# 5. Effect upon policy framework and procedure rules

5.1 The revised constitution will require approval by Cabinet and Council and will be published on the Council's website together with the constitution guides.

# 6. Equality Act 2010 implications

6.1 The protected characteristics identified within the Equality Act 2010, Socioeconomic Duty and the impact on the use of the Welsh language have been
considered in the preparation of this report. As a public body in Wales, the Council
must consider the impact of strategic decisions, such as the development or the
review of policies, strategies, services and functions. It is considered that there will
be no significant or unacceptable equality impacts as a result of this report.

# 7. Well-being of Future Generations (Wales) Act 2015 implications

7.1 The well-being goals identified in the Act were considered in the preparation of this report. It is considered that there is no significant or unacceptable impact upon the achievement of well-being goals/objectives as a result of this report.

# 8.1 Financial implications

8.1 There are no financial implications arising from this report.

# 9. Recommendation

9.1 It is recommended that the Committee consider the model constitution and accompanying guides and agree in principle to them being presented to Cabinet and Council for approval and adoption.

K Watson Chief Officer – Legal and Regulatory Services, HR and Corporate Policy and Monitoring Officer 29 June 2022

**Contact Officer:** L Griffiths

Group Manager - Legal and Democratic Services

**Telephone:** (01656) 643145

**E-mail:** laura.griffiths@bridgend.gov.uk

Postal address: Civic Offices

Angel Street Bridgend CF31 4WB

Background documents: None



BRIDGEND COUNTY BOROUGH COUNCIL CONSTITUTION

# CONTENTS

SE	CTION	l 1	10
1.	Introd	duction	10
	1.1	Purpose and Content of the Constitution	10
	1.2	How the Council Operates	10
SE	CTION	2	12
2.	Purpo	ose, Definition Interpretation and Amendment of the Constitution	12
	2.1	Purpose of the Constitution	12
	2.2	Definitions in the Constitution	122
	2.3	Interpretation of the Constitution	166
	2.4	Duty to Monitor and Review the Constitution	166
	2.5	Protocol for Monitoring and Review of Constitution by Monitoring Officer	166
	2.6	Changes to the Constitution	16
	2.7	Suspension of the Constitution	17
	2.8	Publication	17
SE	CTION	3	18
3.	Gettii	ng Information and Getting Involved	18
	3.1	Getting Information	18
	3.2	Getting Involved	19
	3.3	Getting Involved – Members	21
SE	CTION	4	23
4.	Full (	Council	23
	4.1	Introduction	23
	4.2	The Policy Framework	23
	4.3	The Single Integrated Plan	23
	4.4	The Budget	233
	4.5	Housing Land Transfer	233
	4.6	Functions of the Full Council	233

4.7	Membership	255
4.8	Council Meetings	26
4.9	Rules of Procedure and Debate	26
4.10	Council Procedure Rules – Annual Meeting of the Council	26
4.11	Ordinary Meetings	27
4.12	Extraordinary Meetings	28
4.13	Time, Place and Duration of Meetings	28
4.14	Notice of and Summons to Meetings	29
4.15	Chair of Meeting	29
4.16	Quorum	29
4.17	Remote Attendance	29
4.18	Questions by the Public	29
4.19	Questions by Members	31
4.20	Motions on Notice	33
4.21	Motions without Notice	34
4.22	Rules of Debate	355
4.23	County Borough Debate	38
4.24	Previous Decisions and Motions	38
4.25	Voting	38
4.26	Minutes	39
4.27	Record of Attendance	400
4.28	Exclusion of Public	400
4.29	Members' Conduct	400
4.30	Disturbance by Public	400
4.31	Filming and Use of Social Media During Meetings	411
4.33	Suspension and Amendment of Council Procedure Rules	411
4 34	Application to Committees and Sub-Committees	<b>∆</b> 11

SE	CTION	5	42
5.	The C	Cabinet	. 422
	5.1	Introduction	42
	5.2	Form and Composition of the Cabinet	42
	5.3	Leader	42
	5.4	Deputy Leader	42
	5.5	Other CabinetMembers	42
	5.6	Delegation of Functions	43
	5.7	Assistants to the Executive	. 433
	5.8	Rules of Procedure and Debate	43
	5.9	CabinetProcedure Rules	44
	5.10	How does the Cabinet Operate?	44
	5.11	How are Cabinet Meetings Conducted?	45
	5.12	Disturbance by the Public, Filming and Use of Social Media	46
	5.13	Job sharing by Executive Leaders and Executive Members	47
SE	CTION	6	48
6.	The L	eader	48
	6.1	Election	48
	6.2	Term of Office	48
	6.3	Resignation, Dismissal, Disqualification and Suspension	48
	6.4	Deputy Leader	48
	6.5	Functions and Delegated Authority	48
SE	CTION	7	50
7.	Overv	riew and Scrutiny Committees	50
	7.1	Introduction	50
	7.2	Overview and Scrutiny Committees	50
	7.3	Role, Scope and Membership	50
	7 4	Specific Functions	52

	7.5	Head of Democratic Services	. 54
	7.6	Who May Sit on Overview and Scrutiny Committees?	. 54
	7.7	Co-Optees	. 54
	7.8	Education Representatives	. 55
	7.9	Who Chairs?	. 55
	7.10	Role of the Chair and the Overview and Scrutiny Committees	55
	7.11	Work Programme	. 55
	7.12	Joint Overview and Scrutiny Committees	. 56
	7.13	Rules of Procedure and Debate	. 56
	7.14	What will be the Number and Arrangements for Overview and Scrutiny Committees?	.57
	7.15	Meetings of the Overview and Scrutiny Committees	. 57
	7.16	Quorum	. 57
	7.17	Agenda Items	. 57
	7.18	Policy Review and Development	. 57
	7.19	Reports from the Overview and Scrutiny Committees	. 58
	7.20	Rights of Members of the Overview and Scrutiny Committees to Documents	. 58
	7.21	Members and Officers Giving Account	. 58
	7.22	Attendance by Others	. 59
	7.23	Call-In	. 59
	7.24	The Party Whip	. 61
	7.25	Procedure at Overview and Scrutiny CommitteeMeetings	. 61
	7.26	Matters within the Remit of more than one Overview and Scrutiny Committee	. 62
SE	CTION 8	3	. 63
8.	The St	andards Committee	. 63
	8.1	Membership	. 63
	8.2	Term of Office	. 63
	8.3	Quorum	. 63
	8.4	Voting	. 63

	8.5	Chairing the Committee	63
	8.6	Role and Function	64
	8.7	Annual Report	64
	8.8	Work Programme	65
	8.9	Joint Standards Committee	65
	8.10	Rules of Procedure and Debate	65
SE	CTION	9	66
9.	Regul	atory Committees	66
	9.1	Regulatory and Other Committees	66
	9.2	The Governance and Audit Committee	66
	9.3	The Democratic Services Committee	66
	9.4	Other Committees and Sub-Committees	67
	9.5	Rules of Procedure and Debate	67
SE	CTION	10	68
10.	Joint (	Committees	68
	10.1	Introduction	68
	10.2	Arrangements to Promote Wellbeing	68
	10.3	Joint Arrangements	68
	10.4	Access to Information	68
	10.5	Delegation to and from Other Local Authorities	69
	10.6	Contracting Out	69
SE	CTION	11	70
11.	Corpo	rate Joint Committees	70
	11.1	Introduction	70
	11.2	Corporate Joint Committees	70
SE	CTION	12	71
12.	Office	rs	71
	12 1	Management Structure	71

	12.2	Functions of the Chief Executive	72
	12.3	Functions of the Monitoring Officer	73
	12.4	Functions of the Chief Finance Officer	74
	12.5	Functions of the Head of Democratic Services	75
	12.6 Financ	Duty to Provide Sufficient Resources to the Chief Executive, Monitoring Officer, the Officer and Head of Democratic Services	
	12.7	Conduct	77
	12.8	Employment	77
	12.9	Officer Employment Procedure Rules	77
SEC	CTION	13	83
13.	Financ	ce Contracts and Legal Matters	83
	13.1	Financial Management	83
	13.2	Contracts	83
	13.3	Legal Proceedings	83
	13.4	Authentication of Documents	83
	13.5	Common Seal of the Council	84
SEC	CTION	14	85
14.	Respo	nsibility for Functions - Summary	85
	14.1	Who can be Decision Makers?	85
	14.2	Principles of Decision Making	85
	14.4	Other Bodies	86
	14.5	Who Decides – Non-Executive Functions?	86
	14.6	Who Decides – Executive Functions?	86
	14.7	Removal of Delegation	87
	14.8	Who May Exercise Officer Delegations?	87
SEC	CTION	15	114
15.	Acces	s to Information Procedure Rules	114
	15.1	Scope	114
	15.2	Additional Rights to Information	11/

	15.3	Rights to Attend Meetings	. 114
	15.4	Notices of Meeting	. 114
	15.5	Access to Agenda and Reports Before the Meeting	. 114
	15.6	Supply of Copies	. 114
	15.7	Access to Minutes etc after the Meeting	. 115
	15.8	Background Papers	. 115
	15.9	Summary of Public's Rights	. 115
	15.10	Exclusion of Access by the Public to Meetings	. 116
	15.11	Public Interest Test	. 118
	15.12	Exclusion of Access by the Public to Reports	. 118
	15.13	The Forward Work Programme	. 119
	15.14	Consultation on Proposals to be Considered by the Cabinet	. 119
	15.15	Record of Decisions of the Cabinet	. 120
	15.16	Decisions by An Individual Member of the Cabinet	. 121
	15.17	Overview and Scrutiny Committeeand Members' Access to Documents	. 121
	15.18	Additional Rights of Access for Members of Overview and Scrutiny Committees	. 121
SEC	CTION '	16	. 123
16.	Budge	t and Policy Framework Procedure Rules	. 123
	16.1	The Framework for Executive Decisions	. 123
	16.2	Process for Developing the Framework	. 123
	16.3	Process for Developing the Budget	. 124
	16.4	Decisions Outside the Budget or Policy Framework	. 124
	16.5	Urgent Decisions Outside the Budget or Policy Framework	. 125
	16.6	Virement	. 125
	16.7	In-Year Changes to Policy Framework	. 125
	16.8	Call-In of Decisions Outside the Budget or Policy Framework	. 126
SEC	CTION <sup>2</sup>	17	. 128
17.	Financ	ial Procedure Rules	. 128

SE	CTION 18	. 152
18.	Contract Procedure Rules	. 152
SE	CTION 19	. 173
19.	Code of Conduct for Members	. 173
SE	CTION 20	. 193
20.	Officers' Code of Conduct	. 193
SE	CTION 21	. 199
21.	Protocol on Member / Officer Relations	. 199
SE	CTION 22	. 204
22.	Member Role Descriptions	. 204
	Elected Member Role Description	. 204
	Leader (and Deputy) Role Description	. 206
	Deputy Leader	. 208
	Cabinet Member Role Description	. 208
	Chair of the Council Role Description (Mayor)	. 210
	Vice Chair	. 211
	Chair of Democratic Services Committee Role Description	. 211
	Member of a Democratic Services Committee Role Description	. 213
	Chair of a Regulatory Committee Role Description	. 215
	Member of a Regulatory Committee Role Description	. 216
	Chair of Standards Committee Role Description	. 217
	Member of a Standards Committee Role Description	. 218
	Chair of Governance and Audit Committee Role Description	. 219
	Member of a Governance and Audit Committee Role Description	. 221
	Overview and Scrutiny Chair Role Description	. 223
	Overview and Scrutiny Member Role Description	. 224
	Leader of the Opposition Role Description	. 226
	Role of the Deputy Leader of the Opposition	. 227

Member Champion Purpose and Role	. 227
Member Champion Role Description	. 228
Elected Member Person Specification	. 230
Leader Person Specification	. 230
Cabinet Member Person Specification	. 232
Chair of the Council Person Specification (Mayor)	. 233
Chair of Regulatory Committee Person Specification	. 234
Regulatory Committee Member Person Specification	. 234
Scrutiny Chair Person Specification	. 235
Scrutiny Member Person Specification	. 236
Leader of the Opposition Person Specification	. 237
Member Champion Person Specification	. 238

#### CONSTITUTION

# **SECTION 1**

#### 1. INTRODUCTION

#### 1.1 Purpose and Content of the Constitution

- 1.1.1 The Constitution describes the various bodies that make up the Council, their functions, Membership and procedural rules.
- 1.1.2 In Section 3 (Getting Information and Getting Involved) we have provided information for members of the public and Councillors on how you can get information about the Council, and how you can get involved. We hope that this will help people who have an interest in the Council's work, or a particular matter it is dealing with, understand where they can get more information, and how they can contribute to Council activities.
- 1.1.3 You can get a better understanding of what each of the Council's bodies do in Sections 4 to 10 of this Constitution, including Full Council, Cabinet, Overview and Scrutiny Committees, Standards Committee and Regulatory Committees. Section 12 provides information on the management and Officer structures of the Council. Some Officers have a specific duty to ensure that the Council operates within the law and uses resources wisely. Responsibility for functions at Section 14 says which Council bodies, and which Officers, have authority to make which decisions.
- 1.1.4 The procedural rules that apply to the different Council bodies are contained in the sections relating to those bodies, eg Full Council (Section 4), the Cabinet (Section 5) and Overview and Scrutiny (Section 7). You may find these useful if you want to attend a meeting, particularly if you want to be able to ask questions, have a matter discussed, or put your point of view.
- 1.1.5 Sections **Error! Reference source not found.** to 0 have the Codes of Conduct and Protocols which Officers and Members have agreed to comply with. They set the standards of behaviour.
- 1.1.6 The Contents pages at the beginning of this Constitution provide a guide on what is in the Constitution and where you can find it.

# 1.2 How the Council Operates

- 1.2.1 The Council is comprised of 51 Councillors elected every four years. Each Councillor is democratically accountable to the residents of their electoral division as well as all of those who live in the County Borough of Bridgend. The overriding duty of Councillors is to the whole community, but they have a special duty to their constituents, including those that did not vote for them.
- 1.2.2 All Councillors have agreed to follow a Code of Conduct, to ensure high standards in the way they undertake their duties. The Standards Committee is responsible for training and advising on the Code of Conduct.
- 1.2.3 All Councillors meet together regularly as the Council. Meetings of the Council are normally open to the public. Here Councillors decide the Council's overall Policy Framework and set the Budget each year. The Council appoints the

Leader of the Council at the annual meeting. The Leader then decides the size and Membership of the Cabinet i.e the role of individual Members of the Cabinet and arrangements for the exercise and delegation of Executive Functions. The Cabinet is appointed to carry out all of the Council's functions which are not the responsibility of any other part of the Council, whether by Law or under this Constitution, together with making decisions within the Budget and Policy Framework set by the Council.

If you need any further help please do not hesitate to contact us at: <a href="mailto:talktous@bridgend.gov.uk">talktous@bridgend.gov.uk</a>

Whilst the Constitution is very long we hope that you will find it easy to use. We have spent a lot of time trying to make it as easy to follow as is possible with such a long and complex legal document.

You may find it helpful to read this Constitution in conjunction with the Constitution Guide, which is available on the Council's website.

#### **SECTION 2**

### PURPOSE, DEFINITION INTERPRETATION AND AMENDMENT OF THE CONSTITUTION

#### 2.1 **Purpose of the Constitution**

The purpose of the Constitution is to:

- 2.1.1 enable the Council to provide clear leadership to the Community in partnership with citizens, businesses and other organisations;
- 2.1.2 support the active involvement of citizens in the process of local authority decision making:
- 2.1.3 help Councillors represent their constituents more effectively;
- 2.1.4 enable decisions to be taken efficiently and effectively;
- 2.1.5 create a powerful and effective means of holding decision makers to public account:
- 2.1.6 ensure that no one will scrutinise a decision in which they are directly involved;
- 2.1.7 ensure that those responsible for decision making are clearly identifiable to local people and that they explain the reasons for decisions; and
- 2.1.8 provide a means of improving the delivery of services to the community.

#### 2.2 **Definitions in the Constitution**

- 2.2.1 The Constitution of the Council is this document (Sections 1 to 0).
- 2.2.2 Within the Constitution the following words and phrases have the meaning set out below:

"Budget" the overall revenue and capital budget approved by Full

Council (Section 4):

"Chief Executive" an Officer who must be appointed by law to carry out

certain functions. See Section 12 for more details.

"Chief Officer" any Officer (other than a person whose duties are solely

secretarial or administrative or whose duties are

otherwise in the nature of support services) who:

reports directly to the Chief Executive in respect of

all or most of his/her duties; or

the Chief Executive is directly responsible for;

"Constitution Guide"

a document which explains, in ordinary language, the content of this constitution, published by the Council in accordance with section 37 of the Local Government Act 2000.

"Corporate Joint Committee"

corporate bodies, established via regulation, comprising specified principal councils in Wales, for the purpose of enabling strategic planning and delivery at a regional scale. See section 11 for details of the Corporate Joint Committee that the Council is a member of, and the functions that the Corporate Joint Committee will perform.

"Corporate Team"

the senior management body for Officers (Section 12). It includes those Officers designated by the Chief Executive from time to time as Members of the Corporate Team;

"Councillor"

a person elected to the Council to represent an area (called an electoral division) within Bridgend County Borough Council;

"Data Protection Legislation"

the Data Protection Act 2018, the UK General Data Protection Regulation (UK GDPR) and all other laws relating to the use, protection and privacy of personal data

"Deputy Chief Officer" any Officer (other than a person whose duties are solely secretarial or administrative or whose duties are otherwise in the nature of support services) who reports directly to a Chief Officer in respect of all or most of his/her duties;

"Executive"

the Cabinet or a Member or Members of the Cabinet when exercising Executive Functions;

"Executive Decision"

any decision taken by the Cabinet to exercise or refrain from exercising an Executive Function. It also includes decisions made by persons or Member bodies to whom the Cabinet has delegated Executive Functions to exercise or refrain exercising those functions;

"Executive Function"

- (a) Executive Functions are defined by the Local Government Act 2000, subsidiary legislation and associated guidance. Any function that is not exercisable only by Full Council or delegated to another Member body is an Executive Function.
- (b) Contractual matters, the acquisition and disposal of land and financial support to organisations and individuals are also Executive Functions.

It should be noted that Regulatory functions such as planning licensing and building control, are not Executive Functions;

"Forward Work Programme"

the Forward Work Programme is a document which lists all of the decisions that the Council and the Cabinet intend to take and what business the Overview and Scrutiny Committees will be considering and when those matters will be discussed. This does not prevent urgent or unforeseen matters being considered;

"Full Council"

the body where all Councillors act to exercise functions of the Council;

"Local Choice Functions"

there are some functions which the Council may treat as being the responsibility of the Cabinet (in whole or in part) or as being non-executive, at its discretion;

"Local Government (Wales) Measure 2011" referred to as "The Measure". Legislation introduced, inter alia, to strengthen local democracy, deal with changes to executive arrangements, overview and scrutiny, county councils and Member payments;

"Member"

either a Councillor or a person chosen by the Council to serve on one of its Member Bodies (called "a Co-Opted Member");

"Member Body"

any of the following:

- Full Council;
- Cabinet :
- an Overview and Scrutiny Committee;
- Licensing Committee;
- Standards Committee;
- Development Control Committee;
- Appeals Panel;
- Governance and Audit Committee;
- Democratic Services Committee:

Note - references to Committee also includes Sub-Committee;

"Monitoring Officer"

an Officer who must be appointed by law to carry out certain functions. See Section 12 for a description of those functions. The Officer will usually have other duties and a different job title. See Section 12 for which Officer is the Monitoring Officer;

"Non-Executive Functions"

any function which may only be exercised by Full Council (whether by local choice or as a matter of law) or which is delegated to a Member Body other than the Cabinet

"Petition Scheme"

a scheme prepared and published pursuant to section 42 of the Local Government and Elections (Wales) Act

2021 (and reviewed from time to time) for the purpose of setting out: -

- how a petition may be submitted to the Council;
- how and when the Council will acknowledge receipt of a petition;
- the steps the Council will take in response to a petition received by it;
- the circumstances (in any) in which the Council may take no further action in response to a petition; and
- how any by when the Council will make available its response to a petition to the person who submitted the petition and to the public.

# "Planning Application"

any of the following:

- application for planning permission (including renewal);
- application for approval of reserved matters;
- application for listed building consent;
- application relating to trees;
- proposal to serve an urgent works notice or acquire a listed building in need of repair;
- application for conservation area consent;
- application for advertisement consent;
- application to vary or remove conditions on a planning condition;

"Policy Framework"

See Section 4.2:

"Public Services Board"

the public services board established for the Council's local authority area under Part 4 of the Well-being of Future Generations (Wales) Act 2015;

"Section 151 Officer"

an Officer who must be appointed by law to carry out certain functions in relation to financial administration. See Section 12 for a description of those functions. The Officer will usually have other duties and a different job title. See Section 12 for which Officer is the Section 151 Officer:

"Service"

one of the services provided by the Council;

"Single Integrated Plan (SIP)"

this is the single plan for the area that reflects the needs of the local population. The SIP builds on and brings together in one document the joint work previously developed through the Community Plan, as well as the Community Safety, Children and Young Peoples and the Health Social Core and Wellbeing Partnership Plan.

# 2.3 <u>Interpretation of the Constitution</u>

- 2.3.1 We have tried to make the Constitution as clear and as easy to understand as possible. Inevitably, people will have different views about what certain passages mean.
- 2.3.2 During meetings, the person chairing or presiding at the meeting may interpret the relevant procedure rules.
- 2.3.3 In all other situations, the Monitoring Officer will determine the interpretation and application of the Constitution.

# 2.4 Duty to Monitor and Review the Constitution

The Monitoring Officer will monitor and review the operation of the Constitution to ensure that the aims and principles of the Constitution are given full effect. The Section 151 Officer shall be responsible for keeping under review the Financial Procedure Rules set out in Section 0 of the Constitution and shall make any necessary amendments and revisions as are required from time to time. He/she shall report any amendments made to Section 0 to the next available Council meeting for noting.

### 2.5 Protocol for Monitoring and Review of Constitution by Monitoring Officer

A key role for the Monitoring Officer is to make recommendations for ways in which the Constitution could be amended in order to better achieve the purposes set out in this Section. In undertaking this task, the Monitoring Officer may:

- 2.5.1 observe meetings of different parts of the Member and Officer structure;
- 2.5.2 undertake an audit trail of a sample of decisions;
- 2.5.3 record and analyse issues raised with him/her by Members, Officers, the public and other relevant stakeholders; and,
- 2.5.4 compare practices in this Council with those in comparable authorities, or national examples of best practice.

# 2.6 Changes to the Constitution

#### 2.6.1 Approval

Subject to paragraphs 2.6.2 and 2.6.3 below, changes to the Constitution will only be approved by the Full Council after consideration of a proposal by the Monitoring Officer or on the recommendation of the Cabinet.

#### 2.6.2 Minor Changes

If, in the reasonable opinion of the Monitoring Officer, a change is:

- (a) a minor variation; or
- (b) required to be made to remove any inconsistency, ambiguity or typographical correction; or
- (c) required to be made so as to put into effect any decision of the Council or its committees or the Cabinet,

in which case the Monitoring Officer may make such a change. Any such change made by the Monitoring Officer shall come into force with immediate effect. Such changes shall be reported to the next Full Council meeting for information.

#### 2.6.3 Legislative Change

Any part of the Constitution may be amended by the Monitoring Officer where such amendment is required to be made so as to comply with any legislative provision. Such amendments shall take effect when the Monitoring Officer so decides or the legislation (where relevant) so provides. Such changes shall be reported to the next Full Council meeting for information.

### 2.7 **Suspension of the Constitution**

#### 2.7.1 Limit to Suspension

Any of the procedure rules contained in the Constitution may be suspended to the extent permitted within these rules and the law.

# 2.7.2 Procedure to Suspend

A motion to suspend any Rules will not be moved without notice unless at least one half of the whole number of Councillors is present. The extent and duration of suspension will be proportionate to the result to be achieved, taking account of the purposes of the Constitution set out in this Section.

### 2.8 **Publication**

- 2.8.1 The Monitoring Officer will ensure that copies of this Constitution are available for inspection at Council offices and on the Council's website.
- 2.8.2 The Monitoring Officer will ensure that copies of the Constitution Guide are available for inspection at Council offices and on the Council's website.
- 2.8.3 The Monitoring Officer will give an electronic copy of this Constitution to each Member of the Council upon delivery to him/her of that individual's declaration of acceptance of office on the Member first being elected to the Council and thereafter ensure that an up to date version is available for inspection and published on the Council's website.
- 2.8.4 The Monitoring Officer will ensure that the Constitution is updated as necessary in accordance with paragraph 2.6.

#### **SECTION 3**

### 3. **GETTING INFORMATION AND GETTING INVOLVED**

### 3.1 **Getting Information**

#### 3.1.1 Information Available to Members of the Public

(a) When Meetings of the Member Bodies Will Take Place

A programme of meetings is available by contacting the Council direct or via the website.

# (b) Forward Work Programme

From the Forward Work Programme, see what decisions will be taken by the Cabinet or Council and what issues the Overview and Scrutiny Committees will be considering and when these matters will be discussed.

#### (c) Information Available Prior to a Meeting

At least three clear days before a meeting, the agenda, any report likely to be discussed and background papers to that report shall be available for inspection at the offices of the Council and on the website. If an item is added to the agenda later, the revised agenda will be open to inspection from the time when the item is added to the agenda and any report will be made available to the public as soon as it is available and sent to Councillors.

#### (d) Information Available at a Meeting

The Council will make available to members of the public who are physically present at a meeting a reasonable number of copies of the agenda and of the reports for the meeting (save during any part of the meeting to which the public are excluded).

#### (e) Information Available After a Meeting

For a period of six years the agenda, reports and the minutes of the meeting shall be available for inspection. The background papers shall remain open for inspection for a period of four years.

#### (f) Council's Accounts

The Public may inspect the Council's accounts and make their views known to the external auditor (sections 29 and 30 Public Audit (Wales) Act 2004). Under the Accounts and Audit (Wales) Regulations 201, the accounts will be available for public inspection for twenty working days after the date appointed by the auditor.

Information which is confidential or exempt (as defined in paragraphs 15.10.3 and 15.10.4) will not be disclosed to members of the public at any time.

#### 3.1.2 Information Available to Members of the Council

(a) Members can see any information, which is available to a member of the public.

(b) In addition, a Member may see any information which he or she needs to know in order to fulfil his or her role as a Member of the Council (otherwise known as "need to know"). A Member will not make public information which is confidential or exempt (as defined in Section 15) without the consent of the Council or divulge information given in confidence to anyone other than a Councillor or person(s) or organisation(s) entitled to know it.

#### 3.1.3 Members of an Overview and Scrutiny Committee

A Member of an Overview and Scrutiny Committee may also see any document containing material relating to:

- (a) any business transacted at a meeting of the Cabinet;
- (b) any decision taken by an individual Member of the Cabinet.

An Overview and Scrutiny Committee Member is not be entitled to:

- (c) any document in draft form;
- (d) any part of a document which contains confidential or exempt information unless that information is relevant to an action or decision they are reviewing or scrutinising or intending to scrutinise and which is included in the Committee's Forward Work Programme.

No Member is entitled to see any information relating to a matter in which he or she has a prejudicial interest.

#### 3.1.4 <u>Information Available to Officers</u>

The Monitoring Officer, the Section 151 Officer and the Chief Executive may see any papers or records held by any part of the Council or its Officers. Other Officers may see any information held by the Council provided:

- (a) they need to see the information to do their job; and
- (b) that information is processed lawfully in accordance with the Data Protection Legislation.

# 3.2 **Getting Involved**

3.2.1 The Council must encourage local people to participate in the making of decisions by the Council. See the Council's public participation strategy for more information.

#### 3.2.2 Members of the Public

Members of the public can get involved in the following ways:

(a) Voting for Councillors

If they are over 16 years and registered as a local elector with the Council.

(b) Suggesting Items of Business for Meetings

A member of the public can seek to get a matter included in an agenda by:

- (i) asking the Chair of any Member Body to add an item to the agenda; and/or
- (ii) attending a meeting of the body and suggesting that it looks at an issue when it is considering "items of future business relating to the functions of the Member Body".

#### (c) Submitting Petitions

A member of the public can seek to raise a matter with the Council by submitting a petition in accordance with the Council's Petition Scheme.

#### (d) Taking Part in Meetings

- (i) Members of the public can come to and speak at any meeting which the Council has resolved should include participation by members of the public. The rules on when you may speak and for how long are contained in Section 4.
- (ii) You can also ask formal questions at meetings of Full Council (Section 4).

#### (e) Views of the Public

Under the arrangements put in place by the Council under section 62 of the Local Government (Wales) Measure 2011, members of the public may bring to the attention of the relevant Overview and Scrutiny Committee their views on any matter under consideration by the relevant Committee. The relevant Overview and Scrutiny Committee must take into account any views brought to their attention under these arrangements.

#### (f) When are Meetings Open to the Public?

Meetings will be open to the public wherever possible. The public must be excluded from meetings whenever it is likely that confidential information will be disclosed. The public may be excluded from meetings where it is likely that exempt information will be disclosed. (See section 15.10 for definition of exempt information and section 15.11 for definition of public interest.)

#### (g) Making Comments/Complaints

- (i) A member of the public may comment or complain about Council services by:
  - (A) contacting their local Councillor;
  - (B) contacting the Member of the Cabinet responsible for the service;
  - (C) contacting the Officer responsible for delivering the service or their manager;
  - (D) using the Council's complaints procedure;

- (E) contacting the Public Services Ombudsman for Wales at 1 Ffordd Yr Hen Gae, Pencoed, CF35 5LJ. Telephone: 0300 790 0203 or via the website: www.ombudsman.wales.
- (ii) Comments or complaints can be made about an Officer or Member by:
  - (A) Officer

Contacting the Officer or the Officer's manager.

(B) Members

If the complaint is against a Member, then the complaint should be referred to the Monitoring Officer or the Public Services Ombudsman for Wales (contact details above).

### 3.3 **Getting Involved – Members**

Members can get involved by:

#### 3.3.1 Suggesting Items of Business for the Agenda

As a Member of the Council, you have the same rights as members of the public. In addition to these rights you also have the following rights:

(a) Member bodies in Column A can request that Member bodies in Column B consider or reconsider an issue.

Column A	Column B
Democratic Services Committee	Council
Overview and Scrutiny Committees	Cabinet

(b) Any Member can submit a notice of motion to Council (Section 4) and also ask questions (Section 4).

### 3.3.2 Participating in Meetings

Members of the Council are entitled to attend any formal meeting of the Council, its committees or sub-committees or the Cabinet.

- (a) Members of the Council may attend and speak at any meetings where they are a Member of that body. Where they are not a Member of that body, their attendance and right to speak is at the discretion of the Chair of the body.
- (b) Members of the Cabinet have a special role to play within the Council. They are entitled to exercise any Executive Function provided the Executive Function has been delegated to them by the Leader of the Council.

# 3.3.3 Comments and Complaints

- (a) Members may comment, subject to restrictions in the Code of Conduct for Members on any aspect of Council business by:
  - (i) talking to Officers;
  - (ii) talking to the Leader or Members of the Cabinet;
  - (iii) talking to the Chair of an Overview and Scrutiny Committee.
- (b) If a Member wishes to complain about an:
  - (i) Officer

The procedure set out in the Protocol on Member/Officer Relations may be used.

(ii) Member

The procedure set out in Appendix 3 to Section Error! Reference source not found. may be followed.

#### **SECTION 4**

#### 4. FULL COUNCIL

# 4.1 **Introduction**

The Full Council is a formal meeting of all Councillors. The Full Council is required by law to take certain important decisions including setting the Council's Budget and council tax and approving a number of key plans and strategies, which together form the Policy Framework (listed below). It is responsible for all of the functions not the responsibility of the Cabinet. It will carry out some functions itself, but others will be delegated to Committees or named Officers.

# 4.2 The Policy Framework

The Policy Framework means the plans, schemes and strategies not to be the sole responsibility of the Executive in accordance with current legislation.

# 4.3 **The Single Integrated Plan**

This replaces four of the existing plans and strategies which formed part of the Policy Framework, namely the Community Strategy, the Children and Young People's Plan, the Health, Social Care and Wellbeing Strategy and the Community Safety Partnership Plan.

# 4.4 The Budget

The Budget includes the allocation of financial resources to different services and projects, proposed contingency funds, the council tax base, setting the council tax and decisions relating to the control of the Council's borrowing requirement, the control of its capital expenditure and the setting of virement limits. The Full Council will decide the Council's overall revenue budget and overall capital budget and any changes to these. (See Section 16 for how the Council can change the Policy Framework or Budget referred to it for approval by the Cabinet.)

# 4.5 **Housing Land Transfer**

Housing Land Transfer means the approval or adoption of applications (whether in draft form or not) to the Welsh Ministers for approval of a programme of disposal of land used for residential purposes where approval is required under section 32 or 43 of the Housing Act 1985.

#### 4.6 Functions of the Full Council

Only the Full Council will exercise the following functions:

- 4.6.1 adopting and changing the Constitution;
- 4.6.2 approving or adopting the Single Integrated Plan, Policy Framework, the Budget and any application to the Welsh Ministers in respect of any Housing Land Transfer;
- 4.6.3 subject to the urgency procedure contained in the Access to Information Procedure Rules in Section 15 of this Constitution, making decisions about any matter in the discharge of an Executive Function which is covered by the Policy Framework or the Budget where the decision maker is minded to make it in a

- manner which would be contrary to the Policy Framework or contrary to/or not wholly in accordance with the Budget;
- 4.6.4 appointing and removing the Leader;
- 4.6.5 agreeing and/or amending the terms of reference for Committees, deciding on their composition and making appointments to them (in accordance with the Local Government and Housing Act 1989) unless the appointments have been delegated by the Council;
- 4.6.6 performing the corporate joint committee functions set out in Section 11 of this Constitution, including the making of a corporate joint committee application and the giving of consent to corporate joint committee regulations being made;
- 4.6.7 changing the name of the area or conferring the title of freedom of the Borough;
- 4.6.8 making or confirming the appointment of the Chief Executive; and other Chief Officers;
- 4.6.9 making, amending, revoking, re-enacting or adopting bylaws and promoting or opposing the making of local legislation or personal bills;
- 4.6.10 all Local Choice Functions set out in Section 14 of this Constitution which the Council decides should be undertaken by itself rather than the Cabinet;
- 4.6.11 all matters which by law must be reserved to the Council. For example, appointing the Chair of the Democratic Services Committee, approving the appointment or dismissal of the Chief Executive, determining the level (and any change in the level) of remuneration to the paid to a Chief Officer and approving the pay policy statement;
- 4.6.12 appointing representatives to outside bodies unless the appointment has been delegated by the Council;
- 4.6.13 approving, reviewing and amending the Council's Petition Scheme;
- 4.6.14 consulting local people, other persons carrying on a business in the Council's area, the officers of the Council and every trade union which is recognised by the Council about the extent to which the Council is meeting its performance requirements;
- 4.6.15 preparing an annual report on the extent to which the Council has met its performance requirements;
- 4.6.16 making arrangements for a panel performance assessment and responding to the panel's report;
- 4.6.17 reviewing, approving and amending the Council's wellbeing objectives in accordance with section 9 of the Well-being of Future Generations (Wales) Act 2015 and the accompanying statutory guidance issued by the Welsh Government;
- 4.6.18 considering Chief Executive reports prepared pursuant to section 54(2)(b) of the Local Government and Elections (Wales) Act 2021, within 3 months of them being made;

- 4.6.19 considering annual reports received from the Standards Committee, within 3 months of them being made; and
- 4.6.20 keeping under review the extent to which:
  - (a) it is exercising its functions effectively;
  - (b) using its resources economically, efficiently and effectively; and
  - (c) its governance arrangements are effective for securing the matters set out in a) and b) above.

## 4.7 Membership

- 4.7.1 All Members of the Council shall be Members of Full Council.
- 4.7.2 Substitution is not possible at meetings of the Council.

## 4.7.3 Chairing the Council

- (a) The Councillor elected annually by the Council as its chair will be called the "Mayor".
- (b) The Mayor will cease to be Mayor if they resign, are dismissed by a vote of Full Council, cease to be a Member of the Council, or are unable to act as a Member of the Council. They continue to act as Mayor after an election until their successor has been appointed.

## 4.7.4 Role and Function of the Mayor

The Mayor of the Council and in his/her absence, the Deputy Mayor will have the following roles and functions:

(a) Ceremonial Role

The Mayor of the Council:

- (i) is the civic leader of Bridgend County Borough Council;
- (ii) promotes the interests and reputation of the Council and Bridgend County Borough as a whole and acts as an ambassador for both; and
- (iii) undertakes civic, community and ceremonial activities and fosters community identity and pride.
- (b) Responsibilities of the Mayor
  - (i) to uphold and promote the purpose of the Constitution, and to interpret the Constitution when necessary;
  - to preside over meetings of the Council so that its business can be carried out fairly and efficiently and with regard to the rights of Councillors and the interests of the community;
  - (iii) to ensure that the Council meeting is a forum for the debate of matters of concern to the local community and the place at which Members

who are not on the Cabinet are able to hold the Cabinet and Committee Chair to account;

- (iv) to promote public involvement in the Council's activities;
- (v) to be the conscience of the Council; and
- (vi) to attend such civic and ceremonial functions as the Council and he/she determines appropriate;

## 4.8 Council Meetings

There are three types of Council meeting:

- 4.8.1 the annual meeting;
- 4.8.2 ordinary meetings; and
- 4.8.3 extraordinary meetings.

## 4.9 Rules of Procedure and Debate

The Council Procedure Rules contained in the Sections below will apply to meetings of the Full Council.

## 4.10 Council Procedure Rules - Annual Meeting of the Council

## 4.10.1 Timing and Business

In a year when there is an ordinary election of councillors, the annual meeting will take place within 21 days of the retirement of the outgoing councillors. In any other year, the annual meeting will take place in May. The annual meeting will:

- (a) elect a person to preside if the Mayor and Deputy Mayor of the Council is not present;
- (b) elect the Mayor of the Council;
- (c) elect the Deputy Mayor of the Council;
- (d) approve the minutes of the last meeting;
- (e) receive any announcements from the Mayor and/or the Chief Executive;
- (f) Elect the Leader of the Council;
- (g) to agree the number of Members to be appointed to the Cabinet;
- (h) the Leader to inform Council of the number of Members to be appointed to the Cabinet and announce the Deputy Leader from those Members appointed to the Cabinet and Members to portfolio;
- (i) appoint the Overview and Scrutiny Committees, a Standards Committee and such other committees and sub-committees as the Council considers

- appropriate to deal with matters which are neither reserved to the Council nor are Executive Functions (as set out in Section 14 of this Constitution);
- agree the scheme of delegations or such part of it as the Constitution determines it is for the Council to agree (as set out in Section 14 of this Constitution);
- (k) approve a programme of ordinary meetings of the Council for the year; and
- (I) consider any business set out in the notice convening the meeting.

## 4.10.2 <u>Selection of Councillors on Committees and Outside Bodies</u>

At the annual meeting, the Council meeting will:

- (a) decide which committees and sub-committees to establish for the municipal year;
- (b) decide the size and terms of reference for those committees:
- (c) decide the allocation of seats to political groups in accordance with the political balance rules;
- (d) make appointments to outside bodies except where appointment to those bodies has been delegated by the Council or is exercisable only by the Cabinet.

#### 4.11 **Ordinary Meetings**

Ordinary meetings of the Council will take place in accordance with a programme decided at the Council's annual meeting. The order of business at ordinary meetings will be as follows:

- 4.11.1 elect a person to preside and exercise the powers of the Mayor at the meeting if the Mayor and Deputy Mayor are not present;
- 4.11.2 approve the minutes of the last meeting;
- 4.11.3 receive any declarations of interest from Members;
- 4.11.4 receive any announcements from the Mayor, Chief Executive, the Leader or any Members of the Cabinet;
- 4.11.5 receive a report from the Leader and receive questions and answers on the report;
- 4.11.6 receive a report from the Cabinet and receive questions and answers on the report;
- 4.11.7 receive reports from the Council's committees and receive questions and answers on those reports;
- 4.11.8 receive reports about and receive questions and answers on the business of joint arrangements and external organisations;

- 4.11.9 consider any other business specified in the summons to the meeting, including consideration of proposals from the Cabinet in relation to the Council's Budget and Policy Framework, Single Integrated Plan and reports of the Overview and Scrutiny Committee;
- 4.11.10 consider motions; and
- 4.11.11 deal with questions from Members in accordance with Rule 4.19;
- 4.11.12 receive questions from, and provide answers to, the public in relation to matters which in the opinion of the Mayor are relevant to the business of the meeting.

## 4.12 Extraordinary Meetings

## 4.12.1 Calling Extraordinary Meetings

The Monitoring Officer may call Council meetings in addition to ordinary meetings. Those listed below may request the Monitoring Officer to call additional Council meetings:

- (a) the Council by resolution;
- (b) the Mayor of the Council;
- (c) the Chief Executive;
- (d) the Section 151 Officer;
- (e) any five Members of the Council if they have signed a requisition presented to the Mayor of the Council and he has refused to call a meeting or has failed to call a meeting within seven days of the presentation of the requisition.

## 4.12.2 Business

The business to be conducted at an extraordinary meeting shall be restricted to the item or items of business contained in the request for the extraordinary meeting and there shall be no consideration of previous minutes or reports from committees etc, except that the Mayor may at his/her absolute discretion permit other items of business to be conducted for the efficient discharge of the Council's business.

# 4.13 Time, Place and Duration of Meetings

#### 4.13.1 Time and Place of Meetings

- (a) The time and place of meetings will be determined by the Monitoring Officer and notified in the summons.
- (b) Where the meeting is to be conducted in accordance with the Council's arrangements for multi-location meetings, the summons will also include details of how to access the meeting by remote means.

## 4.14 Notice of and Summons to Meetings

The Monitoring Officer will give notice to the public of the time and place of any meeting in accordance with the Access to Information Procedure Rules in Section 15. At least three

clear days before a meeting, the Monitoring Officer will send a summons signed by him/her to every Member of the Council. The summons will give the date, time and place of each meeting and, where meetings are conducted in accordance with the Council's arrangements for multi-locations meetings, details of how to access the meeting by remote means. The summons will also specify the business to be transacted and will be accompanied by such reports as are available.

## 4.15 Chair of Meeting

The person presiding at the meeting may exercise any power or duty of the Mayor. Where these rules apply to committee and sub-committee meetings, references to the Mayor should instead be read as references to the chair of that committee or sub-committee.

#### 4.16 **Quorum**

4.16.1 The quorum of a meeting will be one quarter of the whole number of Members. During any meeting if the Mayor counts the number of Members present and declares there is not a quorum present, then the meeting will adjourn immediately. Remaining business will be considered at a time and date fixed by the Mayor. If he/she does not fix a date, the remaining business will be considered at the next ordinary meeting.

#### 4.17 Remote Attendance

- 4.17.1 Members will be regarded as present at a meeting where they attend that meeting by remote means in accordance with the Council's arrangements for multi-location meetings providing that they can speak to, hear and be heard by, other attendees at the meeting.
- 4.17.2 For the purposes of section 4.17.1, attendance "by remote means" means attendance in a different physical location to that of other participants and participating through an online meeting platform.

#### 4.18 Questions by the Public

## 4.18.1 General

- (a) Members of the public may ask questions of Members of the Cabinet at ordinary meetings of the Council other than the first ordinary meeting following the annual meeting in a year of ordinary elections of councillors to the Council.
- (b) The total time allocated for questions by the public should be limited to thirty minutes.

## 4.18.2 Order of Questions

Questions will be asked in the order notice of them was received, except that the Mayor may group together similar questions.

## 4.18.3 Notice of Questions

A question may only be asked if notice has been given by delivering it in writing or by electronic mail to the Monitoring Officer no later than midday ten working

days before the day of the meeting. Each question must give the name and address of the questioner.

## 4.18.4 Number of Questions

At any one meeting no person may submit more than one question and no more than one such question may be asked on behalf of one organisation.

#### 4.18.5 Scope of Questions

The Monitoring Officer may reject a question if it:

- (a) is not about a matter for which the Council has a responsibility, or which affects the County Borough;
- (b) is defamatory, frivolous or offensive;
- (c) is substantially the same as a question which has been put at a meeting of the Council in the past six months:
- (d) requires the disclosure of confidential or exempt information.

#### 4.18.6 Record of Questions

- (a) The Monitoring Officer will keep a record of each question in a form that enables them to be made available for public inspection and will immediately send a copy of the question to the Councillor to whom it is to be put. Rejected questions will include reasons for rejection.
- (b) Copies of all questions will be circulated to all Councillors and will be made available to the public attending the meeting.

# 4.18.7 Asking the Question at the Meeting

The Mayor will invite the questioner to put the question to the Councillor named in the notice. If a questioner who has submitted a written question is unable to be present, they may ask the Mayor to put the question on their behalf. The Mayor may ask the question on the questioner's behalf, indicate that a written reply will be given or decide, in the absence of the questioner, that the question will not be dealt with.

#### 4.18.8 Supplementary Question

A questioner who has put a question in person may also put one supplementary question without notice to the Councillor who has replied to his or her original question. A supplementary question must arise directly out of the original question or the reply. The Mayor may reject a supplementary question on any of the grounds set out in Rule 4.18.5 above.

# 4.18.9 Written Answers

Any question which cannot be dealt with during public question time, either because of lack of time or because of the non-attendance of the Councillor to whom it was to be put, will be dealt with by a written answer.

# 4.18.10 Reference of Question to the Cabinet or a Committee

Unless the Mayor decides otherwise, no discussion will take place on any question, but any Member may move that a matter raised by a question be referred to the Cabinet or the appropriate committee or sub-committee. Once seconded, such a motion will be voted on without discussion.

## 4.19 **Questions by Members**

#### 4.19.1 On Reports of the Cabinet or Committees

A Member of the Council may ask the Leader or the Chair of a Committee any question without notice directly arising from an item of the report of the Cabinet or a Committee, when that item is being received or is under consideration by the Council.

## 4.19.2 Questions on Notice at Full Council

Except at the annual meeting or the allocated budget meeting, and subject to Rule 4.19.4, a Member of the Council may ask:

- (a) the Mayor;
- (b) a Member of the Cabinet;
- (c) the chair of any committee or sub-committee;

a question on any matter in relation to which the Council has powers or duties or which affects the County Borough of Bridgend, as an item of business after the consideration of reports.

#### 4.19.3 Questions on Notice at Committees and Sub-Committees

Subject to Rule 4.19.4, a Member of a committee or sub-committee may ask the chair of it a question on any matter in relation to which the Council has powers or duties or which affects the County Borough and which falls within the terms of reference of that committee or sub-committee, as an item of business prior to any reports.

#### 4.19.4 Notice of Questions

A Member may ask a question under Rule 4.19.2 or 4.19.3 if either:

- (a) they have given at least ten clear working days' notice in writing of the question to the Monitoring Officer; or
- (b) the question relates to urgent matters, they have the consent of the Mayor or Member to whom the question is to be put and the content of the question is given to the Monitoring Officer by 10:00am on the day of the meeting in the case of an afternoon meeting or by 1:00pm on the working day preceding the day of the meeting in the case of a morning meeting.

## 4.19.5 <u>Maximum Number of Questions</u>

A Member may ask only one question under Rule 4.19.2 or 4.19.3 except with the consent of the Mayor of the Council, committee or sub-committee. The maximum number of questions that may be asked at any meeting of the Full Council is one.

## 4.19.6 Order of Questions

Questions of which notice has been given under Rule 4.19.2 or 4.19.3 will be listed on the agenda in the order determined by the Mayor of the Council, committee or sub-committee.

#### 4.19.7 Content of Questions

Questions under Rule 4.19.2 or 4.19.3 must, in the opinion of the Mayor:

- (a) contain no expressions of opinion;
- (b) relate to matters on which the Council has or may determine a policy;
- (c) not relate to questions of fact.

## 4.19.8 Response

An answer may take the form of:

- (a) a direct oral answer at the meeting;
- (b) where the desired information is in a publication of the Council or other published work, a reference to that publication; or
- (c) where the reply cannot conveniently be given orally, a written answer circulated within ten working days to the questioner.

## 4.19.9 <u>Supplementary Question</u>

A Member asking a question under Rule 4.19.2 or 4.19.3 may ask one supplementary question without notice of the Member to whom the first question was asked. The supplemental question must arise directly out of the original question or the reply.

#### 4.19.10 Length of Speeches

A Member asking a question under Rule 4.19.2 or 4.19.3 and a Member answering such a question may speak for no longer than three minutes unless the Mayor consents to a longer period.

## 4.19.11 Time Allowed for Questions at Council Meetings

- (a) The time allowed for consideration of questions submitted under Rule 4.19.2 shall not, without the consent of the Council, exceed thirty minutes.
- (b) Any remaining questions shall be responded to in writing before the next ordinary meeting of the Council.

#### 4.20 Motions on Notice

#### 4.20.1 Notice

Except for motions which can be moved without notice under Rule 4.21 and in cases of urgency, written notice of every motion, must be delivered to the Monitoring Officer no later than 10:00am on the tenth working day before the Council meeting at which it is to be considered. Motions received will be kept in a form that enables them to be made available for public inspection.

# 4.20.2 Motion Set Out in Agenda

Motions for which notice has been given will be listed on the agenda in the order in which notice was received, unless the Member giving notice states, in writing, that they propose to move it to a later meeting or withdraw it.

## 4.20.3 <u>Scope</u>

Motions must be about matters for which the Council has a responsibility or which affect the County Borough.

#### 4.20.4 Motion to Remove the Leader

- (a) A motion to remove the Leader cannot be moved unless the notice of motion is signed by a number of councillors which is at least equivalent to 15% of the total number of councillors on the Council and which includes councillors from at least two political groups. See Section 6.3.2.
- (b) In order for such a motion to be carried it must have the support of at least two thirds of those Members voting and present at the time the question was put.
- (c) A motion to remove the Leader cannot be moved more than once in any rolling twelve month period.

#### 4.20.5 One Motion per Member

No Member may give notice of more than one motion for any Council meeting, except with the consent of the Mayor.

#### 4.20.6 Time Allowed for Motions

The time allowed for consideration of motions submitted under this Rule shall not, without the consent of the Council, exceed thirty minutes. At the conclusion of the speech being delivered at the expiry of five minutes (or such longer period to which the Council has consented) from the commencement of the Council's consideration of the first such motion, the Mayor shall put to the vote, without further discussion, all the questions necessary to dispose of the motion then under debate provided that:

(a) if the speech to be concluded is a speech proposing a motion, the Mayor shall allow the motion to be formally seconded (without comment);

- (b) if the speech to be concluded is a speech moving an amendment, the Mayor shall allow the amendment to be formally seconded (without comment) and the mover of the motion to exercise his right of reply; and
- (c) otherwise, the Mayor shall allow the mover of the motion to exercise his right of reply.

Any remaining motions submitted under this Rule shall be deferred to the next ordinary meeting of the Council and shall be dealt with at that meeting in the same order and before any other motions of which notice is given for that meeting.

## 4.21 Motions without Notice

The following motions may be moved without notice:

- 4.21.1 to appoint a Chair of the meeting at which the motion is moved;
- 4.21.2 in relation to the accuracy of the minutes;
- 4.21.3 to change the order of business in the agenda;
- 4.21.4 to refer something to an appropriate body or individual;
- 4.21.5 to appoint a committee or Member arising from an item on the summons for the meeting;
- 4.21.6 to receive reports or adoption of recommendations of committees or Officers and any resolutions following from them;
- 4.21.7 to withdraw a motion;
- 4.21.8 to amend a motion;
- 4.21.9 to proceed to the next business:
- 4.21.10 that the question be now put;
- 4.21.11 to adjourn a debate;
- 4.21.12 to adjourn a meeting;
- 4.21.13 to suspend a particular Council Procedure Rule;
- 4.21.14 to exclude the public and press in accordance with the Access to Information Procedure Rules;
- 4.21.15 to not hear further a Member named under Rule 4.29.3 or to exclude them from the meeting under Rule 4.29.4; and
- 4.21.16 to give the consent of the Council where its consent is required by this Constitution.

#### 4.22 Rules of Debate

## 4.22.1 No Speeches until Motion Seconded

No speeches may be made after the mover has moved a proposal and explained the purpose of it until the motion has been seconded.

#### 4.22.2 Right to Require Motion in Writing

Unless notice of the motion has already been given, the Mayor may require it to be written down and handed to him before it is discussed.

#### 4.22.3 Seconder's Speech

When seconding a motion or amendment, a Member may reserve their speech until later in the debate.

## 4.22.4 Content and Length of Speeches

Speeches must be directed to the question under discussion or to a personal explanation or point of order and no speech may exceed five minutes without the consent of the Mayor.

## 4.22.5 When a Member may Speak Again

A Member who has spoken on a motion may not speak again whilst it is the subject of debate, except:

- (a) to speak once on an amendment moved by another Member;
- (b) to move a further amendment if the motion has been amended since he last spoke;
- if his first speech was on an amendment moved by another Member, to speak on the main issue (whether or not the amendment on which he spoke was carried);
- (d) in exercise of a right of reply;
- (e) on a point of order; and
- (f) by way of personal explanation.

#### 4.22.6 Amendments to Motions

- (a) An amendment to a motion must be relevant to the motion and will either be:
  - (i) to refer the matter to an appropriate body or individual for consideration or reconsideration;
  - (ii) to leave out words:
  - (iii) to leave out words and insert or add others; or
  - (iv) to insert or add words

- as long as the effect of 4.22.6(a)(ii) to 4.22.6(a)(iv) is not to negate the motion.
- (b) Only one amendment may be moved and discussed at any one time. No further amendment may be moved until the amendment under discussion has been decided.
- (c) If an amendment is not carried, other amendments to the original motion may be moved.
- (d) If an amendment is carried, the motion as amended takes the place of the original motion. This becomes the substantive motion to which any further amendments are moved.
- (e) After an amendment has been carried, the Mayor will read out the amended motion before accepting any further amendments, or if there are none, put it to the vote.

#### 4.22.7 Alteration of Motion

- (a) A Member may alter a motion of which he has given notice with the consent of the meeting. The meeting's consent will be signified without discussion.
- (b) A Member may alter a motion which he has moved without notice with the consent of both the meeting and the seconder. The meeting's consent will be signified without discussion.
- (c) Only alterations which could be made as an amendment may be made.

## 4.22.8 Withdrawal of Motion

A Member may withdraw a motion which he has moved with the consent of both the meeting and the seconder. The meeting's consent will be signified without discussion. No Member may speak on the motion after the mover has asked permission to withdraw it unless permission is refused.

# 4.22.9 Right of Reply

- (a) The mover of a motion has a right to reply at the end of the debate on the motion, immediately before it is put to the vote.
- (b) If an amendment is moved, the mover of the original motion has the right of reply at the close of the debate on the amendment but may not otherwise speak on it.
- (c) The mover of the amendment has no right of reply to the debate on his amendment.

## 4.22.10 Motions which may be Moved During Debate

When a motion is under debate, no other motion may be moved except the following procedural motions:

- (a) to withdraw a motion;
- (b) to amend a motion;

- (c) to proceed to the next business;
- (d) that the question be now put;
- (e) to adjourn a debate;
- (f) to adjourn a meeting;
- (g) to exclude the public and press in accordance with the Access to Information Procedure Rules; and
- (h) to not hear further a Member named under Rule 4.29.3 or to exclude them from the meeting under Rule 4.29.4.

## 4.22.11 Closure Motions

- (a) A Member may move, without comment, the following motions at the end of a speech of another Member;
  - (i) to proceed to the next business;
  - (ii) to act that the question be now put;
  - (iii) to adjourn a debate; or
  - (iv) to adjourn a meeting.
- (b) If a motion to proceed to next business is seconded and the Mayor thinks the item has been sufficiently discussed, he will give the mover of the original motion a right of reply and then put the procedural motion to the vote.
- (c) If a motion that the question be now put is seconded and the Mayor thinks the item has been sufficiently discussed, he will put the procedural motion to the vote. If it is passed, he will give the mover of the original motion a right of reply before putting his motion to the vote.
- (d) If a motion to adjourn the debate or to adjourn the meeting is seconded and the Mayor thinks the item has not been sufficiently discussed and cannot reasonably be so discussed on that occasion, he will put the procedural motion to the vote without giving the mover of the original motion the right of reply.

## 4.22.12 Point of Order

A point of order is a request from a Member to the Mayor to rule on an alleged irregularity in the procedure of the meeting. A Member may raise a point of order at any time. The Mayor will hear them immediately. A point of order may only relate to an alleged breach of these Council Rules of Procedure or the law. The Member must indicate the Rule or law and the way in which he considers it has been broken. The ruling of the Mayor on the matter will be final.

#### 4.22.13 Personal Explanation

A Member may make a personal explanation at any time. A personal explanation may only relate to some material part of an earlier speech by the Member which may appear to have been misunderstood in the present debate.

The ruling of the Mayor on the admissibility of a personal explanation will be final.

## 4.23 State of the County Borough Debate

## 4.23.1 Calling of Debate

If called, a County Borough debate can be held quarterly on a date and in a form to be agreed with the Mayor.

## 4.23.2 Form of Debate

The Leader and Leaders of the political groups will decide the form of the debate with the aim of enabling the widest possible public involvement and publicity. Democratic Services should be notified of the topic and relevant invitees four weeks in advance of the proposed meeting.

## 4.23.3 Chairing of Debate

The debate will be Chaired by the Mayor.

#### 4.23.4 Results of Debate

The results of the debate will be disseminated as widely as possible within the community and to agencies and organisations in the area which work in active partnership with the Council and considered by the Leader in proposing the Budget and Policy Framework to the Council for the coming year.

## 4.24 **Previous Decisions and Motions**

#### 4.24.1 Motion to Rescind a Previous Decision

A motion or amendment to rescind a decision made at a meeting of Council within the past six months cannot be moved unless the notice of motion is signed by at least five Members, except in the case of new information becoming available.

## 4.24.2 Motion Similar to One Previously Rejected

A motion or amendment in similar terms to one that has been rejected at a meeting of Council in the past six months cannot be moved unless the notice of motion or amendment is signed by at least five Members. Once the motion or amendment is dealt with, no one can propose a similar motion or amendment for six months.

#### 4.25 **Voting**

# 4.25.1 Majority

Unless this Constitution provides otherwise, any matter will be decided by a simple majority of those Members voting and present at the time the question was put.

#### 4.25.2 Mayor's Casting Vote

If there are equal numbers of votes for and against, the Mayor will have a second or casting vote. There will be no restriction on how the Mayor chooses to exercise a casting vote.

## 4.25.3 Method of Voting

Unless a recorded vote is demanded under Rule 4.25.4 the Mayor will take the vote by show of hands, or by use of the Council's electronic voting system if any, or if there is no dissent, by the affirmation of the meeting.

#### 4.25.4 Recorded Vote

If a Member present at the meeting demands it and it is seconded, the names for and against the motion or amendment or abstaining from voting will be taken down in writing and entered into the minutes.

## 4.25.5 Right to Require Individual Vote to be Recorded

Where any Member requests it immediately after the vote is taken, their vote will be so recorded in the minutes to show whether they voted for or against the motion or abstained from voting.

## 4.25.6 Voting on Appointments

If there are more than two people nominated for any position to be filled and there is not a clear majority of votes in favour of one person, then the name of the person with the least number of votes will be taken off the list and a new vote taken. The process will continue until there is a majority of votes for one person. The Mayor's casting vote will not apply.

#### 4.26 Minutes

#### 4.26.1 Signing the Minutes

The Mayor will sign the minutes of the proceedings at the next suitable meeting. The Chair will move that the minutes of the previous meeting be signed as a correct record. The only part of the minutes that can be discussed is their accuracy.

#### 4.26.2 No Requirement to Sign Minutes of Previous Meeting at Extraordinary Meeting

Where in relation to any meeting, the next meeting for the purpose of signing the minutes is a meeting called under paragraph 3 of Schedule 12 to the Local Government Act 1972 (an extraordinary meeting), then the next following meeting (being a meeting called otherwise than under that paragraph) will be treated as a suitable meeting for the purposes of paragraph 41(1) and (2) of Schedule 12 relating to signing of minutes.

#### 4.26.3 Form of Minutes

Minutes will contain all motions and amendments in the form and order the Mayor put them.

## 4.27 Record of Attendance

All Members physically present during the whole or part of a meeting must sign their names on the attendance sheets before the conclusion of every meeting to assist with the record of attendance. All Members attending a meeting by remote means must confirm their attendance to Democratic Services before the conclusion of every meeting.

## 4.28 Exclusion of Public

Members of the public and press may only be excluded either in accordance with the Access to Information Procedure Rules in Section 15 of this Constitution or Rule 4.30 (Disturbance by Public).

#### 4.29 Members' Conduct

#### 4.29.1 Speaking at Meetings

When a Member speaks at Full Council he/she must address the meeting through the Mayor. If more than one Member signifies their intention to speak, the Mayor will ask one to speak. Other Members must remain silent whilst a Member is speaking unless they wish to make a point of order or a point of personal explanation.

#### 4.29.2 Mayor Speaking

When the Mayor speaks during a debate, any Member speaking at the time must stop.

#### 4.29.3 Member not to be Heard Further

If a Member persistently disregards the ruling of the Mayor by behaving improperly or offensively or deliberately obstructs business, the Mayor may move that the Member be not heard further. If seconded, the motion will be voted on without discussion.

#### 4.29.4 Member to Leave the Meeting

If the Member continues to behave improperly after such a motion is carried, the Mayor may move that either the Member leaves the meeting or that the meeting is adjourned for a specified period. If seconded, the motion will be voted on without discussion.

#### 4.29.5 General Disturbance

If there is a general disturbance making orderly business impossible, the Mayor may adjourn the meeting for as long as he thinks necessary.

## 4.30 Disturbance by Public

#### 4.30.1 Removal of Member of the Public

If a member of the public interrupts proceedings, the Mayor will warn the person concerned. If they continue to interrupt, the Mayor will order their removal from the meeting room or their removal from the online platform by which they are accessing the meeting from another location.

#### 4.30.2 Clearance of Part of Meeting Room

If there is a general disturbance in any part of the meeting room open to the public, the Mayor may call for that part to be cleared.

## 4.30.3 <u>Clearance of Online Meeting Platform</u>

If there is a general disturbance on the online meeting platform, the Mayor may call for the online meeting platform to be temporarily suspended.

## 4.31 Filming and Use of Social Media During Meetings

Filming and use of social media is permitted during meetings so long as there is no disturbance to the conduct of the meeting.

## 4.32 Electronic Broadcasts of Meetings

4.32.1 Ordinary meetings of the Council will be broadcast live electronically on the Council's website so that members of the public not in attendance at the meeting can see and hear the proceedings. Copies of the meeting will be available for six months after the meeting.

# 4.33 Suspension and Amendment of Council Procedure Rules

## 4.33.1 Suspension

All of these Council Rules of Procedure except Rule 4.20.5, 4.25.5 and 4.26.2 may be suspended by motion on notice or without notice if at least one half of the whole number of Members of the Council are present. Suspension can only be for the duration of the meeting. Rule 4.20.5 can only be suspended by motion on notice and the motion must have the support of at least two thirds of those Members present and voting.

# 4.33.2 Amendment

Any motion to add to, vary or revoke these Council Rules of Procedure will, when proposed and seconded, stand adjourned without discussion to the next ordinary meeting of the Council.

#### 4.34 Application to Committees and Sub-Committees

All of the Council Rules of Procedure apply to meetings of Full Council. None of the rules apply to meetings of the Cabinet. Only Rules 4.13 to 4.16, 4.19 to 4.22, 4.24 to 4.34 (but not Rule 4.29.1) apply to meetings of committees and sub-committees.

## **SECTION 5**

## 5. THE CABINET

## 5.1 Introduction

The Cabinet is appointed to carry out all of the Council's functions which are not the responsibility of any other part of the Council, whether by Law or under this Constitution.

#### 5.2 Form and Composition of the Cabinet

The Cabinet will consist of:

- 5.2.1 the Leader of the Council (the "Leader"); and
- 5.2.2 at least two but not more than nine other Councillors appointed to the Cabinet by the Leader subject to the provisions outlined at 5.13 (Job Sharing).

## 5.3 Leader

## 5.3.1 Election

(a) The Leader will be a Councillor elected to the position of Leader by the Council.

# 5.3.2 Term of Office

The Leader is appointed on an annual basis at the annual meeting of the Council.

#### 5.3.3 Role of the Leader

The Leader will Chair meetings of the Cabinet and determine the portfolios of Members of the Cabinet. Further information is contained in Section 6.

## 5.4 **Deputy Leader**

- 5.4.1 The Leader will appoint a Deputy Leader to act as Leader in the Leader's absence and may also if he/she thinks fit remove the Deputy Leader from Office at any time.
- 5.4.2 The Deputy Leader may exercise all of the functions of the Leader where the position is vacant or where the Leader is absent or otherwise unable to act.

## 5.5 Other Cabinet Members

Other Cabinet Members will be Councillors elected to the position of Cabinet Member by the Leader. Each Cabinet Member shall hold office until:

- 5.5.1 he/she resigns from that office; or
- 5.5.2 he/she is removed from that office by the Leader upon such notice (if any) as the Leader considers appropriate; or
- 5.5.3 he/she ceases to be a Councillor.

The Leader may at any time appoint a Cabinet Member to fill any vacancies.

## 5.6 **Delegation of Functions**

The Leader may exercise Executive Functions himself/herself or may otherwise make arrangements to delegate responsibility for their discharge. The Leader may delegate Executive Functions to:

- 5.6.1 the Cabinet as a whole;
- 5.6.2 a Committee of the Cabinet (comprising executive Members only);
- 5.6.3 an individual Cabinet Member:
- 5.6.4 a joint committee;
- 5.6.5 another local authority or the executive of another local authority;
- 5.6.6 a delegated Officer.

## 5.7 Assistants to the Executive

- 5.7.1 Other Councillors may, from time to time, be designated by the Leader as Assistants to the Executive. Such Councillors will not be:
  - (a) a Member of the Cabinet
  - (b) The Chairman and Vice-Chairman of the Council
- 5.7.2 Assistants to the Executive will not participate in Executive Decision making, but may work closely with a Cabinet Member. He/she will not be a Member of an Overview and Scrutiny Committee relating to the specific responsibilities of the Cabinet Member he/she is assisting or any other areas to which they are assigned.
- An Assistant to the Executive may support the Cabinet Member through the delegation of tasks as agreed for his/her area of responsibility, including attending/chairing meetings; speaking/opening events; reading and commenting on papers; meeting Officers; agreeing press releases/comments and carrying out interview; representative the Council on appropriate groups. An Executive Support Member will not, however, have delegated powers and will not be entitled vote at Cabinet Meetings or Cabinet Committee Meetings nor deputise for the Cabinet Member when the Cabinet Member is called to appear at an Overview and Scrutiny Committee.
- 5.7.4 Assistants to the Executive will be entitled to attend, and speak at, any meeting of the Cabinet or a committee of the Cabinet.

#### 5.8 Rules of Procedure and Debate

The proceedings of the Cabinet shall take place in accordance with the Executive Procedure Rules in Section 5.10 below.

## 5.9 Cabinet Procedure Rules

## 5.10 How does the Cabinet Operate?

Following the annual meeting the Monitoring Officer, in consultation with the Leader will prepare and submit for approval a Cabinet Scheme of Delegations setting out the delegations of Executive Functions, including, where the principle has been approved as part of the Council's Executive Arrangements the delegated authority of each Member of the Cabinet, and thereafter keep this under review and submit updates to the Scheme as appropriate. In addition, the Leader has powers under s.15(4) of the Local Government Act 2000 to discharge personally or to arrange for discharge under others' delegated powers any Executive functions not covered by the Scheme of Delegations for the time being.

## 5.10.1 <u>Sub-Delegation of Executive Functions</u>

- (a) Where the Cabinet, a committee of the Cabinet or an individual Member of the Cabinet is responsible for an Executive Function, they may delegate further to joint arrangements or an Officer.
- (b) Unless the Leader directs otherwise, a committee of the Cabinet to whom functions have been delegated by the Leader may delegate further to an Officer.
- (c) Where Executive Functions have been delegated, that fact does not prevent the discharge of delegated functions by the person or body who delegated.

#### 5.10.2 Conflicts of Interest

- (a) Where the Leader has a conflict of interest this should be dealt with as set out in the Council's Code of Conduct for Members
- (b) If any Member of the Cabinet has a conflict of interest this should be dealt with as set out in the Council's Code of Conduct for Members.
- (c) If the exercise of an Executive Function has been delegated to a committee of the Cabinet, an individual Member or an Officer, and should a conflict of interest arise, then the function will be exercised in the first instance by the person or body by whom the delegation was made and otherwise as set out in the Council's Code of Conduct for Members.

# 5.10.3 <u>Cabinet Meetings – When and Where?</u>

The frequency and timing of meetings of the Cabinet will be determined by the Cabinet. The Cabinet will meet at the Council's main offices or another location. The Council's arrangements for multi-location meetings apply to meetings of the Cabinet.

#### 5.10.4 Public or Private Meetings of the Cabinet?

The Cabinet will hold its meetings in public, except in the circumstances set out in the Access to Information Procedure Rules in Section 15, for example where confidential or exempt information is being discussed.

## 5.10.5 Quorum

- (a) The quorum for a meeting of the Cabinet shall be one quarter of the total number of members of the Cabinet or three, whichever is the larger including the Leader or Deputy Leader.
- (b) The quorum for a meeting of a committee of the Cabinet shall be a quarter of the total number of Members of the committee or three, whichever is the larger.

## 5.10.6 Remote Attendance

- (a) Members will be regarded as present at a meeting of the Cabinet or a committee of the Cabinet where they attend that meeting by remote means in accordance with the Council's arrangements for multi-location meetings providing that they can speak to, and be heard by, other attendees at the meeting.
- (b) For the purposes of section 5.10.8, attendance "by remote means" means attendance in a different physical location to that of other participants and participating through an online meeting platform.

#### 5.10.7 How are Decisions to be Taken by the Cabinet?

- (a) Executive Decisions made by the Cabinet as a whole will be taken at a meeting convened in accordance with the Access to Information Procedure Rules in Section 15 of this Constitution.
- (b) Where Executive Decisions are delegated to a committee of the Cabinet, the rules applying to Executive Decisions taken by them shall be the same as those applying to those taken by the Cabinet as a whole.

## 5.11 How are Cabinet Meetings Conducted?

#### 5.11.1 Who Presides?

The Leader will preside at any meeting of the Cabinet or its committees at which he is present. In his absence, the Deputy Leader will preside. In his/her absence, then a person appointed to do so by those present shall preside.

#### 5.11.2 Who May Attend?

These details are set out in the Access to Information Procedure Rules in Section 15 of this Constitution. See also Section 3 in relation to Member participation in meetings.

## 5.11.3 What Business?

At each meeting of the Cabinet the following business will be conducted:

- (a) consideration of the minutes of the last meeting;
- (b) declarations of interest, if any;
- (c) matters referred to the Cabinet (whether by Overview and Scrutiny Committees or by the Council) for reconsideration by the Cabinet in accordance with the provisions contained in the Overview and Scrutiny Procedure Rules or the Budget and Policy Framework Procedure Rules set out in Sections 7 and 16 of this Constitution;
- (d) consideration of reports from Overview and Scrutiny Committees;
- (e) consideration of reports from Cabinet Committees;
- (f) reports from Officers of the Authority.

## 5.11.4 Consultation

All reports to the Cabinet from any Member of the Cabinet or an Officer on proposals relating to the Budget and Policy Framework must contain details of the nature and extent of consultation undertaken with stakeholders and with the Overview and Scrutiny Committees and the outcome of that consultation. Reports about other matters will set out the details and outcome of consultation as appropriate. The level of consultation required will be appropriate to the nature of the matter under consideration.

## 5.11.5 Who can put Items on the Cabinet Agenda?

- (a) The Leader will decide upon the schedule for meetings of the Cabinet. He/she may put any matter on the agenda of any Cabinet meeting whether or not authority has been delegated to the Cabinet, a committee of it or any Member or Officer in respect of that matter.
- (b) Any Member of the Cabinet may require the Monitoring Officer to make sure that an item is placed on the agenda of the next available meeting of the Cabinet for consideration.
- (c) The Chief Executive, the Monitoring Officer and/or the Chief Financial Officer may include an item for consideration on the agenda of a Cabinet meeting and may require that such a meeting be convened in pursuance of their statutory duties.
- (d) In other circumstances, where any two of the Chief Executive, Chief Finance Officer and Monitoring Officer are of the opinion that a meeting of the Cabinet needs to be called to consider a matter that requires a decision they may jointly include an item on the agenda of a Cabinet meeting. If there is no meeting to deal with the issue in question, then the person(s) entitled to include an item on the agenda may also require that a meeting be considered at which the matter will be considered.

## 5.12 <u>Disturbance by the Public, Filming and Use of Social Media</u>

5.12.1 The provisions in Council Procedure Rules in Section 4.30 in relation to disturbance by the public apply to meetings of the Cabinet.

5.12.2 The provisions in Council Procedure Rules in Section 4.31 relating to filming and use of social media apply to meetings of the Cabinet.

## 5.13 **Job sharing by Executive Leaders and Executive Members**

- 5.13.1 Any election or appointment to the Cabinet (including the Leader) may include the election of two or more Councillors to share office.
- 5.13.2 Where two or more Members have been elected or appointed to share the same office on the Cabinet the maximum number of members of the Cabinet, including the Leader, will be:
  - (a) 12, where at least two of the members have been elected or appointed to share office; or
  - (b) 13, where at least three of the members have been elected or appointed to share office.
- 5.13.3 The members of the Cabinet who share the same office will have one vote between them in respect of any matter on which they have a right to vote because they are a member of the Cabinet.
- 5.13.4 Where any meeting is attended by more than one of the members who share the same office and those members are attending in their capacity as a member of the Cabinet, they together count only as one person for the purpose of determining whether the meeting is quorate.

#### **SECTION 6**

#### 6. THE LEADER

## 6.1 Election

- 6.1.1 The Leader will be elected by Full Council.
- 6.1.2 The Leader will usually be elected on an annual basis at the Annual Council meeting. However, where the post of Leader becomes vacant between Council elections the Leader will be elected at the next meeting of the Full Council.

## 6.2 Term of Office

The Leader is appointed on an annual basis at the annual meeting of the Council.

## 6.3 Resignation, Dismissal, Disqualification and Suspension

- 6.3.1 The Leader may resign the position of Leader by writing to the Mayor.
- 6.3.2 The Leader can be dismissed where the Council passes a resolution removing him/her from office in accordance with the Rules below.
- 6.3.3 The Leader shall cease to be Leader if he/she is suspended or disqualified as a Councillor, or, for other such reasons, cannot fulfil the role of Leader.
- 6.3.4 The Leader will cease to be Leader upon death or upon suffering any disability which will, or is likely to, prevent them from undertaking the role of Leader for a period of three months or more.

#### 6.4 **Deputy Leader**

## 6.4.1 Appointment

The Leader may designate one of the Members of the Cabinet as Deputy Leader.

## 6.4.2 Duties of the Deputy Leader

The Deputy Leader may exercise all the functions of the Leader where the position is vacant or where the Leader is absent or is otherwise unable to act.

#### 6.4.3 Removal from Office

The Leader may, if he/she think fit, remove the Deputy Leader from office at any time.

#### 6.5 Functions and Delegated Authority

## 6.5.1 Membership of the Cabinet

The Leader appoints and dismisses the Members of the Cabinet subject only to their being a minimum of two, and a maximum of nine, Members of the Cabinet (not counting the Leader) at any time. This number may be increased pursuant to section 5.13.2(b) in the event of a job share to 13.

#### 6.5.2 Role of the Leader

The Leader will chair meetings of the Cabinet and determine the portfolios of Members of the Cabinet.

## 6.5.3 <u>The Cabinet Scheme of Delegations</u>

The Monitoring Officer, in consultation with the Leader, will prepare and submit for approval a Cabinet Scheme of Delegations setting out the delegations of Executive Functions, including where the principle has been approved as part of the Council's Executive arrangements, the delegated authority of each member of the Cabinet and thereafter keep this under review and submit updates to the Scheme, as appropriate. In addition, the Leader has powers under section 15(4) of the Local Government Act 2000 to discharge personally or to arrange for discharge under other delegated powers, any Executive Functions not covered by the Scheme for the time being.

## 6.5.4 Meetings of the Cabinet

Subject to the requirement to publish notice of each meeting three clear days before it takes place, the Leader can call meetings of the Cabinet at such times and places as he/she chooses.

## 6.5.5 Chairing Cabinet Meetings

The Leader shall chair Cabinet meetings. In the Leader's absence the Deputy Leader will chair. If the Deputy Leader is not available the Cabinet will appoint a Member of the Cabinet to Chair the meeting on their behalf by informing the Proper Officer of the person chosen subject to quorum.

## 6.5.6 Acting as one of the Council's Representatives on the Public Services Board

- (a) The Leader shall be one of the Council's two representatives at meetings of the Public Services Board.
- (b) The Leader may designate another member of the Cabinet to attend a meeting of the Public Services Board in their absence. The Leader shall designate the Deputy Leader to attend in their absence unless the Deputy Leader is also not available in which case the Leader shall designate another Member of the Cabinet.

# 6.5.7 <u>Acting as the Council member of the South East Wales Corporate Joint Committee</u>

The Leader shall be the Council member of the South East Wales Corporate Joint Committee. Where the Leader is unable to discharge their functions in respect of the Corporate Joint Committee, the Council shall appoint another member of the Cabinet to discharge those functions on behalf of the Council.

#### 6.5.8 Appointments of Representatives on Outside Bodies

The Leader has authority to appoint representatives of the Council on outside bodies where those outside bodies relate to Executive Functions of the Council.

#### **SECTION 7**

#### 7. OVERVIEW AND SCRUTINY COMMITTEES

# 7.1 Introduction

- 7.1.1 The Council is required by Law to discharge certain overview and scrutiny functions. These functions are an essential component of local democracy. Overview and Scrutiny Committees should be powerful committees that can contribute to the development of Council policies and also hold the Cabinet to account for its decisions. Another key part of the overview and scrutiny role is to review existing policies, consider proposals for new policies and suggest new policies.
- 7.1.2 Overview and scrutiny should be carried out in a constructive way and should aim to contribute to the delivery of efficient and effective services that meet the needs and aspirations of local inhabitants. Overview and Scrutiny Committees should not shy away from the need to challenge and question decisions and make constructive criticism.

## 7.2 Overview and Scrutiny Committees

In order to achieve this, the Council have appointed four Overview and Scrutiny Committees which between them will:

- 7.2.1 review or scrutinise decisions made, or actions taken, in connection with the discharge of any of the Council's functions whether by the Cabinet or another part of the Council;
- 7.2.2 make reports or recommendations to the Council or the Cabinet in connection with the discharge of any functions;
- 7.2.3 review or scrutinise decisions made, or actions taken, by the Public Services Board;
- 7.2.4 consider any matter which affects the Council's area or its inhabitants; and
- 7.2.5 exercise the right to call in for reconsideration decisions made but not yet implemented by the Cabinet and Officers.

## 7.3 Role, Scope and Membership

The role, scope and Membership of the Overview and Scrutiny Committees are described in the table below:

Committee and Membership	Role and Scope
Corporate Overview and Scrutiny Committee Twelve County Borough Councillors (for consideration of Education matters to include 5 Education Representatives).	<ul> <li>To consider the service provision, planning, management and performance relating to corporate performance and governance;</li> <li>To consider policies, protocols and plans relating to corporate performance and governance;</li> <li>To co-ordinate the recommendations from each Overview and Scrutiny Committee upon</li> </ul>

Committee and Membership	Role and Scope
	the draft annual MTFS, including the budget savings proposals and comments upon the consultation process;  To scrutinise the performance and budget monitoring of all Directorates in the achievement of the corporate priorities;  To consider reports regarding any recommendations made by the Public Service Board Scrutiny Panel regarding the performance of Bridgend's Public Service Board (PSB) in accordance with the Wellbeing of Future Generations (Wales) Act 2015 and Guidance;  To scrutinise the Community Safety Partnership and its constituent bodies in respect of their community safety roles in accordance with the Crime and Disorder guidance;  To consider the Council's Corporate Plan and monitor progress against the Corporate Priorities;  To develop and implement a Forward Work Programme for the Committee having regard to the Council's Corporate Priorities and Risk Management Framework and have oversight and coordination of the Forward Work Programmes for the Overview and Scrutiny Committees.
Subject Overview and Scrutiny Committee 1 (Education themed but not exclusive to) Twelve County Borough Councillors (For consideration of education matters to include 5 Education Representatives).	<ul> <li>To consider the service provision, planning, management and performance relating to the relevant subject area prioritised and allocated by the Corporate Overview and Scrutiny Committee;</li> <li>To consider policies, protocols and plans relating to the relevant subject area prioritised and allocated by the Corporate Overview and Scrutiny Committee;</li> <li>To contribute to the annual MTFS budget consultation process in relation to the relevant subject area;</li> <li>To propose items for the Forward Work Programme having regard for the Council's Corporate Priorities and Risk Management Framework and subject to oversight and coordination by the Corporate Overview and Scrutiny Committee.</li> </ul>
Subject Overview and Scrutiny Committee 2 (Social Services and Wellbeing themed but not exclusive to)	To consider the service provision, planning, management and performance relating to the relevant subject area prioritised and allocated

Committee and Membership	Role and Scope
Twelve County Borough Councillors (For consideration of education matters to include 5 Education Representatives).	<ul> <li>by the Corporate Overview and Scrutiny Committee;</li> <li>To consider policies, protocols and plans relating to the relevant subject area prioritised and allocated by the Corporate Overview and Scrutiny Committee;</li> <li>To contribute to the annual MTFS budget consultation process in relation to the relevant subject area;</li> <li>To propose items for the Forward Work Programme having regard for the Council's Corporate Priorities and Risk Management Framework and subject to oversight and coordination by the the Corporate Overview and Scrutiny Committee.</li> </ul>
Subject Overview and Scrutiny Committee 3  Twelve County Borough Councillors (For consideration of education matters to include 5 Education Representatives).	<ul> <li>To consider the service provision, planning, management and performance relating to the relevant subject area prioritised and allocated by the Corporate Overview and Scrutiny Committee;</li> <li>To consider policies, protocols and plans relating to the relevant subject area prioritised and allocated by the Corporate Overview and Scrutiny Committee;</li> <li>To contribute to the annual MTFS budget consultation process in relation to the relevant subject area;</li> <li>To propose items for the Forward Work Programme having regard for the Council's Corporate Priorities and Risk Management Framework and subject to oversight and coordination by the Corporate Overview and Scrutiny Committee.</li> </ul>

# 7.4 **Specific Functions**

# 7.4.1 Policy Development and Review

The Overview and Scrutiny Committees may:

- (a) assist the Council and the Cabinet in the development of its Budget and Policy Framework by in depth analysis of policy issues;
- (b) conduct research, community and other consultation in the analysis of policy issues and possible options;
- (c) question Members of the Cabinet and/or Committees and Chief Officers from the Council about their views on issues and proposals affecting the area;

- (d) liaise with other external organisations operating in the area, whether national, regional or local, to ensure that the interest of local people are enhanced by collaborative working; and
- (e) consider the impact of policies to assess if they have made a difference.

## 7.4.2 Scrutiny

The Overview and Scrutiny Committees may:

- review and scrutinise the decisions by and performance of the Cabinet and/or Committees and Council Officers in relation to individual decisions and over time:
- (b) review and scrutinise the performance of the Council in relation to its policy objectives, performance targets and/or particular service areas;
- (c) question Members of the Cabinet and/or Committees and Chief Officers from the Council about their decisions and performance, whether generally in comparison with service plans and targets over a period of time, or in relation to particular decisions, initiatives or project;
- (d) make recommendations to the Cabinet and/or appropriate Committee and/or Council arising from the outcome of the scrutiny process;
- review and scrutinise the performance of other public bodies in the area and invite reports from them by requesting them to address the Overview and Scrutiny Committees and local people about their activities and performance; and
- (f) question and gather evidence from any person (with their consent).

#### 7.4.3 Scrutiny of the Public Services Board

The Corporate Overview and Scrutiny Committee has the additional functions of overseeing the work of the Public Service Board in accordance with the Wellbeing of Future Generations (Wales) Act 2015. The Committee designate this responsibility to the Public Service Board Scrutiny Panel to be able to carry out the following functions:

- (a) review or scrutinise decisions made, or other action taken, by the Public Services Board:
- (b) review or scrutinise the Public Services Board's governance arrangements;
- (c) make reports or recommendations to the Public Services Board with respect to the board's functions or governance arrangements;
- (d) consider such matters relating to the board as the Welsh Ministers may refer to it and to report to the Welsh Ministers accordingly; and
- (e) copy reports to the Public Services Board regarding the board's functions and governance arrangements to: -
  - (i) the Welsh Ministers:

- (ii) the Future Generations Commissioner for Wales; and
- (iii) the Auditor General for Wales.
- (f) require one or more of the attendees at the Public Services Board, or anyone designated by such a person, to attend a meeting of the Public Services Board Scrutiny Panel and provide it with explanations of such matters as it may specify.
- 7.4.4 The Membership of the Scrutiny Panel is determined annually by the nomination of three Members from the Corporate Overview and Scrutiny Committee, one Member from each of the Subject Overview and Scrutiny Committees and invitees from counterpart representatives that sit on the Public Service Board. Any recommendations made by the Panel are presented to the Corporate Overview and Scrutiny Committee for approval prior to submission to the Board.

## 7.4.5 Annual Report

The Scrutiny Annual Report will be considered and endorsed by the Corporate Overview and Scrutiny Committee before being reported to Full Council on their work with recommendations for the coming year and amended working methods if appropriate.

## 7.5 **Head of Democratic Services**

One of the roles of the Head of Democratic Services under section 8 of The Measure is to promote the role of the Council's Overview and Scrutiny Committees and to promote support and guidance to Council Members and Officers generally about the functions of the Overview and Scrutiny Committees.

## 7.6 Who May Sit on Overview and Scrutiny Committees?

All Councillors except Members of the Cabinet and Assistants to the Executive may be Members of the Overview and Scrutiny Committees. However, no Member may be involved in scrutinising on decisions in which he has been directly involved.

#### 7.7 Co-Optees

Each Overview and Scrutiny Committee shall be entitled to recommend to Council the appointment of a number of people as non-voting co-optees subject to the number of co-opted members on the Committee shall not exceed one third of the total membership of the Committee. In exercising or deciding whether to exercise a co-option, the Authority must, under section 76 of The Measure, have regard to guidance given by the Welsh Ministers and comply with directions given by them.

Co-optees' membership may be for:

- (1) the life of the Committee;
- (2) until such time as it is terminated by the Council, upon the recommendation of the Committee; or
- (3) for the purposes of a particular review or performance monitoring exercise.

#### 7.8 Education Representatives

The Overview and Scrutiny Committee considering education matters shall include in its Membership voting representatives of religious faiths and of parent governors, as required by law and guidance from Senedd Cymru:

- (1) One Church in Wales representative
- (2) One Roman Catholic Church representative
- (3) Three parent governor representatives (one from a primary school, one from a secondary school and one from a special school

A Co-opted Member of an Overview and Scrutiny Committee or sub-committee will be entitled to vote at a meeting of the Committee or sub-committee on any matter which relates to any education functions which are the responsibility of the Cabinet and which falls to be decided at the meeting; but will not otherwise be entitled to vote.

# 7.9 Who Chairs?

The arrangements included in sections 66-75 of the Local Government (Wales) Measure 2011 will be followed for appointing persons to chair Overview and Scrutiny Committees.

#### 7.10 Role of the Chair and the Overview and Scrutiny Committees

- 7.10.1 The Chairs will liaise with the Cabinet and supervise the Work Programme and identify suitable priority areas for Scrutiny.
- 7.10.2 In summary, therefore, the Chair will:
  - (a) meet regularly to monitor Work Programmes;
  - (b) liaise with the Cabinet on issues affecting the Scrutiny Work Programme.

#### 7.11 Work Programme

At the beginning of the municipal year each Overview and Scrutiny Committee will be asked to identify issues for consideration during the year, using pre-determined criteria which emphasises the need to consider issues such as impact, risk, performance, budget and community perception when identifying topics for investigation. These items will be presented to the Corporate Overview and Scrutiny Committee for prioritisation and designation to each Overview and Scrutiny Committee. There is an understanding that the Work Programme will remain flexible and be revisited at each meeting of the Corporate Overview and Scrutiny Committee to consider all potential items and reprioritise with input from all the Committees.

- 7.11.1 Meetings of the Overview and Scrutiny Committees shall be called by the Monitoring Officer and may be required to do so by the Chair of the relevant Overview and Scrutiny Committee or by any five members of the Committee.
- 7.11.2 Extraordinary meetings may be called from time to time in order to deal with call-ins (Section 7.23) where the Chair of an Overview and Scrutiny Committee and the Monitoring Officer agree it is necessary for that Overview and Scrutiny Committee to consider the called in decision before the Committee's next programmed meeting.

#### 7.12 **Joint Overview and Scrutiny Committees**

Under section 58 of The Measure, regulations may be made to permit or require two or more local authorities to appoint a joint Overview and Scrutiny Committee. This is set out in the Local Authorities (Joint Overview and Scrutiny) (Wales) Regulations 2013.

## 7.13 Rules of Procedure and Debate

The Overview and Scrutiny Procedure Rules will apply to meetings of the Overview and Scrutiny Committees.

## 7.14 What will be the Number and Arrangements for Overview and Scrutiny Committees?

- 7.14.1 The Council will have four Overview and Scrutiny Committees set out in the table in Section 7.3 and will appoint to them as it considers appropriate from time to time. The Committees may appoint smaller groups to carry out detailed examination of particular topics for report back to them. Such groups may be appointed for a fixed period on the expiry of which they shall cease to exist.
- 7.14.2 The terms of reference of the various Overview and Scrutiny Committees will be as set out in the table in Section 7.3.
- 7.14.3 Each Subject Overview and Scrutiny Committee will be Chaired by a Chair appointed from the membership of that Overview and Scrutiny Committee. The Chair of the Corporate Overview and Scrutiny Committee will be appointed from one of the Chairs of the Subject Overview and Scrutiny Committees in accordance with the political balance rules:
  - (a) there will be cross party Membership of all Overview and Scrutiny Committees;
  - (b) the Overview and Scrutiny Committees shall undertake the following:
    - investigate or review a particular matter in depth and without delay, reporting their conclusions and making any recommendations to the Council or Cabinet as appropriate;
    - (ii) conduct research, community (and other) consultation for the purposes of analysing issues and developing where appropriate; possible options, through liaison with the area/community partnerships;
    - consider and report on mechanisms to encourage and enhance community participation in the development of service delivery options;
    - (iv) question Cabinet Members and Officers about their views and actions on issues and proposals affecting the County Borough;
    - (v) liaise with other external organisations operating in the area, whether national, regional or local, to ensure that the interests of local people are enhanced by collaborative working; and
    - (vi) question and gather evidence from any person (with his/her consent) whilst conducting investigative and reporting processes.

#### 7.15 Meetings of the Overview and Scrutiny Committees

The Council may determine a cycle of meetings for the Overview and Scrutiny Committees. If the Council do not set the cycle, each such Committee shall determine their own cycle of meetings. The Chair may change the date or cancel meetings, or call additional meetings as they consider necessary to deal with the Committee's work programme. A meeting of an Overview and Scrutiny Committee may be called by the Chair or by the Monitoring Officer, if he/she considers it necessary or appropriate.

## 7.16 **Quorum**

The quorum for an Overview and Scrutiny Committee shall be a quarter of all Members of the Committee or 3 Members whichever is greater.

## 7.17 Agenda Items

- 7.17.1 Any Member of a particular Overview and Scrutiny Committee shall be entitled to give notice to the Monitoring Officer that he wishes an item relevant to the functions of the that Committee to be included on the agenda for the next available meeting. Ten working days' notice of the item should be given to the Monitoring Officer together with sufficient information to enable the Officer to advise about the nature and purpose of the item.
- 7.17.2 On receipt of such a request, so long as it is an appropriate matter to be considered, the Monitoring Officer will ensure that it is included on the next available agenda.
- 7.17.3 An Overview and Scrutiny Committee shall also respond, as soon as their work programme permits, to requests from the Council and/or the Cabinet to review particular areas of Council activity. Where they do so, the particular Overview and Scrutiny Committee shall report their findings and any recommendations back to the Cabinet and/or Council. The Council and/or the Cabinet shall consider the report of the Overview and Scrutiny Committee within one month of receiving it.

#### 7.18 **Policy Review and Development**

- 7.18.1 The role of the Overview and Scrutiny Committees in relation to the development of the Council's Budget and Policy Framework is set out in detail in the Budget and Policy Framework Procedure Rules in Section 15.
- 7.18.2 In relation to the development of the Council's approach to other matters not forming part of its Budget and Policy Framework, an Overview and Scrutiny Committee may make proposals to the Cabinet for developments in so far as they relate to matters within their terms of reference.
- 7.18.3 An Overview and Scrutiny Committee may hold enquiries and investigate the available options for future direction in policy development and may appoint advisers and assessors to assist them in this process. They may go on site visits, conduct public surveys, hold public meetings, commission research and do all other things that they reasonably consider necessary to inform their deliberations. They may ask witnesses to attend to address them on any matter under consideration and may pay to any advisers, assessors and witnesses a reasonable fee and expenses for doing so.

#### 7.19 Reports from the Overview and Scrutiny Committees

- 7.19.1 All formal reports from the Overview and Scrutiny Committees will be submitted to the Monitoring Officer for consideration by the Cabinet (if the proposals are consistent with the existing Budget and Policy Framework), or to the Council as appropriate (eg if the recommendation would require a departure from or a change to the agreed Budget and Policy Framework).
- 7.19.2 If an Overview and Scrutiny Committee cannot agree on one single final report to the Council or Cabinet as appropriate, one minority report may be prepared and submitted for consideration by the Council or Cabinet with the majority report.
- 7.19.3 The Council or Cabinet shall consider the report of an Overview and Scrutiny Committee within one month of it being submitted to the Monitoring Officer.

# 7.20 Rights of Members of the Overview and Scrutiny Committees to Documents

- 7.20.1 In addition to their rights as Councillors, Members of the Overview and Scrutiny Committees have the additional right to documents, and to notice of meetings as set out in the Access to Information Procedure Rules in Section 15 of this Constitution.
- 7.20.2 Nothing in this paragraph prevents more detailed liaison between the Cabinet and the Overview and Scrutiny Committees as appropriate depending on the particular matter under consideration.

#### 7.21 Members and Officers Giving Account

- 7.21.1 The Overview and Scrutiny Committees may scrutinise and review decisions made or actions taken in connection with the discharge of any Council functions. As well as reviewing documentation, in fulfilling the Scrutiny role, it may require any Member of the Cabinet, the Chief Executive and/or any senior Officer to attend before it to explain in relation to matters within their remit:
  - (a) any particular decision or series of decisions;
  - (b) the extent to which the actions taken implement Council policy; and/or
  - (c) their performance
  - and it is the duty of those persons to attend if so required.
- 7.21.2 For this purpose, senior Officer includes any Chief Officer, deputy Chief Officer and other appropriate senior Officer. Where there are concerns about the appropriateness of the Officer who should attend, the relevant Chief Officer shall discuss this with the appropriate Overview and Scrutiny Chair or Vice Chair with a view to achieving consensus.
- 7.21.3 Where any Member or Officer is required to attend an Overview and Scrutiny Committee under this provision, the Chair of that Committee will inform the Monitoring Officer. The Monitoring Officer shall inform the Member or Officer, if necessary in writing, giving at least five working days' notice of the meeting at which he or she is required to attend (unless agreed otherwise). Any notice will state the nature of the item on which he or she is required to attend to give

- account and whether any papers are required to be produced for the Committee.
- 7.21.4 Where the account to be given to an Overview and Scrutiny Committee will require the production of a report, then the Member or Officer concerned will be given sufficient notice to allow for preparation of that documentation.
- 7.21.5 Where, in exceptional circumstances, the Member or Officer is unable to attend on the required date, then the Overview and Scrutiny Committee shall in consultation with the Member or Officer arrange an alternative date for attendance.

## 7.22 Attendance by Others

An Overview and Scrutiny Committee may invite people other than those people referred to in paragraph 7.212 above to address it, discuss issues of local concern and/or answer questions. It may for example wish to hear from residents, stakeholders and Members and Officers in other parts of the public sector and shall invite such people to attend.

#### 7.23 **Call-In**

## 7.23.1 Rules

- (a) Where a decision is made by the Cabinet, an individual Member of the Cabinet or a Committee of the Cabinet or under joint arrangements, the decision shall be published by the Monitoring Officer, including where possible by electronic means, and shall be available at the main offices of the Council as soon as reasonably practicable after being made. All Members of the Overview and Scrutiny Committees will be sent copies of the records of all such decisions within the same time scale, by the person responsible for publishing the decision.
- (b) That notice will bear the date on which it is published and will specify that the decision will come into force and may then be implemented, on the expiry of five clear working days after the publication of the decision, unless an Overview and Scrutiny Committee objects to it and calls it in for review.
- (c) During that period the Monitoring Officer shall call-in a decision for scrutiny by an Overview and Scrutiny Committee if so requested in the specified format by any three Members of that Overview and Scrutiny Committee and the Chair and, shall then notify the decision taker of the call-in. He/she shall call a meeting of that Overview and Scrutiny Committee on such a date as he/she may determine, where possible after consultation with the Chair of that Overview and Scrutiny Committee, and in any case within five clear working days of the decision to call-in (only in exceptional circumstances will the Chair of the Overview and Scrutiny Committee consider extending this time limit).
- (d) If, having considered the decision, the Overview and Scrutiny Committee is still concerned about it, then it may refer it back to the decision making person or body for reconsideration, setting out in writing the nature of its concerns. If referred to the decision maker they shall then reconsider within a further seven clear working days, amending the decision or not, before adopting a final decision.

- (e) If following an objection to the decision, an Overview and Scrutiny Committee does not meet within the period set out above, or does meet but does not refer the matter back to the decision making person or body, the decision shall take effect on the date of the Overview and Scrutiny Committee meeting, or the expiry of that further five working day period (or agreed extended period), whichever is the earlier.
- (f) In order to ensure that call-in is not abused, nor causes unreasonable delay, certain limitations are to be placed on its use. These are:
  - (i) an Overview and Scrutiny Committee may only call-in a total of six decisions per year;
  - (ii) three Members of the Overview and Scrutiny Committee and a Chair are needed for a decision to be called in:
  - (iii) once a Member has signed a request for call-in under this paragraph, he/she may not do so again until the period of three months has expired;
  - (iv) No Education Co-opted Members may request a decision be called in.
- (g) The Monitoring Officer may veto any request for call-in if it falls outside the remit of this scheme and/or:
  - (i) It is not clear which decision is being called-in;
  - (ii) The decision is exempt from call-in on account of the urgency provisions as outlined below;
  - (iii) The call-in request provides too little information to enable the Scrutiny Committee members or the decision maker to adequately prepare for the call-in meeting;
  - (iv) The decision being called-in, or broadly the same decision, has been called in during the last six months;
  - (v) The call-in request is trivial or without substantial merit.
  - (j) Save in exceptional circumstances all Members requesting a matter be called in must attend the meeting at which the matter is being considered.

#### 7.23.2 Call-In and Urgency

(a) The call-in procedure set out above shall not apply where the decision being taken by the Cabinet, Cabinet Committee, Cabinet Member or Officer is urgent and not subject to call-in. A decision will be urgent and not subject to call-in if any delay likely to be caused by the call-in process would, for example, seriously prejudice the Council's or other public interests. The record of the decision, and notice by which it is made public, shall state whether in the opinion of the decision making person or body, the decision is an urgent one, and therefore not subject to call-in. Consultation will be sought of all Chairs of the Overview and Scrutiny Committees to agree that the decision proposed is reasonable in all the circumstances and to it being treated as a matter of urgency. The agreement of one Chair is sufficient for

the matter to be treated as urgent. In the absence of all Chairs the Mayor or the Deputy Mayor's consent shall be required. In the absence of both, the Chief Executive or his/her nominee's consent should be required. Decisions taken as a matter of urgency must be reported at the next available meeting of the Council, together with the reasons for urgency.

(b) The operation of the provisions relating to call-in and urgency shall be monitored annually and a report submitted to Council with proposals for review if necessary.

### 7.24 The Party Whip

If a Member of an Overview and Scrutiny Committee is subject to a party whip in respect of an issue to be considered by it, that Member must declare the existence of the whip and the nature of it before the commencement of deliberations on the matter. The declaration, and the detail of the whipping arrangements, shall be recorded in the minutes of the meeting.

#### 7.25 Procedure at Overview and Scrutiny Committee Meetings

- 7.25.1 An Overview and Scrutiny Committee shall consider the following business:
  - (a) minutes of the last meeting;
  - (b) declarations of interest;
  - (c) consideration of any matter referred to that Overview and Scrutiny Committee for a decision in relation to call in of a decision;
  - (d) responses of the Cabinet to reports of that Overview and Scrutiny Committee:
  - (e) the business otherwise set out on the agenda for the meeting.

The Rules of Procedure at an Overview and Scrutiny Committee will be the same as the Council Procedure Rules except that the Chair of the meeting may allow the rules of debate to be relaxed to enable a full contribution by those attending the meeting whether as Members of an Overview and Scrutiny Committee or in any other capacity which allows them to contribute to the worth of the meeting.

- 7.25.2 An Overview and Scrutiny Committee may ask people to attend to give evidence or answer questions about any items on their agenda. Meetings should be conducted in accordance with the following principles:
  - that the business be conducted fairly and all Members of the Overview and Scrutiny Committee be given the opportunity to ask questions of attendees, and to contribute and speak;
  - (b) that those assisting by giving evidence be treated with respect and courtesy;
  - (c) that the business be conducted as efficiently as possible.
- 7.25.3 Following any investigation or review, an Overview and Scrutiny Committee shall prepare a report, for submission to the Cabinet and/or Council as appropriate and shall make its report and findings public.

## 7.26 Matters within the Remit of more than one Overview and Scrutiny Committee

Where a matter for consideration by an Overview and Scrutiny Committee also falls within the remit of one or more other Committees, the decision as to which Committee will consider it will be resolved by the respective Chairs or, if they fail to agree, the Chair of the Corporate Overview and Scrutiny Committee.

#### 8. THE STANDARDS COMMITTEE

#### **Composition**

## 8.1 Membership

The Standards Committee is composed of seven Members. Its Membership includes:

- 8.1.1 Four "independent" Members, who are not either a Councillor or an Officer or the spouse of a Councillor or an Officer of this Council or any other relevant Authority as defined by the Local Government Act 2000, appointed in accordance with the procedure set out in the Standards Committees (Wales) Regulations 2001 (as amended);
- 8.1.2 Two Councillors other than the Leader and not more than one Member of the Executive; and
- 8.1.3 One Community Council Member.

#### 8.2 Term of Office

- 8.2.1 Independent Members are appointed for a period of not less than four and not more than six years and may be reappointed for a consecutive term.
- 8.2.2 Members of local authorities who are Members of the Standards Committee will have a term of office until the next ordinary local government election following their appointment. They may be reappointed for one further consecutive term.

#### 8.3 Quorum

A meeting of the Standards Committee shall only be quorate when:

- 8.3.1 at least three Members, including the Chairperson, are present; and
- 8.3.2 at least half the Members present (including the Chairperson) are Independent Members.

#### 8.4 **Voting**

Independent Members and Community Council Members will be entitled to vote at meetings.

#### 8.5 Chairing the Committee

- 8.5.1 Only an Independent Member of the Standards Committee may be the Chair.
- 8.5.2 The Chair and Vice Chair will be elected by the Members of the Standards Committee for whichever is the shortest period of:
  - (i) not less than four years or no more than six years; or
  - (ii) until the term of office of the Independent Member comes to an end.

#### 8.6 Role and Function

The Standards Committee will have the following roles and functions:

- 8.6.1 promoting and maintaining high standards of conduct by Councillors and coopted Members of the Authority;
- 8.6.2 assisting the Councillors and co-opted Members to observe the Members' Code of Conduct:
- 8.6.3 advising the Council on the adoption or revision of the Members' Code of Conduct;
- 8.6.4 monitoring the operation of the Members' Code of Conduct;
- 8.6.5 advising, training or arranging to train Councillors and co-opted Members on matters relating to the Members' Code of Conduct;
- 8.6.6 to monitor the operation of the Council's Whistleblowing Policy;
- 8.6.7 to grant dispensations from the prohibitions contained in the Members' Code of Conduct in accordance with the Regulations;
- 8.6.8 Consider reports submitted by the Public Ombudsman for Wales and the Council's Monitoring Officer;
- 8.6.9 monitoring compliance by leaders of political groups on the Council with their duties to promote and maintain high standards of conduct by the members of the group;
- 8.6.10 Advising, training or arranging to train leaders of political groups on the Council about matters as outlined in 8.6.9.

#### 8.7 Annual Report

- 8.7.1 As soon as possible after the end of each financial year, the Standards Committee must make an annual report to the Council.
- 8.7.2 The annual report must include: -
  - (a) a description of how the Standards Committee has discharged its functions;
  - (b) a summary of any reports and recommendations that were referred to the Standards Committee under Chapter 3 of Part 3 of the Local Government Act 2000:
  - (c) a summary of the actions that the Standards Committee has taken following consideration of the reports and recommendations referred to in (b) above;
  - (d) a summary of any notices that were given to the Standards Committee under Chapter 4 of Part 4 of the Local Government Act 2000; and
  - (e) the Standards Committee's assessment of the extent to which leaders of political groups on the Council have complied with their duties to promote and maintain high standards of conduct by members of their group and to

- cooperate with the Standards Committee in the exercise of the Standards Committee's functions; and
- (f) any recommendations which the Standards Committee considers it appropriate to make to the Council about any matter which falls within the Committee's functions.

## 8.8 Work Programme

The Committee will prepare a work programme, which will be reviewed and approved by the Committee.

## 8.9 **Joint Standards Committee**

The Council may establish a joint standards committee with one or more local authorities for the purpose of discharging the statutory functions of the Standards Committee.

### 8.10 Rules of Procedure and Debate

- 8.10.1 The Council Procedure Rules at Section 4 will apply to the meetings of the Standards Committee Members.
- 8.10.2 When considering the conduct of individual Councillors, the procedures outlined in Appendix 3 to Section **Error! Reference source not found.** will apply.

### 9. **REGULATORY COMMITTEES**

## 9.1 Regulatory and Other Committees

The Council will appoint the Committees to discharge the functions set out in Section 14 of this Constitution.

#### 9.2 The Governance and Audit Committee

- 9.2.1 The Council will appoint a Governance and Audit Committee to discharge the functions described in Section 14 of this Constitution and in accordance with sections 81-87 of The Measure.
- 9.2.2 The Committee shall comprise of two-thirds Councillor Members and one-third members who are not a Member of the Council (lay members). No more than one member of the Committee may be a member of the Cabinet (which Cabinet Member must not be the Leader) or an Assistant to the Executive, although the Committee may have no Cabinet Members or Assistants to the Executive among its membership.
- 9.2.3 The Council shall appoint Members to the Audit Committee in accordance with the political balance rules.
- 9.2.4 The Chair of the Audit Committee and Deputy Chair are appointed by it. The person appointed Chair of the Audit Committee must be a lay member. The person appointed as Deputy Chair cannot be a Member of the Cabinet or an Assistant to the Executive.
- 9.2.5 A meeting of the Audit Committee is to be chaired:
  - (a) by the Chair of the Audit Committee, or
  - (b) if the Chair of the Audit Committee is absent, by the Deputy Chair.
  - (c) If both the Chair of the Audit Committee and the Deputy Chair are absent, the Committee may appoint a member of the Committee who is not a member of the Council's Cabinet, or an Assistant to the Cabinet, to chair the meeting.
- 9.2.6 Members of the Audit Committee may vote on any matter from the Committee.

#### 9.3 The Democratic Services Committee

- 9.3.1 The Council will appoint a Democratic Services Committee to discharge the functions described in Section 14 of this Constitution.
- 9.3.2 The Committee shall comprise of Councillor Members but no more than one Member of the Cabinet (which Cabinet Member must not be the Leaders).
- 9.3.3 The Council shall appoint Members to the Democratic Services Committee in accordance with the political balance rules.
- 9.3.4 The Chair of the Democratic Services Committee is appointed by Full Council and must not be the Cabinet Member.

## 9.4 Other Committees and Sub-Committees

- 9.4.1 The Council will appoint such other Committees as it considers appropriate to the exercise of its functions. These will include a Development Control Committee and Licensing Committee.
- 9.4.2 Any Committee appointed by the Council may at any time appoint additional Sub-Committees and panels throughout the year. The terms of reference and delegation of powers to them shall be explicit and within the appointment Committees terms of reference.

## 9.5 Rules of Procedure and Debate

- 9.5.1 The Council Procedure Rules in Section 4 will apply, except as set out in paragraph 9.5.2 below.
- 9.5.2 A meeting of the Development Control Committee shall only be quorate when at least half of the total number of members of the Committee, rounded to the nearest whole number, are present.

#### 10. **JOINT COMMITTEES**

#### 10.1 Introduction

There are a number of circumstances where the Council or the Cabinet is entitled to carry out certain functions jointly with another local authority.

#### 10.2 **Arrangements to Promote Wellbeing**

The Cabinet in order to promote the economic, social, or environmental wellbeing of its area may:

- 10.2.1 enter into arrangements or agreements with any person or body;
- 10.2.2 co-operate with, or facilitate or co-ordinate the activities of any person or body; and
- 10.2.3 exercise on behalf of that person or body any functions of that person or body.

#### 10.3 **Joint Arrangements**

- 10.3.1 The Council may establish joint arrangements with one or more local authorities and/or their executives to exercise functions which are not Executive Functions in any of the participating authorities, or advise the Council. Such arrangements may involve the appointment of a joint committee with these other local authorities.
- 10.3.2 The Cabinet may establish joint arrangements with one or more local authorities to exercise functions which are Executive Functions. Such arrangements may involve the appointment of joint committees with these other local authorities. Except as set out below, or as permitted or required by Law, the Cabinet may only appoint Cabinet Members to such joint committees and those Members need not reflect the political composition of the Council as a whole.
- 10.3.3 The Cabinet may appoint Members to a Joint Committee from outside the Cabinet where the Joint Committee has functions for only part of the area of the Council and that area is smaller than two fifths of that local authority, by area or population. In such cases, the Cabinet may appoint to the Joint Committee any Councillor who is Member for an electoral division contained within the area. Political balance requirements do not apply to such appointments.
- 10.3.4 When considering whether to establish a joint committee, the Council and the Cabinet shall have regard to any guidance issued by the Welsh Ministers about establishing joint committees and the circumstances in which it is appropriate to do so.

#### 10.4 Access to Information

10.4.1 The Access to Information Procedure Rules in Section 15 apply.

- 10.4.2 If all the Members of a Joint Committee are Members of the Cabinet in each of the participating authorities, then its access to information regime is the same as that applied to the Cabinet.
- 10.4.3 If the Joint Committee contains Members who are not on the Cabinet of any participating authority, then the Access to Information Rules in part VA of the Local Government Act 1972 (as amended) will apply.

### 10.5 <u>Delegation to and from Other Local Authorities</u>

- 10.5.1 The Council can delegate Non-Executive Functions to another local authority or, where those functions are the responsibility of the executive of another local authority, to that executive.
- 10.5.2 The Cabinet can delegate Executive Functions to another local authority or the executive of another local authority in certain circumstances.
- 10.5.3 The decision whether or not to accept such a delegation from another local authority is reserved to the Full Council.

#### 10.6 Contracting Out

The Council (in respect of Non-Executive Functions) and the Cabinet (in respect of Executive Functions) may contract out to another body or organisation functions:

- 10.6.1 which may be exercised by an Officer and which are subject to an order under section 70 of the Deregulation and Contracting Out Act 1994; or
- 10.6.2 under contracting arrangements where the Contractor acts as the Council's Agent under usual contracting principles, provided there is no delegation of the Council's discretionary decision making.

### 11. CORPORATE JOINT COMMITTEES

#### 11.1 Introduction

Corporate joint committees may be established: -

- 11.1.1 at the instigation of the Welsh Ministers, to undertake the Council's transport, strategic planning, economic development and/or improving education functions; or
- 11.1.2 at the instigation of the Council, to undertake any of the Council's functions.

#### 11.2 Corporate Joint Committees

- 11.2.1 The Council may, jointly with one or more other local authorities make a joint committee application to the Welsh Ministers, asking them to consider making regulations under section 72 of the Local Government and Elections (Wales) Act 2021 establishing a corporate joint committee to exercise a function of those authorities or the economic wellbeing function in relation to the principal areas of those authorities.
- 11.2.2 The Council shall not make such an application unless and until it has consulted: -
  - (a) local people in the Council's area;
  - (b) community councils in the Council's area;
  - (c) the Public Services Board;
  - (d) every trade union which is recognised (within the meaning of the Trade Union and Labour Relations (Consolidation) Act 1992) by the Council; and
  - (e) such other persons as the Council considers appropriate.
- 11.2.3 The Council may, jointly with all other local authorities for the principal areas in a corporate joint committee's area, make an application to the Welsh Ministers asking them to consider making regulations under section 80 of the Local Government and Elections (Wales) Act 2021 to amend or revoke joint committee regulations which establish a corporate joint committee of which the Council is a member.
- 11.2.4 The Council may not exercise the power in section 11.2.3 to ask the Welsh Ministers to amend corporate joint committee regulations so as to omit or modify a function which relates to the Council's transport, strategic planning, economic development or improving education functions.

## 12. **OFFICERS**

## 12.1 Management Structure

## 12.1.1 General

The Full Council may engage such staff (referred to as Officers) as it considers necessary to carry out its functions.

## 12.1.2 Chief Officers

The Full Council will engage persons for the following posts, who will be designated Chief Officers, which designation includes persons acting temporarily in such capacity:

Post	Functions and Areas of Responsibility	
Chief Executive	Overall corporate management and operational responsibility including overall management responsibility for all Officers.	
	Principal adviser to the Council on general policy.	
	The provision of professional and impartial advice to all parties in the decision making process to the Cabinet, to Overview and Scrutiny Committees, the Full Council and other Committees).	
	Together with the Monitoring Officer, responsibility for a system of record keeping for all the Authority's decisions (executive or otherwise).	
	Representing the Council on partnership and external bodies (as required by statute or the Council).	
	Service to the whole Council, on a politically neutral basis.	
Corporate Director Communities	Management and overall responsibility for taking the lead role in strategy and policy development in respect of and ensuring effective service co-ordination of the following services: regeneration and development, neighbourhood services.	
Corporate Director Education and Family Support	Management and overall responsibility for taking the lead role in strategy and policy development in respect of and ensuring effective service co-ordination of the following services: family support, learning.	
Corporate Director Social Services and Wellbeing and	Management and overall responsibility for taking the lead role in strategy and policy development in respect of and ensuring effective co-ordination of the following services: wellbeing, safeguarding.	

Post	Functions and Areas of Responsibility
Director of Social Services	
Chief Officer - Legal and Regulatory Services, HR and	Legal and Democratic Services, Member Support Services, Democratic Services, Registrars, Shared Regulatory Services, Land Charges.
Corporate Policy	Employee Relations, Organisation and Employee Development, Health and Safety, Equal Opportunities.
	Corporate Policy.

## 12.1.3 Chief Executive, Monitoring Officer and Chief Finance Officer

The Council will designate the following posts as shown:

Post	Designation
Chief Executive	Chief Executive
Chief Officer – Legal and Regulatory Services, Human Resources and Corporate Policy.	Monitoring Officer
Chief Officer – Finance, Performance and Change	Chief Finance Officer / Section 151 Officer
Democratic Services Manager	Head of Democratic Services

Such posts will have the functions described in Sections 12.2 to 12.5.

## 12.2 Functions of the Chief Executive

## 12.2.1 <u>Discharge of Functions by the Council</u>

- (a) Section 4 of the Local Government and Housing Act 1989 imposes a duty on authorities to designate one of their officers as Chief Executive. The Chief Executive will report to Full Council on the manner in which the discharge of the Council's functions is co-ordinated, the number and grade of Staff required for the discharge of functions, the organisation of the authority's staff and the appointment and proper management of the authority's staff.
- (b) The Chief Executive must keep the following matters under review:
  - (i) the manner in which the exercise by the Council of its different functions is co-ordinated;
  - (ii) the council's arrangements in relation to-

- (A) financial planning,
- (B) asset management, and
- (C) risk management;
- (iii) the number and grades of staff required by the Council for the exercise of its functions;
- (iv) the organisation of the Council's staff;
- (v) the appointment of the Council's staff; and
- (vi) the arrangements for the management of the Council's staff (including arrangements for training and development).
- 12.2.2 If the Chief Executive considers it appropriate to do so, they must make a report to the Full Council setting out their approach to these matters. As soon as possible after preparing a report, the Chief Executive must arrange for a report to be sent to each Member of the Council.
- 12.2.3 Acting as one of the Council's Representatives on the Public Services Board

The Chief Executive shall be one of the Council's two representatives at meetings of the Public Services Board.

## 12.2.4 Restrictions on Functions

The Chief Executive may not be the Monitoring Officer or the Head of Democratic Services but may hold the post of Chief Finance Officer if a qualified accountant.

## 12.3 Functions of the Monitoring Officer

These are set out in section 5 of the Local Government and Housing Act 1989 as amended.

#### 12.3.1 Maintaining the Constitution

The Monitoring Officer will maintain an up-to-date version of the Constitution and will ensure that it is widely available for consultation by Members, staff and the public.

#### 12.3.2 Ensuring Lawfulness and Fairness of Decision Making

After consulting with the Chief Executive and Chief Finance Officer, the Monitoring Officer will report to the Full Council or to the Cabinet in relation to any Function if he or she considers that any proposal, decision or omission would give rise to unlawfulness or if any decision or omission has given rise to maladministration. Such a report will have the effect of stopping the proposal or decision being implemented until the report has been considered.

#### 12.3.3 <u>Supporting the Standards Committee</u>

The Monitoring Officer will contribute to the promotion and maintenance of high standards of conduct through provision of support to the Standards Committee.

#### 12.3.4 Receiving Reports

The Monitoring Officer will receive and act on reports made by the Public Services Ombudsman for Wales and decisions of the case tribunals.

### 12.3.5 Conducting Investigations

The Monitoring Officer will conduct investigations into matters referred by the Public Services Ombudsman for Wales and make reports or recommendations in respect of them to the Standards Committee.

#### 12.3.6 Proper Officer for Access to Information

The Monitoring Officer will ensure that Executive decisions, together with the reasons for those decisions and relevant Officer reports and background papers are made publicly available as soon as possible.

# 12.3.7 <u>Advising whether decisions of the Cabinet are within the Budget and Policy</u> Framework

The Monitoring Officer will, in conjunction with the Chief Finance Officer, advise whether decisions of the Cabinet are in accordance with the Budget and Policy Framework.

## 12.3.8 Providing Advice

The Monitoring Officer will provide advice on the scope of powers and authority to take decisions, maladministration, financial impropriety, probity and Budget and Policy Framework issues to the Councillors.

#### 12.3.9 Restrictions on Posts

The Monitoring Officer cannot be the Chief Finance Officer or the Chief Executive.

#### 12.4 Functions of the Chief Finance Officer

These are set out in section 6 of the Local Government and Housing Act 1989.

## 12.4.1 Ensuring Lawfulness and Financial Prudence of Decision Making

After consulting with the Chief Executive and the Monitoring Officer, the Chief Finance Officer will report to the Full Council or to the Cabinet in relation to an Executive Function and the Council's external auditor if he or she considers that any proposal, decision or course of action will involve incurring unlawful expenditure, or is unlawful and is likely to cause a loss or deficiency or if the Council is about to enter an item of account unlawfully.

#### 12.4.2 Administration of Financial Affairs

The Chief Finance Officer will have responsibility for the administration of the financial affairs of the Council.

#### 12.4.3 Contributing to Corporate Management

The Chief Finance Officer will contribute to the corporate management of the Council, in particular through the provision of professional financial advice.

#### 12.4.4 Providing Advice

The Chief Finance Officer will provide advice on the scope of powers and authority to take decisions, maladministration, financial impropriety, probity and Budget and Policy Framework issues to all Councillors and will support and advise Councillors and Officers in their respective roles.

#### 12.4.5 Give Financial Information

The Chief Finance Officer will provide financial information to the media, members of the public and the community.

# 12.4.6 Advising whether Decisions of the Cabinetare within the Budget and Policy Framework

The Chief Finance Officer will, in conjunction with the Monitoring Officer, advise whether decisions of the Cabinet are in accordance with the Budget and Policy Framework.

#### 12.4.7 Restrictions on Posts

- (a) The Chief Finance Officer cannot be the Monitoring Officer or the Head of Democratic Services.
- (b) The Chief Finance Officer must be a member of one or more of the following professional bodies: -
  - (i) the Institute of Chartered Accountants in England and Wales;
  - (ii) the Chartered Association of Certified Accountants:
  - (iii) the Chartered Institute of Public Finance and Accountancy;
  - (iv) the Chartered Institute of Management Accountants; or
  - (v) any other body of accountants established in the United Kingdom and for the time being approved by the Welsh Ministers for this purpose.

#### 12.5 Functions of the Head of Democratic Services

These are set out in section 9 of The Measure. The functions of the Head of Democratic Services are:

- 12.5.1 to provide support and advice to the Authority in relation to its meetings, subject paragraph 12.5.10;
- 12.5.2 to provide support and advice to committees of the Authority (other than the committees mentioned in paragraph 12.5.5) and the members of those committees, subject to paragraph 12.5.10;

- 12.5.3 to provide support and advice to any joint committee which a local authority is responsible for organising and the members of that committee, subject to paragraph 12.5.10;
- 12.5.4 to promote the role of the Authority's Overview and Scrutiny Committee;
- 12.5.5 to provide support and advice to:
  - (a) the Authority's Overview and Scrutiny Committee and the members of those Committees; and
  - (b) the Authority's Democratic Services Committee and the members of that committee;
  - (c) to provide support and advice in relation to the functions of the Authority's Overview and Scrutiny Committee to each of the following:
    - (i) members of the Authority;
    - (ii) members of the executive of the Authority;
    - (iii) officers of the Authority;
- 12.5.6 to provide support and advice to each member of the Authority in carrying out the role of member of the Authority, subject to paragraph 12.5.11;
- 12.5.7 to make reports and recommendations in respect of any of the following:
  - (a) the number and grades of staff required to discharge democratic services functions;
  - (b) the appointment of staff to discharge democratic services functions;
  - (c) the organisation and proper management of staff discharging democratic services functions;
- 12.5.8 such other functions as may be prescribed by law.
- 12.5.9 Restrictions on Posts

The Head of Democratic Services cannot be the Chief Executive, or the Chief Finance Officer.

- 12.5.10 The function of providing advice about whether or how the Authority's functions should be, or should have been exercised, only applies to advice concerning the functions of the Overview and Scrutiny Committees and Democratic Services Committee.
- 12.5.11 Advice to a Member does not include advice in connection with their role as an executive Member and does not include advice about a matter being or to be considered at a meeting (other than a meeting of an Overview and Scrutiny Committee or Democratic Services Committee).

# 12.6 <u>Duty to Provide Sufficient Resources to the Chief Executive, Monitoring Officer, Chief Finance Officer and Head of Democratic Services</u>

The Council will provide the Chief Executive, the Monitoring Officer, the Chief Finance Officer and the Head of Democratic Services with such Officers, accommodation and other resources as are in their opinion sufficient to allow their duties to be performed.

#### 12.7 **Conduct**

Officers will comply with the Officers' Code of Conduct and the Protocol on Officer/Member Relations set out in this Constitution.

#### 12.8 **Employment**

The recruitment, selection and dismissal of Officers will comply with the Officer Employment Rules set out below.

## 12.9 Officer Employment Procedure Rules

## 1. Interpretation

(1) "Chief Officer" means:

Chief Executive

Corporate Director – Education and Family Support

Corporate Director - Communities;

Corporate Director – Social Services and Wellbeing;

Solicitor to the Council and Monitoring Officer;

Chief Finance Officer / Section 151 Officer.

#### 2. Declarations

- (1) The Authority will draw up a statement requiring any candidate for appointment as an officer to state in writing whether they are the parent, grandparent, partner, child, stepchild, adopted child, grandchild, brother, sister, uncle, aunt, nephew or niece of an existing Councillor or officer of the Authority; or of the partner of such persons;
  - (a) No candidate so related to a Councillor or an officer will be appointed without the authority of the relevant Chief Officer or an officer nominated by him/her.
- (2) Seeking support for appointment: -
  - (a) Subject to sub-paragraph (c) below, the Authority will disqualify any applicant who directly or indirectly seeks the support of any Councillor for any appointment with the Authority. The content of this sub-paragraph will be included in any recruitment information;
  - (b) Subject to sub-paragraph (c) below, no Councillor will seek support for any person for any appointment with the Authority;
  - (c) Nothing in sub-paragraphs (a) and (b) above will preclude a Councillor from giving a written reference for a candidate for submission with an application for appointment.

#### 3. Appointments

- (1) The Authority must take the steps set out in sub-paragraph (2) where it proposes to appoint a Chief Officer, and the remuneration which it proposes to pay to the Chief Officer is £100,000 or more per annum;
- (2) The steps are to:
  - (a) draw up a statement specifying-
    - (i) the duties of the officer concerned, and
    - (ii) any qualifications or qualities to be sought in the person to be appointed;
  - (b) make arrangements for the post to be publicly advertised in such a way as is likely to bring it to the attention of persons who are qualified to apply for it; and
  - (c) make arrangements for a copy of the statement mentioned in paragraph (a) to be sent to any person on request.
- (3) Where sub-paragraph (1) applies, the Authority is not required to take the steps set out in sub-paragraph (2) (b) if it proposes to appoint the chief officer for a period of no longer than 12 months.
- (4) (a) Where a post has been advertised as provided in paragraph 2(b), the Authority must-
  - (i) interview all qualified applicants for the post, or
  - (ii) select a short list of such qualified applicants and interview those included on the short list.
  - (b) Where no qualified person has applied, or if the Authority decide to re-advertise the appointment, the Authority may make further arrangements for advertisement in accordance with paragraph 2(b).

#### 4. Appointment of Chief Officers and Heads of Service

- (1) The Authority will appoint Chief Officers, Heads and Service and the Head of Democratic Services.
- Where a committee, sub-committee or officer is discharging, on behalf of the Authority, the function of the appointment of an officer designated as the Chief Executive, the Authority must approve that appointment before an offer of appointment is made. The power to approve the appointment of the Chief Executive must be exercised by the Authority itself and accordingly section 101 of the Local Government Act 1972 (arrangements for discharge of functions by local authorities) does not apply to the exercise of that power.
- (3) Where a committee or a sub-committee of the Authority is discharging, on behalf of the authority, the function of the appointment of a Chief Officer, a Deputy Chief Officer, Head of Service or Head of Democratic Services
  - (a) at least one member of the executive must be a member of that committee or subcommittee; and
  - (b) not more than half of the members of that committee or sub-committee are to be members of the executive of the Authority.

(4) The Authority must determine the level, and any change in the level, of the remuneration to be paid to a Chief Officer.

## 5. Appointment of Assistants to Political Groups

The appointment of assistants to political groups shall be made in accordance with the wishes of the political groups concerned.

#### 6. Appointment of Officers other than Chief Officers and Heads of Service

The function of appointment of members of staff of the Authority, other than Chief Officers, Deputy Chief Officers, Heads of Service and the Head of Democratic Services, must be discharged, on behalf of the Authority, by the officer designated as the Chief Executive or by an officer nominated by the Chief Executive.

Nothing in the above paragraph prevents a person from serving as a member of any committee or sub-committee established by the Authority to consider an appeal by another person against any decision relating to the appointment of that other person as a member of staff of the Authority.

## 7. Disciplinary Action and Dismissal

#### (1) Chief Officers and Heads of Service

- (a) Disciplinary action against and dismissal of Chief Officers, Deputy Chief Officers, Heads of Service, and Head of Democratic Services must be discharged by the Authority.
- (b) Where a committee, sub-committee or officer is discharging, on behalf of the Authority, the function of the dismissal of an officer designated as the Chief Executive, the Authority must approve that dismissal before a notice of dismissal is given. The power to approve the dismissal of the Chief Executive must be exercised by the Authority itself and accordingly section 101 of the Local Government Act 1972 (arrangements for discharge of functions by local authorities) does not apply to the exercise of that power.
- (c) Where a committee or a sub-committee of the Authority is discharging, on behalf of the authority, the function of the dismissal of Chief Officers, Deputy Chief Officer, Heads of Service or Head of Democratic Services-
  - (i) at least one member of the executive must be a member of that committee or sub-committee; and
  - (ii) not more than half of the members of that committee or sub-committee are to be members of the executive of the Authority.

#### (2) Chief Executive, Monitoring Officer, Chief Finance Officer

- (a) No disciplinary action (other than action to which paragraph (c) applies) in respect of the Chief Executive, or its Monitoring Officer, its Chief Finance Officer, its Head of Democratic Services or any other officer referred to in paragraph (b) may be taken by the Authority, or by a committee, a sub-committee, a joint committee on which the Authority is represented or any other person acting on behalf of the Authority, other than in accordance with a recommendation in a report made by a designated independent person under regulation 9 of the Local Authorities (Standing Orders) (Wales) Regulations 2006 (investigation of alleged misconduct).
- (b) An officer in relation to whom disciplinary action is proposed where-

- (i) the officer was, but at the time of the proposed disciplinary action no longer is, an officer referred to in (a) and;
- (ii) the alleged misconduct or, as the case may be, the reason for the proposal for dismissal, occurred during the period when the officer was an officer referred to in paragraph (a).
- (c) The action to which this paragraph applies is suspension of the officer for the purpose of investigating the alleged misconduct occasioning the action; provided such suspension is on full pay and terminates no later than the expiry of two months beginning on the day on which the suspension takes effect.

#### (3) Officers other than Chief Officers, Heads of Service and Head of Democratic Services

The function of dismissal of, and taking disciplinary action against a member of staff of the Authority other than Chief Officers, Deputy Chief Officers, Heads of Service and the Head of Democratic Services must be discharged, on behalf of the Authority, by the officer designated as the Chief Executive or by an officer nominated by the Chief Executive.

- (b) Paragraph (a) does not apply to the dismissal of, or disciplinary action against, an officer in relation to whom disciplinary action is proposed-
- (i) who was, but at the time of the proposed disciplinary action no longer is, a Chief Officer, Deputy Chief Officer, Head of Democratic Services, and (ii) where the alleged misconduct, or, as the case may be, the reason for the proposal for dismissal occurred during the period when the officer was an officer referred to in (i).
- (c) Nothing in paragraph (a) prevents a person from serving as a member of any committee or sub-committee established by the relevant Authority to consider an appeal by a member of staff of the relevant Authority against any decision relating to the dismissal of, or taking disciplinary action against, that member of staff.

# 8. Chief Executive, Monitoring Officer, Chief Finance Officer, Head of Democratic Services - Investigation of alleged misconduct

- (1) Where it appears to the Authority that an allegation of misconduct which may lead to disciplinary action has been made against a relevant officer, the authority must appoint a committee ("an investigation committee") to consider the alleged misconduct. A 'relevant officer' for this purpose is—
  - (a) the Chief Executive;
  - (b) its Monitoring Officer; or
  - (c) its Chief Finance Officer,
  - (d) its Head of Democratic Services or
  - (e) an officer who was, but at the time the investigation committee is appointed no longer is, an officer referred to in sub-paragraphs (a) to (d) where the alleged misconduct occurred during the period when the officer was an officer referred to in those sub-paragraphs.
- (2) The investigation committee must:

- (a) consist of a minimum of 3 members of the Authority;
- (b) be politically balanced in accordance with section 15 of the Local Government and Housing Act 1989 Act; and must, within 1 month of its appointment, consider the allegation of misconduct and decide whether it should be further investigated.
- (3) For the purpose of considering the allegation of misconduct, the investigation committee:
  - (a) may make such enquiries of the relevant officer or any other person it considers appropriate;
  - (b) may request the relevant officer or any other person it considers appropriate to provide it with such information, explanation or documents as it considers necessary within a specified time limit; and
  - (c) may receive written or oral representations from the relevant officer or any other person it considers appropriate.
  - (4) Where it appears to the investigation committee that an allegation of misconduct by the relevant officer should be further investigated, it must appoint a person ("the designated independent person") for the purpose of paragraph 7(2)(a).
  - (5) The designated independent person who is appointed-
    - (a) must be such person as may be agreed between the Authority and the relevant officer within 1 month of the date on which the requirement to appoint the designated independent person arose; or
    - (b) where there is no such agreement, must be such person as is nominated for the purpose by the Welsh Ministers.
  - (6) The designated independent person-
    - (a) may direct-
      - (i) that the Authority terminate any suspension of the relevant officer;
      - (ii) that any such suspension is to continue after the expiry of the period referred to in paragraph 7 (2) (b);
      - (iii) that the terms on which any such suspension has taken place are to be varied in accordance with the direction; or
      - (iv) that no steps (whether by the Authority or any committee, sub-committee or officer acting on behalf of the authority) by way of disciplinary action or further disciplinary action against the relevant officer, other than steps taken in the presence, or with the agreement, of the designated independent person, are to be taken before a report is made under sub-paragraph (d);
    - (b) may inspect any documents relating to the conduct of the relevant officer which are in the possession of the authority, or which the Authority has power to authorise the designated independent person to inspect;
    - (c) may require any member or member of staff of the Authority to answer questions concerning the conduct of the relevant officer;

- (d) must make a report to the Authority-
  - (i) stating an opinion as to whether (and, if so, the extent to which) the evidence obtained supports any allegation of misconduct against the relevant officer; and
  - (ii) recommending any disciplinary action which appears appropriate for the Authority to take against the relevant officer, and
- (e) must no later than the time at which the report is made under sub-paragraph d), send a copy of the report to the relevant officer.
- (7) Subject to paragraph (8), the relevant officer and the Authority must, after consulting the designated independent person, attempt to agree a timetable within which the designated independent person is to undertake the investigation.
- (8) Where there is no agreement under paragraph (7), the designated independent person must set a timetable as that person considers appropriate within which the investigation is to be undertaken.
- (9) The Authority must consider the report prepared under paragraph (6) (d) within 1 month of receipt of that report.
- (10) The Authority must pay reasonable remuneration to a designated independent person appointed by the investigation committee and any costs incurred by, or in connection with, the discharge of functions under this regulation.

### 13. FINANCE CONTRACTS AND LEGAL MATTERS

#### 13.1 Financial Management

The management of the Council's financial affairs will be conducted in accordance with the financial rules set out in Section 0 of this Constitution.

#### 13.2 Contracts

Every contract made by the Council will comply with the Contracts Procedure Rules set out in Section 0 of this Constitution.

#### 13.3 **Legal Proceedings**

- 13.3.1 The Monitoring Officer is authorised to institute, defend, participate in or settle any legal proceedings and take all necessary steps in any case where such action is necessary to give effect to decisions of the Council or in any case where the Monitoring Officer considers that such action is necessary to protect the Council's interests.
- 13.3.2 The Monitoring Officer has delegated powers to authorise Officers to appear in court on the Council's behalf.

#### 13.4 Authentication of Documents

- 13.4.1 Where any document is necessary to any legal procedure or proceedings on behalf of the Council, it will be signed by the Monitoring Officer or other person authorised by him/her, unless any enactment or otherwise authorises or requires, or the Council has given requisite authority to some other person.
- 13.4.2 Every contract not exceeding £500,000 in amount or value shall be made in writing in a form approved by the Monitoring Officer, such form being the Form of Tender. Subject to the Contracts Procedure Rules, such contracts must be signed by the Appropriate Chief Officer or by an officer in writing authorised to approve on behalf of the Appropriate Chief Officer.
- 13.4.3 In addition to any other person who may be authorised by resolution of the Council, the Proper Officer for the purposes of authentication of documents under the Local Government Acts shall be:
  - (a) the Chief Executive;
  - (b) the Monitoring Officer;
  - (c) any Chief Officer of the Council concerned with the matter to which the document relates; or,
  - (d) any Officer authorised in writing by such Chief Officer.

#### 13.5 Common Seal of the Council

#### 13.5.1 Common Seal

- (a) The Common Seal of the Council shall be kept in a safe place in the custody of the Monitoring Officer.
- (b) A decision of the Council, including decisions under delegated powers, will be sufficient authority for sealing any document necessary to give effect to the decision.

## 13.5.2 <u>Sealing and Execution of Documents</u>

- (a) The Monitoring Officer shall have authority to affix the Common Seal and execute under Seal any deed or document.
- (b) The Monitoring Officer shall have authority to execute any deed or document not required by law to be under seal which is necessary to effect the decisions of the Council.
- (c) This function can be delegated further by the named person.

## 13.5.3 Record of Sealing of Documents

Any entry of the sealing of every deed or document to which the Common Seal has been affixed shall be made by the Monitoring Officer and consecutively numbered in a book to be provided for the purpose.

## 14. RESPONSIBILITY FOR FUNCTIONS - SUMMARY

The purpose of this Section of the Constitution is to set out who is responsible for making the various decisions in the Council.

#### 14.1 Who can be Decision Makers?

Under this constitution, there are a number of different decision makers:

- 14.1.1 Full Council;
- 14.1.2 a Committee or Sub-Committee of the Council;
- 14.1.3 the Executive Leader;
- 14.1.4 the Cabinet;
- 14.1.5 a Committee of Cabinet;
- 14.1.6 an individual Cabinet Member;
- 14.1.7 a Joint Committee:
- 14.1.8 a Corporate Joint Committee;
- 14.1.9 an Officer.

The Council will issue and keep up to date a record of which individual has responsibility for particular types of decisions. This record is set out in Section 14 of this Constitution.

#### 14.2 **Principles of Decision Making**

All decisions of the Council will be made in accordance with the following principles:

- 14.2.1 proportionately (ie the action must be proportionate to the desired outcome);
- 14.2.2 due consideration and the taking of professional advice from Officers;
- 14.2.3 respect for human rights;
- 14.2.4 a presumption in favour of openness;
- 14.2.5 clarity of aims and desired outcomes;
- 14.2.6 consideration of any alternative options; and
- 14.2.7 the giving and recording of reasons for the decision and the proper recording of these reasons.
- 14.3 Functions fall into the following categories:

#### 14.3.1 Functions of the Full Council

The Council is the supreme decision making body and may, with some exceptions, exercise any of the functions vested in the Council by Law. It may also delegate many of those functions to a Committee, Sub-Committee or Officer. The functions of the Full Council are set out in Section 4.

#### 14.3.2 Non-Executive Functions

These are functions which, by law, may not be the responsibility of the Cabinet. In some cases, only the Full Council meeting may take the decision. In other cases, the Council may delegate the responsibility for taking the decision to a Committee or an Officer.

#### 14.3.3 "Local Choice" Functions

There are some functions which the Council may treat as being the responsibility of the Cabinet (in whole or in part) or as being non-executive, at its discretion.

#### 14.3.4 Executive Functions

All other functions are Executive Functions.

#### 14.4 Other Bodies

#### 14.4.1 Advisory Bodies

The Council and/or the Leader can also set up Advisory Committees and Joint Advisory Committees.

#### 14.4.2 Overview and Scrutiny Committees

Overview and Scrutiny Committees are responsible for the overview and scrutiny function. They cannot exercise other functions and make decisions.

#### 14.4.3 Officer Delegations

Officer delegations are published in the Scheme of Delegation of Functions, available on the Council's website.

### 14.5 Who Decides - Non-Executive Functions?

- 14.5.1 The Council may decide whether to delegate Non-Executive Functions to a committee, sub-committee, delegated Officer or joint committee.
- 14.5.2 Where a non-Executive Function has been delegated to a committee, the committee may further delegate to a sub-committee or delegated Officer.
- 14.5.3 Where a non-Executive Function has been delegated to a sub-committee, the sub-committee may further delegate to a delegated Officer.

#### 14.6 Who Decides – Executive Functions?

The Leader may decide whether to delegate Executive Functions to a committee of the Cabinet, an individual Cabinet Member, a delegated Officer or a joint committee.

## 14.7 Removal of Delegation

- 14.7.1 Where a function has been delegated, the body that delegated the function may withdraw the delegation generally or in any particular case, and may exercise the function itself.
- 14.7.2 Where a function has been delegated, the decision maker is not required to exercise the delegation and may refer any particular matter to the body that made the delegation or any other body that has power to exercise the function.

## 14.8 Who May Exercise Officer Delegations?

Where a function has been delegated to an Officer(s) ("delegated Officer(s)"), the decision may be taken in the name of (but not necessarily personally by) such delegated Officer(s) ("authorised Officer(s)") in accordance with arrangements made from time to time by such delegated Officer(s) for this purpose. The Officer with delegated powers can only delegate to a third party if that Officer is given delegated powers to "delegate on" that decision making.

## **APPENDIX 1 TO SECTION 14**

## **Non-Executive Functions**

Member Body	Membership	Functions
Appeals Panel	Chairperson and/or Vice Chairperson together with one or two County Borough Councillors drawn on a rota basis from a panel of ten County Borough Councillors.	To hear and determine appeals under the following policies and procedures of the Council:
Appointments Committee	The Appointments Committee is comprised of 8 Elected Member representatives:  • Leader (Chairperson)  • Deputy Leader  • Cabinet Member (of relevant portfolio to the post)  • 2 x Bridgend County Independent Members  • 2 x Democratic Alliance Members  1 x Labour Member	1. To undertake the appointment process of JNC officers (other than the Chief Executive).  2. To undertake the appointment process of the Chief Executive and recommend to full Council an appointable candidate(s).  Substitution of Appointments Committee Members is permissible but only for the whole of an appointments process.  3. To facilitate the JNC Determination and JNC Appeals panels. These will comprise of 3 members with the Leader or Deputy Leader chairing the panel supported by 2 other members of the committee.  The JNC Panels cannot be substituted and must be comprised from the original membership of the Appointments Committee.

Member Body	Membership	Functions
Governance and Audit Committee	8 County Borough Councillors and 4 Lay Members.	<ul> <li>Review, scrutinise and issue reports and recommendations in relation to the Council's financial affairs,</li> <li>Review, scrutinise and issue reports and recommendations on the appropriateness of the Council's risk management, internal control, arrangements to secure value for money and corporate governance arrangements.</li> <li>To consider the report on the annual risk assessment, any interim reports and the Corporate Risk Management Policy.</li> <li>Oversee the Council's internal and external audit arrangements (including the performance of external providers of Internal Audit) and review its financial statements.</li> <li>To approve the Internal Audit Charter.</li> <li>To approve the risk-based internal audit plan and to approve significant interim changes to the risk-based internal audit plan.</li> <li>To contribute to the Quality Assurance and Improvement Programme and in particular, to the external quality assessment of internal audit that takes place at least once every five years.</li> <li>To consider and approve the Head of Internal Audit's annual report and opinion, a summary of internal audit activity (actual and proposed) and the level of assurance it can give over the Council's corporate governance arrangements.</li> <li>To consider and approve reports from Internal Audit on the adequacy of internal Audit on the adequacy of internal control.</li> </ul>

Member Body	Membership	Functions
		<ul> <li>To consider and approve reports dealing with the management and performance of the providers of internal audit services.</li> <li>To consider reports from Internal Audit on agreed recommendations not implemented within a reasonable timescale and approve necessary remedial action.</li> <li>To identify issues from the Annual Improvement Report by Audit Wales.</li> <li>To identify areas for examination by Internal and External Audit.</li> <li>To be responsible for ensuring effective scrutiny of the Treasury Management function and policies, in accordance with the Treasury Statement and Treasury Management Indicators.</li> <li>To maintain an overview of the Council's Constitution in respect of Contract and Finance Procedure Rules.</li> <li>To review any issue referred to it by the Chief Executive Officer, the Monitoring Officer and the Section 151 Officer.</li> <li>To monitor the Council's Anti-Fraud and Bribery Strategy, Anti-Money Laundering Policy and Anti-Tax Evasion Policy.</li> <li>To review and approve the Council's Annual Governance Statement and Code of Corporate Governance.</li> <li>To assess the Council's compliance with its own and other published standards and controls.</li> <li>To review and approve the Annual Statement of Accounts. Specifically, to consider whether appropriate accounting policies have been followed and whether there are concerns arising from the</li> </ul>

Member Body	Membership	Functions
		financial statements or from the audit that need to be brought to the attention of the Council.  To consider and approve the External Auditor's report in relation to those charged with governance on issues arising from the audit of the accounts.  To review and consider reports from the External Auditor on the Council's performance, financial probity and corporate governance and providing the opportunity for direct discussion with the auditor on these.  To receive reports from the External Regulators as appropriate.  To attend relevant training sessions in accordance with the Members Training programme e.g. Treasury Management.  To review and assess the Council's ability to handle complaints effectively;  To make reports and recommendations in relation to the Council's ability to handle complaints effectively;  To review the Council's draft self-assessment report on its performance and, if deemed necessary, make recommendations for changes to the conclusions;  To receive the Council's self-assessment report in respect of a financial year as soon as reasonably practicable after the end of that financial year;  At least once during the period between two consecutive ordinary elections of councillors to the Council, consider the panel performance assessment report into

Member Body	Membership	Functions
		which the Council is meeting its performance requirements;  To receive and review the Council's draft response to the report of the panel performance assessment and, if deemed necessary, make recommendations for changes to the statements made in the draft response;  To consider the assurance framework, including partnerships and collaboration arrangements;  To support the ethical framework of the Council
Democratic Services Committee	11 County Borough Councillors	<ol> <li>To designate an officer as the Head of Democratic Services,</li> <li>To review the adequacy of provision of staff, accommodation and other resources to discharge democratic services functions, and</li> <li>To make reports and recommendations to Council, at least annually, in relation to such provision.</li> <li>At the request of the Council, review any matter relevant to the support and advice available to members of the Council, and the terms and conditions of office of those members.</li> <li>To make reports and recommendations to the Council following a review.</li> </ol>
Democratic Services Sub- Committee	3 County Borough Councillors that are members of the Committee	A Panel constituted under The Family Absence for Members of Local Authorities (Wales) Regulations 2013  a) Determine a complaint
		made by a Member

Member Body	Membership	Functions
		regarding cancellation of family absence by the authority
		b) The Sub-Committee may confirm a decision made or substitute its own decision as to the Member's entitlement to a period of family absence in accordance with the 2013 Regulations.
		c) Determine a complaint made by a Member on maternity absence or parental absence regarding a decision made by the chair of the authority as to the Member attending any meeting or performing any duty;
		d) The Sub-Committee may confirm the decision of the chair of the authority or substitute its own decision as to the Member attending any meeting or performing any duty;
		e) The decision of the Sub- Committee is final.
Development Control Committee	18 County Borough Councillors	<ol> <li>To make recommendations to the Council in respect of Departure Applications which the Committee are minded to approve and Applications which accord with Council policy which the Committee are minded to refuse;</li> <li>To determine applications for planning permission including applications by the Council for deemed planning permission other than Departure Applications which the</li> </ol>

Member Body	Membership	Functions
		Committee are minded to approve and Applications which accord with Council policy which the Committee are minded to refuse;
		3. To deal with all matters relating to or arising under the regulations for the time being in force governing the control of advertisements;
		4. To authorise the service of notices and the making of orders in accordance with the powers conferred upon the Council as local planning authority by Parts III, IV, VI, VII, VIII and XIII of the Planning (Listed Buildings and Conservation Areas) Act 1990;
		5. To discharge the Council's functions pursuant to Part I, Chapters II, III, IV and VI, Part II, Sections 72-75, Part III and Part IV of the Planning (Listed Buildings and Conservation Areas) Act 1990;
		6. To discharge the Council's functions pursuant to the Planning (Hazardous Substances) Act 1990;
		7. To deal with all matters relating to or arising under the regulations for the time being in force governing European Nature Conservation Sites;
		To discharge the Council's functions under the Building Regulations;
		9. To be responsible for:

Member Body	Membership	Functions
		<ul> <li>a) The making of Tree Preservation Orders;</li> <li>b) The confirmation of Tree Preservation Orders in respect of which there are no subsisting objections or representations;</li> <li>c) The determination of all applications for consent under confirmed Tree Preservation Orders;</li> <li>d) The making of observations on tree felling licences proposed to be granted by National Resources Wales;</li> <li>e) The determination of applications under the Woodland Grant Scheme;</li> <li>10. To approve design briefs and advice notes relating to the control of development;</li> <li>11. To accept tenders for the execution of work, the performance of services or the supply of goods or materials in connection with the exercise of the functions of the Committee;</li> </ul>
		12. To exercise those functions relating to town and country planning and development control specified:  (a) In column (1) of Schedule 1 to the Local Authorities Executive Arrangements (Functions and Responsibilities) (Wales) Regulations 2001; and In Regulations 2(2), 2(3) or 2(4) of those Regulations; which are not specified in paragraphs 1 to 11 above, unless the responsibility for exercising any of those functions has been delegated by the Council

Member Body	Membership	Functions
		to any other committee, sub- committee, panel or other body.
Rights of Way Sub-Committee	6 County Borough Councillors (plus one observer from each: Ramblers' Association; the British Horse Society; and a Footpath Secretary).	<ol> <li>To approve the making of applications for and the making, modification or variation of orders relating to rights of way in pursuance of any provision contained in the following enactment (or any statutory modification, re-enactment or amendment thereof):         Town and Country Planning Act 1990;</li> <li>To confirm, where the Council has power to do so, any proposed Order made in accordance with paragraph 1 above to which there are no objections or in respect of</li> </ol>
		which any objections made are withdrawn;
		3. Where the Council does not have power to determine any proposed Order to refer the proposed Order to such determining body (i.e. the National Assembly for Wales, the Magistrates' Court or the County Court) as is appropriate in the circumstances.
The Licensing Committee	14 County Borough Councillors	To determine from time to time any standard conditions applicable to and detailed policies governing the issue of the following licences, permits and consents; to determine the amount of the fees to be charged from time to time in respect of those licenses, permits and consents; and to determine objections in relation to proposed fee revisions:  a) Hackney Carriage and Private Hire Vehicle Licences, Drivers'

Member Body	Membership	Functions
		Licences and Operators' Licences;
		b) Licences for Pleasure Boats and Vessels;
		c) Street Trading Licences and Consents;
		d) Sex Establishments;
		e) Street Collections;
		f) House to House Collections;
		g) Safety of Sports Grounds Act 1975 and the Fire Safety and Safety of Places of Sports Act 1987.
		h) Section 26(1)(bb) of the Marriage Act 1949, (as amended)
		To prescribe guidelines, conditions, limitations or restrictions governing the grant of applications for Hackney Carriage and Private Hire Vehicle Drivers' Licences by Council officers under the power delegated to them;
		To resolve to designate any street within the County Borough under the street trading provisions contained in Schedule 4 of the Local Government (Miscellaneous Provisions) Act 1982 and to deal with any matter arising there from, including licence and consent issues.

Member Body	Membership	Functions
The Licensing Sub-Committee	Two Panels sitting on a rota basis each consisting of seven county borough councillors, and chaired by the Chairperson and Vice Chairperson of the Licensing Committee	To hear and determine applications and to deal with all detailed matters (including the suspension and revocation of licences) relating to: a. Hackney Carriage and Private Hire Vehicle Licences, Drivers' Licences and Operators' Licences;
		b. Licences for Pleasure Boats and Vessels;
		c. Street Trading Licences and Consents;
		d. Sex Establishments;
		e. Street Collections;
		f. House to House collections;
		To determine any matters in relation to the Drivers Awareness Course for Hackney Carriage and Private Hire Vehicle Drivers, including the making of any charges deemed appropriate.
		To determine all applications in respect of both designated grounds and regulated stands and grounds under the Safety of Sports Grounds Act 1975 and the Fire Safety and Safety of Places of Sports Act 1987.
		To hear and determine applications for the grant or renewal of approvals of premises for the solemnisation of marriages in pursuance of Section 26(1)(bb) of the Marriage Act 1949, (as amended) or to revoke such

Member Body	Membership	Functions
		approvals, in circumstances in which the Council Officer to whom the power to determine such applications, or to revoke such approvals, has been delegated has declined to exercise the delegated power.
The Licensing Act 2003 Committee	14 County Borough Councillors	All matters relating to the discharge by the licensing authority of its licensing functions under the Licensing Act 2003 other than any function conferred by section 5 in respect of its statement of licensing policy;
		All matters relating to the discharge by the Licensing Authority of its licensing functions under the Gambling Act 2005 other than any function conferred by section 349 in respect of its statement of Licensing Policy and section 166 in respect of a resolution not to issue a casino license.
The Licensing Act 2003 Sub- Committee(s)	Ad-hoc panels of 3 Licensing Act 2003 Committee members sitting on a rota basis chaired by the Chairperson or Vice- Chairperson of the Licensing Act 2003 Committee or in their absence a member of the panel to be elected as	To hear and determine or make decisions (as appropriate) under the Licensing Act 2003 regarding the following matters:
		(a) Applications for personal licences (if police objection):
	chairperson for that meeting	(b) Applications for personal licences with unspent convictions:
		<ul><li>(c) Applications for premises licences/club premises certificates (if relevant representation made);</li></ul>
		(d) Applications for provisional statements (if relevant representation made);
		(e) Applications to vary premises licences/club premises certificates (if

Member Body	Membership	Functions
		relevant representation made);
		(f) Applications to vary designated premises supervisor (if police objection);
		(g) Applications for transfer of premises licence (if police objection);
		(h) Applications for interim authorities (if police objection);
		(i) Applications to review premises licences/club premises certificate;
		(j) Whether to object when the authority is a consultee and not the relevant authority considering an application;
		(k) Determination of police objections (counter notices) to temporary event notices;
		(I) Revocations of licences where convictions come to light.
		2. To determine applications received in respect of the Legislative Reform (Minor variations to Premises Licences and Club Premises Certificates) Order 2009.
		To determine applications received in respect of the

Member Body	Membership	Functions	
		Legislative Reform (Supervision of Alcohol Sales in Church and Village Halls &c.) Order 2009.	
		4. To hear and determine or make decisions (as appropriate) under the Gambling Act 2005 regarding the following matters:	
		(a) Applications for Premises Licence;	
		(b) Application for variation of Premises License;	
		(c) Application for transfer of Premises Licence;	
		(d) Application for Provisional Statement;	
		(e) Review of Premises License;	
		(f) Application for Club Gaming / Club Machine permit;	
		(g) Cancellation of Club Gaming / Club Machine permit;	
		(h) Cancellation of Licensed Premises Gaming Machine permit;	
		(i) Consideration of temporary use notice;	
		<ul><li>(j) Decision to give counter notice to a temporary use notice;</li></ul>	
		(k) Decision to refuse application for small lottery registration;	
		(I) Decision to disapply s282 - automatic entitlement to gaming machines in licensed premises.	

Member Body Membership		Functions
		Other matters where the Council Officer to whom the power to determine or revoke has been delegated declines to exercise the delegated power.
Town and Community Council Forum	Nineteen County Borough Councillors and one Town and Community Councillor representing each Town and Community Council	To consult with representatives of Town and Community Councils within the County Borough on matters of mutual interest.

Certain functions have been delegated to Officers. Details of those delegations can be found in the Scheme of Delegation of Functions on the Council's website.

The Committees set out above may from time to time add to or alter those delegations, as appropriate, within their functional responsibilities.

#### **APPENDIX 2 TO SECTION 14**

## **Responsibility for Executive Functions**

The Leader is responsible for discharge of Executive Functions and may delegate those functions as set out in Section 5 of this Constitution.

The Leader has allocated portfolios (lead responsibilities) for particular Executive Functions among individual Member of the Cabinet. Normally, decisions will be taken collectively by the Cabinet but individual Cabinet Members can make decisions within their portfolios which they consider to be urgent to protect the interest of the Council, subject to prior consultation with the appropriate Chief Officer(s) and, where necessary, other Cabinet Members. If there is uncertainty as to whether a matter falls within a particular portfolio, the Leader will make a determination as to which is the relevant Cabinet Member. The Leader can make urgent decisions in the absence of the appropriate portfolio holder.

The table below indicates how the Leader has allocated particular Executive Functions by a Committee of Cabinet:

Member Body	Membership	Functions.
Equalities Committee	17 elected members, 7 of whom are Cabinet members and have voting rights.	<ul> <li>To ensure that appropriate equality and Welsh language policies and procedures are in place to fulfil the Authority's statutory duties.</li> <li>To consider and make recommendations to Cabinet on proposals from the Corporate Equality Management Group on the development and implementation of the Authority's Corporate Equality Scheme and Welsh language Scheme.</li> <li>To monitor and review community and employee involvement in the development and implementation of the Corporate Equality Scheme.</li> <li>To request and receive information and statistical monitoring reports about the authority's workforce related to those groups of people protected by anti-discrimination</li> </ul>

		legislation, (including: gender, marital status, age, race, disability, Welsh language, religion, belief and sexual orientation).  To monitor and assess the Authority's performance against the Equality Improvement Framework.  To report annually to Cabinet on the Authority's progress against the objective set out in the Corporate Equality Scheme and Welsh language scheme in line with the Wales Programme for Improvement.  To make recommendations to Cabinet for improvement in performance in fulfilment of the Authority's statutory duties.
Corporate Parenting	12 elected members, 7 of whom are Cabinet members and have voting rights.	<ul> <li>To ensure that Looked After Children are seen as a priority by the whole of the Authority and by the Children and Young People's Partnership.</li> <li>To seek the views of Children and Young People in shaping and influencing the parenting they receive.</li> <li>To determine appropriate policies, opportunities and procedures.</li> <li>To monitor and evaluate the effectiveness of the Authority in its role as corporate parent against Welsh Assembly guidance.</li> </ul>

- To approve the Permanence Policy in relation to Looked After Children.
   To approve individual
- To approve individual Statements of Purpose for the service areas of Fostering, Adoption and Children's Homes.
- To receive information specifically relating to Looked After Children and the Authority's Corporate Parenting Role.
- To approve the Placement Strategy.
- To approve Support Plans and Recruitment Campaigns for Foster Parents.
- To approve Schemes in relation to work experience, training and apprenticeships for Looked After Children and former Looked After Children.
- To approve increases in allowances which can be funded within existing budgets and amendments to the method of assessing financial support relating to Looked After Children.
- To approve Residential Reviews relating to Looked After Children

# **APPENDIX 3 TO SECTION 14**

# **RESPONSIBILITY FOR LOCAL CHOICE FUNCTIONS**

Functions	Member Body	Membership	Delegation of Functions
Section 6 (Coalmining operations: Performance bonds).	Council, through the Development Control Committee	18 County Borough Councillors	Delegated to Corporate Director - Communities (see Schemes of Delegation of Functions).
Section 8 (Council to be trust corporation).	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	Allocated to Corporate Director – Education and Family Support (see Schemes of Delegation of Functions).
Section 9 (Access for fire brigade).	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	Allocated to Corporate Director - Communities (see Schemes of Delegation of Functions).
Section 10 (Dust, etc., from building operations).	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	Allocated to Corporate Director - Communities (see Schemes of Delegation of Functions).
Section 11 (Dust from the movement of coal).	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	Allocated to Corporate Director - Communities (see Schemes of Delegation of Functions).
Section 12 (Cost of weatherproofing walls).	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	Allocated to Corporate Director - Communities (see Schemes of Delegation of Functions).

Functions	Member Body	Membership	Delegation of Functions
Section 13 (Retaining walls).	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	Allocated to Corporate Director - Communities (see Schemes of Delegation of Functions).
Section 14 (Plans, etc., of new sewers).	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	Allocated to Corporate Director - Communities (see Schemes of Delegation of Functions).
Section 15 (Seizure of stray animals).	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	Allocated to Solicitor to the Council (see Schemes of Delegation of Functions).
Section 16 (Expenses of executing demolition orders).	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	Allocated to Corporate Director - Communities (see Schemes of Delegation of Functions).
Section 17 (Prohibition of parking of certain vehicles in residential streets).	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	Allocated to Corporate Director - Communities (see Schemes of Delegation of Functions).
Section 18 (Restriction on use of dustbins, etc.).	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	Allocated to Corporate Director - Communities (see Schemes of Delegation of Functions).
Section 19 (Repair of walls, etc., of yards).	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	Allocated to the Solicitor to the Council(see Schemes of Delegation of Functions).

Functions	Member Body	Membership	Delegation of Functions
Section 22 (Protection of seashore and adjoining land from pollution).	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	Allocated to Corporate Director - Communities (see Schemes of Delegation of Functions).
Section 23 (Power to employ lifeguards).	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	Allocated to Corporate Director – Social Services and Wellbeing (see Schemes of Delegation of Functions).
Part V (Porthcawl Harbour).	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	Allocated to Corporate Director – Communities (see Schemes of Delegation of Functions).
Section 67(1) of, and Schedule 18 to, the School Standards and Framework Act 1998 (Making of arrangements for appeals against exclusion of pupils).	Council	51 County Borough Councillors.	None.
Section 94(1) and (4) of, and Schedule 24 to, the School Standards and Framework Act 1998 (Making of arrangements for admission appeals).	Council	51 County Borough Councillors.	None.
Section 95(2) of, and Schedule 25 to, the School Standards and Framework Act 1998 (Making of arrangements regarding appeals by governing bodies).	Council	51 County Borough Councillors.	None.
Section 20 of the Police Act 1996 (Making of	Council	51 County Borough Councillors.	None.

Functions	Member Body	Membership	Delegation of Functions
arrangements for questions on police matters at Council meetings).			
Paragraphs 2 to 4 of Schedule 2 to the Police Act 1996 (Making of appointments to police authorities).	Council	51 County Borough Councillors.	None.
Any function relating to contaminated land.	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	Allocated to Solicitor to the Council (see Schemes of Delegation of Functions).
Any function relating to the control of pollution or the management of air quality.	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	Allocated to Solicitor to the Council (see Schemes of Delegation of Functions).
The service of an abatement notice in respect of a statutory nuisance.	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	Allocated to Solicitor to the Council(see Schemes of Delegation of Functions).
Passing of resolution that Schedule 2 to the Noise and Statutory Nuisance Act 1993 shall apply to the authority's area.	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	None.
Inspection of the Authority's area to detect any statutory nuisance.	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	Allocated to Solicitor to the Council(see Schemes of Delegation of Functions).
Investigation of a complaint as to the existence of a statutory nuisance.	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	Allocated to Solicitor to the Council(see Schemes of Delegation of Functions).

Functions	Member Body	Membership	Delegation of Functions
Section 330 of the Town and Country Planning act 1990 (Obtaining of information as to interests in land).	Council	51 County Borough Councillors.	Delegated to Corporate Director - Communities (see Schemes of Delegation of Functions).
Section 16 of the Local Government (Miscellaneous Provisions) Act 1976 (Obtaining of particulars of persons interested in land).	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	Allocated to Corporate Director - Communities and the Solicitor to the Council(see Schemes of Delegation of Functions). Functions).
Making of agreements for the execution of highway works.	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	Allocated to Corporate Director - Communities (see Schemes of Delegation of Functions).
The appointment of any individual:  (a) To any office other than an office in which he/she is employed by the Council;  (b) To any body other than:  (i) the Council;  (ii) a joint committee of two or more authorities; or  c) a committee or sub-committee of such a body, and the revocation of any such appointment.	Cabinet (in respect of appointments to offices or bodies discharging or performing any role in connection with functions which are the responsibility of the Cabinet); otherwise the Council	Leader, Deputy Leader and four other County Borough Councillors (Cabinet); otherwise 51 County Borough Councillors (Council).	None.
Power to make payments or provide other benefits in cases of maladministration etc.	Council	51 County Borough Councillors.	None.

Functions	Member Body	Membership	Delegation of Functions
Discharge of any function by an authority acting as a harbour authority.	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	Allocated to Corporate Director – Communities (see Schemes of Delegation of Functions)
Licensing Functions in accordance with: Part 2 of the Licensing Act 2003 except Section 6	Council	51 County Borough Councillors	None
The functions contained in sections 25 and 26 of the Highways Act 1980 (Creation of Highways).	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	Specified functions allocated to the Corporate Director - Communities (see Schemes of Delegation of Functions).
The functions contained in sections 116, 117, 118, 118ZA, 118A, 118B, 118C, 119, 119ZA, 119A, 119B, 119C, 119D, 120, 121B of the Highways Act 1980 (Stopping up and diversion of Highways).	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	Specified functions allocated to the Corporate Director - Communities (see Schemes of Delegation of Functions).
The functions contained in sections 130, 139, 140, 140A, 142, 147, 147ZA, 149, 169, 171, 171A, 172, 173, 178, 179, 180 of the Highways Act 1980 (Lawful and unlawful interference with Highways and Streets).	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	Specified functions allocated to the Corporate Director - Communities (see Schemes of Delegation of Functions).
The functions contained in section 34 of the Wildlife and Countryside Act	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	Specified functions allocated to the Corporate Director - Communities (see Schemes of

Functions	Member Body	Membership	Delegation of Functions
1981 (Limestone pavement orders).			Delegation of Functions).
The functions contained in section 53 of the Wildlife and Countryside Act 1981 (Duty to keep definitive map and statement under continuous review).	Cabinet	Leader, Deputy Leader and four other County Borough Councillors.	Specified functions allocated to the Corporate Director - Communities (see Schemes of Delegation of Functions).
Functions in respect of gambling under sections 29, 30, 166, 212, 284, 304, 346, 349, 350 and Part 5 of Schedule 11 of the Gambling Act 2005.	Council	51 County Borough Councillors.	Specified functions allocated to the Licensing Committee and to the Solicitor to the Council(see Constitution and Schemes of Delegation of Functions).
To determine all appeals against decisions made by or on behalf of the Authority except where there is other specific statutory requirement.	Council	51 County Borough Councillors.	To the Appeals Panel Chairperson and/or Vice Chairperson together with one or two County Borough Councillors drawn on a rota basis from a panel of ten County Borough Councillors.
Functions in respect of the calculation of council tax base in accordance with any of the following—	Council	51 County Borough Councillors	None
(a) the determination of an item for T in section 33(1) and 44(1) of the Local Government Finance Act 1992;			
(b) the determination of an amount for item TP in sections 34(3),			

Functions	Member Body	Membership	Delegation of Functions
45(3) 48(3) and 48(4) of the Local Government Finance Act 1992;			
(c) the determination of an amount required for determining an amount for the item mentioned in paragraph (a) or (b) above.			
The functions set out in the Scrap Metal Dealers Act 2013.	Cabinet	Leader, Deputy leader and four other County Borough Councillors.	Specified functions allocated to the Solicitor to the Council (see Schemes of Delegation of Functions).

#### **SECTION 15**

#### 15. ACCESS TO INFORMATION PROCEDURE RULES

## 15.1 **Scope**

These rules apply to all meetings of the Council, the Cabinet, Overview and Scrutiny Committee, the Standards Committee, and Regulatory Committees.

#### 15.2 Additional Rights to Information

These rules do not affect any more specific rights to information contained elsewhere in this Constitution or the law.

## 15.3 Rights to Attend Meetings

Members of the public may attend all meetings subject only to the exceptions in these rules.

# 15.4 Notices of Meeting

- 15.4.1 Unless a meeting is convened at short notice, the Council will give at least three clear days' notice of any meeting by posting details of the meeting on its website.
- 15.4.2 Where the meeting or part of the meeting is open to the public and is held wholly through remote means, the notice must include details of the time of the meeting, and how to access it.
- 15.4.3 Where the meeting or part of the meeting is open to the public and is held partly through remote means, the notice must include details of the time and place of the meeting and how to access it.
- 15.4.4 Where the meeting is not open to the public and is held partly through remote means or not held through remote means, the notice must include details of the time and place of the meeting and the fact that it is not open to the public.
- 15.4.5 Where the meeting is not open to the public and is held through remote means only, the notice must include details of the time of the meeting, the fact that it is being held through remote means only and the fact that it is not open to the public.

#### 15.5 Access to Agenda and Reports Before the Meeting

The Council will make copies of the agenda and reports open to the public available for inspection at the designated office and on its website at least three clear days before the meeting. If an item is added to the agenda later, the revised agenda (where reports are prepared after the summons has been sent out, the designated Officer shall make each such report available to the public as soon as the report is completed and sent to Councillors) will be open to inspection from the time the item was added to the agenda.

#### 15.6 Supply of Copies

The Council will supply copies of:

15.6.1 any agenda and reports which are open to public inspection;

- 15.6.2 any further statements or particulars necessary to indicate the nature of the items in the agenda; and
- 15.6.3 if the Monitoring Officer thinks fit, copies of any other documents supplied to Councillors in connection with an item

to any person either electronically or on payment of a charge for postage and any other printing and handling costs.

#### 15.7 Access to Minutes etc after the Meeting

The Council will make available either electronic or hard copies of the following for a period of six years after the date of a meeting:

- 15.7.1 the minutes of the meeting or record of decisions taken by the Cabinet, excluding any part of the minutes of proceedings when the meeting was not open to the public because exempt or confidential information was being considered;
- 15.7.2 a summary of any proceedings not open to the public, where the minutes open to inspection would not provide a reasonably fair and coherent record;
- 15.7.3 the agenda for the meeting;
- 15.7.4 reports relating to items when the meeting was open to the public.

#### 15.8 Background Papers

#### 15.8.1 <u>List of Background Papers</u>

The Officer preparing the report will set out in every report a list of those documents (called background papers) relating to the subject matter of the report which in his/her opinion:

- (a) disclose any facts or matters on which the report or an important part of the report is based; and
- (b) which have been relied on to a material extent in preparing the report

but does not include published works or those which disclose exempt or confidential information as defined in Rule 15.10.

#### 15.8.2 <u>Public Inspection of Background Papers</u>

The Council will make available for public inspection for four years after the date of the meeting one copy of each of the documents on the list of background papers.

#### 15.9 **Summary of Public's Rights**

A written summary of the public's rights to attend meetings and to inspect and copy documents will be kept at, and made available to the public at the Civic Offices, Angel Street, Bridgend, CF31 4WB and published on the Council's website.

#### 15.10 Exclusion of Access by the Public to Meetings

#### 15.10.1 Confidential Information – Requirement to Exclude Public

(a) The public must be excluded from meetings whenever it is likely in view of the nature of the business to be transacted, or the nature of the proceedings, that confidential information would be disclosed.

## 15.10.2 Exempt Information – Discretion to Exclude Public

- (a) The public may be excluded from meetings whenever it is likely in view of the nature of the business to be transacted, or the nature of the proceedings, that exempt information would be disclosed.
- (b) Where the meeting will determine any person's civil rights or obligations, or adversely affect their possessions, article 6 of the Human Rights Act 1998 establishes a presumption that the meeting will be held in public, unless a private hearing is necessary for one of the reasons specified in article 6.

## 15.10.3 Meaning of Confidential Information

Confidential information means information given to the Council by a Government Department on terms which forbid its public disclosure or information which cannot be publicly disclosed by Court Order.

### 15.10.4 Meaning of Exempt Information

Exempt information means information falling within the following seven categories (subject to any condition):

Information is not exempt information if it relates to proposed development for which the local planning authority may grant itself planning permission pursuant to regulation 3 of the Town and Country Planning General Regulations 1992.

No.	Category of Exempt Information	Condition
1.	Information relating to any individual.	Public interest test applies (see below).
2.	Information which is likely to reveal the identity of an individual.	Public interest test applies (see below).
3.	Information relating to the financial or business affairs of any particular person (including the authority holding that information).	Information falling within paragraph 3 is not exempt information by virtue of that paragraph if it is required to be registered under:
		The Companies Act 1985;
		The Friendly Societies Act 1974;
		The Friendly Societies Act 1992;

No.	Category of Exempt Information	Condition
		The Industrial and Provident Societies Acts 1965 to 1978;
		• The Building Societies Act 1986; or
		The Charities Act 1993.
		Public interest test applies (see below).
4.	Information relating to any consultations or negotiations, or contemplated consultations or negotiations, in connection with any labour relations matter arising between the authority or a Welsh Minister or a Minister of the Crown and employees of, or office holders under, the authority.	Public interest test applies (see below).
5.	Information in respect of which a claim to legal professional privilege could be maintained in legal proceedings.	
6.	Information which reveals that the authority proposes:	Public interest test applies (see below).
	<ul> <li>to give under any enactment a notice under or by virtue of which requirements are imposed on a person; or</li> </ul>	
	to make an order or direction under any enactment.	
7.	Information relating to any action taken or to be taken in connection with the prevention, investigation or prosecution of crime. Public interest test applies (see below).	Public interest test applies (see below).
	(In relation to a meeting of a Standards Committee, or a Sub-Committee of a Standards Committee, which is convened to consider a matter referred under the provisions of section 70(4) or (5) or 71(2) of the Local Government Act 2000):	
	7A Information which is subject to any obligations of confidentiality.	

No.	Categ	ory of Exempt Information	Condition
	7B	Information which relates in any way to matters concerning national security.	
	7C	The deliberations of a Standards Committee or of a Sub-Committee of a Standards Committee established under the provisions of Part 3 of the Local Government Act 2000 in reaching any finding on a matter referred to it.	

#### 15.11 Public Interest Test

#### 15.11.1 Information which:

- (a) falls within any of paragraphs 1 to 4, 6 and 7 above; and
- (b) is not prevented from being exempt by virtue of the "qualifications" above,

is exempt information if and so long, as in all the circumstances of the case, the public interest in maintaining the exemption outweighs the public interest in disclosing the information.

- 15.11.2 The starting point is that there is a general public interest in release and the public authority has to decide whether in any particular case it would serve the interest of the public better to either disclose or withhold the information.
- 15.11.3 There is no legal definition of what the public interest is, but the following have been identified as some of the relevant considerations:
  - (a) There is a distinction between public interest and what merely interests the public.
  - (b) Does it further the understanding of and participation in the public debate of issues of the day?
  - (c) Does it promote accessibility and transparency by public authorities for decisions taken by them or in the spending of public money?
  - (d) Does it allow individuals and companies to understand decisions made by public authorities affecting their lives?
  - (e) Does it bring to light information affecting public health and public safety?

#### 15.12 Exclusion of Access by the Public to Reports

If the Monitoring Officer thinks fit, the Council may exclude access by the public to reports which, in his or her opinion, relate to items during which, in accordance with Rule 15.10, the meeting is likely not to be open to the public. Such reports will be marked "Not for

Publication", together with the category of information likely to be disclosed and if applicable, why it is in the public interest it is considered that the information should not be disclosed.

#### 15.13 The Forward Work Programme

#### 15.13.1 Period of Forward Work Programme

The Forward Work Programme will be prepared by the Monitoring Officer to cover a period of four months except when ordinary elections of Councillors occur, in which case the forward work programme will cover the period up to the date of the elections. The forward work programme will be updated quarterly.

#### 15.13.2 Contents of Forward Work Programme

- (a) The Forward Work Programme will contain matters which the Cabinet, Overview and Scrutiny Committee and Full Council are likely to consider. It will contain information on:
  - the timetable for considering the Budget and any plans forming part of the Policy Framework and requiring Council approval, and which body is to consider them;
  - (ii) the timetable for considering any plans, policies or strategies which are the responsibility of the Cabinet;
  - (iii) any individual matters on which the Cabinet intends to consult in advance of taking a decision, and the timetable for consultation and decision;
  - (iv) the work programme of the Overview and Scrutiny Committees, to the extent that it is known.
- (b) The Forward Work Programme will be published at least fourteen days before the start of the period covered. The Monitoring Officer will publish a notice in at least one newspaper circulating in the area, stating that the Forward Work Programme has been published and giving details of where it may be consulted or obtained.

## 15.14 Consultation on Proposals to be Considered by the Cabinet

- 15.14.1 At least four weeks should be permitted in the Forward Work Programme for consultation with the Overview and Scrutiny Committees where a matter is to be considered by the Cabinet and is not urgent (as defined below) or confidential or exempt (as defined in paragraph 15.10).
- 15.14.2 A matter may be considered urgent where the events to which it is addressed were unforeseen at the time that the last Forward Work Programme was produced and a decision is required within four weeks.
- 15.14.3 A decision can only be treated as urgent if the decision taker (if an individual) or the Chair of the body making the decision obtains the agreement of the Chair of the relevant Overview and Scrutiny Committee that the taking of the decision cannot be reasonably deferred. If there is no Chair of the relevant Overview

and Scrutiny Committee, or if the Chair of the Overview and Scrutiny Committee is unable to act, then the agreement of the Mayor or, in his/her absence, the Deputy Mayor will suffice. Any decisions taken under this urgency procedure will be recorded as having been taken in the absence of consultation in the decision record.

#### 15.15 Record of Decisions of the Cabinet

#### 15.15.1 The Decision Record

- (a) A written record will be made of every Executive Decision made by the Cabinet and its Committees (if any) and by individual Cabinet Members, and Joint Committees and Joint Sub-Committees whose Members are all Members of a local authority executive.
- (b) This decision record will include a statement, for each decision, of:
  - (i) the decision made:
  - (ii) the date the decision was made:
  - (iii) the reasons for that decision;
  - (iv) any personal interest declared;
  - (v) any dispensation to speak granted by the Authority's Standards Committee;
  - (vi) any consultation undertaken prior to the decision.

#### 15.15.2 Preparing the Decision Record

- (a) The Monitoring Officer or his or her representative shall attend any meeting of the Cabinet, a Committee of the Cabinet or a Joint Committee or joint Sub-Committee where all its Members are Members of a local authority Executive, and shall, as soon as reasonably practicable after the meeting, produce a decision record.
- (b) Where an individual Cabinet Member has made any Executive Decision:
  - (i) that Member shall as soon as reasonably practicable instruct the Chief Executive to produce a decision record; and
  - (ii) the decision shall not be implemented until that decision record has been produced, subject to paragraph (c) below.
- (c) Where the date by which an Executive Decision made by an individual Cabinet Member must be implemented makes compliance with (b)(b)(ii) above impracticable, the decision may be implemented if the decision maker has the agreement of
  - (i) the Chair of the relevant Overview and Scrutiny Committee; or
  - (ii) if there is no such person or that person is unable to act, the Mayor;or

(iii) if there is none of the above then the Deputy Mayor,

that the making of the decision is urgent and cannot reasonably be deferred.

# 15.16 Decisions by An Individual Member of the Cabinet

15.16.1 An individual Cabinet Member may only make a decision within the terms of the Scheme of Delegation of Functions for the time being in force. The Cabinet Member will comply with the procedural requirements of the Scheme of Delegation of Functions. The decision recording rules in paragraph 15.15 will apply.

#### 15.17 Overview and Scrutiny Committee and Members' Access to Documents

#### 15.17.1 Rights of Access

Subject to paragraph 15.17.2 below, Overview and Scrutiny Committees will be entitled to access to any document which is in the possession or control of the Cabinet or its Committees and which contains material relating to:

- (a) any business transacted at a meeting of the Cabinet or its Committees; or
- (b) any decision taken by an individual Member of the Cabinet.

# 15.17.2 Limit on Rights

Overview and Scrutiny Committee will not be entitled to any part of a document that contains:

- (a) confidential or exempt information; or
- (b) advice provided by a political advisor or assistant

unless that information is relevant to an action or decision that is being reviewed or scrutinised or any review contained in a programme of work of the Committees.

#### 15.18 Additional Rights of Access for Members of Overview and Scrutiny Committees

#### 15.18.1 Rights of Access

All Members will be entitled (except where a Member has a prejudicial interest in any matter) to inspect any document which is in the possession or under the control of the Cabinet or its Committees and contains material relating to any business transacted at a meeting of a decision making body of the Council or by an individual Member of the Cabinet.

## 15.18.2 <u>Limitation on Rights</u>

A Member will not be entitled to any part of a document where:

- (a) it would disclose exempt information falling within paragraphs 12 to 18 of Part 4 of Schedule 12A of the Local Government Act 1972; or
- (b) it would disclose advice of a political advisor or assistant.

# 15.18.3 Nature of Rights

These rights of a Member are additional to any other right he/she may have.

#### **SECTION 16**

## 16. BUDGET AND POLICY FRAMEWORK PROCEDURE RULES

## 16.1 The Framework for Executive Decisions

The Council will be responsible for the adoption of its Budget and Policy Framework as set out in Section 4. Once a Budget or a Policy Framework is in place, it will be the responsibility of the Cabinet to implement it.

#### 16.2 Process for Developing the Framework

- 16.2.1 The Cabinet will publicise by including it in the Forward Work Programme a timetable for making proposals to the Council for the approval of the Budget, and its arrangements for consultation upon its initial Budget proposals. The Chairpersons of the Overview and Scrutiny Committees will also be notified of the initial Budget process. The consultation period shall be not less than four weeks. If an Overview and Scrutiny Committee wishes to respond to the Cabinet in that consultation process then it may do so. At the end of the consultation period and in accordance with the published timetable, the Cabinet will draw up firm proposals having regard to the responses to that consultation.
- 16.2.2 The Cabinet, following detailed reports from respective Chief Officers, (following appropriate consultation with stakeholders, including any response from the Corporate Overview and Scrutiny Committee) will present to Council at the earliest opportunity, proposed plans, policies and the associated annual budgets. This will be done allowing adequate time for Council to deal with the matter and, if needed, refer the matter back to the Cabinet for further consideration.
- 16.2.3 The Council will consider the proposals of the Cabinet and may adopt them, amend them, refer them back to the Cabinet for further consideration, or substitute its own proposals in their place (see 16.2.4 below). In considering the matter, the Council shall have before it the Cabinet's proposals and any related report from an Overview and Scrutiny Committee.
- 16.2.4 Any amendments to the proposals of the Cabinet to be proposed by Members at Council may not be considered by Council unless notice of the proposed amendment has been given to the Monitoring Officer and Chief Finance Officer in writing and signed by the proposer and seconder no later than 10:00am at least three clear working days before the date of the Council meeting. Any amendment will only be acceptable if it proposes a budget which balances overall income with expenditure for the coming year. The source of funding any proposals must be robust and defined and accepted as such by the Chief Finance Officer.
- 16.2.5 The Council's decision will be publicised and a copy shall be given to the Leader. The notice of decision shall be dated and shall state either that the decision shall be effective immediately (if the Council accepts the Cabinet's proposals without amendment) or (if the Cabinet's proposals are not accepted without amendment), that the Council's decision will become effective on the expiry of five working days after the publication of the notice of decision, unless the Leader objects to it in that period.

- 16.2.6 If the Leader objects to the decision of the Council, he/she shall give written notice to the Chief Executive to that effect, prior to the date upon which the decision is to be effective. The written notification must state the reasons for the objection. Where such notification is received, the Chief Executive shall convene a further meeting of the Council to reconsider its decision and the decision shall not be effective pending that meeting.
- 16.2.7 The Council meeting must take place within five working days of the receipt of the Leader's written objection. At that Council meeting, the decision of the Council shall be reconsidered in the light of the objection, which shall be available in writing for the Council.
- 16.2.8 The Council shall, at that meeting, make its final decision on the matter on the basis of a simple majority. The decision shall be made public and shall be implemented immediately.

# 16.3 Process for Developing the Budget

- 16.3.1 The Cabinet will follow the process set out in the Financial Procedure Rules in Section 0 of the Constitution. In addition the following process shall apply to the development of the Budget:
  - (a) The Cabinet will publicise a timetable for making proposals to the Council for the adoption of the Budget, and its arrangements for consultation after publication of the initial proposals. The Chair of the Corporate Overview and Scrutiny Committee will also be notified.
  - (b) At the end of the consultation period, the Cabinet will draw up firm proposals having regard to the responses to the consultation. If the Overview and Scrutiny Committee wishes to respond to the Cabinet in that consultation process then it may do so. The Overview and Scrutiny Committee shall report to the Cabinet on the outcome of its deliberations.
- 16.3.2 The Cabinet will take any response from the relevant Overview and Scrutiny Committee account in drawing up firm proposals for submission to the Council, and its report to Council will reflect the comments made by consultees and the Cabinet's response.
- 16.3.3 Once the Cabinet has approved the firm proposals, it will refer them at the earliest opportunity to the Council.

#### 16.4 Decisions Outside the Budget or Policy Framework

- 16.4.1 Subject to the provisions of paragraph 16.6 (Virement) the Cabinet, committees of the Cabinet, individual Members of the Cabinet or any Officers or joint arrangements discharging Executive Functions may only take decisions which are in line with the Budget and Policy Framework. If any of these bodies or persons wishes to make a decision which is contrary to the Policy Framework, or contrary to or not wholly in accordance with the Budget approved by Full Council, then that decision may only be taken by the Council, subject to paragraph 16.5 below.
- 16.4.2 If the Cabinet, committees of the executive, individual Members of the Cabinet or any Officers or joint arrangements discharging Executive Functions want to make such a decision, they shall take advice from the Monitoring Officer and/or

the Chief Finance Officer as to whether the decision they want to make would be contrary to the Policy Framework, or contrary to or not wholly in accordance with the budget.

16.4.3 If the advice of either of those Officers is that the decision would not be in line with the existing Budget and/or Policy Framework, then the decision must be referred by that body or person to the Council for decision, unless the decision is a matter of urgency, in which case the provisions in paragraph 16.5 (urgent decisions outside the Budget and Policy Framework) shall apply.

## 16.5 <u>Urgent Decisions Outside the Budget or Policy Framework</u>

- 16.5.1 The Cabinet, a committee of the Cabinet, an individual Member of the Cabinet or Officers or joint arrangements discharging Executive Functions may take a decision which is contrary to the Council's Policy Framework or contrary to or not wholly in accordance with the budget approved by Full Council if the decision is a matter of urgency. However, the decision may only be taken:
  - (a) if it is not practical to convene a quorate meeting of the Full Council; and
  - (b) if the Chair of the relevant Overview and Scrutiny Committee agrees that the decision is a matter of urgency.
- 16.5.2 The reasons why it is not practical to convene a quorate meeting of Full Council and the Chair of the relevant Overview and Scrutiny Committee's consent to the decision being taken as a matter of urgency must be noted on the record of the decision. In the absence of the Chair of the relevant Overview and Scrutiny Committee the consent of the Mayor and in the absence of both the Deputy Mayor of the Council will be sufficient.
- 16.5.3 Following the decision, the decision taker will provide a full report to the next available Council meeting explaining the decision, the reasons for it and why the decision was treated as a matter of urgency.

#### 16.6 Virement

Steps taken by the Cabinet, a committee of the Cabinet, individual Members of the Cabinet or Officers or joint arrangements discharging Executive Functions to implement Council policy, shall not exceed those budgets allocated to each budget head. In the event that virement across budget heads or budgets becomes necessary or desirable, it shall be carried out in accordance with the Financial Procedure Rules in Section 0 of this Constitution.

#### 16.7 In-Year Changes to Policy Framework

The responsibility for agreeing the Budget and Policy Framework lies with the Council and decisions by the Cabinet, a committee of the Cabinet, an individual Member of the Cabinet or Officers or joint arrangements discharging Executive Functions must be in line with it. No changes to any policy or strategy which make up the Policy Framework may be made by those bodies or individuals except those changes:

16.7.1 which will result in the closure or discontinuance of a service or part of service to meet a budgetary constraint;

- 16.7.2 necessary to ensure compliance with the law, ministerial direction or government guidance;
- 16.7.3 in relation to the Policy Framework in respect of a plan, policy or strategy which would normally be agreed annually by the Council following consultation, but where the existing policy document is silent on the matter under consideration;
- 16.7.4 which relate to policy in relation to schools, where the majority of school governing bodies agree with the proposed change; or
- 16.7.5 for which provision is made within the relevant budget or policy.

#### 16.8 Call-In of Decisions Outside the Budget or Policy Framework

- 16.8.1 Where the relevant Overview and Scrutiny Committee is of the opinion that an Executive Decision is, or if made would be, contrary to the Policy Framework, or contrary to or not wholly in accordance with the Council's Budget, then it shall seek advice from the Monitoring Officer and/or Chief Finance Officer.
- 16.8.2 In respect of functions which are the responsibility of the Cabinet, the Monitoring Officer's report and/or Chief Finance Officer's report shall be to the Cabinet with a copy to every Member of the Council. Regardless of whether the decision is delegated or not, the Cabinet must meet to decide what action to take in respect of the Monitoring Officer's or Chief Finance Officer's report and to prepare a report to Council in the event that the Monitoring Officer or the Chief Finance Officer conclude that the decision was a departure, and to the relevant Overview and Scrutiny Committee if the Monitoring Officer or the Chief Finance Officer conclude that the decision was not a departure.
- 16.8.3 If the decision has yet to be made, or has been made but not yet implemented, and the advice from the Monitoring Officer and/or the Chief Finance Officer is that the decision is or would be contrary to the Policy Framework or contrary to or not wholly in accordance with the Budget, the relevant Overview and Scrutiny Committee may refer the matter to Council. In such cases, no further action will be taken in respect of the decision or its implementation until the Council has met and considered the matter. The Council shall meet within ten working days of the request by the Overview and Scrutiny Committee. At the meeting it will receive a report of the decision or proposals and the advice of the Monitoring Officer and/or the Chief Finance Officer. The Council may either:
  - (a) endorse a decision or proposal of the Executive Decision taker as falling within the existing Budget and Policy Framework. In this case no further action is required, save that the decision of the Council be minuted and circulated to all councillors in the normal way; or
  - (b) amend the Council's Budget or the plan, policy or strategy concerned to encompass the decision or proposal of the body or individual responsible for that Executive Function and agree to the decision with immediate effect. In this case, no further action is required save that the decision of the Council be minuted and circulated to all councillors in the normal way; or
  - (c) where the Council accepts that the decision or proposal is contrary to the Policy Framework or contrary to or not wholly in accordance with the Budget, and does not amend the existing Budget or the plan, policy or strategy concerned to accommodate it, require the Cabinet to reconsider the matter

in accordance with the advice of either the Monitoring Officer or Chief Finance Officer.

#### **SECTION 17**

## FINANCIAL PROCEDURE RULES

#### <u>Introduction</u>

#### What are Financial Procedure Rules?

Financial Procedures provide the framework for managing the Council's financial affairs. They are supported by more detailed specific guidance and procedure notes in a range of areas. The procedures identify the financial responsibilities of the Full Council, the Cabinet and Officers.

#### Why are they important?

- (a) To conduct its business effectively, the Council needs to ensure that sound financial management arrangements are in place and that they are strictly adhered to in practice. Part of this process is the establishment of Financial Procedures which set out the financial responsibilities of the Council. These procedures have been devised as a control to help the Council manage its financial matters properly in compliance with all necessary requirements.
- (b) Good, sound financial management is a key element of the Council's Corporate Governance framework which helps to ensure that the Council is doing the right things, in the right way, for the right people, in a timely, inclusive, open, honest and accountable manner.
- (c) Good financial management secures value for money, controls spending, ensures due probity of transactions and allows decisions to be informed by accurate accounting information.
- (d) Good financial management requires secure and reliable records and systems to process transactions and information and substantiate the effective use of public money.
- (e) Financial Procedures should not be seen in isolation, but rather as part of the overall regulatory framework of the Council as set out in this Constitution.

# Who do Financial Procedures apply to?

- (f) Financial Procedures apply to every Member and Officer of the Council and anyone acting on its behalf. Members and Officers have a general responsibility for taking reasonable action to provide for the security and use of the resources and assets under their control, and for ensuring that the use of such resources and assets is legal, is consistent with Council policies and priorities, is properly authorised, provides value for money and achieves best value.
- (g) Separate financial procedures have been incorporated into the Council's Financial Scheme for Schools and relate to those matters where decisions have been delegated to school governing bodies.
- (h) These Financial Procedures shall apply in relation to any partnership for which the Council is the accountable body, unless the Council expressly agrees otherwise.

(i) Failure to observe Financial Procedures may result in action under the Council's disciplinary procedures.

#### Who is responsible for ensuring that they are applied?

- (j) Chief Officers and Heads of Service are ultimately responsible to the Council for ensuring that Financial Procedures are applied and observed by his/her staff and contractors providing services on the Council's behalf and for reporting to the Chief Finance Officer any known or suspected breaches of the procedures.
- (k) The Chief Finance Officer is responsible for maintaining a continuous review of the Financial Procedures and submitting any additions or changes necessary to the Council for approval. The Chief Finance Officer is also responsible for reporting, where appropriate, breaches of the Financial Procedures to the Council and/or to the Cabinet Members.
- (I) The Chief Finance Officer is responsible for issuing advice and guidance to underpin the Financial Procedures which Members, Chief Officers and others acting on behalf of the Council are required to follow.
- (m) Where any Chief Officer considers that complying with Financial Procedures in a particular situation might conflict with the achievement of value for money or the principles of Best Value or the best interests of the Council, he/she shall raise the issue with the Chief Finance Officer who will, if he/she considers necessary and appropriate, seek formal approval from the Council for a specific waiver of the procedures, or an amendment to the procedures themselves.

#### **Section A - Financial Control**

#### 1. General

- 1.1 For the purposes of these Rules:
  - (a) "the Chief Executive Officer" means the officer designated by the Council as Head of Paid Service under Section 4 of the Local Government & Housing Act 1989.
  - (b) "the Chief Finance Officer" means the officer designated by the Council as the officer responsible for the administration of its financial affairs under Section 151 of the Local Government Act 1972.
  - (c) "Chief Officer" means any of the following:
    - The Chief Executive Officer:
    - The Corporate Director Social Services and Wellbeing;
    - The Corporate Director Education and Family Support;
    - The Corporate Director Communities;
    - Chief Officer Legal and Regulatory Services, Human Resources and Corporate Policy.
    - Chief Officer Finance, Performance and Change.
  - (d) "the Monitoring Officer" means the officer designated as such by the Council under Section 5 of the Local Government & Housing Act 1989.

- (e) "the Cabinet" means the Executive established under Part II of the Local Government Act 2000.
- (f) "Cabinet Member" means the Leader and any of the members of the Cabinet.
- (g) "Budget Head" means the budget for a particular service/services.
- (h) Words importing the singular number only shall include the plural and vice versa, words importing the masculine gender include the feminine and vice versa.
- 1.2 A Chief Officer may nominate any suitably qualified officer in their Directorate to undertake any of the duties placed upon him or exercise any power granted to him by these Rules.
- 1.3 Each Chief Officer shall consult with the Chief Finance Officer on any matter which is liable to materially affect the finances of the Council before any commitment is incurred and before reporting thereon to the Cabinet or the Council.
- 1.4 (a) Chief Officers shall be responsible for bringing these Financial Procedure Rules to the attention of staff and for ensuring the observance of these Rules throughout their respective Directorates;
  - (b) Staff who fail to observe these Rules may be subject to disciplinary action.
- 1.5 Each Chief Officer shall be responsible for the accountability of staff, and the security, custody and control of all other resources including plant, buildings, materials, cash and stores appertaining to their individual Directorates in accordance with the procedures agreed with the Chief Finance Officer.
- 1.6 Each Chief Officer shall have a duty to endeavour to maximise the value for money obtained in running the activities within their purview.
- 1.7 Each Chief Officer shall have a duty to consult the Chief Finance Officer if he has reason to believe that any matter within their purview may result in:
  - (a) The Council incurring expenditure in excess of its approved budget;
  - (b) The Council incurring expenditure for which it has no statutory power to incur;
  - (c) The Council failing to comply with the financially related provisions of any relevant European and UK legislation;
  - (d) The Council failing to comply with the financially related provisions of any code of practice adopted by the Council.
- 1.8 Whenever any matter arises which involves or is thought to involve irregularities concerning cash, stores or other property of the Council or held on trust by the Council, the Chief Officer concerned shall notify the Chief Finance Officer who shall take such steps as he considers necessary by way of investigation and report.
- 1.9 Each Chief Officer shall have a duty to inform the Chief Finance Officer and the Monitoring Officer if he suspects that the Council or its officers are exceeding the Council's statutory powers.
- 1.10 No Chief Officer may recommend to the Council or the Cabinet that expenditure be incurred or any other action taken on the assumption that sufficient statutory power is provided by either

- Sections 137 of the Local Government Act 1972 and Section 2 of the Local Government Act 2000, without prior approval of the Chief Finance Officer and the Monitoring Officer.
- 1.11 The Chief Finance Officer in consultation with the Monitoring Officer shall be authorised to vary from time to time any amount included in the Rules, but any alteration shall be reported to the next meeting of Council.
- 1.12 Separate Financial Regulations shall be issued by the Chief Finance Officer for use by the Council's schools, drawing on, and supplementing this document, as appropriate.

## 2. Accounting Systems

- 2.1 All accounting procedures and accounting records of the Council and its Officers shall be subject to the approval of the Chief Finance Officer.
- 2.2 All accounts and accounting records of the Council shall be compiled by, or under the direction of, the Chief Finance Officer.
- 2.3 The following principles shall be observed in the allocation of accounting duties:
  - (a) The duties of providing information regarding sums due to or from the Council and of calculating, checking and recording these sums, shall be separated as completely as possible from the duty of collecting or disbursing them;
  - (b) Officers charged with the duty of examining and checking the accounts of cash transactions shall not themselves be engaged in any of these transactions.
- 2.4 A Chief Officer shall be responsible for the financial management and audit of any private fund entrusted to him as part of their duties and will apply regulations to such funds as stipulated by the Chief Finance Officer.
- 2.5 Notwithstanding the duties of a Chief Officer in Rule 2.4, the Chief Finance Officer shall have the right to audit any private fund at any time.
- 2.6 For the purposes of these Rules "private fund" shall mean any fund in the management of which a Chief Officer of the Council is concerned and which may affect any person or property for which the Council has a responsibility, notwithstanding that contributions towards such fund shall have been made by another party other than the Council.

#### 3. Budgeting and budgetary control

#### 3.1 Council's Budget

- 3.1.1 The Budget shall be approved by the Council in accordance with the Budget and Policy Framework Procedure Rules (Section 16);
- 3.1.2 In referring the Cabinet's Budget Proposals to Council the proper officer will forward to the Council a report of the Chief Finance Officer incorporating the recommendations of the Cabinet and detailing for approval of the Council:
  - (a) a probable out-turn for the current year;

- a revenue budget for the forthcoming year detailing the Budget Heads over which that budget is allocated to specific services and service programme areas;
- (c) a forward indication of the medium term financial strategy which will comprise the revenue budget for the first year and indicative budgets for the subsequent three years and a capital budget for the forthcoming year incorporating a capital programme of at least three years, but no more than ten years;
- (d) an assessment of any major financial risks which may impact on the budget;
- (e) an analysis of reserves and balances and movements over the forthcoming financial year;
- (f) the proposed increase in fees and charges for the forthcoming year;
- (g) a recommendation of the level of Council Tax to be levied for the forthcoming year.
- 3.1.3 The detailed form of the Budget will be determined by the Chief Finance Officer within the general direction of the Council and Cabinet after consultation with Chief Officers.

## 3.2 Amendments to the agreed budget (virements and technical adjustments)

- 3.2.1. Each Chief Officer shall have the authority to incur expenditure on any activity under their control up to the amounts specified in respect of that activity in the revenue or capital budgets approved by the Council for the financial year, unless directed otherwise by the Chief Finance Officer.
- 3.2.2 Each Chief Officer, subject to the agreement of the Chief Finance Officer, may approve virements between specific revenue Budget Heads which do not amend any individual Budget Head by more than £100,000 from that approved by the Council. The relevant Chief Officer and Chief Finance Officer must jointly report to Cabinet on any virements which amend individual Budget Heads by more than £100,000.
- 3.2.3 Cabinet may, following a report of the appropriate Chief Officer in consultation with the Chief Finance Officer, approve virements between individual revenue Budget Heads which do not amend any individual budget head by more than £500,000 from that approved by the Council.
- 3.2.4 All approved virements over £100,000 must be reported to the Council for information as soon as reasonably practicable after their approval.
- 3.2.5 Any variations to the capital programme, other than those permitted under paragraphs 3.4.7 and 3.4.9, shall require the approval of the Council following a report of the Chief Finance Officer after taking into consideration the recommendations of the Cabinet.
- 3.2.6 All proposals for revenue and capital budget virements that exceed the above limits will need to be approved by the Council on receipt of a report of the appropriate Chief Officer in consultation with the Chief Finance Officer following consideration by the Cabinet. Virement proposals must also be approved by the Council if they:
  - a) imply a change in a plan, policy or strategy which would be contrary to the existing policy framework;

- b) have a major operational impact on existing service provision;
- c) are contrary to or not wholly in accordance with the Budget;
- d) imply any additional revenue commitment in future years.
- 3.2.7 There are also technical adjustments to budgets as a result of the Council adhering to the Code of Practice on Local Authority Accounting. These could include, but are not limited to, the reallocation of budgets and spend for central support services to other areas within the Council or the allocation of capital charges across the Council. These are not subject to the authorisation limits of budget virements and the Chief Finance Officer or nominated representative can authorise these.

### 3.3 Budgetary control

- 3.3.1 The Chief Finance Officer shall provide each Chief Officer with regular information relating to income and expenditure under each approved budget head and any other relevant information available. Each Chief Officer shall be responsible for ensuring control of expenditure and income against the approved budget (attention is drawn to Rule 6.1).
- 3.3.2 A Chief Officer in consultation with the Cabinet Member responsible for the function in respect of which the service is provided will be expected to manage their services within the approved cash limited budget and to provide the Chief Finance Officer with such information as is required to facilitate and monitor budgetary control. A Chief Officer or Cabinet Member may not incur expenditure or forego income if this will result in the approved budget being exceeded.
- 3.3.3 A Chief Officer may delegate responsibility for management of budgets within their control to other senior officers within their Directorate. Such delegation shall be within defined parameters and shall be recorded in writing as determined by the Chief Finance Officer.
- 3.3.4 At year end, consideration will be given to the overall financial position of the Council including the final outturn, any accrued Council Tax income, the Council's reserve levels and any new pressures or risks that need to be provided for. At that time, in line with the Council's Reserves and Balances Protocol, a Chief Officer will be invited to submit earmarked reserve requests to meet any specific unfunded one-off expenditure that they expect to arise in the following financial year and these will be considered by the Chief Finance Officer in the context of the Directorate outturn position as well as that of the Council as a whole. Chief Officers will be notified of successful earmarked reserves.
- 3.3.5 Any over-spending against budget may be carried forward at the discretion of the Chief Finance Officer, either against the specific directorate or Council wide budgets. In no circumstances should this provision be seen as giving a Chief Officer power to overspend against approved budgets. Any such overspend will be treated as a breach of the Financial Procedure Rules and the respective Chief Officer held accountable. No Chief Officer or Cabinet Member may budget for a deficit. Where a net overspending occurs this will be a first call on the following year's budget, unless the Chief Finance Officer gives approval to meet this from usable reserves.
- 3.3.6 Urgent expenditure not included in any budget approval may only be incurred with the approval of the Chief Finance Officer.

- 3.3.7 The cash limited budget for a Chief Officer only relates to the "controllable" elements of their budget and excludes, amongst other things:
  - Central capital charges
  - Central support service charges
  - Centrally controlled office accommodation budgets
  - Joint Committee precepts and levies

# 3.4 Capital programme

- 3.4.1 As part of the budget process the Chief Finance Officer or appropriate Cabinet Member will annually present to the Council a capital programme which shall include:
  - (a) those capital expenditure items proposed to commence during the next three years as a minimum;
  - (b) an estimate of the capital costs of those schemes together with the associated proposed funding.
- 3.4.2 Schemes for which external funding has been approved will be added to the capital programme once the funding has been accepted and included in the next capital programme report to Council.
- 3.4.3 A Chief Officer, before submitting a scheme for inclusion in the capital programme, shall satisfy himself that the scheme is in line with the Council's Capital Strategy, and that:
  - (a) land purchases, design planning consents and relevant studies are sufficiently advanced to ensure that the proposed year of start of a scheme is feasible:
  - (b) the level of expenditure envisaged is realistic, following a full feasibility assessment, and taking into account the Chief Finance Officer's forecast of capital resources available and the ability of each service directorate to meet the consequential costs resulting from prudential borrowing, if applicable, and any on-going maintenance costs;
  - (c) each scheme in the programme has been fully appraised to ensure it is the most economic method of satisfying an identified need.
- 3.4.4 The Capital Programme upon approval by the Council shall:
  - (a) confer authorisation upon the Chief Officer concerned to take steps to enable design work to be completed and land to be acquired in due time;
  - (b) form the basis of the annual capital estimates.
- 3.4.5 The inclusion of any item in the approved capital estimates shall not confer authority to incur any expenditure (except on design work and land acquisition) until:
  - (a) all necessary statutory approvals have been obtained;
  - (b) any external funding contribution to the project has been secured; and

- (c) a tender or quotation has been received and accepted in accordance with the Contract Procedure Rules, which does not exceed that part of the total cost included in the capital estimate in respect of the main contract work for the project by more than 10% or £100,000, whichever is lower.
- 3.4.6 If the tender or quotation exceeds the criteria in 3.4.5(c) above, its acceptance will be subject to the approval of the Chief Finance Officer, in consultation with the Cabinet Member(s) responsible for the function in respect of which the decision is required, to a diversion of money from other approved schemes within the appropriate Chief Officer's control sufficient to meet any additional cost to be borne within the first year.
- 3.4.7 Chief Officers shall monitor both the progress of schemes and the totality of capital expenditure with the aim of avoiding under or overspending against the approved capital estimates. Should such a situation appear likely a Chief Officer in conjunction with the Chief Finance Officer shall recommend to the Council the remedial action necessary to accelerate or retard existing schemes within the approved capital programme.
- 3.4.8 Remedial action which necessitates the retardation or deletion of a scheme within the first year of the programme shall be subject to the prior approval of the Council based on a joint report of the Chief Finance Officer and Chief Officer.
- 3.4.9 The Chief Finance Officer shall give Chief Officers information relating to actual payments made for each scheme in such detail and at such time as arranged between them in order that they may carry out their responsibilities under paragraph 3.4.7.
- 3.4.10 The capital programme includes an annual allocation for capital minor works. Allocation of this funding to individual schemes is the responsibility of the Corporate Landlord Group (or its successor) and approval on individual schemes will not be sought from Council insofar as the overall funding allocated to schemes does not exceed the funding agreed by Council in the capital programme for that financial year.
- 3.4.11 Any additional capital expenditure from within the capital programme for new or existing schemes which, for reasons of urgency, cannot await the next meeting of Council, may only be incurred with the approval of the Chief Executive Officer in consultation with the Chief Finance Officer, subject to a maximum value of £100,000, under the Scheme of Delegation of Functions, Scheme B1 paragraph 2.2.
- 3.4.12 Urgent expenditure not included in any budget approval, which needs to be agreed prior to the next meeting of Council, may only be incurred with the approval of the Chief Finance Officer. Any such decision requires approval by either the Chief Executive Officer or Solicitor to the Council under the Scheme of Delegation of Functions, Scheme B1 paragraph 2.1.

## 3.5 Capital Strategy

- 3.5.1 The Chief Finance Officer will be responsible for preparing a Capital Strategy for submission to Council for approval prior to the start of each financial year. The Strategy must demonstrate that capital expenditure and investment decisions are in line with service objectives and properly take account of stewardship, value for money, prudence, sustainability and affordability.
- 3.5.2 The Capital Strategy will need to comply with the requirements of the Chartered Institute of Public Finance and Accountancy (CIPFA)'s Prudential Code for Capital Finance in Local

Authorities. It will incorporate the Prudential Indicators that will need to be approved by Council.

3.5.3 The Chief Finance Officer shall report quarterly to Cabinet and Council with an update on the Capital Strategy and the Prudential Indicators.

#### 4. Internal audit

- 4.1 The Chief Finance Officer shall have a duty to maintain an effective internal audit of the Council's operations in order to review, evaluate and test the adequacy of the Council's systems of internal control as contributions to the proper, economic, efficient and effective use of resources.
- 4.2 The Chief Finance Officer shall have the responsibility to review, appraise and report to Council, the Governance and Audit Committee, Cabinet, Cabinet Member, or Chief Officer as appropriate upon:
  - 4.2.1 The soundness, adequacy and application of financial and other related operations of the Council.
  - 4.2.2 The extent of compliance with, and financial effect of, established policies, plans and procedures.
  - 4.2.3 The extent to which the Council's assets and interests are accounted for and safeguarded from losses of all kinds arising from:
    - (a) fraud and other offences;
    - (b) waste, extravagance, poor value for money or other cause.
  - 4.2.4 The suitability and reliability of financial and other related management data developed within the Council.
- 4.3 The Chief Finance Officer or an authorised representative of the Chief Finance Officer shall have the authority to require any officer to:
  - 4.3.1 Provide access to any Council premises or land under their control. Where sites are in the possession of a contractor or subject to any tenancy or licence to occupy, such entry will be governed by the conditions of the contract or other legal agreement.
  - 4.3.2 Produce any records, documents and correspondence in their possession.
  - 4.3.3 Provide explanations of matters arising from an audit.
  - 4.3.4 Produce and account for any cash, stores or other Council property under their control.
- 4.4 The Chief Finance Officer shall have the authority to request the immediate suspension from duty of any officers who:
  - (a) they have reasonable grounds to suspect of misappropriation of Council funds or other property;
  - (b) they believe present a threat of further misappropriation or hindering of any investigation.

4.5 Notwithstanding the duty of the Chief Finance Officer for the control and direction of Internal Audit, it shall be the duty of the Chief Internal Auditor to report direct to the Chief Executive Officer and to the Cabinet Member for Resources on any matter in which the Chief Finance Officer appears to be personally involved.

# 5. Contracts of building, construction or engineering work

- 5.1 Contracts for the execution of capital works shall only be entered into for those schemes which are included in the approved Capital Programme.
- 5.2 Each Chief Officer will maintain contract registers showing for each contract under their control which has a value greater than £50,000:
  - (a) the contract sum;
  - (b) the value of any extras or variations to the contract;
  - (c) the amounts and dates of any instalments made;
  - (d) the amount of any retentions held or bonds taken under any contract;
  - (e) the balance outstanding to the Contractor.
- 5.3 Payments to contractors on account of contracts shall be made only on a certificate issued by the responsible officer.
- 5.4 When authorising any extra or variation to a contract the appropriate Chief Officer shall:
  - (a) estimate the cost of the variation;
  - (b) issue written instructions to the Contractor to carry out the work, except in cases of urgency where a Chief Officer may issue verbal instructions but shall confirm them in writing as soon as possible, but in any case within 7 days; and
  - (c) ensure that such variation is in accordance with the Council's Contract Procedure Rules.
- 5.5 The Chief Finance Officer may, at their discretion, decide to audit the final account of a contract, before certifying the final payment.
- 5.6 Claims from contractors in respect of matters not clearly within the terms of any existing contract shall be referred to the Solicitor to the Council for consideration of the Council's legal liability and, where necessary, to the Chief Finance Officer for financial consideration before a settlement is reached.
- 5.7 Each Chief Officer shall as soon as practical report to Cabinet the total of extras or variations to any contract which exceeds 10% of the original contract sum, or £100,000 whichever is the greater.
- 5.8 If the final account of any contract has not been agreed within eighteen months of the completion of works, then the Chief Officer controlling that contract shall report to the Cabinet on the outstanding items. Any report under this regulation shall include a comparison of the final or likely cost with the original contract sum together with reasons for any differences.

## 6. Orders for work, goods and services

- 6.1 Each Chief Officer has a duty to issue official orders for all works and services to be carried out, and all goods supplied, within their Directorate with the exception of:
  - (a) public utilities;
  - (b) periodical payments;
  - (c) petty cash purchases;
  - (d) purchases made using Purchasing Cards;
  - (e) works, goods and services subject to formal written contracts, or excepted from this requirement by the Chief Finance Officer.
  - (f) Framework contracts.

Before issuing official orders, Officers authorising them must be satisfied that there is provision in the estimates and that the Contract Procedure Rules have been complied with.

- 6.2 Each official order shall be in a form prescribed and approved by the Chief Finance Officer and shall include:
  - (a) a description of the works, goods and services ordered;
  - (b) the name and address of the supplier:
  - (c) an estimate of the cost;
  - (d) the financial code to which the cost is to be charged;
  - (e) name of the authorising officer.
- 6.3 Verbal orders may only be placed in exceptional circumstances the details of which must be confirmed by e-mail or other appropriate means on the same day and confirmed by written official order within 3 working days.
- 6.4 Electronic orders will be treated in the same way as other official orders. Changes to on-line authorising officers shall be notified to the Chief Finance Officer immediately.

#### 7. Leasing

- 7.1 The Chief Finance Officer shall be responsible for making leasing arrangements for the use of buildings, plant, vehicles and equipment. No officer other than the Chief Finance Officer may enter into any type of leasing arrangements unless agreed in writing or delegated by the Chief Finance Officer.
- 7.2 Due to complex capital accounting requirements regarding expenditure in relation to leases, all leases, hire, rental, hire purchase agreements, deferred purchase agreements and other

- arrangements where the use of an asset is being acquired without the ownership of it must be referred to the Chief Finance Officer.
- 7.3 Provision within revenue estimates to cover the annual costs of leasing should be made only following consultation with the Chief Finance Officer.

### 8. Grant Funding and Third Party Funding

- 8.1 It is essential that proper management of external funding or grants received, or funding awarded by the Council, is undertaken in order to safeguard the financial position of the Council and also to maximise the benefits to service delivery from the receipt of these additional funds.
- 8.2 The Council has a Grants Management Policy which all Chief Officers and their staff must adhere to in order to ensure the proper management of grant funding and other internal and external funds. Failure to comply could result in funding being withheld by the funding body or recovered at a later date. Any failure to comply with the Grants Management Policy, which results in funding being reduced to the Council, may be reported to Governance and Audit Committee, and any shortfall in the funding borne by the responsible directorate.

### **Section B - Expenditure and Income**

### 9. Payment of accounts

- 9.1 Excluding payments in the form of petty cash from imprest accounts, the preferred method of payment of money due from the Council shall be by automated transfers from the Council's bank accounts overseen by the Chief Finance Officer.
- 9.2 The Chief Finance Officer shall be the authorised signatory of any joint bank account opened in the name of the Council and any other party and will be responsible for the payment of any cheques from that joint bank account.
- 9.3 Each Chief Officer is responsible for examining, verifying and authorising invoices and any other payment vouchers or accounts arising from sources in their Directorate including the use of purchasing cards. Authorised officers can discharge this responsibility on behalf of the Chief Officer. Each Chief Officer will supply the names and specimen signatures for authorised officers together with authorisation limits to the Chief Finance Officer and will need to confirm the list on an annual basis.
- 9.4 The Chief Finance Officer, in consultation with a Cabinet Member, can give approval for payment in advance of goods or services in exceptional circumstances and on request from a Chief Officer.
- 9.5 Before authorising an account, the authorising officer shall be satisfied that:
  - (a) the work, goods or services to which the account relates have been received, carried out, examined and approved;
  - (b) the invoice satisfies VAT regulations and that prices, extensions, calculations, trade discounts, other allowances and credits are correct;
  - (c) the relevant expenditure has been properly incurred and is within the relevant estimate provision;

- (d) the financial code to which the expenditure is to be charged is correct and sufficient budget is available;
- (e) the account or invoice indicates the official order number relating to the goods or services provided, or if there is no such order the reason for the omission unless it relates to services not subject to orders such as care contracts;
- (f) appropriate entries have been made in inventories, stores records or stock books as required;
- (g) the account has not been previously passed for payment and is a proper liability of the Council:
- (h) in the case of accounts for the supply of public utility services and other periodic payments, relevant expenditure, and where appropriate, units of energy consumed, have been entered in records approved by the Chief Finance Officer.
- 9.6 Accounts authorised manually should be passed for payment to the Chief Finance Officer, unless alternative arrangements have been specifically agreed with the Chief Finance Officer. All payments should be processed within 30 days.
- 9.7 Where payments are electronically uploaded into the financial system via a feeder file, appropriate controls, including standardising of files and validity checks, are put in place to ensure their integrity. For any new feeder files, Internal Audit will be asked to review and give assurance on the process for authorisation before use.
- 9.8 For the Council to reclaim VAT on individual payments, the Council is required under VAT Regulations to obtain the supplier's VAT registration number and invoice to be addressed to Bridgend County Borough Council.
- 9.9 The Chief Finance Officer shall set down procedures for the retention of financial documentation. Such documentation must not be disposed of without the specific approval of the Chief Finance Officer. All invoices shall be retained for at least 6 years plus the current financial year. In the case of invoices relating to grant claims, these must be kept until after the grant claim has been audited even if this exceeds 6 years. It is the responsibility of the Chief Officer of the grant to ensure the correct retention period is maintained. The same retention periods apply to electronic copies of all original invoices.
- 9.10 Changes to on-line authorising officers shall be notified to the Chief Finance Officer immediately.
- 9.11 Each Chief Officer shall, no later than a date specified at the financial year end, notify the Chief Finance Officer of any outstanding expenditure relating to the previous financial year to be accrued in the Statement of Accounts.

## 10. Imprest accounts

- 10.1 Where appropriate, the Chief Finance Officer shall provide imprest accounts for such officers of the Council as may need them for the purpose of defraying petty cash and other expenses. Such accounts shall be maintained in accordance with the notes of guidance issued to imprest holders by the Chief Finance Officer.
- 10.2 Where appropriate, the Chief Finance Officer shall open an account with the Council's bankers for use by the imprest holder, who shall not deliberately cause the account to be overdrawn. Any prolonged overdrawn balance on an imprest holder's bank account shall be reported to

- the Chief Finance Officer. Where an officer holds a cheque book in respect of any account he must ensure that it is securely stored and all cancelled cheques are crossed and retained with the counterfoils of the original cheque book.
- 10.3 Any officer responsible for an imprest account shall be personally responsible for making good any deficiencies in that account.
- 10.4 No income received on behalf of the Council may be paid into an imprest account but must be dealt with in accordance with paragraph 15.3.
- 10.5 Except as otherwise agreed between the Chief Finance Officer and the Chief Officer concerned, payments out of the accounts shall be limited to petty disbursements and shall not include sums due to any tradesman with whom the Council has an account for the supply of goods and services, nor any account for goods exceeding in value a sum to be agreed from time to time by the Chief Finance Officer nor travelling expenses other than those of a casual nature.
- 10.6 An officer responsible for an imprest shall on a regular basis as specified by the Chief Finance Officer, or at any other time if so requested, provide a certified statement as to the state of the account.
- 10.7 Whenever an officer who is an imprest holder leaves the employment of the Council, or ceases to be entitled to hold an imprest advance, the officer shall account to the Chief Finance Officer for the amount advanced to him.

### 11. Purchasing Cards

- 11.1 Where appropriate, a purchasing card will be issued to support the current Purchase to Pay invoice process and should not be considered as a replacement mechanism to bypass the formal requisition and approval process. Expenditure shall be made in accordance with the purchasing card guidelines.
- 11.2 Each card has an individual monthly expenditure limit, individual transaction limit and restricted category types for expenditure. The cardholder will be making financial commitments on behalf of the Council and will be responsible for obtaining value for money in accordance with Contract Procedure Rules and Corporate Contracts.
- 11.3 Except, as otherwise agreed with the Chief Finance Officer, payments by purchasing card shall be limited to approved disbursements and shall not include sums due to any tradesman with whom the Council has an account. Where appropriate, invoices should be paid via the Financial System. Travelling expenses, such as train fares, can be pre-booked and paid with the purchasing card, however, the purchasing card cannot be used to reimburse travelling expenses nor subsistence expenses other than those of a casual nature.
- 11.4 The cardholder is responsible for updating the Barclaycard Spend Management system with costing, VAT and narrative details of the expenditure in a timely manner and in accordance with the purchasing card guidelines.
- 11.5 Whenever a member of staff who is a purchasing card holder leaves the employment of the Council, or ceases to be entitled to hold a purchasing card, the member of staff must return the purchasing card immediately on cessation of entitlement to the Corporate Procurement Manager for cancellation. Any replacement member of staff who requires a purchasing card must apply for a new card via the Corporate Procurement Manager.

- 11.6 Any person holding a purchasing card shall be personally responsible to notify the Corporate Procurement Manager of any unauthorised / unrecognised spend made on the card. The cardholder shall be personally responsible for making good any unauthorised spend that they incur on the card.
- 11.7 Where appropriate, the Chief Finance Officer shall provide a purchasing card enabled for cash withdrawal to give services access to cash for the purpose of defraying petty expenditure. Such cards shall be maintained in accordance with the notes of guidance to card holders issued by the Corporate Procurement Manager. There is an administration charge for cash withdrawals. Detailed records of monies withdrawn, spending of this money, including VAT amount and costing information, must be kept and be available for examination on request.
- 11.8 Except as otherwise agreed with the Chief Finance Officer, payments out of the cash withdrawn from the bank by purchasing card shall be limited to petty disbursements and shall not include sums due to any tradesman with whom the authority has an account, nor travelling expenses, nor subsistence expenses other than those of a casual nature.
- 11.9 A person responsible for the cash balance held shall, if so requested, give to the Chief Finance Officer certification as to the state of the funds.
- 11.10 Whenever a member of staff who is a cash balance administrator leaves the employment of the authority, or ceases to be entitled to administer the cash balance, the member of staff shall repay to the Chief Finance Officer the unexpended cash balance of the withdrawn amount, or shall transfer monies, records and vouchers relating to the account to the new cash administrator. When a purchasing card holder leaves, the card in their name must be returned to the Corporate Procurement Manager for cancellation and a new card applied for via the Corporate Procurement Manager unless the new cash administrator is not to be allocated a purchasing card.
- 11.11 Any person holding a cash balance obtained by a purchasing card cash withdrawal shall be personally responsible for making good any deficiencies in that account.

## 12. Construction Industry Scheme (CIS) – Tax Deduction

12.1 For certain types of construction work, the Council is obliged by HMRC's Construction Industry Scheme to deduct tax at source from sub-contractor payments, at the appropriate rate, from the invoiced amount. Those CIS sub-contractors paid under deduction must have the labour/material split shown on the invoice and entered into the Financial System. Further guidance can be obtained from the Purchase Ledger Team in Finance.

# 13. Salaries, wages and Members' allowances

- 13.1 The payment of all salaries, wages, allowances, expenses or other emoluments to all employees or members shall be made by, or under arrangements approved and controlled by, the Chief Finance Officer.
- 13.2 Each Chief Officer shall notify the Chief Finance Officer as soon as possible and in the prescribed form, of all matters affecting the payment of such emoluments and in particular:
  - (a) appointments, resignations, dismissals, suspensions, secondments, and transfers;
  - (b) absences from duty for sickness or other reason, apart from approved leave with pay;
  - (c) changes in remuneration, other than normal increments and pay awards;

- (d) information necessary to maintain records of service for superannuation, income tax, national insurance etc.;
- (e) any failure to comply with Council or statutory regulations.
- 13.3 Appointments of all employees shall be made in accordance with the regulations of the Council and the approved establishment, gradings and rates of pay.
- 13.4 All time records and other pay documents shall be in a form prescribed or approved by the Chief Finance Officer and shall be certified by or on behalf of the Chief Officer. The names of the officers authorised to certify such records shall be sent to the Chief Finance Officer by each Chief Officer together with specimen signatures and shall be amended on the occasion of any change. Changes to on-line authorising officers shall be notified to the Chief Finance Officer immediately.
- 13.5 No payments to staff may be made other than through the normal payroll system without the specific approval of the Chief Finance Officer. In particular, casual staff must be recorded on the payroll.
- 13.6 Where an Officer is hiring or engaging a staff member who is not on the Council payroll there is a legal requirement to determine whether it is the responsibility of the Council to deduct tax and national insurance at source, in accordance with the requirements of the Social Security Contributions (Intermediaries) Regulations 2000, as amended (IR35).

### 14. Officers fees, travelling and subsistence allowances

- 14.1 All claims for payment of fees, car allowances, subsistence allowances and travelling expenses shall be submitted, duly certified, in a format approved by the Chief Finance Officer, to the appropriate Chief Officer in accordance with the timetable set down by the Chief Finance Officer. A record of all officers authorised to approve such records shall be sent to the Chief Finance Officer together with specimen signatures and shall be amended on the occasion of any change. This applies to both manual and electronic expenses systems.
- 14.2 The certification by, or on behalf of, the Chief Officer shall be taken to mean that the certifying officer is satisfied that the journeys were authorised, the expenses properly and necessarily incurred, the vehicle used had appropriate insurance cover, the necessary receipts of expenditure retained and that the fees and allowances are properly payable by the Council.
- 14.3 Claims submitted more than three months after the expenses were incurred must be accompanied by a letter of explanation for the delay and shall be paid only with the express approval of the Chief Finance Officer who shall refer the matter to Cabinet if necessary.
- 14.4 All claims for payment of incidental expenses such as removal, lodging and disturbance allowances and training course fees etc. shall be made on the appropriate form and shall be in accordance with the appropriate scheme approved by the Council.

# 15. Banking arrangements and cheques

- 15.1 All arrangements with the Council's bankers shall be made by or under arrangements approved by the Chief Finance Officer, who shall be authorised to operate such bank accounts as considered necessary.
- 15.2 All cheques shall be ordered only on the authority of the Chief Finance Officer who shall make proper arrangements for their safe custody, including cheque books issued automatically by the bank.

- 15.3 Any such bank account opened in respect of monies held on behalf of the Council shall be arranged in consultation with or under the direction of the Chief Finance Officer. The Chief Finance Officer shall maintain a list of authorised signatories, along with specimen signatures, for all such accounts. Any change in signatory shall be reported to the Chief Finance Officer immediately.
- 15.4 No Standing Orders or Direct Debits from the Council's bank accounts shall be set up unless arranged in consultation with or under the direction of the Chief Finance Officer. The Chief Finance Officer shall maintain a list of all officers authorised to set up Direct Debits and Standing Orders.
- 15.5 The Chief Finance Officer shall be authorised to:
  - (a) nominate officers empowered to authorise payment out of these bank accounts;
  - (b) negotiate charges with the Council's bankers.
- 15.6 The Chief Finance Officer will be responsible for ensuring that there are adequate controls in place to ensure that all payment methods, whether physical or electronic, have appropriate authorisations, approvals and signatures as necessary.
- 15.7 The Chief Finance Officer shall ensure that appropriate arrangements are in place to facilitate the monthly reconciliation of the Council's Bank Accounts.

#### 16. Income

- 16.1 Each Chief Officer will be responsible for the prompt and accurate billing, collection and banking of all income due to the Council in connection with the Directorate's activities, except where in the interests of efficiency or security it is agreed with the Chief Finance Officer that all or part of the duties should be discharged by some other Chief Officer.
- 16.2 Except as agreed between the Chief Finance Officer and the Chief Officer concerned, all receipt books and other such items shall be ordered and supplied to Directorates by the Chief Finance Officer, who shall be satisfied as to the arrangements for their control.
- 16.3 In carrying out this function, each Chief Officer shall ensure that any officer of the staff engaged in the collection of money:
  - (a) maintains a record of receipts and bankings in a form approved by the Chief Finance Officer;
  - (b) pays without delay any money collected either:
    - i) to the approved officer;
    - ii) to a security firm employed by the Chief Finance Officer for this purpose; or
    - to one of the Council's main bank accounts or via outlets of the nominated collector(s) as agreed by the Council.
  - (c) makes no deduction from monies collected, except with the specific and exceptional approval of the Chief Finance Officer;
  - (d) ensures income is correctly coded onto the financial system nominated by the Chief Finance Officer and satisfies VAT regulations;

- (e) makes good any shortfalls in cash collected and identifies any surpluses;
- (f) in the case of cheques received, enters on any paying-in slips details with the cheque or provides supporting evidence and a reference to the related debt.
- 16.4 Personal cheques shall not be cashed out of the money held on behalf of the Council.
- Outstanding debts which are found to be irrecoverable may be written off by the Chief Finance Officer, except where it is considered that there are matters of principle or policy which should be referred to the Cabinet.
- 16.6 Any officer holding a cash float shall be personally responsible for making good any deficiencies in that account.
- 16.7 Where an officer takes cash for works, goods or services that is either:
  - i. in excess of the level stipulated within the Council's Anti-Money Laundering Policy; or
  - ii. where there is anything suspicious regarding, but not limited to, multiple use of high denomination notes, multiple and frequent disaggregation of payment of a higher value outstanding debt;

then the officer must report it immediately to the Money Laundering Reporting Officer (MLRO) in accordance with the Policy.

# 17. Fees and Charges

- 17.1 The Council has a Fees and Charging Policy to support the Medium Term Financial Strategy. It is intended to provide a consistent and co-ordinated approach to charging across the Council, setting out the key principles for charging and for reviewing charges and must be adhered to by all Chief Officers and their staff.
- 17.2 When setting charges, there is recognition that where possible, the full cost of the service should be recovered. However, there is also a recognition that in some instances, there will be a conscious decision not to fully charge or a statutory charge or limit set, and the remaining cost will be met by the Council Tax payer.
- 17.3 In line with the Medium Term Financial Strategy, a review of fees and charges should be undertaken at least annually and any new or increased charges must receive approval from Cabinet or delegated authority under the Scheme of Delegation, Scheme A paragraph 1.6. Where a proposal is made not to increase charges in a financial year, approval must be sought from the appropriate Chief Officer and Chief Finance Officer.

### 18. Value Added Tax

18.1 The Chief Finance Officer shall establish appropriate systems of control for the proper accounting of Value Added Tax (VAT) inputs and outputs. The Chief Finance Officer will produce and distribute a VAT guidance document to all Directorates and schools to enable officers to comply with HMRC VAT regulations.

## **Section C - Security and Assets**

# 19. Security

- 19.1 The Chief Finance Officer shall have overall responsibility to the Council for security of Council property.
- 19.2 Each Chief Officer shall be responsible for maintaining proper security at all times for all buildings, stocks, stores, furniture, equipment, cash, title deeds, securities, etc. under their custody.
- 19.3 Maximum limits for cash holdings shall be agreed with the Chief Finance Officer.
- 19.4 Every transfer of official money from one member of staff to another will be evidenced in the records of the Directorate concerned by the signature of the receiving officer.
- 19.5 Keys to safes and similar receptacles for the safeguarding of money or other valuables are to be carried on the person of those responsible; the loss of any such keys must be reported to the Chief Finance Officer.

#### 20. Data Protection

- 20.1 The Council's Data Protection Officer shall be responsible for undertaking the tasks set out in Section 71 of the Data Protection Act 2018 and Article 39 of the UK GDPR.
- 20.2 Each officer shall be responsible for maintaining proper security, privacy and compliance with the Data Protection Act 2018 and UK GDPR in respect of information held.
- 20.3 Each Chief Officer shall have a duty to notify the Data Protection Officer of any computer system in their Directorate which holds personal data.
- 20.4 Each Chief Officer shall have a duty to ensure the safekeeping and prevention of improper use of any information held in the Directorate, regardless of the media on which it is held.
- 20.5 The Chief Finance Officer, or an authorised representative of the Chief Finance Officer, shall have access to all computer systems and records and may require and receive such explanations as are necessary, for the purposes of the Data Protection Act 2018 and UK GDPR.
- 20.6 The security of financial systems, e.g. use of passwords, shall be maintained by adhering to instructions issued by the Chief Finance Officer and in line with the ICT Code of Conduct.

### 21. Stocks and stores

- 21.1 Each Chief Officer shall be responsible for the custody of the stocks and stores in the Directorate and shall have a duty to:
  - (a) ensure that stocks are adequate but not excessive for the purpose envisaged;
  - (b) maintain accurate and up to date records of such stocks and stores and ensure that a stocktake is undertaken at year end and the certificate is submitted to Finance;
  - (c) provide the Chief Finance Officer with such information as he requires in relation to stores for accounting, costing and financial records.
- 21.2 A Chief Officer shall arrange for periodical test examinations of stocks by persons other than storekeepers and shall ensure that all stocks are checked at least once in every year. Any

- surpluses or deficiencies revealed by such periodical test examinations shall be reported to the Chief Finance Officer and, after consultation with the Chief Officer, the Chief Finance Officer will decide what action to take.
- 21.3 Losses due to theft of stocks shall be reported to the Chief Finance Officer as soon as possible and, where found to be irrecoverable, shall be written off.
- 21.4 In all other instances write offs need the consent of the Chief Finance Officer following the submission of a report by the Chief Officer outlining the reasons for write off.

# 22. Investments, borrowings and trust funds

- 22.1 The Council has adopted the Chartered Institute of Public Finance and Accountancy (CIPFA)'s Code of Practice on Treasury Management in the Public Services.
- 22.2 The Chief Finance Officer will be responsible for preparing and presenting an annual Treasury Management Strategy to the Governance and Audit Committee prior to submission to Council for approval in advance of the start of each financial year. The Strategy must include a Borrowing Strategy, Investment Strategy and set the Council's Treasury Management Indicators for the forthcoming financial year.
- 22.3 All investment and borrowing transactions shall be undertaken in accordance with the Treasury Management Strategy with due regard to the requirements of CIPFA's Code of Practice on Treasury Management in the Public Services.
- 22.4 All investments of money under its control shall be made in the name of the Council. Any borrowing activity must have regard to CIPFA's Prudential Code for Capital Finance in Local Authorities and the Code of Practice on Treasury Management in the Public Services.
- 22.5 The Chief Finance Officer shall report quarterly to the Cabinet, summarising borrowing and investment activity and indicating compliance with any statutory or Council approved guidelines together with a half yearly and an annual report to Council.
- 22.6 The Chief Finance Officer, or an agent nominated by the Chief Finance Officer, will be the Council's Registrar of loan instruments and shall maintain records of all borrowing of money by the Council.
- 22.7 The Chief Finance Officer will have a duty to ensure a proper, efficient and effective mix of borrowing and investments.
- 22.8 The Governance and Audit Committee is responsible for ensuring effective scrutiny of the Treasury Management policies and practices. The Committee will review the Annual Report on Treasury Management as well as the Treasury Management Strategy. The Committee will make any recommendations for improvements on treasury management to the Chief Finance Officer and raise any concerns that the Council is exceeding its legal powers with the Monitoring Officer.

### 23. Inventories

- 23.1 Each Chief Officer shall be responsible for the plant, vehicles, machinery, equipment, tools, furniture and other non-consumable property in the Directorate and shall have a duty to:
  - (a) maintain an up to date inventory of such goods:

- (b) carry out an annual physical check on goods listed in the inventory;
- (c) as far as is practicable see that such goods are marked as Council property;
- (d) ensure that such goods are not removed or used except in accordance with the ordinary course of the Council's business.
- 23.2 Every transfer of items contained in the inventory from one establishment to another shall be evidenced in the records of the establishment concerned by the signature of the receiving officer.

#### 24. Insurances

- 24.1 The Chief Finance Officer shall have a duty to:
  - (a) affect adequate insurance cover in accordance with the Council's Insurance Strategy contained in the Risk Management Policy. He will arrange insurance for losses which would have a significant impact on budgets and the provision of services, where it must be bought by law and where the insurance provides additional benefits which enable an activity to take place;
  - (b) promptly claim any insurance loss the Council has suffered and liaise with the Council's Insurers to settle any claim brought by another party, if it is assessed that the Council has a legal liability to do so;
  - (c) ensure that sums insured and limits of indemnity are regularly reviewed;
  - (d) maintain comprehensive records of insurance including policy documents and understand what coverage they provide;
  - (e) employ the services of a professional insurance broking company who can provide expert advice;
  - (f) only transfer risks to Insurance Companies which are financially strong;
  - (g) comply with the "duty of fair presentation" contained in the Insurance Act 2015. The Council must disclose to its Insurers all information, facts and circumstances which are, or ought to be, known to it, which are material to the risk. A material circumstance is one which would influence the judgement of a prudent Underwriter in considering whether to provide insurance and, if so, on what basis and cost.
- 24.2 A Chief Officer will have a duty to:
  - (a) promptly notify the Chief Finance Officer of any new risks, insurable assets or liabilities which are required to be insured under the Insurance Strategy.;
  - (b) promptly notify the Chief Finance Officer in writing of any loss, liability, damage or any event likely to lead to a claim and where appropriate inform the Police;
  - (c) promptly provide any information required by the Chief Finance Officer or the Council's Insurers to progress a claim;
  - (d) assist the Chief Finance Officer to comply with the "duty of fair presentation contained in the Insurance Act 2015 by disclosing all relevant information.

- 24.3 The Chief Finance Officer shall review of all insurances at least annually, in consultation with other Chief Officers as appropriate.
- 24.4 No indemnity shall be given in the name of the Council without the prior approval of the Chief Finance Officer.

### 25. Prevention of Theft, Fraud and Corruption

- 25.1 The Chief Finance Officer is responsible for advising on effective systems of internal control to prevent fraud and corruption.
- 25.2 The Chief Finance Officer is responsible for developing, maintaining and implementing an Anti-Fraud and Bribery Policy. If a Chief Officer suspects any irregularities concerning cash, stores or other property of the Council or held on trust by the Council, they will notify the Chief Finance Officer who will take such steps as considered necessary by way of investigation and report.
- 25.3 The Chief Finance Officer is responsible for developing, maintaining and implementing an Anti-Money Laundering Policy. The key message of this Policy is to make staff aware of their responsibilities and if they suspect that money laundering activity may be taking place or proposed, they must disclose those suspicions to the Council's Money Laundering Reporting Officer who is the Chief Finance Officer.
- 25.4 The Council employs a corporate Senior Fraud Investigator, so should any fraud or irregularity be identified, whilst it would still be reported to the 'Chief Finance Officer' initially, certain matters may be referred to the Senior Fraud Investigator for investigation where appropriate.
- 25.5 The Chief Finance Officer is also responsible for developing, maintaining and implementing the Anti-Tax Evasion Policy. The policy addresses the prevention of tax evasion and will provide a coherent and consistent approach for all employees and any person who performs services for and on behalf of Bridgend County Borough Council.
- 25.6 All Chief Officers are responsible for ensuring compliance with the Anti-Fraud and Bribery Policy, Anti-Money Laundering Policy and the Anti-Tax Evasion Policy and with systems of internal control.

### 26. Estates

- 26.1 The Chief Executive Officer shall maintain a register and appropriate records detailing all properties owned by the Council including:
  - (a) purpose for which held;
  - (b) location, extent, and plan reference;
  - (c) particulars of nature of interest held;
  - (d) purchase details;
  - (e) rents payable;
  - (f) particulars of tenancies granted.
- 26.2 The Chief Finance Officer will maintain an asset register of all Council assets.

26.3 Where land and/or buildings are found to be surplus to requirements the responsible Chief Officer shall, as soon as possible, submit a report to the appropriate Cabinet Member for appropriate action in accordance with the Schemes of Delegation.

# 27. Asset Disposal

- 27.1 Directors are responsible for ensuring that all property assets which are surplus to requirements are referred to the Corporate Landlord section, who will determine the appropriate action to be taken in conjunction with Directorates. In accordance with the Council's disposal strategy the following shall be considered:
  - restrictions relating to the ownership of the asset;
  - opportunities for using the asset elsewhere in the Council;
  - market opportunities;
  - appropriate use of tendering procedures for disposal;
  - fairness in the disposal process;
  - asset security;
  - the costs of disposal in relation to the expected income.
- 27.2 All assets, including assets declared surplus where disposal is proposed must be notified to the Chief Finance Officer for approval and appropriate action in accordance with the Council's disposal strategy and Scheme of Delegation of Functions.
- 27.3 All assets, whether land, property, buildings, vehicles, plant or equipment, must be notified to the Capital Accountant to ensure their disposal is accurately recorded on the Council's Asset Register in a timely manner, and as soon after disposal as possible.

## 28. Protection of private property

- 28.1 The Chief Officer shall in any known case where steps are necessary to prevent or mitigate loss of or damage to moveable property, prepare in a form agreed with the Chief Finance Officer, an itemised inventory in each case prepared in the presence of two officers.
- 28.2 All valuables such as jewellery, watches and other small articles of a similar nature and documents of title deposited with the Council for safe custody shall be recorded in a form agreed by the Chief Finance Officer.
- 28.3 All monies deposited with the Council for safekeeping shall be dealt with in accordance with guidelines agreed by the Chief Officer and the Chief Finance Officer.
- 28.4 Where a Chief Officer is required as part of their duties to hold in trust any property, valuables or cash belonging to a third party he should do so in accordance with guidelines agreed with the Chief Finance Officer.

## 29. Risk Management

- 29.1 The Chief Finance Officer is responsible for preparing the Council's Risk Management Strategy and its promotion throughout the Council and for advising of strategic, financial and operational risks.
- 29.2 Chief Officers shall be responsible for the identification, classification and control of all risks falling within their areas of responsibility. The risks identified shall be incorporated into a Departmental Risk Register, and any risks that exceed the Council's risk threshold shall be

- notified to the Chief Finance Officer for incorporation into the Risk Register, which shall be subject to periodic review at no more than annual intervals.
- 29.3 Chief Officers shall take responsibility for risk management within their areas of responsibility, having regard to advice from the Chief Finance Officer and other specialist Officers (e.g. crime prevention, fire prevention, health and safety), and shall undertake regular reviews of risk within their own Directorates.

### **SECTION 18**

## **CONTRACT PROCEDURE RULES**

## 1. Interpretation

In these Contract Procedure Rules:-

"Appropriate Body" means the Cabinet, and any committee, panel or other body to which power has been delegated to accept tenders for Contracts of the category for which tenders are to be or have been invited.

"Appropriate Chief Officer" means whichever of the following is responsible in any particular case for inviting tenders or placing an official order:-

Chief Executive

Chief Officer Legal and Regulatory Services, Human Resources and Corporate Policy and Monitoring Officer

Corporate Director – Education and Family Support

Corporate Director - Social Services and Wellbeing

Corporate Director – Communities.

This definition also includes any Officer to which functions have been delegated under the Council's Scheme of Delegation of Functions.

"Cabinet" means the executive body of the Council established in accordance with Part II of the Local Government Act 2000.

"Cabinet Functions" means the functions for which the Cabinet is responsible.

"Cabinet Members" means the leader and the members of the Cabinet.

"Chief Audit Officer" means the Head of Audit Service for the Council.

"Chief Finance Officer" means the nominated Section 151 Officer under the Local Government Act 1972 for the time being having statutory responsibility for the administration of the Council's financial affairs.

"Community Benefits" shall include any benefit to the local community which does not directly positively impact on the operation of the Council. This can include the provision of additional facilities/resources or the training of primarily inactive people through programmes which may include apprenticeships. "Core" benefits are those which are directly related to the purpose of the Contract and can be measured as an outcome. "Non-Core" benefits are those which are not essential in the delivery of the service and will not be assessed as part of the tender evaluation process.

"Contract" means any form of contract or agreement entered into by the Council with any other party for the supply of goods, services or works, or any combination thereof.

"Contractor" means any person or organisation that undertakes a Contractual relationship with the Council.

"Contract Value" means the total value of the Contract over the whole term of the Contract. If there is no total price then the value of the Contract is to be calculated on the expected aggregate spend over the four year period, if there is no fixed term for the Contract. In the case where there is a Contract term but no total price then the value of the Contract is to be calculated on the expected expenditure over the whole Contract period, in accordance with the Regulations. All Contract values are exclusive of VAT.

"Corporate Contracts Register" means a live database of contracts let by the Council.

"Corporate Procurement Manager" means the lead Officer for the Corporate Procurement Unit, who has functional responsibility for the delivery of Procurement strategy and guidance.

"Council" means Bridgend County Borough Council.

"Data Protection Legislation" means (i) the UK GDPR and any applicable national implementing Law, as amended from time to time; (ii) the Data Protection Act 2018 to the extent that it relates to processing of personal data and privacy; and (iii) all applicable Law about the processing of personal data and privacy.

"EU" means the European Union.

"Form of Tender" means a covering document signed by the tenderer to indicate that it understands the tender and accepts the terms and conditions and other requirements of participating in the exercise a copy of which can be found on the Corporate Procurement Page.

"Framework Agreement" means an agreement between one or more contracting authorities and one or more economic operators, the purpose of which is to establish the terms governing Contracts to be awarded during a given period, in particular with regard to price and, where appropriate, the quantity envisaged.

"Internal Providers" means an organisation managed, funded and staffed by the Council, which provides a service to various directorates and schools as its core business.

"Legal Requirements" means all UK and Welsh law that governs and or relates to the procurement of goods, services and works and the letting of concession contracts.

"Monitoring Officer" means the officer for the time being designated as such under Section 5 of the Local Government and Housing Act 1989.

"Most Economically Advantageous Tender" (MEAT) means criterion which enables the Council to take account of criteria that reflect qualitative, technical and sustainable aspects of the tender submission as well as price when reaching an award decision

"Parties" means collectively the Council and any other party or more than one party entering into a Contract or agreement.

"Pre-tender Client Information Sheet" means a document that requires completion by the client before going out to tender or to obtain quotations unless a Cabinet Report has been approved a copy of which can be found on the Corporate Procurement Page.

"Regulations" means the Public Contract Regulations 2015;

"Trade Journal" means a newspaper or journal circulating among such persons or bodies as undertake Contracts of the category for which tenders are to be invited.

"UK GDPR" means the UK General Data Protection Regulation.

Words importing the singular shall include the plural and vice versa.

Any reference to any statutory provision shall include reference to any statutory modification amendment or re-enactment thereof and any subordinate legislation made under them.

### 2. Compliance with Contract Procedure Rules and Relevant European and UK Legislation

- 2.1 Every Contract shall:
  - (a) Comply with the Council's Constitution, including in particular the Financial Procedure Rules, the Scheme of Delegation of Functions and these Contract Procedure Rules;
  - (b) Comply with all relevant statutory provisions including in particular, but not limited to, the Local Government Act 1988 Part II, the Local Government (Contracts) Act 1997, the Local Government Act 1999 Part I, and the Regulations;
  - (c) Achieve Most Economically Advantageous Tender;
  - (d) Be consistent with the highest standards of integrity, having regard to the Bribery Act 2010 and the Competition Act 1998;
  - (e) Ensure fairness in allocating public Contracts;
  - (f) Comply with all Legal Requirements;
  - (g) Comply with the Equality Act 2010;
  - (h) Comply with the Modern Slavery Act 2015;
  - (i) Comply with relevant employment legislation;
  - (j) Support the Council's corporate and departmental aims and policies and comply with any Council policy;
  - (k) Comply with Data Protection Legislation:
  - (I) Apply the Welsh Government approved approach as standard to Supplier selection;
  - (m) Be procured in accordance with all procurement planning guidance issued by the Corporate Procurement Manager.
- 2.2 For the avoidance of doubt, the Appropriate Chief Officer shall ensure that where the estimated Contract Value or amount of a proposed Contract (or series of Contracts) is likely to be equal to or to exceed the thresholds set by the Legal Requirements for Contracts, then the tendering procedures to be adopted must be in accordance with the Regulations, and any statutory provision relating thereto.
- 2.3 Contracts for the procurement of works, goods or services may be subdivided into lots where appropriate and in accordance with the Regulations but lots should not be used where the result would be the avoidance or effect the application of the Contract Procedure Rules.
- 2.4 The procurement of goods, services and works shall not be artificially over estimated or underestimated which would result in either the avoidance of the Contract Procedure Rules or which would affect the application of the Contract Procedure Rules.
- 2.5 It shall be a condition of any contract between the Council and any person (not being an Officer of the Council) who is required to supervise a Contract on the Council's behalf that, in relation to the Contract, they shall comply with the requirements of these Contract Procedure Rules and the Council's Financial Procedure Rules.

- 2.6 The Monitoring Officer with the agreement of the Chief Finance Officer may issue procurement guidance or protocols, and all Chief Officers and any person who is required to supervise a Contract on behalf of the Council shall have due regard to any such guidance or protocol.
- 2.7 The use of the Council's in-house services and Internal Providers is not covered by or subject to these Contract Procedure Rules. Any contract subsequently let by the in-house services and Internal Providers to sub-contractors will be let in accordance with these Contract Procedure Rules.
- 2.8 Any failure by officers to comply with any of the provisions of these Contract Procedure Rules or associated guidance adopted by the Council may result in disciplinary action.
- 2.9 Appropriate authority must be in place before any procurement is carried out on behalf of the Council.
- 2.10 Any dispute regarding the interpretation of these Contract Procedure Rules shall be referred to the Chief Officer Legal and Regulatory Services, Human Resources and Corporate Policy, for resolution in consultation with the Corporate Procurement Manager.
- 2.11 These Contract Procedure Rules deal only with procurements conducted under the Regulations. Where a procurement is required to be conducted under the Concession Contracts Regulations 2016 and the Utilities Contracts Regulations 2016, authority must be sought from the Appropriate Body by way of an exemption to these Contract Procedure Rules in order to conduct such a procurement pursuant to Rule 3.1.7.

### 3. Exemptions and Waiver

#### **Exemptions to the Contract Procedure Rules**

- 3.1 The Contract Procedure Rules do not apply to the situations set out in Rules 3.1.1 to 3.1.7.
  - 3.1.1 For the acquisition or disposal of land, unless it involves a development agreement which would require a tender process.
  - 3.1.2 In the case of professional arrangements in accordance with Regulation 10 of the Regulations between the Councils Solicitor and Counsel or external Solicitors, the Chief Officer Legal and Regulatory Services, Human Resources and Corporate Policy, and Monitoring Officer will not be expected to comply with the requirements of this Rule when instructing and briefing Counsel or Solicitors. The Chief Officer Legal and Regulatory Services, Human Resources and Corporate Policy and Monitoring Officer will make arrangements to ensure that a record is kept of all such instructions or briefs together with a record of Counsel's/ Solicitors fees.
  - 3.1.3 For the carrying out of security works where the publication of documents in accordance with the tendering procedure could prejudice the security of the work to be done, provided the method of tendering has been approved by the Appropriate Body.
  - 3.1.4 Employment Contracts which make an individual a direct employee of the Council.
  - 3.1.5 Contracts between entities within the public sector as prescribed in Regulation 12 of the Regulations, including those Contracts where Regulation 12 would be applicable if the Contract Value was not below the financial thresholds specified in or by the Regulations, provided the arrangement has been approved by the Appropriate Body.

- 3.1.6 In an emergency situation in which injury or damage to persons or property is threatened in which case the Appropriate Chief Officer may take any necessary action but must report the circumstances and the action taken to the next meeting of the Appropriate Body.
- 3.1.7 Where a procurement is required to be conducted under the Concession Contracts Regulations 2016 or the Utilities Contracts Regulations 2016, provided the use of the Concession Contracts Regulations 2016 or the Utilities Contracts Regulations 2016 has been approved by the Appropriate Body.

#### **Waivers**

- 3.2 Waiver from obtaining quotations or tendering will only apply to the criteria listed in Rules 3.2.1 to 3.2.12 below and any waiver must be obtained, either:
- (a) in respect of any new Contract from the Appropriate Body where the value of the Contract exceeds £100,000; or
- (b) in respect of any new Contract in accordance with the Council's Scheme of Delegation of Functions where the value of the Contract does not exceed £100,000.
  - 3.2.1 Where no quotes or tenders have been received in response to the Council's procedure for obtaining competitive offers. This is to include the open procedure, restricted procedure, competitive dialogue, competitive procedure with negotiation, innovative partnerships procedures where procurement falls within the scope of the Regulations.
  - 3.2.2 The Contract involved is purely for the purpose of research, experiment, study or development under the conditions stated in the Regulations.
  - 3.2.3 The works/goods/services can be provided only by a particular economic operator for any of the following reasons:—
    - (i) the aim of the procurement is the creation or acquisition of a unique work of art or artistic performance.
    - (ii) competition is absent for technical reasons,
    - (iii) the protection of exclusive rights, including intellectual property rights,

but only, in the case of paragraphs (ii) and (iii), where no reasonable alternative or substitute exists and the absence of competition is not the result of an artificial narrowing down of the parameters of the procurement.

- 3.2.4 Extreme urgency brought about by events unforeseeable by the Council and in accordance with the strict conditions contained in the Regulations.
- 3.2.5 New works/services, constituting a repetition of existing works/services and ordered in accordance with the strict conditions contained in the Regulations, subject to approval in accordance with Rule 3.5.
- 3.2.6 A Contract awarded to the successful tenderer or one of them after a design contest.
- 3.2.7 For supplies quoted and purchased on a commodity market.

- 3.2.8 For the purchase of supplies on particularly advantageous terms from a Supplier which is definitely winding up its business activities, or from the receivers or liquidators of an insolvency, bankruptcy, an arrangement with creditors or a similar procedure.
- 3.2.9 In the case of Contracts for services entered into in pursuance of powers under the National Health Service and Community Care Act 1990, the Children Act 1989, Children's (Leaving Care) Act 2000 and Social Services and Wellbeing (Wales) Act 2014 for the benefit of an individual client, the Corporate Director Social Services and Wellbeing, insofar as it is consistent with the law, will make appropriate arrangements for provision.
- 3.2.10 Where a tendering procedure is prescribed by legislation.
- 3.2.11 Contracts which can only be performed by a statutory undertaker.
- 3.2.12 Where goods, services and works are procured by another public body on behalf of the Council or through collaborative arrangements awarded via the National Procurement Service or the Regional Delivery Group or by the Council through a Contract properly let by another such organisation, then those organisations Procurement / Contract Procedure Rules will apply subject to compliance with European Union and Procurement Law of England and Wales.

### **Modifications to existing Contracts and Framework Agreements**

- 3.3. Contracts and frameworks agreements may be modified without the requirement for a new procurement procedure in accordance with Rules 3.3.1 to Rules 3.3.6:
  - 3.3.1 Where the modifications in a Contract or Framework Agreement have been provided for in the initial procurement documents in clear, precise and unequivocal review clauses, which may include price revision clauses or options, provided that such clauses:
    - (a) state the scope and nature of possible modifications or options as well as the conditions under which they may be used, and
    - (b) do not provide for modifications or options that would alter the overall nature of the Contract or the Framework Agreement;
  - 3.3.2 Subject to Rule 3. 4 and Rule 3.5 where the modification(s) in a Contract or Framework Agreement is for additional works, services or supplies by the original Contractor that have become necessary and were not included in the initial procurement, and where a change of Contractor:
    - cannot be made for economic or technical reasons such as requirements of interchangeability or interoperability with existing equipment, services or installations procured under the initial procurement, and
    - (ii) would cause significant inconvenience or substantial duplication of costs for the Council.

provided that any increase in price does not exceed 50% of the value of the original Contract;

For the purpose of the calculation of the price the updated figure shall be the reference figure when the Contract includes an indexation clause.

- 3.3.3 Where all of the following conditions are fulfilled:
  - (i) the need for modification has been brought about by circumstances which the Council having been duly diligent could not have foreseen;
  - (ii) the modification does not alter the overall nature of the Contract;
  - (iii) any increase in price does not exceed 50% of the value of the original Contract or Framework Agreement.

For the purpose of the calculation of the price the updated figure shall be the reference figure when the Contract includes an indexation clause.

- 3.3.4 Where a new Contractor replaces the one to which the Council had initially awarded the Contract as a consequence of:
  - (i) an unequivocal review clause or option in conformity with Rule 3.3.1, or
  - (ii) universal or partial succession into the position of the initial Contractor, following corporate restructuring, including takeover, merger, acquisition or insolvency, of another economic operator that fulfils the criteria for qualitative selection initially established, provided that this does not entail other substantial modifications to the Contract and is not aimed at circumventing the application of the Regulations.
- 3.3.5 Where the modifications, irrespective of their value, are not substantial. A modification shall be considered substantial where one or more of the following conditions is met:
  - (i.) the modification renders the Contract or the Framework Agreement materially different in character from the one initially concluded; or
  - (ii.) the modification introduces conditions which, had they been part of the initial procurement procedure, would have—
    - (a) allowed for the admission of other candidates than those initially selected,
    - (b) allowed for the acceptance of a tender other than that originally accepted, or
    - (c) attracted additional participants in the procurement procedure;
  - (iii.) the modification changes the economic balance of the Contract or the Framework Agreement in favour of the Contractor in a manner which was not provided for in the initial Contract or Framework Agreement;
  - (iv.) the modification extends the scope of the Contract or Framework Agreement considerably;
  - (v.) a new Contractor replaces the one to which the Contracting authority had initially awarded the Contract in cases other than those provided for in Rule 3.3.4

- 3.3.6 Where the value of the modification is below both of the following values:—
  - (i) the relevant applicable financial thresholds contained in the Regulations, and
  - (ii) 10% of the initial Contract Value for service and supply Contracts and 15% of the initial Contract Value for works Contracts,

provided that the modification does not alter the overall nature of the Contract or Framework Agreement.

Where several successive modifications are made, the value shall be the net cumulative value of the successive modifications and the updated figure shall be the reference figure when the Contract includes an indexation clause.

- 3.4 Where several successive modifications are made:—
  - (a) the limitations imposed by the proviso at the end of Rule 3.3.2 and by Rule 3.3.3 (iii) shall apply to the value of each modification; and
  - (b) such successive modifications shall not be aimed at circumventing the Regulations.
- 3.5 Where the Council has modified a Contract in accordance with Rule 3.3.2 or Rule 3.3.3 the Council is required to send a notice to that effect, in accordance with Regulation 51 of the Regulations for publication. Such a notice shall contain the information set out in Regulation 72.4 of the Regulations.
- 3.6 Approval to any modification of an existing Contract or Framework Agreement must be obtained, either:
  - (a) from the Appropriate Body where the value of the modification of a Contract or Framework Agreement exceeds £100,000; or
  - (b) in accordance with the Council's Scheme of Delegation of Functions where the value of the modification of a Contract or Framework Agreement does not exceed £100,000.

### **Notification of Waiver and Modification**

- 3.7 If the Appropriate Chief Officer is seeking from the Appropriate Body a waiver from the Council's Contract Procedure Rules under Rule 3.2 or a modification to an existing Contract or Framework Agreement under Rule 3.3, the report to the Appropriate Body shall set out the reason for requiring the waiver or modification and the Contract Procedure Rule(s) from which the waiver or modification is required. The report shall include a legal and financial statement and shall highlight any future commitments (whether or not of a financial character) which the proposed Contract might entail.
- 3.8 Where the value of any waiver or modification does not exceed £100,000, the reasons for requiring the waiver or modification must be approved by the Corporate Procurement Manager, and recorded in accordance with the Scheme of Delegation of Functions.
- 3.9 Where a waiver is sought by the Appropriate Chief Officer in respect of obtaining quotations or tendering, the report to the Appropriate Body or record of the decision under the Council's Scheme of Delegation of Functions shall in addition to the requirements of Rules 3.7 and 3.8

- justify the use of an alternative method of selection so that propriety, value for money and compliance with Legal requirements can be demonstrated.
- 3.10 Purchasing arrangements made for the Council by the Corporate Procurement Manager do not constitute exceptions as such, because they are made under the full provisions of these Rules.
- 3.11 Correct use of these arrangements, as advised by the Corporate Procurement Manager will be sufficient to ensure officers meet their responsibilities for compliance. Such arrangements are mandatory and include, but are not limited to:
  - 3.11.1 Purchasing Frameworks endorsed for use by the Council, where items are specified and costed and the option of a Direct Award has been provided for.i.e. no requirement to undertake a further mini competition.
  - 3.11.2 Collaborative procurement arrangements undertaken by, or in consultation with the Corporate Procurement Manager.
  - 3.11.3 Electronic systems and catalogues endorsed for use by the Council in procurement.
- 3.12 The Appropriate Chief Officer must ascertain whether there is a Framework Agreement approved by the Corporate Procurement Manager for use by the Council. Where a Framework Agreement is available, it must be used in accordance with the guidance provided by the Corporate Procurement Manager.
- 3.13 Any joint procurement arrangements with other local authorities and or public sector bodies including membership or use of any consortia must be approved by the Corporate Procurement Manager.
- 3.14 Established framework and consortia arrangements endorsed by the Corporate Procurement Manager for use by the Council shall be mandatory except where there is an applicable waiver or exemption, a copy of which shall be sent to the Corporate Procurement Manager.
- 3.15 Contracts to be awarded on the basis of a Framework Agreement must be in conformity with the Regulations. Where it is necessary to run a further mini competition, any subsequent minicompetition must be in accordance with the terms of the original Framework Agreement (except for the requirement to advertise and those stages in the procurement that are not applicable to a framework).
- 3.16 These Contract Procedure Rules do not apply to Contracts let by school or college governing bodies acting in accordance with their approved schemes of delegation or contract procedure rules.

#### 4. Delegated Authority to Enter into Contracts

- 4.1 Each Chief Officer has been allocated the following functions under Scheme B2 of the Council's Scheme of Delegation of Functions (Functions delegated to Chief Officers without consultation and call-in) which shall be exercised where any proposed Contract administered by their Directorate/Office has an estimated value not exceeding £1,000,000.00:
  - (a) To authorise the invitation of tenders;
  - (b) To accept the Most Economically Advantageous Tender received and enter into a Contract and any further deeds and documents which are supplemental to the Contract.

- 4.2 Each Cabinet Member has been allocated the following functions under Scheme A of the Council's Scheme of Delegation of Functions (Cabinet Functions delegated to Individual Cabinet Members) which shall be exercised where any proposed Contract falling within the Cabinet Member's portfolio has an estimated value exceeding £1,000,000.00 but not exceeding £5.000.000.00:
  - (a) To authorise the invitation of tenders;
  - (b) To accept the Most Economically Advantageous Tender received and enter into a Contract\_and any further deeds and documents which are supplemental to the Contract.
- 4.3 Where the estimated value of any proposed Contract exceeds £5,000.00.00, the authority to do any of the following shall be sought from the Appropriate Body:
  - (a) To authorise the invitation of tenders;
  - (b) To accept the Most Economically Advantageous Tender received and enter into a Contract and any further deeds and documents which are supplemental to the Contract.

#### 5. Declaration of Interest

- 5.1 No member, employee or agent of the Council shall improperly use their position to obtain any personal or private benefit from any procurement entered into by the Council.
- 5.2 Members and employees of the Council shall comply with the requirements of Section 117 of the Local Government Act 1972, the Bribery Act 2010, and the Officers and Members' Code of Conduct set out in the Constitution in respect of the declaration of interests in Contracts with the Council.
- 5.3 Such interests must be declared to the Monitoring Officer for inclusion in the appropriate registers
- 5.4 Officers must complete a conflict of interest form and declaration of interest form prior to participating in any tender exercise and a copy can be found on the Corporate Procurement Page.

#### 6. Selective Tendering, Invitation of Tenders and Quotations

- 6.1 Subject to any overriding Legislative Requirements, all procurements shall be invited in accordance with either Rules 7, 8 or 9.
- 6.2 The appropriate service budget holder shall prepare the relevant part of the Pre-tender Client Information Sheet for all invitations to tender and quotations and submit that to the Corporate Procurement Manager for approval.
- 6.3 A record of tenders and quotations shall be maintained by the Chief Officer Legal and Regulatory Services, Human Resources and Corporate Policy and Monitoring Officer, on the Corporate Contracts Register.
- In the case of procurements that are required to be undertaken in accordance with the Regulations, the advice of the Corporate Procurement Manager must be sought at the outset and at all subsequent stages of the procurement process, and the Pre-Tender Client Information Sheet must be completed. The Regulations will take precedence over the Contract Procedure Rules.

- The value of a procurement means the estimated total monetary value over its full duration, including any extension options. Where there is no fixed term for the Contract, the value of the Contract shall be taken to be the estimated value of the Contract over a period of four years. No procurement may be artificially split to avoid compliance with these Contract Procedure Rules and Legal Requirements. This Rule shall, however, not be construed as to avoid the requirements relating to "lots" under the Regulations.
- 6.6 For works Contracts all Contractors must be registered and verified on Constructionline. Constructionline show the Contractors and categories which have been pre-assessed with a recommended contract value. The recommended contract value is based on a financial risk assessment and is the maximum limit of work that must be placed with a Contractor without additional safeguards.
- 6.7 The basis of selection for companies on Constructionline shall be by way of rotation and/or the suitability of Contractors to undertake the project. Reasons for such choice must be recorded

#### **Quotations**

- 6.8 The invitation to submit quotations must include the specification, terms and conditions and evaluation criteria/weightings.
- 6.9 A reasonable time must be allowed for Suppliers to return bids. Unless otherwise dictated by urgency, a minimum time period of not less than 14 days must be allowed for the return of quotations.
- 6.10 In accordance with the requirements of the Scheme of Delegation of Functions a record shall be kept by the Appropriate Chief Officer in the prescribed form. In any case in which at least three alternative written offers or quotations were not obtained, the Appropriate Chief Officer shall record the reasons why it was considered impracticable to do so. This should then be recorded on the Corporate Contracts Register.

### **Tenders**

- 6.11 All of the invitation to tender documentation, including specification, evaluation criteria/weightings, must be approved by the Corporate Procurement Manager.
- 6.12 A minimum time period of not less than 14 days must be allowed for the return of tenders, except where the tender is subject to the Regulations in which case the time periods specified in the Regulations are applicable.
- 6.13 For all Contracts the Council must consider how what is proposed to be procured might improve the economic, social and environmental and cultural well-being of the area and how, in conducting the procurement process, it might act with a view to securing that improvement. The Council must consider the sustainable development principles and the well-being objectives that the Council has adopted in accordance with the Well-being of Future Generations (Wales) Act 2015. Guidance on delivering social value through procurement is available on the Corporate Procurement Page on the Intranet.
- 6.14 Where appropriate the Council will need to include Community Benefits clauses to maximise the benefits that such clauses can bring. Where Community Benefits are included in any tender issued by the Council, this must be mentioned in the tender advertisement, indicating whether they are Core or Non-Core.

- 6.15 Every written Contract shall include clauses in a form approved by the Monitoring Officer relating to the following matters as a minimum.
  - 6.15.1 Data Protection;
  - 6.15.2 Freedom of Information;
  - 6.15.3 Modern Slavery;
  - 6.15.4 Bribery and Corruption;
  - 6.15.5 Equalities and Human Rights;
  - 6.15.6 Welsh Language Standards;
  - 6.15.7 Audit.

Template provisions for the aforementioned matters can be found on the Corporate Procurement Page on the Intranet.

- 6.16 Where a relevant specification issued by the British Standards Institution is current at the date of the tender all Council specifications shall require as a minimum that the goods and materials used in the execution of the Contract shall be in accordance with that specification or an equivalent.
- 6.17 The Council is a signatory to the UK Steel Charter. The Appropriate Chief Officer must consider whether fulfilment of any of the steps contained within the Charter are applicable to the Contract being procured and if so liaise with the Corporate Procurement Manager to ensure that appropriate provisions are contained within the procurement documents. A copy of the UK Steel Charter can be found on the Corporate Procurement Intranet page.
- 6.18 The Council is a signatory to the Welsh Government Code of Practice Ethical Employment in the Supply Chain. The Appropriate Chief Officer must ensure that employment practices are considered as part of the procurement process and liaise with the Corporate Procurement Manager to ensure that appropriate provisions are contained within the procurement documents. A copy of the Welsh Government Code of Practice Ethical Employment in the Supply Chain can be found on the Corporate Procurement Intranet page.
- 6.19 The Council has adopted the principles contained within the Welsh Government Policy on Project Bank Accounts. The Appropriate Chief Officer must consider whether payment through a project bank account (PBA) is appropriate for any construction or infrastructure Contract (or other Contract with a significant sub-contracting supply chain) of over 6 months duration and valued at £2 million or more and where there are sub-contracting opportunities valued at 1% or more of the contract value. The use of payment through a PBA for such Contracts will be required unless there are compelling reasons not to do so. The use of payment through a PBA must be applied to all construction and infrastructure projects and any other appropriate contracts valued at £2m or more which are part or fully funded by Welsh Government where the terms of such funding require a PBA to be applied.

# 7. Thresholds and Advertising Requirements

- 7.1 Thresholds for procurement do not apply to existing Framework Agreements endorsed for use by the Council.
- 7.2 Thresholds for all goods & services (except works Contracts) and advertising requirements are as follows and are set out in the Summary of Advertising (a copy of which can be found on the Corporate Procurement Page).
  - 7.2.1 Up to £5,000 it is necessary only to demonstrate that value for money is being achieved, unless any grant funding conditions dictate otherwise.

- 7.2.2 Over £5,000 to £25,000 Minimum of 3 quotes to be requested from businesses preferably registered on Sell2Wales, unless any grant funding conditions dictate otherwise. All quotes must be conducted electronically.
- 7.2.3 Over £25,000 to £100,000 all requirements to be openly advertised on Sell2Wales where practical and providing it represents best value for money. A minimum of three quotations must be invited when following a restricted procedure or when an advert has not been issued, unless any grant funding conditions dictate otherwise. All quotes must be conducted electronically.
- 7.2.4 Over £100,000 to the Regulations thresholds, requirements must be tendered, and advertised on Sell2Wales (as a minimum). Where the restricted procedure is used a minimum of five tenders must be invited, unless any grant funding conditions dictate otherwise. All tenders must be conducted electronically.
- 7.2.5 Above the Regulations threshold, requirements must be tendered and advertised on Find a Tender, together with such other advertising as directed by the Corporate Procurement Manager. All tenders must be conducted electronically.
- 7.3 Thresholds for Works Contracts and advertising requirements are as follows and are set out in the Summary of Advertising (a copy of which can be found on the Corporate Procurement Page):
  - 7.3.1 Contract Up to £5,000 it is necessary only to demonstrate that value for money is being achieved, using contractors with Safety Schemes in Procurement (SSIP) accreditation and registered and verified on Constructionline unless any Grant conditions dictate otherwise.
  - 7.3.2 Over £5,000 to £25,000 quotations are to be requested from a minimum of three prospective Tenderers with Safety Schemes in Procurement (SSIP) accreditation and registered and verified on Constructionline, unless any grant conditions dictate otherwise. Where the minimum number of Tenderers required through Constructionline are not available, the Corporate Procurement Unit will put an advertisement on Sell2Wales for the requirement. All quotes must be conducted electronically.
  - 7.3.3 Over £25,000 to £250,000 quotations are to be requested from a minimum of five prospective Tenderers with Safety Schemes in Procurement (SSIP) accreditation and registered and verified on Constructionline, the minimum number of Tenderers required through Constructionline are not available, the Corporate Procurement Unit will put an advertisement on Sell2Wales for the requirement. All quotes must be conducted electronically.
  - 7.3.4 Over £250,000 to the Regulations thresholds requirements must be tendered and advertised on Sell2Wales, unless any grant funding conditions dictate otherwise. All tenders must be conducted electronically.
  - 7.3.5 Above the Regulations thresholds, requirements must be tendered and advertised on Find a Tender, together with such other advertising as directed by the Corporate Procurement Manager. All tenders must be conducted electronically.
- 7.4 Where it is considered appropriate to use additional advertising e.g. specialist requirements, they should be placed in one or more of the following:
  - > A local newspaper.

- Trade Journal.
- 7.5 Where additional advertisements have been placed as outlined in Rule 7.4 the advertisement shall:
  - Specify details of the Contract into which the Council wishes to enter.
  - > Invite persons or bodies interested to apply for permission to tender.
  - Specify a time limit of not less than 14 days within which such applications or tenders are to be received by the Council, except where legislation or funding conditions dictates a different time period. Indicate the criteria and weighting to be used to select Tenderer and award the Contract, unless this is specified in the ITT.

#### 8. Pre-Tender Market Research and Consultation

8.1 The Council may conduct market consultation prior to the issue of any Tender. This may involve seeking or accepting advice from the market. This advice can be used in the planning and conducting of the procurement process as long as it does not distort competition or violate the principles of non-discrimination and transparency.

### 9. Procurement by Consultants

- 9.1 Where the Council uses consultants to act on its behalf in relation to any procurement, then the Appropriate Chief Officer shall ensure that the consultants carry out any procurement in accordance with these Contract Procedure Rules.
- 9.2 All decisions must be made in accordance with the Council's Scheme of Delegation of Functions.
  - 9.3 No consultant shall make any decision on whether to award a Contract or who a Contract should be awarded to. The Appropriate Chief Officer shall ensure that the consultant's performance in relation to procurement is in accordance with these Contract Procedure Rules and legislation.
  - 9.4 Where the Council uses consultants to act on its behalf in relation to any procurement the consultant must declare any potential conflict of interest that may arise to the relevant Appropriate Chief Officer prior to the commencement of the procurement process or at such time that the Contractor becomes aware of such a potential conflict of interest.
  - 9.5 Where the Appropriate Chief Officer considers that such a conflict of interest is significant the consultant shall not be allowed to participate in the procurement process.

#### 10. Other Procurement Procedures

- 10.1 The use of the negotiated procedure without prior publication, innovative partnerships, competitive procedure with negotiation and competitive dialogue as defined in the Regulations, can only be used in exceptional circumstances as directed by the Chief Finance Officer and with prior consultation with the appropriate Cabinet Member and the Monitoring Officer.
- 10.2 The use of reserved, concessionary contracts and dynamic purchasing systems may only be used following consultation and prior approval by the Corporate Procurement Manager.

# 11. Naming Products Sub-Contractors and Suppliers

- 11.1 The principle of non-discriminatory and equal treatment is undermined by the naming of products, Suppliers or sub-Contractors. Equivalents are nearly always able to be specified and will be advertised with the wording "or equivalent" where possible.
- 11.2 Where the Council has determined that a particular type of product or provision of service will be stipulated as an essential requirement of a Contract then this must be approved by the Appropriate Chief Officer following consultation with the Corporate Procurement Manager.

# 12. Electronic Tendering

- 12.1 An electronic quote and tender facility now supersedes the use of paper tenders in line with the Regulations. All electronic quotes and tenders must be issued using a portal approved by the Corporate Procurement Manager. All relevant staff must be fully trained in the use of the system and set up to use the facility by the Corporate Procurement Manager.
- 12.2 For all quotes and tenders issued, the awarded Contract must be uploaded to the Corporate Contracts Register on the approved portal.
- 12.3 An advertisement must be placed in accordance with Rule 7. Any advert placed on sell2wales must alert prospective bidders to the requirement and the need to register on the approved etenderwales portal. The advert should state that bidders must have JAVA to access the etenderwales portal and respond electronically.
- 12.4 All correspondence between the Council and bidders must be concluded via the approved portal to maintain a defined audit trail. Under no circumstances must email correspondence outside the portal be used. Any verbal communication specific to the quote or tender must only be communicated through the relevant category specialist and must be confirmed via the portal. All clarification questions must be broadcast to all prospective bidders, where applicable i.e. not commercially sensitive.
- 12.5 The requirements for the submissions of bids must be set out in the Invitation to Tender (ITT) document issued to bidders via the portal. This will specify the return date and time, which will normally be 12 noon on the agreed date. The time and date when bidders upload their completed documents onto the system is recorded and forms part of an audit record.
- 12.6 Any late bids submitted will be recorded by the system and must be disqualified. A report will be required from the system administrator of the portal that there has been no system malfunction if a bidder challenges the disqualification.
- 12.7 Tenders must be opened electronically in accordance with Rule 14.

### 13. Submission and Custody of Tenders and Quotations

- 13.1 Where tenders are invited in accordance with these Contract Procedure Rules or otherwise as determined by the Appropriate Body, the requirements for submission of tenders shall be set out in the invitation to tender document.
- 13.2 Electronic tenders and quotes are held in the approved portal and are inaccessible until the specified closing date and time.

## 14. Opening of Electronic Tenders

- 14.1 Quotations or tenders for requirements having an estimated value or amount of £5,000 and not exceeding £100,000 shall be opened via the approved portal at the same time in the presence of one officer from the corporate procurement team, this is to include the technical and commercial envelope.
- 14.2 Tenders for Contracts having an estimated value or amount over £100,000 and not exceeding £1,000,000 shall be opened at the same time in the presence of two officers, one designated by the Monitoring Officer and the other an officer from the corporate procurement team, this is to include the technical and commercial envelope.
- 14.3 Tenders, including the technical and commercial envelope, having an aggregated estimated value or amount exceeding £1,000,000 shall be opened at the same time in the presence of the Monitoring Officer or their nominated representative and two further officers usually an officer from Democratic Services and an officer from the corporate procurement team.
- 14.4 All tenders opened in accordance with rule 14 shall include the Technical and Commercial envelope within the approved portal.
- 14.5 A record shall be kept of all tenders and quotations received through the approved portal approved by the Monitoring Officer, including any reasons for disqualification.

#### 15. Amendment of Tenders

- 15.1 If an error is identified in the invitation to tender documentation before the closing date for the return of tenders, all Tenderers shall be appraised of the error and invited to adjust their tenders.
- 15.2 If an error in the invitation to tender documentation is identified after the closing date for the return of tenders, but prior to opening of tenders, all Tenderers shall be given details of the error and afforded the opportunity of withdrawing the tender or submitting an amended tender.
- 15.3 Except as provided in this Rule 15, a tender may not be amended by the Tenderer after it has been received unless examination of it reveals errors or discrepancies which would affect the tender figure in an otherwise successful tender, in which case:
  - 15.3.1 If the error is not arithmetical the Tenderer shall be given details of the error(s) and shall be given the opportunity of confirming without amendment or withdrawing the tender, or
  - 15.3.2 If the error is arithmetical the Appropriate Chief Officer shall correct the error(s) provided that, apart from these genuine arithmetic error(s), no other adjustment revision or qualification is permitted.

#### 16. Post-Tender Negotiation and Clarification

- 16.1 Where procurement is conducted pursuant to the Regulations through either the open, restricted, innovative partnerships or competitive dialogue procedures, no post-tender negotiations are permitted. Any negotiation in respect of the competitive procedure with negotiation must comply with the requirements of the Regulations.
- 16.2 The responsible officer may seek clarification from tenderers where appropriate in consultation with the Corporate Procurement Manager.

- 16.3 Where procurements do not exceed the Regulations thresholds the Appropriate Chief Officer may authorise negotiations in exceptional circumstances, following the opening of tenders and before acceptance, subject to compliance with any legislative requirements, and only when:
  - (a) Amendments have been made to any relevant specification since the invitation of tenders, which could not have been foreseen at the invitation stage, or
  - (b) The volume of goods, works or services required by the Council has substantially increased or decreased since the invitation of tenders, or
  - (c) Only one tender has been received, or
  - (d) Tender prices exceed financial targets predetermined by the Council, the Cabinet or the appropriate Committee,

In all instances, the possibility of such negotiations must have been indicated in the invitation to tender.

16.4 Negotiations shall be conducted in a manner which treats all Tenderers equally and is non-discriminatory and transparent at all times during the procurement process, and a full record of the negotiations shall be made by the Appropriate Chief Officer which shall include the names of the officers involved, minutes of all meetings and copies of all relevant correspondence and other documents.

#### 17. Evaluation of Tenders

- 17.1 The Appropriate Chief Officer shall examine tenders in accordance with predetermined evaluation criteria.
- 17.2 In respect of all tenders the tender criteria shall be Most Economically Advantageous Tender (which can be the lowest/highest price only where appropriate) and the evaluation criteria shall as a minimum be listed in the Invitation To Tender documentation in order of importance. Any particular scoring or weighting attributable to any criteria and sub-criteria must be clearly stated. In addition, the criteria must be strictly observed and remain unchanged at all times throughout the Contract award procedure.

## 18. Acceptance and Award of Tenders

- 18.1A Contract shall only be awarded subject to the tender evaluation criteria specified in the invitation to tender document.
- 18.2 Tenders for Contracts having an estimated value exceeding £5,000,000 may be invited and accepted only by the Appropriate Body unless the power to do so has been specifically allocated or delegated to the appropriate Cabinet Member or Appropriate Chief Officer.
- 18.3 No tender relating to a Contract to which Sections 2 to 9 of the Local Government (Contracts) Act 1997 apply, shall be accepted without the approval of the Monitoring Officer.
- 18.4 Before a tender is accepted other than for the sale of goods the Appropriate Chief Officer must obtain the Chief Finance Officer's confirmation that the Tenderer's financial standing is satisfactory, unless the Tenderer has already been approved using the prequalification of the tender process.
- 18.5 The award of any Contract shall be in accordance with the Regulations and any other Legal Requirements.

- 18.6 If the award of any Contract is subject to the Regulations then there shall be a standstill period of at least ten calendar days between the communication of the decision of the intention to award the Contract being notified to those parties who were selected to tender, or who applied to be selected to tender, and the entering into of that Contract.
- 18.7 Standstill letters issued as part of a tender process under the Regulations must not be issued prior to confirmation of award in accordance with the Council's constitution.
- 18.8 Feedback to unsuccessful tenderers must be included in the standstill letters and must include the characteristics and relative advantages of the tender selected. For Contracts subject to the Regulations, the requirements of the Regulations must also be complied with.
- 18.9 Where a tender above the Regulations threshold has been tendered and advertised on Sell2Wales, the Council shall publish a Contract Award Notice on Sell2Wales which will also be included on Find a Tender, as soon as possible after the decision to award and in any event no later than 30 calendar days after the date of award of the Contract.
- 18.10 All correspondence notifying evaluation outcomes must be communicated through the approved portal

#### 19. Letters of Intent

- 19.1 Letters of Intent shall only be used with the agreement of the Chief Officer Legal and Regulatory Services, HR and Corporate Policy and Monitoring Officer and only in very exceptional circumstances.
- 19.2 Where the terms and conditions of the Contract are not fully agreed no Contractor shall be allowed on site to begin preliminary works until a full risk assessment has been carried out by the Appropriate Chief Officer as to possible implications to the Council by the Contractor being allowed to start on site before the Contract terms and conditions have been finalised.

## 20. Execution of Contracts

- 20.1 Every Contract not exceeding £500,000 in amount or value shall be in writing in a form approved by the Monitoring Officer such form being the Form of Tender. However, the Form of Tender should not be used in the following situations and a formal Contract must be entered into:
  - (a) where the appropriate Directorate has assessed the level of risk associated with the Contract and it deems the level of risk to be high; or
  - (b) where the Contract must be executed under seal.

The Form of Tender must be issued as part of the tender and must be signed and returned by the bidder. On contract award the signed Form of Tender must be issued with the award letter, signed by either the Appropriate Chief Officer of the Monitoring Officer, or another officer authorised in writing to sign on behalf of the Appropriate Chief Officer or the Monitoring Officer. Both Parties are to keep a signed Form of Tender.

- 20.2 Every Contract, including frameworks which exceeds £500,000 in amount or value shall be referred to the Solicitor for the Council for approval.
- 20.3 Contracts in writing or official orders not exceeding £500,000 in amount or value may be signed either by the Appropriate Chief Officer or by an officer in writing authorised to approve on behalf of the Appropriate Chief Officer.

- 20.4 Every Contract for Works (other than Framework Agreements) exceeding £1,000,000 shall require a performance bond and for Contracts below this limit the Appropriate Chief Officer shall determine the necessity for a Performance Bond based on an assessment of risk for that Contract. A Surety shall be approved by the Chief Finance Officer.
- 20.5 A risk assessment shall be conducted for all service Contracts to determine the necessity for a performance bond. The approval of the Chief Finance Officer is required where a performance bond is not deemed necessary for service Contracts in excess of £1,000,000 in value.
- 20.6 Contracts not exceeding £500,000 may be signed electronically except in the following situations:
  - (a) where the appropriate Directorate has assessed the level of risk associated with the Contract and it deems the level of risk to be high; or
  - (b) where the Contract must be executed under seal.

The use of electronic signature must be in accordance with the procedure set down by the Corporate Procurement Manager.

## 21. Assignment and Novation

21.1 Any Contract subject to potential assignment or novation must be referred to the Monitoring Officer at the earliest possible instance.

#### 22. Termination of Contracts

22.1 For any Contract, or part thereof, exceeding £100,000 in value, early termination shall be approved by the Appropriate Chief Officer in consultation with the Monitoring Officer and the Chief Finance Officer and must be in accordance with the termination provisions set out in the Contract. Contracts of a lesser value may be terminated early by agreement prior to the expiry date or in accordance with the termination provisions set out in the Contract.

## 23. Financial Limits

23.1 With the exception of the financial limit specified in Rule 20, which may be varied only by the Council, the financial limits specified in these Contract Procedure Rules or any of them may be varied at any time by the Chief Finance Officer in consultation with the Monitoring Officer, but any alterations shall be reported to the next meeting of the Council.

#### 24. Record and Document Retention and Control

- 24.1 A full report of all Contracts awarded should be written and kept electronically. (A checklist of what the report should contain can be found on the Corporate Procurement Page.)
- 24.2 A full record of all documents used including the Delegated Power to invite and accept, any other internal approvals, should be scanned and uploaded into the approved portal or procurements 'o' drive against the specific Contract reference number.
- 24.3 All Contracts under seal i.e. Deeds (inc. Tenders) must be retained for a minimum of 12 years after the term of the Contract has expired. Contracts which have not been sealed must be

- retained for a minimum of 6 years after the term of the Contract has expired, unless a longer period is required by legislation or grant conditions.
- 24.4 Unsuccessful tender and pre-qualifying documents which have been accepted for tender purposes will be retained for a minimum period of 2 years after the start of the Contract within the approved portal.
- 24.5 Any tender which is rejected or disqualified at the time of opening will be notified via the approved portal.
- 24.6 All request for quotations must be retained for a minimum period of 2 years after the term of the Contract has expired.
- 24.7 Where Contracts are subject to the Regulations, they must also comply with the record, document retention and control requirements of the Regulations.

#### 25. Contract Management

- 25.1 A Contract must not start until the Contract has been signed or sealed by both parties and dated. Directorates must allow time for this in their forward work plan.
- 25.2 Within ten calendar days of the start of the Contract, the directorate must add the details of the Contract onto the Corporate Contracts Register and record all the required information within etender wales.
  - These details must be maintained and updated as and when required to do so. Officers should contact the Corporate Procurement Manager if training is required.
- 25.3 The Directorate shall be the Contract manager for the Contracts for which they are responsible for up to the Regulations threshold, but may authorise another officer to have day-to-day responsibility for managing the Contract. For Contracts over the Regulations threshold, the directorate must appoint a dedicated Contract manager with appropriate skills and experience.
- 25.4 The Contract manager is responsible for:
  - (a) Managing the Contract and ensuring that it is carried out in accordance with its terms and conditions;
  - (b) Monitoring the Contractor's performance and carrying out mid-Contract reviews to determine ongoing value for money:
  - (c) Ensuring that the Contractor maintains the insurance policies required by the Contract; and
  - (d) Keeping a record of all payments, claims, monitoring, changes and certificates under the Contract.
- 25.5 The Contract manager must consult the Monitoring Officer for consideration of the Council's legal position:
  - (a) Before any Contract is to be terminated or suspended; or
  - (b) In the event of a claim for payment not clearly within the terms and conditions of Contract.

#### 26. Corporate Contracts Register

26.1 The Corporate Contracts Register is held within the approved portal and all quotations and tenders conducted must be created into a Contract. It is the responsibility of each Directorate to create and maintain their own Contracts within the portal.

- 26.2 It is the responsibility of each Directorate to update the register following a quotation or tender process within the approved portal.
- 26.3 Any officer that requires training must contact the corporate procurement unit.

#### 27. Review and Amendment of Contract Procedure Rules

- 27.1 The Corporate Procurement Manager shall continually review these Contract Procedure Rules and shall undertake a formal review at least every three years.
- 27.2 With the exception of Rule 23.1 the Monitoring Officer shall have authority to make alterations of a minor nature or such as are required to be enacted immediately to ensure compliance with statutory obligations. Any minor changes will be reported to the next meeting of the Council.

## **SECTION 19**

## 19. CODE OF CONDUCT FOR MEMBERS

#### Part 1 - Interpretation

1.—(1) In this code —

"co-opted member" ("aelod cyfetholedig") in relation to a relevant authority, means a person who is not a member of the authority but who —

is a member of any committee or sub-committee of the authority, or

is a member of, and represents the authority on, any joint committee or joint sub-committee of the authority, and who is entitled to vote on any question which falls to be decided at any meeting of that committee or sub-committee;

"meeting" ("cyfarfod") means any meeting —

(a) of the relevant authority,

of any executive or board of the relevant authority,

of any committee, sub-committee, joint committee or joint sub-committee of the relevant authority or of any such committee, sub-committee, joint committee or joint-committee of any executive or board of the authority, or

where members or officers of the relevant authority are present other than a meeting of a political group constituted in accordance with regulation 8 of the Local Government (Committee and Political Groups) Regulations 1990,

and includes circumstances in which a member of an executive or board or an officer acting alone exercises a function of an authority;

"member" ("aelod") includes, unless the context requires otherwise, a co-opted member;

"registered society" means a society, other than a society registered as a credit union, which is-

- (a) a registered society within the meaning given by section 1(1) of the Co-operative and Community Benefit Societies Act 2014; or
- (b) a society registered or deemed to be registered under the Industrial and Provident Societies Act (Northern Ireland) 1969;

"register of members' interests" ("cofrestr o fuddiannau'r aelodau") means the register established and maintained under section 81 of the Local Government Act 2000;

"relevant authority" ("awdordod perthnasol")means—

- (b) a county council,
- a county borough council.
- a community council,
- a fire and rescue authority constituted by a scheme under section 2 of the Fire and Rescue Services Act 2004 or a scheme to which section 4 of that Act applies,
- a National Park authority established under section 63 of the Environment Act 1995;
  - "you" ("chi") means you as a member or co-opted member of a relevant authority; and "your authority" ("eich awdurdod") means the relevant authority of which you are a member or co-opted member.
- (1) In relation to a community council
- "proper officer" ("swyddog pridol") means an officer of the Council within the meaning of section 270(3) of the Local Government Act 1972; and
- "standards committee" ("pwyllgor safonau") means the standards committee of the county borough council which has functions in relation to the community council for which it is responsible under section 56(1) and (2) of the Local Government Act 2000.

#### Part 2 - General Provisions

2.(1) Save where paragraph 3(a) applies, you must observe this code of conduct —

- (a) whenever you conduct the business, or are present at a meeting, of your authority;
- (b) whenever you act, claim to act or give the impression you are acting in the role of member to which you were elected or appointed;
- (c) whenever you act, claim or act or give the impression you are acting as a representative of your authority; or
- (d) at all times and in any capacity, in respect of conduct identified in paragraphs 6(1)(a) and 7.
- (2) You should read this code together with the general principles prescribed under section 49(2) of the Local Government Act 2000 in relation to Wales.
  - 3. Where you are elected, appointed or nominated by your authority to serve
    - (a) on another relevant authority, or any other body, which includes a Local Health Board you
      must, when acting for that other authority or body, comply with the code of conduct of that
      other authority or body; or
    - (b) on any other body which does not have a code relating to the conduct of its members, you must, when acting for that other body, comply with this code of conduct, except and insofar as it conflicts with any other lawful obligations to which that other body may be subject.

#### 4.You must —

- (a) carry out your duties and responsibilities with due regard to the principle that there should be equality of opportunity for all people, regardless of their gender, race, disability, sexual orientation, age or religion;
- (b) show respect and consideration for others:
- (c) not use bullying behaviour or harass any person; and
- (d) not do anything which compromises, or which is likely to compromise, the impartiality of those who work for, or on behalf of, your authority.

#### 5. You must not —

- (a) disclose confidential information or information which should reasonably be regarded as being of a confidential nature, without the express consent of a person authorised to give such consent, or unless required by law to do so;
- (b) prevent any person from gaining access to information to which that person is entitled by law.

#### **6.**(1) You must —

- (a) not conduct yourself in a manner which could reasonably be regarded as bringing your office or authority into disrepute;
- (b) report, whether through your authority's confidential reporting procedure or direct to the proper authority, any conduct by another member or anyone who works for, or on behalf of, your authority which you reasonably believe involves or is likely to involve criminal behaviour (which for the purposes of this paragraph does not include offences or behaviour capable of punishment by way of a fixed penalty);
- (c) report to your authority's monitoring officer any conduct by another member which you reasonably believe breaches this code of conduct;
- (d) not make vexatious, malicious or frivolous complaints against other members or anyone who works for, or on behalf of, your authority.
- (2) You must comply with any request of your authority's monitoring officer, or the Public Services Ombudsman for Wales, in connection with an investigation conducted in accordance with their respective statutory powers.

#### 7. You must not —

- (a) in your official capacity or otherwise, use or attempt to use your position improperly to confer on or secure for yourself, or any other person, an advantage or create or avoid for yourself, or any other person, a disadvantage;
- (b) use, or authorise others to use, the resources of your authority
  - (i) imprudently;
  - (ii) in breach of your authority's requirements;
  - (iii) unlawfully;
  - (iv) other than in a manner which is calculated to facilitate, or to be conducive to, the discharge of the functions of the authority or of the office to which you have been elected or appointed;
  - (v) improperly for political purposes; or
  - (vi) improperly for private purposes.

#### 8. You must —

- (a) when participating in meetings or reaching decisions regarding the business of your authority, do so on the basis of the merits of the circumstances involved and in the public interest having regard to any relevant advice provided by your authority's officers, in particular by
  - (i) the Chief Executive;
  - (ii) the authority's chief finance officer;
  - (iii) the authority's monitoring officer;
  - (iv) the authority's chief legal officer (who should be consulted when there is any doubt as to the authority's power to act, as to whether the action proposed lies within the policy framework agreed by the authority or where the legal consequences of action or failure to act by the authority might have important repercussions);
- (b) give reasons for all decisions in accordance with any statutory requirements and any reasonable additional requirements imposed by your authority.

#### 9. You must —

- (a) observe the law and your authority's rules governing the claiming of expenses and allowances in connection with your duties as a member;
- (b) avoid accepting from anyone gifts, hospitality (other than official hospitality, such as a civic reception or a working lunch duly authorised by your authority), material benefits or services for yourself or any person which might place you, or reasonably appear to place you, under an improper obligation.

#### Part 3 - Interests

#### Personal Interests

- **10**(1) You must in all matters consider whether you have a personal interest, and whether this code of conduct requires you to disclose that interest.
- (2) You must regard yourself as having a personal interest in any business of your authority if
  - (a) it relates to, or is likely to affect
    - (i) any employment or business carried on by you;
    - (ii) any person who employs or has appointed you, any firm in which you are a partner or any company for which you are a remunerated director;

- (iii) any person, other than your authority, who has made a payment to you in respect of your election or any expenses incurred by you in carrying out your duties as a member;
- (iv) any corporate body which has a place of business or land in your authority's area, and in which you have a beneficial interest in a class of securities of that body that exceeds the nominal value of £25,000 or one hundredth of the total issued share capital of that body;
- (v) any contract for goods, services or works made between your authority and you or a firm in which you are a partner, a company of which you are a remunerated director, or a body of the description specified in sub-paragraph (iv) above;
- (vi) any land in which you have a beneficial interest and which is in the area of your authority;
- (vii) any land where the landlord is your authority and the tenant is a firm in which you are a partner, a company of which you are a remunerated director, or a body of the description specified in sub-paragraph (iv) above;
- (viii) any body to which you have been elected, appointed or nominated by your authority;
- (ix) any
  - (aa) public authority or body exercising functions of a public nature;
  - (bb) company, industrial and provident society, charity, or body directed to charitable purposes;
  - (cc) body whose principal purposes include the influence of public opinion or policy;
  - (dd) trade union or professional association; or
  - (ee) private club, society or association operating within your authority's area, in which you have membership or hold a position of general control or management;
- (x) any land in your authority's area in which you have a licence (alone or jointly with others) to occupy for 28 days or longer;

Note: Subparagraph (b) is omitted.

- (c) a decision upon it might reasonably be regarded as affecting
  - (xi) your well-being or financial position, or that of a person with whom you live, or any person with whom you have a close personal association;
  - (xii) any employment or business carried on by persons as described in 10(2)(c)(i);
- (xiii) any person who employs or has appointed such persons described in 10(2)(c)(i), any firm in which they are a partner, or any company of which they are directors;
- (xiv) any corporate body in which persons as described in 10(2)(c)(i) have a beneficial interest in a class of securities exceeding the nominal value of £5,000; or
- (xv) any body listed in paragraphs 10(2)(a)(ix)(aa) to (ee) in which persons described in 10(2)(c)(i) hold a position of general control or management, to a greater extent than the majority of—
  - (aa) in the case of an authority with electoral divisions or wards, other council tax payers, rate payers or inhabitants of the electoral division or ward, as the case may be, affected by the decision; or
  - (bb) in all other cases, other council tax payers, ratepayers or inhabitants of the authority's area.

#### Disclosure of Personal Interests

11.—(1) Where you have a personal interest in any business of your authority and you attend a meeting at which that business is considered, you must disclose orally to that meeting the existence

and nature of that interest before or at the commencement of that consideration, or when the interest becomes apparent.

- (2) Where you have a personal interest in any business of your authority and you make-
  - (a) written representations (whether by letter, facsimile or some other form of electronic communication) to a member or officer of your authority regarding that business, you should include details of that interest in the written communication; or
  - (b) oral representations (whether in person or some form of electronic communication) to a member or officer of your authority you should disclose the interest at the commencement of such representations, or when it becomes apparent to you that you have such an interest, and confirm the representation and interest in writing within 14 days of the representation.
- (3) Subject to paragraph 14(1)(b) below, where you have a personal interest in any business of your authority and you have made a decision in exercising a function of an executive or board, you must in relation to that business ensure that any written statement of that decision records the existence and nature of your interest.
- (4) You must, in respect of a personal interest not previously disclosed, before or immediately after the close of a meeting where the disclosure is made pursuant to sub-paragraph 11(1), give written notification to your authority in accordance with any requirements identified by your authority's monitoring officer, or in relation to a community council, your authority's proper officer from time to time but, as a minimum containing
  - (a) details of the personal interest;
  - (b) details of the business to which the personal interest relates; and
  - (c) your signature.
- (5) Where you have agreement from your monitoring officer that the information relating to your personal interest is sensitive information, pursuant to paragraph 16(1), your obligations under this paragraph 11 to disclose such information, whether orally or in writing, are to be replaced with an obligation to disclose the existence of a personal interest and to confirm that your monitoring officer has agreed that the nature of such personal interest is sensitive information.
- (6) For the purposes of sub-paragraph (4), a personal interest will only be deemed to have been previously disclosed if written notification has been provided in accordance with this code since the last date on which you were elected, appointed or nominated as a member of your authority.
- (7) For the purposes of sub-paragraph (3), where no written notice is provided in accordance with that paragraph you will be deemed as not to have declared a personal interest in accordance with this code.

#### **Prejudicial Interests**

- **12.**(1) Subject to sub-paragraph (2) below, where you have a personal interest in any business of your authority you also have a prejudicial interest in that business if the interest is one which a member of the public with knowledge of the relevant facts would reasonably regard as so significant that it is likely to prejudice your judgement of the public interest.
- (2) Subject to sub-paragraph (3), you will not be regarded as having a prejudicial interest in any business where that business—
  - (a) relates to
    - (i) another relevant authority of which you are also a member;
    - (ii) another public authority or body exercising functions of a public nature in which you hold a position of general control or management;
    - (iii) a body to which you have been elected, appointed or nominated by your authority;
    - (iv) your role as a school governor (where not appointed or nominated by your authority) unless it relates particularly to the school of which you are a governor;

- (v) your role as a member of a Local Health Board where you have not been appointed or nominated by your authority;
- (b) relates to
  - (i) the housing functions of your authority where you hold a tenancy or lease with your authority, provided that you do not have arrears of rent with your authority of more than two months, and provided that those functions do not relate particularly to your tenancy or lease:
  - (ii) the functions of your authority in respect of school meals, transport and travelling expenses, where you are a guardian, parent, grandparent or have parental responsibility (as defined in section 3 of the Children Act 1989) of a child in full time education, unless it relates particularly to the school which that child attends;
  - (iii) the functions of your authority in respect of statutory sick pay under Part XI of the Social Security Contributions and Benefits Act 1992, where you are in receipt of, or are entitled to the receipt of such pay from your authority;
  - (iv) the functions of your authority in respect of an allowance or payment made in accordance with the Provisions of Part 8 of the Local Government (Wales) Measure 2011, or an allowance or pension provided under section 18 of the Local Government and Housing Act 1989;
- (c)your role as a community councillor in relation to a grant, loan or other form of financial assistance made by your community council to community or voluntary organisations up to a maximum of £500.
- (3) The exemptions in subparagraph (2)(a) do not apply where the business relates to the determination of any approval, consent, licence, permission or registration.

## **Overview and Scrutiny Committees**

- **13.**You also have a prejudicial interest in any business before an overview and scrutiny committee of your authority (or of a sub-committee of such a committee) where—
  - (a) that business relates to a decision made (whether implemented or not) or action taken by your authority's executive, board or another of your authority's committees, sub-committees, joint committees or joint sub-committees; and
  - (b) at the time the decision was made or action was taken, you were a member of the executive, board, committee, sub-committee, joint-committee or joint sub-committee mentioned in sub-paragraph (a) and you were present when that decision was made or action was taken.

## Participation in Relation to Disclosed Interests

- 14 (1)Subject to sub-paragraphs (2), (2A), (3) and (4), where you have a prejudicial interest in any business of your authority you must, unless you have obtained a dispensation from your authority's standards committee
  - (a) withdraw from the room, chamber or place where a meeting considering the business is being held—
    - (i) where sub-paragraph (2) applies, immediately after the period for making representations, answering questions or giving evidence relating to the business has ended and in any event before further consideration of the business begins, whether or not the public are allowed to remain in attendance for such consideration; or
    - ii) in any other case, whenever it becomes apparent that that business is being considered at that meeting;
  - (b) not exercise executive or board functions in relation to that business;
  - c) not seek to influence a decision about that business;

- (d) not make any written representations (whether by letter, facsimile or some other form of electronic communication) in relation to that business; and
- e) not make any oral representations (whether in person or some form of electronic communication) in respect of that business or immediately cease to make such oral representations when the prejudicial interest becomes apparent.
- (2) Where you have a prejudicial interest in any business of your authority you may attend a meeting but only for the purpose of making representations, answering questions or giving evidence relating to the business, provided that the public are also allowed to attend the meeting for the same purpose, whether under a statutory right or otherwise.
- (2A) Where you have a prejudicial interest in any business of your authority you may submit written representations to a meeting relating to that business, provided that the public are allowed to attend the meeting for the purpose of making representations, answering questions or giving evidence relating to the business, whether under statutory right or otherwise.
- (2B) when submitting written representations under sub-paragraph (2A) you must comply with any procedure that your authority may adopt for the submission of such representations.
  - (3) Sub-paragraph (1) does not prevent you attending and participating in a meeting if —
- (a) you are required to attend a meeting of an overview or scrutiny committee, by such committee exercising its statutory powers; or
- (b) you have the benefit of a dispensation provided that you
  - (i) state at the meeting that you are relying on the dispensation; and
  - (ii) before or immediately after the close of the meeting give written notification to your authority containing
    - (aa) details of the prejudicial interest;
    - (bb) details of the business to which the prejudicial interest relates;
    - (cc) details of, and the date on which, the dispensation was granted; and
    - (dd) your signature.
  - (4) Where you have a prejudicial interest and are making written or oral representations to your authority in reliance upon a dispensation, you must provide details of the dispensation within any such written or oral representation and, in the latter case, provide written notification to your authority within 14 days of making the representation.

## Part 4 - The Register of Members' Interests

14.

#### Registration of Personal Interests

- 15 (1) Subject to sub-paragraph (4), you must, within 28 days of—
  - (a) your authority's code of conduct being adopted or the mandatory provisions of this model code being applied to your authority; or
- (b) your election or appointment to office (if that is later), register your personal interests, where they fall within a category mentioned in paragraph 10(2)(a), in your authority's register of members' interests by providing written notification to your authority's monitoring officer.
  - (2) Subject to sub-paragraph (4), you must, within 28 days of becoming aware of any new personal interest falling within a category mentioned in paragraph 10(2)(a), register that new personal interest in your authority's register of members' interests by providing written notification to your authority's monitoring officer.

- (3) Subject to sub-paragraph (4) you must, within 28 days of becoming aware of any change to a registered personal interest falling within a category mentioned in paragraph 10(2)(a), register that change in your authority's monitoring officer, or in the case of a community council to your authority's proper officer.
- (4) Sub-paragraphs (1), (2) and (3) do not apply to sensitive information determined in accordance with paragraph 16(1).
- (5) Sub-paragraphs (1) and (2) do not apply if you are a member of a relevant authority which is a community council when you act in your capacity as a member of such an authority.
- (6)You must, when disclosing a personal interest in accordance with paragraph 11 for the first time, register that personal interest in your authority's register of members' interests by providing written notification to your authority's monitoring officer, or in the case of a community council to your authority's proper officer.

#### Sensitive information

- 16 (1) Where you consider that the information relating to any of your personal interests is sensitive information, and your authority's monitoring officer agrees, you need not include that information when registering that interest, or, as the case may be, a change to the interest under paragraph 15.
- (2) You must, within 28 days of becoming aware of any change of circumstances which means that information excluded under sub-paragraph (1) is no longer sensitive information, notify your authority's monitoring officer, or in relation to a community council, your authority's proper officer asking that the information be included in your authority's register of members' interests.
- (3) In this code, "sensitive information" ("gwybodaeth sensitif") means information whose availability for inspection by the public creates, or is likely to create, a serious risk that you or a person who lives with you may be subjected to violence or intimidation.

#### Registration of Gifts and Hospitality

17. You must, within 28 days of receiving any gift, hospitality material benefit or advantage above a value specified in a resolution of your authority, provide written notification to your authority's monitoring officer, or in relation to a community council, to your authority's proper officer of the existence and nature of that gift, hospitality material benefit or advantage.

# APPENDIX 1 TO SECTION ERROR! REFERENCE SOURCE NOT FOUND.

## **Conduct of Members - The Principles**

## 1. SELFLESSNESS

Members must act solely in the public interest. They must never use their position as Members to improperly confer advantage on themselves or to improperly confer advantage or disadvantage on others.

#### 2. HONESTY

Members must declare any private interests relevant to their public duties and take steps to resolve any conflict in a way that protects the public interest.

## 3. INTEGRITY AND PROPRIETY

Members must not put themselves in a position where their integrity is called into question by any financial or other obligation to individuals or organisations that might seek to influence them in the performance of their duties. Members must on all occasions avoid the appearance of such behaviour.

# 4. **DUTY TO UPHOLD THE LAW**

Members must act to uphold the law and act on all occasions in accordance with the trust that the public has placed in them.

#### 5. **STEWARDSHIP**

In discharging their duties and responsibilities Members must ensure that their authority's resources are used both lawfully and prudently.

## 6. **OBJECTIVITY IN DECISION-MAKING**

In carrying out their responsibilities including making appointments, awarding contracts, or recommending individuals for rewards and benefits, Members must make decisions on merit. Whilst Members must have regard to the professional advice of officers and may properly take account of the views of others, including their political groups, it is their responsibility to decide what view to take and, if appropriate, how to vote on any issue.

# 7. EQUALITY AND RESPECT

Members must carry out their duties and responsibilities with due regard to the need to promote equality of opportunity for all people, regardless of their gender, race, disability, sex, sexual orientation, marital status, age or religion, and show respect and consideration for others.

## 8. OPENNESS

Members must be as open as possible about all their actions and those of their authority. They must seek to ensure that disclosure of information is restricted only in accordance with the law.

# 9. **ACCOUNTABILITY**

Members are accountable to the electorate and the public generally for their actions and for the way they carry out their responsibilities as a Member. They must be prepared to submit themselves to such scrutiny as is appropriate to their responsibilities.

# 10. **LEADERSHIP**

Members must promote and support these principles by leadership and example so as to promote public confidence in their role and in the authority. They must respect the impartiality and integrity of the authority's statutory officers and its other employees.

#### APPENDIX 2 TO SECTION ERROR! REFERENCE SOURCE NOT FOUND.

#### **Protocol - Standard of Conduct Expected by Members**

This protocol sets out the standards of conduct expected from Members within Bridgend County Borough Council in dealing with each other. It should be read in conjunction with the Members Code of Conduct and the Member/Officer Protocol. It adds to these documents and does not detract from them.

Members are expected to:

## 1. PUBLIC BEHAVIOUR

- 1.1 show respect to each other;
- 1.2 not to make personal abusive comments about each other;
- 1.3 not to publish anything insulting about each other;
- 1.4 not to make malicious allegations against each other:
- 1.5 not to publish or spread any false information about each other;
- 1.6 show respect to diversity and equality;

## 2. **BEHAVIOUR IN MEETINGS**

- 2.1 behave with dignity;
- 2.2 show respect to the Chair and obey his/her decisions;
- 2.3 not to use indecent language nor make racial remarks or remarks which prejudice any section of society;

# 3. **CONFIDENTIALITY**

- 3.1 keep the confidentiality of exempt papers and any other documents which are not public;
- 3.2 not to release confidential information to the press or the public;
- 3.3 not to use confidential information for purposes other than intended;

## 4. LOCAL MEMBERS

- 4.1 work with Members of adjoining electoral divisions for the benefit of the locality;
- 4.2 if dealing with any matter relating to another electoral division:
  - 4.2.1 explain to anyone seeking assistance that he/she is not the local Member;
  - 4.2.2 inform the local Member, unless it would lead to a breach of confidentiality.

# APPENDIX 3 TO SECTION ERROR! REFERENCE SOURCE NOT FOUND.

# <u>Procedure for Dealing with Allegations made against Councillors and referred to the Standards Committee</u>

## 1. **INTRODUCTION**

This document sets out the procedure that the Council's Standards Committee will follow where it is required to make decisions about the conduct of Councillors following investigations by the Public Services Ombudsman for Wales or the Council's Monitoring Officer under Part III of the Local Government Act 2000 and related regulations. If there is any conflict between this document and any statutory requirements then those statutory requirements will prevail.

#### 2. INTERPRETATION

In this procedure:

- 2.1 the "Act" means the Local Government Act 2000:
- 2.2 the "Council" means Bridgend County Borough Council;
- 2.3 the "Code of Conduct" means the code of conduct for members adopted by the Council or the community councils within the Council's area in 2008 in accordance with section 51 of the Act, including any revisions;
- 2.4 the "Complainant" means any person who made any allegation which gave rise to the investigation;
- 2.5 the "Investigating Officer" means the person who conducted an investigation into any alleged breach of the Code of Conduct and produced the investigation report, being either the Ombudsman (or a person acting on his or her behalf) or the Monitoring Officer;
- an "investigation report" means a report on the outcome of an investigation into any alleged breach of the Code of Conduct produced either by the Ombudsman under s71(2) of the Act or by the Monitoring Officer under the Regulations;
- 2.7 the "Member" means any person who is the subject of an investigation into any alleged breach of the Code of Conduct;
- the "Monitoring Officer" means the officer for the time being appointed by the Council under section 5 of the Local Government and Housing Act 1989;
- 2.9 the "Ombudsman" means the Public Services Ombudsman for Wales;
- 2.10 the "Regulations" means the Local Government Investigations (Functions of Monitoring Officers and Standards Committees) (Wales) Regulations 2001 as amended;
- 2.11 the "Standards Officer" means the officer for the time being appointed by the Council to support the work of the Standards Committee.

# 3. SUMMARY OF THE PROCEDURE

3.1 Under section 69 of the Act, the Ombudsman may investigate any alleged breach of the Code of Conduct by Members or Co-Opted Members (or former Members or co-opted Members) of the Council or a community council in the Council's area.

- 3.2 Under section 70(4) of the Act, where the Ombudsman ceases such an investigation before it is completed, he/she may refer the matters which are the subject of the investigation to the Monitoring Officer. The Monitoring Officer will then investigate matters in accordance with the Regulations before reporting and, if appropriate, making recommendations to the Standards Committee.
- 3.3 Alternatively, under section 71(2) of the Act, where the Ombudsman decides <u>after</u> investigating that it is appropriate, he/she will produce a report on the outcome of the investigation and send it to the Monitoring Officer and the Council's Standards Committee. The Monitoring Officer will then consider the report of the Ombudsman in accordance with the Regulations, before, if appropriate, making recommendations to the Standards Committee.
- 3.4 The Standards Committee will then make an initial determination either:
  - 3.4.1 that there is no evidence of any failure to comply with the Code of Conduct; or
  - 3.4.2 that the Member should be given the opportunity to make representations, either orally or in writing.
- 3.5 Where the Member is given an opportunity to make representations, the Standards Committee will convene a hearing to consider any response made by the Member and it must determine under regulation 9(1) of the Regulations either that:
  - 3.5.1 there is no evidence of any failure to comply with the Code of Conduct and that therefore no action needs to be taken;
  - 3.5.2 the Member has failed to comply with the Code of Conduct but that no action needs to be taken in respect of that failure;
  - 3.5.3 the Member has failed to comply with the Code of Conduct and should be censured; or
  - 3.5.4 the Member has failed to comply with the Code of Conduct and should be suspended or partially suspended from being a member or co-opted Member of his/her authority for a period not exceeding six months

and take any such action accordingly.

# 4. <u>INVESTIGATIONS BY THE MONITORING OFFICER (REFERRALS UNDER SECTION 70(4) OF THE ACT)</u>

- 4.1 Where the Ombudsman ceases his/her investigation before it is completed and refers the matters which are the subject of the investigation to the Monitoring Officer under section 70(4) of the Act, the Monitoring Officer must:
  - 4.1.1 conduct and investigation; and
  - 4.1.2 report, and if appropriate, make recommendations to the Council's Standards Committee.
- 4.2 The Monitoring Officer will investigate in accordance with the Regulations and may follow such procedures as he or she considers appropriate in the circumstances of the case.
- 4.3 After concluding an investigation, the Monitoring Officer must:

- 4.3.1 produce a report on the findings of his or her investigation and, if appropriate, may make recommendations to the Standards Committee;
- 4.3.2 send a copy of the report to the Member; and
- 4.3.3 take reasonable steps to send a copy of the report to the Complainant.
- 4.4 The Standards Committee will consider the Monitoring Officer's report and any recommendations in accordance with the procedure set out below.

# 5. <u>INVESTIGATIONS BY THE OMBUDSMAN (REFERRALS UNDER SECTION 71(2) OF THE ACT)</u>

- 5.1 Where the Ombudsman completes his or her investigation and sends a report to the Monitoring Officer and the Council's Standards Committee under section 71(2) of the Act, the Monitoring Officer must consider the Ombudsman's report and, if appropriate, make recommendations to the Council's Standards Committee.
- 5.2 The Standards Committee will consider the Ombudsman's report together with any recommendations made by the Monitoring Officer in accordance with the procedure set out below.

#### 6. THE FIRST MEETING OF THE STANDARDS COMMITTEE - INITIAL DETERMINATION

- 6.1 After the Monitoring Officer has:
  - 6.1.1 produced an investigation report in accordance with paragraph 4.3; or
  - 6.1.2 considered the Ombudsman's investigation report in accordance with paragraph 5.1

he/she will arrange for a meeting of the Standards Committee to be convened as soon as possible and for a copy of the investigation report, together with the Monitoring Officer's recommendations (if any), to be sent to each of the members of the Standards Committee.

- 6.2 Notice of the time and place of the meeting will be given in accordance with Part VA of the Local Government Act 1972 as amended by the Standards Committees (Wales) Regulations 2001.
- 6.3 If the investigation report is produced by the Ombudsman, the Monitoring Officer will advise the Standards Committee. If the investigation report is produced by the Monitoring Officer, the Standards Officer or some other suitably qualified person will advise the Standards Committee.
- 6.4 The business of the Standards Committee meeting will be limited to considering the investigation report and the Monitoring Officer's recommendations (if any) and to making an initial determination either:
  - 6.4.1 that there is no evidence of any failure to comply with the Code of Conduct; or
  - 6.4.2 that the Member should be given the opportunity to make representations,

either orally or in writing in respect of the findings of the investigation and any allegation that he or she has failed, or may have failed, to comply with the Code of Conduct.

#### 7. AFTER THE FIRST MEETING OF THE STANDARDS COMMITTEE

- 7.1 Where the Standards Committee decides that there is no evidence of any failure to comply with the Code of Conduct, the Standards Officer will accordingly notify the Member, the Complainant and the Ombudsman.
- 7.2 Where the Standards Committee decides that the Member should be given the opportunity to make representations, the Standards Officer will notify the Member of the Committee's decision and the procedure which the Committee proposes to adopt to receive and consider any representations that he or she may wish to make.

#### 8. PREPARING FOR THE HEARING TO CONSIDER THE MEMBER'S REPRESENTATIONS

- 8.1 The Standards Officer, in consultation with the Chair of the Standards Committee, will write to the Member to propose a date for a hearing to consider any representations that the Member may wish to make and to ask the Member to respond in writing within 14 days to confirm whether he/she:
  - 8.1.1 is able to attend the hearing;
  - 8.1.2 wants to make representations, whether orally or in writing and if so, to include any written representations in his or her response;
  - 8.1.3 disagrees with any of the findings of fact in the investigation report, and if so, which matters he or she disagrees with and the reasons for any disagreements;
  - 8.1.4 wants to appear before the Committee in person or be represented at the hearing by a solicitor, barrister or any other person, in accordance with his/her right under the Regulations;
  - 8.1.5 wants to give evidence to the Standards Committee, either orally or in writing;
  - 8.1.6 wants to call relevant witnesses to give evidence to the Standards Committee;
  - 8.1.7 wants any part of the meeting to be held in private;
  - 8.1.8 wants any part of the investigation report or other relevant documents to be withheld from the public
- 8.2 The Standards Officer will notify the Investigating Officer of the proposed hearing date and ask whether he or she will be attending the hearing.
- 8.3 The Standards Officer will send a copy of the Member's response under paragraph 8.1 to the Investigating Officer and will ask him/her to confirm in writing within seven days whether he/she:
  - 8.3.1 has any comments on the Member's response;
  - 8.3.2 wants to be represented at the hearing;
  - 8.3.3 wants to call relevant witnesses to give evidence to the Standards Committee;
  - 8.3.4 wants any part of the meeting to be held in private; and
  - 8.3.5 wants any part of the investigation report or other relevant documents to be withheld from the public.

- 8.4 The Standards Officer will write to the members of the Committee, the Member and the Investigating Officer at least two weeks before the hearing to:
  - 8.4.1 confirm the date, time and place for the hearing;
  - 8.4.2 summarise the allegation;
  - 8.4.3 outline the main facts of the case that are agreed;
  - 8.4.4 outline the main facts which are not agreed;
  - 8.4.5 note whether the Member or the Investigating Officer will attend or be represented at the hearing;
  - 8.4.6 list those witnesses, if any, who will be asked to give evidence;
  - 8.4.7 enclose the investigation report, any relevant documents, the Member's response and any further response from the Investigating Officer; and
  - 8.4.8 outline the proposed procedure for the meeting.

# 9. POWERS OF THE STANDARDS COMMITTEE

- 9.1 The Standards Committee may, in accordance with the requirements of natural justice, conduct the meeting in the manner it considers most suitable to the clarification of the issues before it and generally to the just handling of the proceedings. It must so far as appears to it appropriate seek to avoid formality and inflexibility in its proceedings. The Standards Committee will decide factual evidence on the balance of probabilities.
- 9.2 The Member or the Investigating Officer may be represented or accompanied whether or not legally qualified but if in any particular case the Standards Committee is satisfied that there is a good reason, it may refuse to permit a particular person to assist or represent a party at the hearing.
- 9.3 The Standards Committee may take legal advice from a Council officer appointed for this purpose at any time during the meeting or while they are considering the outcome. The substance of any legal advice given to the Committee will be shared with the Member and the Investigating Officer if they are present.
- 9.4 Where appropriate, and in accordance with the Regulations, the Standards Committee has power to censure the Member, or suspend or partially suspend the Member for a period not exceeding six months.

## 10. PROCEDURE AT THE HEARING

- 10.1 The hearing will be held in public unless the Standards Committee is persuaded that there is a good reason to exclude the public.
- 10.2 The procedure at the meeting shall be as set out below, subject to the Chair making such changes as he or she thinks fit in order to ensure a fair and efficient hearing.

#### 10.3 **Introduction**

The Chair of the Standards Committee will introduce those persons present and will explain the manner and order of proceedings

#### 10.4 First Stage - Preliminary Procedural Issues

The Standards Committee will then resolve any issues or disagreements about how the hearing should continue, which have not been resolved during the prehearing process.

#### 10.5 **Second Stage - Making Findings of Fact**

The Standards Committee will then consider whether or not there are any significant disagreements about the facts contained in the investigation report.

- 10.5.1 If there is a disagreement as to the facts:
  - (a) the Investigating Officer, if present, will be invited to make any necessary representations to support the relevant findings of fact in the investigation report;
  - (b) the Investigating Officer may call any necessary supporting witnesses to give evidence, with the Standards Committee's permission and the Committee shall give the Member an opportunity to challenge any evidence put forward by any witness called by the Investigating Officer;
  - (c) the Member will then be invited to make representations to support his or her version of the facts;
  - (d) the Member may call any necessary witnesses to give evidence, with the Standards Committee's permission and the Committee shall give the Investigating Officer an opportunity to challenge any evidence put forward by any witness called by the Member.
- 10.5.2 At any time, the Standards Committee may question any of the people involved or any of the witnesses.
- 10.5.3 If the Member disagrees with any relevant fact in the investigation report, without having given prior notice of the disagreement, he or she must give good reasons for not mentioning it before the hearing. If the Investigating Officer is not present, the Standards Committee will consider whether or not it would be in the public interest to continue in his or her absence. After considering the Member's explanation for not raising the issue at an earlier stage, the Committee may then:
  - (a) continue with the hearing, relying on the information in the investigation report;
  - (b) allow the Member to make representations about the issue, and invite the Investigating Officer to respond and call any witnesses, as necessary; or
  - (c) postpone the hearing to arrange for appropriate witnesses to be present, or for the Investigating Officer to be present if he or she is not already.
- 10.5.4 At the conclusion of the representations as to matters of fact, the Standards Committee will retire to deliberate in private on the representations, after which the Chair of the Standards Committee will announce their findings of fact.

## 10.6 Third Stage - Deciding whether the Member has failed to comply with the Code

- 10.6.1 The Standards Committee will then consider whether, based on the facts it has found, the Member has failed to comply with the Code.
- 10.6.2 The Standards Committee will invite the Investigating Officer to make representations as to whether or not, based on the facts the Committee has found, the Member has failed to comply with the Code of Conduct.
- 10.6.3 The Standards Committee will invite the Member to respond to the representations of the Investigating Officer and to make representations as to whether or not, based on the facts the Committee has found, he or she has failed to comply with the Code of Conduct.
- 10.6.4 The Standards Committee may, at any time, question anyone involved on any point they raise in their representations.
- 10.6.5 The Member will be invited to make any final relevant points.
- 10.6.6 The Standards Committee will retire to deliberate in private on the representations and decide whether or not the Member has failed to comply with the Code of Conduct, after which the Chair of the Standards Committee will announce their findings.

#### 10.7 Fourth Stage - Action to be Taken

- 10.7.1 If the Standards Committee decides that the Member has not failed to comply with the Code of Conduct, it will formerly record that there is no evidence of any failure by the Member to comply with the Code of Conduct and that therefore no action needs to be taken.
- 10.7.2 If the Standards Committee decides that the Member has failed to comply with the Code of Conduct it will invite the Member and the Investigating Officer to make representations as to:
  - (a) whether or not the Committee should apply a sanction; and
  - (b) what form any sanction should take.
- 10.7.3 The Standards Committee will retire to deliberate in private on the representations and decide either that:
  - (a) no action needs to be taken in respect of the failure to comply with the Code of Conduct;
  - (b) the Member should be censured; or
  - (c) the Member should be suspended or partially suspended from being a member or Co-Opted Member of his or her authority for a period not exceeding six months,

after which the Chair of the Standards Committee will announce their decision.

10.7.4 After making a decision the Standards Committee will instruct the Standards Officer to confirm the decision and the reasons for the decision in writing and to send a copy of the written decision (including details of the Member's right of

appeal) to the Member, the Complainant and the Ombudsman as soon as reasonably practicable.

# 11. FAILURE TO MAKE REPRESENTATIONS / ATTEND THE HEARING

- 11.1 If the Member fails to make representations, the Standards Committee may:
  - 11.1.1 unless it is satisfied that there is sufficient reason for such failure, consider the investigation report and make a determination in the Member's absence; or
  - 11.1.2 give the Member a further opportunity to make representations.
- 11.2 If a party fails to be present or represented at a hearing, the Standards Committee may, if it is satisfied that the party was duly notified of the hearing and that there is no good reason for such absence:
  - 11.2.1 hear and decide the matter in the party's absence; or
  - 11.2.2 adjourn the hearing.

## 12. ILLNESS OR INCAPACITY

If the Standards Committee is satisfied that any party is unable, through physical or mental sickness or impairment, to attend the hearing and that the party's inability is likely to continue for a long time, the Standards Committee may make such arrangements as may appear best suited, in all the circumstances of the case, for disposing fairly of the matter.

## 13. SUSPENSION

A period of suspension or partial suspension will commence on the day after:

- 13.1 the expiry of the time allowed to lodge a notice of appeal to an appeals tribunal under the Regulations (ie within 21 days of receiving notification of the Standards Committee's determination);
- 13.2 receipt of notification of the conclusion of any appeal in accordance with the Regulations;
- 13.3 a further determination by the Standards Committee made after receiving a recommendation from an appeals tribunal under the Regulations,

whichever occurs last.

#### 14. REFERRAL BY AN APPEALS TRIBUNAL

- 14.1 Where the Standards Committee determines that the Member has failed to comply with the Code of Conduct, the Member may appeal against the determination to an appeals tribunal drawn from the Adjudication Panel for Wales.
- 14.2 An appeals tribunal may endorse the decision of the Standards Committee, refer a matter back to it recommending it impose a different penalty, or overturn the decision.
- 14.3 If:
  - 14.3.1 the Standards Committee determines that the Member failed to comply with the Code of Conduct:

- 14.3.2 the Member appeals to an appeals tribunal drawn from the Adjudication Panel for Wales; and
- 14.3.3 the said tribunal refers the matter back to the Standards Committee with a recommendation that a different penalty be imposed,

the Standards Committee shall meet as soon as reasonably practicable to consider the recommendation of the appeals tribunal and will determine whether or not it should uphold its original determination or accept the recommendation.

14.4 After making its determination the Standards Committee will instruct the Standards Officer to confirm the decision and the reasons for the decision in writing and to send a copy of the written decision to the Member, the Complainant, the Ombudsman and the president of the Adjudication Panel for Wales as soon as reasonably practicable.

## 15. PUBLICATION OF THE STANDARDS COMMITTEE'S REPORT

- 15.1 The Standards Committee will cause to be produced within 14 days after:
  - 15.1.1 the expiry of the time allowed to lodge a notice of appeal under the Regulations, or
  - 15.1.2 receipt of notification of the conclusion of any appeal in accordance with the Regulations, or
  - 15.1.3 a further determination by the Standards Committee made after receiving a recommendation from an appeals tribunal under the Regulations,

whichever occurs last, a report on the outcome of the investigation and send a copy to the Ombudsman, the Monitoring Officer, the Member and take reasonable steps to send a copy to the Complainant.

- 15.2 Upon receipt of the report of the Standards Committee, the Monitoring Officer shall:
  - 15.2.1 for a period of 21 days publish the report on the Council's website and make copies available for inspection by the public without charge at all reasonable hours at one or more of the Council's offices, where any person shall be entitled to take copies of, or extracts from, the report when made so available;
  - 15.2.2 supply a copy of the report to any person on request if he or she pays such charge as the Council may reasonably require, and
  - 15.2.3 not later than seven days after the report is received from the Standards Committee, give public notice, by advertisement in newspapers circulating in the area and such other ways as appear to him or her to be appropriate, that copies of the report will be available as provided by paragraphs 15.2.1 and 15.2.2 above, and shall specify the date (being a date not more than seven days after public notice is first given) from which the period of 21 days will begin.

## 16. **COSTS**

The Standards Committee has no power to make an award of any costs or expenses arising from any of its proceedings.

#### **SECTION 20**

## OFFICERS' CODE OF CODE OF CONDUCT

# Code of Conduct for Qualifying Employees of Relevant Authorities in Wales

# **General Principles**

The public is entitled to expect the highest standards of conduct from all qualifying employees of relevant authorities. The role of such employees is to serve their employing authority in providing advice, implementing its policies, and delivering services to the local community. In performing their duties, they must act with integrity, honesty, impartiality and objectivity.

## **Accountability**

Qualifying employees of relevant authorities work for their employing authority and serve the whole of that authority. They are accountable to, and owe a duty to that authority. They must act in accordance with the principles set out in this Code, recognising the duty of all public sector employees to discharge public functions reasonably and according to the law.

## **Political Neutrality**

Qualifying employees of relevant authorities, whether or not politically restricted, must follow every lawfully expressed policy of the authority and must not allow their own personal or political opinions to interfere with their work. Where qualifying employees are politically restricted (by reason of the post they hold, the nature of the work they do, or the salary they are paid), they must comply with any statutory restrictions on their political activities.

## Relations with Members, the Public and Other Employees

Mutual respect between qualifying employees and members is essential to good local government, and working relationships should be kept on a professional basis.

Qualifying employees of relevant authorities should deal with the public, members and other employees sympathetically, efficiently, and without bias.

#### **Equality**

Qualifying employees of relevant authorities must comply with policies relating to equality issues, as agreed by the authority, in addition to the requirements of the law.

## **Stewardship**

Qualifying employees of relevant authorities must ensure that they use public funds entrusted to them in a responsible and lawful manner, and must not utilise property, vehicles or other facilities of the authority for personal use unless authorised to do so.

#### **Personal Interests**

Whilst qualifying employees' private lives are their own concern, they must not allow their private interests to conflict with their public duty. They must not misuse their official position or information acquired in the course of their employment to further their private interests, or the interests of others. In particular, they must comply with:

(1) any rules of their relevant authority on the registration and declaration by employees of financial and non-financial interests;

(2) any rules of their relevant authority on the declaration by employees of hospitality or gifts offered to or received by them, from any person or organisation doing or seeking to do business, or otherwise benefiting or seeking to benefit from a relationship with the authority. Qualifying employees must not accept benefits from a third party unless authorised to do so by their relevant authority.

#### **Whistleblowing**

In the event that a qualifying employee becomes aware of activities which that employee believes to be illegal, improper, unethical or otherwise inconsistent with this Code, the employee should report the matter, acting in accordance with the employee's rights under the Public Interest Disclosure Act 1998, and with the relevant authority's confidential reporting procedure, or any other procedure designed for this purpose.

#### **Treatment of Information**

Openness in the dissemination of information and decision making should be the norm in relevant authorities. However, certain information may be confidential or sensitive and therefore not appropriate for a wide audience. Where confidentiality is necessary to protect the privacy or other rights of individuals or bodies, information should not be released to anyone other than a member, relevant authority employee or other person who is entitled to receive it, or needs to have access to it for the proper discharge of their functions. Nothing in this Code can be taken as overriding existing statutory or common law obligations to keep certain information confidential, or to divulge certain information.

#### **Appointment of Staff**

Qualifying employees of relevant authorities involved in the recruitment and appointment of staff must ensure that appointments are made on the basis of merit. In order to avoid any possible accusation of bias, such employees must not be involved in any appointment, or any other decisions relating to discipline, promotion or pay and conditions for any other employee, or prospective employee, to whom they are related, or with whom they have a close personal relationship outside work.

#### **Investigations by Monitoring Officers**

Where a monitoring officer is undertaking an investigation in accordance with regulations made under section 73(1) of the Local Government Act 2000 a qualifying employee must comply with any requirement made by that monitoring officer in connection with such an investigation.

#### **CODE OF CONDUCT FOR COUNCIL EMPLOYEES**

#### **Standards**

Local government employees are expected to give the highest possible standard of service to the public, and where it is part of their duties, to provide appropriate advice to Councillors and fellow employees with impartiality. Employees will be expected, through the procedures set out in the Council's Whistleblowing Policy, and without fear of recrimination, to bring to the attention of the appropriate level of management any irregularity or deficiency in the provision of service. Employees must report to the appropriate manager any impropriety or breach of procedure.

#### **Disclosure of Information**

Bridgend County Borough Council is committed to open government. The law requires that certain types of information must be available to members, auditors, government departments, service users and the public. Subject to any obligations of confidentiality it is required to observe, the Council itself may decide to be open about other types of information. Employees must be aware of which information the Council is and is not open about, and act accordingly.

Employees should ensure that the individual or personal circumstances of clients and users of Council services are treated confidentially.

Employees should not use any information obtained in the course of their employment for personal gain or benefit, nor should they pass it on to others who might use it in such a way. Any particular information received by an employee from a Councillor which is personal to that Councillor and does not belong to the Council should not be divulged by the employee without the prior approval of that Councillor, except where such disclosure is required or sanctioned by the law.

# **Political Neutrality**

Employees serve the Council as a whole. It follows they must serve all Councillors and not just those of the controlling group, and must ensure that the individual rights of all Councillors are respected.

Subject to the Council's conventions, employees may also be required to advise political groups. They must do so in ways which do not compromise their political neutrality.

Employees, whether or not politically restricted, must follow every lawful expressed policy of the Council and must not allow their own personal or political opinions to interfere with their work.

## Relationships

#### Councillors

Employees are responsible to the Council through its senior managers. For some, their role is to give advice to Councillors and senior managers and all are there to carry out the Council's work. Mutual respect between employees and Councillors is essential to good local government. Close personal familiarity between employees and individual Councillors can damage the relationship and prove embarrassing to other employees and Councillors and should therefore be avoided.

Employees should comply with the protocol on Member/Officer Relations set out in Constitution.

#### The Local Community and Service Users

Employees should always remember their responsibilities to the community they serve and ensure courteous, efficient and impartial service delivery to all groups and individuals within that community as defined by the Council's policies.

#### **Contractors**

All relationships of a business or private nature with external contractors, or potential contractors, should be made known to the appropriate manager. Orders and contracts must be awarded on merit, by fair competition against other tenders, and no special favour should be shown to businesses run by, for example, friends, partners or relatives in the tendering process. No part of the local community should be discriminated against.

Employees who engage or supervise contractors or have any other official relationship with contractors and have previously had or currently have a relationship in a private or domestic capacity with contractors, should declare that relationship to the appropriate manager.

## **Appointments and Other Employment Matters**

Employees involved in appointments should ensure that these are made on the basis of merit. It would be unlawful for an employee to make an appointment which is based on anything other than the ability of the candidate to undertake the duties of the post. In order to avoid any possible accusation of bias, employees should not be involved in an appointment where they are related to an applicant, or have a close personal relationship outside work with him or her.

Similarly, employees should not be involved in decisions relating to discipline, promotion or pay adjustment for any other employee who is a relative, partner or close friend.

#### **Outside Commitments**

Some employees have conditions of service which require them to obtain written consent to take any outside employment. All employees should be clear about their contractual obligations and should not take outside employment or become involved in any other activity, whether paid or not, which conflicts with the Council's interests.

Employees should seek advice on the ownership of intellectual property or copyright created during their employment.

## **Personal Interests**

Employees must declare to their immediate manager or publicly whenever they are present at any meeting of the Authority (whenever an Officer and a Member are present) any financial or non-financial interests which could conflict with the Council's interests.

Examples of non-financial interests include acting as a school governor in schools maintained by the Council, involvement with an organisation receiving grant aid from the Council, membership of an NHS Trust Board or involvement with an organisation or pressure group which may seek to influence the Council's policies. Membership of a political party or trade union need not be declared.

Employees should declare to an appropriate manager membership of any organisation not open to the public without formal membership and commitment of allegiance and which has secrecy about rules or membership or conduct.

# **Equality Issues**

All local government employees should ensure that policies relating to equality issues as agreed by the Council are complied with in addition to the requirements of the law. All members of the local community, customers and other employees have a right to be treated with fairness and equity.

## **Separation of Roles During Tendering**

Employees involved in the tendering process and dealing with contractors should be clear on the separation of client and contractor roles within the Council. Senior employees who have both a client and contractor responsibility must be aware of the need for accountability and openness.

Employees in contractor or client units must exercise fairness and impartiality when dealing with all customers, suppliers, other contractors and sub-contractors.

Employees who are privy to confidential information on tenders or costs for either internal or external contractors should not disclose that information to any unauthorised party or organisation.

Employees contemplating a management buyout should, as soon as they have formed a definite intent, inform the appropriate manager and withdraw from the contract awarding processes.

Employees should be especially careful to ensure that no special favour is shown to current or former employees or their partners, close relatives or associates in awarding contracts to businesses run by them or employing them in a senior or relevant managerial capacity.

## Corruption

Employees must be aware that it is a serious criminal offence for them corruptly to receive or give any gift, loan, fee, reward or advantage for doing, or not doing, anything or showing favour, or disfavour, to any person in their official capacity. If an allegation is made it is for the employee to demonstrate that any such rewards have not been corruptly obtained.

#### **Use of Financial Resources**

Employees must ensure that they use public funds entrusted to them in a responsible and lawful manner. They should strive to ensure value for money to the local community and to avoid legal challenge to the Council.

## **Gifts and Hospitality**

Employees should not accept significant personal gifts from contractors and outside suppliers, although the Council will allow employees to keep insignificant items of modest value such as mugs, calendars, pens, diaries, etc. up to a value of £20.

Apart from participating in concessionary schemes arranged by the Council, trade unions or other groups specifically for the benefit of their own members, or taking advantage of discounts generally available to any member of the public, employees should not acquire materials, labour or plant at cost, trade or discount prices from or through a contractor employed by the Council.

Employees responsible for the purchase of goods and supplies on behalf of the Council should note that any promotional offers given by suppliers are the property of the Council.

Employees should only accept offers of hospitality if there is a genuine need to receive or impart information or represent the local authority in the community. Offers to attend purely social or sporting functions should be accepted only when these are part of the life of the community or where the Council should be seen to be represented. They should be properly authorised and recorded.

When gifts or hospitality have to be declined those making the offer should be courteously but firmly informed of the procedures and standards operating within the Council.

When receiving authorised hospitality employees should be particularly sensitive as to its timing in relation to decisions which the Council may be taking affecting those providing the hospitality.

Acceptance by employees of hospitality through attendance at relevant conferences, seminars and courses is acceptable where it is clear the hospitality is corporate rather than personal, where consent is given by or on behalf of the Council and where purchasing decisions are not compromised. Where visits to inspect equipment, or for the purpose of investigating any matter

relating to the award of any contract, are required, employees should ensure that the Council meets the cost of such visits to avoid jeopardising the integrity of subsequent purchasing decisions.

The Monitoring Officer will maintain a register recording details of gifts and hospitality offered to employees. Employees are to notify him within 28 days of the offer being made using the appropriate form. Authorisation will be considered for the acceptance of the gift or hospitality. All relevant details will be added to the register which shall be available for public inspection.

## **Sponsorship - Giving and Receiving**

Where an outside organisation wishes to sponsor or is seeking to sponsor a local government activity, whether by invitation, tender, negotiation or voluntarily, the basic conventions concerning acceptance of gifts or hospitality apply. Particular care must be taken when dealing with contractors or potential contractors.

Where the Council wishes to sponsor an event or service neither an employee nor any partner, spouse or relative must benefit from such sponsorship in a direct way without there being full disclosure to an appropriate manager of any such interest. Similarly, where the Council through sponsorship, grant aid, financial or other means, gives support in the community, employees should ensure that impartial advice is given and that there is no conflict of interest involved.

#### **Breaches of the Code**

Failure to meet the standards and requirements of this Code may result in disciplinary action against an employee or action for breach of their employment contract with the Council.

#### **Review of the Code**

The operation of the Code will be evaluated from time to time and will be subject to review as necessary.

## **SECTION 21**

#### **PROTOCOL ON MEMBER / OFFICER RELATIONS**

#### Introduction

- 1.1 The purpose of this Protocol is to guide Members and Officers of Bridgend County Borough Council in their relations with one another in such a way as to ensure the smooth running of the Council.
- 1.2 Given the variety and complexity of such relations, this Protocol does not seek to be either prescriptive or comprehensive. It seeks simply to offer guidance on some of the issues which most commonly arise. It is hoped, however, that the approach which it adopts to these issues will serve as a guide to dealing with other circumstances.
- 1.3 This Protocol is to a large extent a written statement of current practice and convention. It seeks to promote greater clarity and certainty. If the Protocol is followed it should ensure that Members receive objective and impartial advice and that Officers are protected from accusations of bias and any undue influence from Members.
- 1.4 It also seeks to reflect the principles underlying the respective Codes of Conduct which apply to Members and Officers. The shared object of these codes is to enhance and maintain the integrity (real and perceived) of local government and the Codes, therefore, demand very high standards of personal conduct.
- 1.5 The National Assembly for Wales issued a Code of Conduct for Qualifying Employees of Relevant Authorities. This covers all the Council's employees. Paragraph 4 of the Code is reproduced below:
  - "Mutual respect between qualifying employees and members is essential to good local government, and working relationships should be kept on a professional basis."
  - Paragraph 4.1.1 of the Council's own Code of Conduct for its Employees contains similar wording.
- 1.6 In line with the reference in the Code to "mutual respect" it is important that any dealings between Members and Officers should observe standards of courtesy and that neither party should seek to take unfair advantage of their position.
- 1.7 A Member should not raise matters relating to the conduct or capability of an Officer at meetings held in public. This is a long-standing tradition in public service. An Officer has no means of responding to such criticisms in public. If a Member feels they have not been treated with proper respect, courtesy or have any concern about the conduct or capability of an Officer, and fails to resolve it through direct discussion with the Officer, they should raise the matter with the relevant Chief Officer. The Chief Officer will then look into the facts and report back to the Member. If the Member continues to feel concern, s/he should then report the facts to the Chief Executive who will look into the matter afresh. Any action taken against an Officer in respect of a complaint, will be in accordance with the provisions of the Council's Disciplinary Policy and Procedure.
- 1.8 Where an Officer feels that s/he has not been properly treated with respect and courtesy they should raise the matter with the Chief Officer or the Chief Executive as appropriate, especially if they do not feel able to discuss it directly with the Member concerned. In these circumstances the Chief Officer or Chief Executive will take appropriate action either by approaching the individual Member and/or the relevant group leader.

# 2. Officer Advice to Members and Party Groups

- 2.1 It must be recognised by all Officers and Members that in discharging their duties and responsibilities, Officers serve the Council as a whole and not exclusively any political group, combination of groups or any individual Member of the Council.
- 2.2 Paragraph 4 (Promotion of Equality and Respect for Others) of the Council's Code of Conduct for Members and Co-opted Members states that:

#### "Members:

- (a) .....
- (b) must not do anything which compromises, or which is likely to compromise, the impartiality of the authority's employees."

Similarly, paragraph 3 (Political Neutrality) of the Council's Code of Conduct for Council Employees states as follows:

- "3.1 Employees serve the Council as a whole. It follows they must serve all councillors and not just those of the controlling group, and must ensure that the individual rights of all councillors are respected.
- 1.1 Subject to the Council's conventions, employees may also be required to advise political groups. They must do so in ways which do not compromise their political neutrality.
- 1.2 Employees, whether or not politically restricted, must follow every lawful expressed policy of the Council and must not allow their own personal or political opinions to interfere with their work."
- 2.3 There is now statutory recognition for party groups and it is common practice for such groups to give preliminary consideration to matters of Council business in advance of such matters being considered by the relevant Council decision making body. Officers may properly be called upon to support and contribute to such deliberations by party groups but must at all times maintain political neutrality. All Officers must, in their dealings with political groups and individual Members, treat them in a fair and even-handed manner.
- 2.4 The support provided by Officers can take many forms ranging from a briefing meeting with Cabinet Members or a Chairperson prior to a meeting of the Cabinet, a Committee or other decision making body to a presentation to a full party group meeting. Whilst in practice such Officer support is likely to be in most demand from whichever party group is for the time being in control of the Council, such support should be available to all party groups.
- 2.5 Certain points must, however, be clearly understood by all those participating in this type of process, Members and Officers alike. In particular:
  - 2.5.1 Officer support must not extend beyond providing information and advice in relation to matters of Council business. Officers must not be involved in advising on matters of party business. The observance of this distinction will be assisted if Officers are not expected to be present at meetings or parts of meetings, when matters of party business are to be discussed;
  - 2.5.2 Party group meetings, whilst they form part of the preliminaries to Council decision making, are not empowered to make decisions on behalf of the Council. Conclusions reached at such meetings do not therefore rank as Council decisions and it is essential that they are not interpreted or acted upon as such; and

- 2.5.3 Similarly, where Officers provide information and advice to a party group meeting in relation to a matter of Council business, this cannot act as a substitute for providing all necessary information and advice to the Council, the Cabinet, the relevant Committee or other decision making body when the matter in question is considered.
- 2.6 Special care needs to be exercised whenever Officers are involved in providing information and advice to a party group meeting which includes persons who are not Members of the Council. Such persons are not bound by the Code of Conduct for Members and Co-opted Members (in particular, the provisions concerning the declaration of interests and confidentiality) and for this and other reasons Officers may not attend and give advice at such a group meeting.
- 2.7 Officers must respect the confidentiality of any party group discussions at which they are present in the sense that they should not relay the content of any such discussion to another party group.
- 2.8 Whilst any Member may ask a relevant Chief Officer or the Chief Executive for written factual information about a Directorate or service, such requests must be reasonable and not seek information relating, for instance, to case work. Requests will be met subject to any overriding legal considerations (which will be determined by the Monitoring Officer, or if the recipient of any request considers the cost of providing the information requested or the nature of the request to be unreasonable). If that is the case, s/he will raise the matter with the Chief Executive who will discuss the issue with the relevant group leader(s).
- 2.9 Where a Member requests factual information (usually written) from an Officer as set out in paragraph 2.8 above, that information will also be supplied to the appropriate Cabinet Member or Chairperson of the appropriate Committee and relevant Officers.
- 2.10 Where a Member requests policy advice (usually oral) that advice will not be supplied to the relevant Cabinet Member or Chairperson of the appropriate Committee without the consent of the requesting Member.
- 2.11 In relation to Budget proposals:
  - (a) the Cabinet shall be entitled to confidential discussions with Officers regarding options and proposals. These will remain confidential until the Cabinet's Budget proposals are published; and
  - (b) the opposition groups shall also be entitled to confidential discussions with Officers to enable them to formulate alternative Budget proposals. These will remain confidential until determined by the respective opposition groups or until published whichever is the earlier.
- 2.12 It must not be assumed by any party group or Member that any Officer is supportive of any plan, policy or strategy developed because of that Officer's assistance in the formulation of that plan, policy or strategy.
- 2.13 Any particular cases of difficulty or uncertainty in this area of Officer advice to party groups should be raised with the Chief Executive who will discuss them with the relevant group leader(s).

# 3. Support Services to Members and Party Groups

3.1 The only basis on which the Council can lawfully provide support services (e.g. stationery, typing, printing, photocopying, transport etc.) to Members is to assist them in discharging their role as Members of the Council. Such support services must therefore only be used on Council business. They should never be used in connection with party political or campaigning activity or for private purposes.

# 4. Members' Access to Information and to Council Documents

- 4.1 Members are free to approach any Directorate of the Council to ask for information in accordance with paragraph 2.8. This right extends to such information, explanation or advice as they may reasonably need in order to assist them in discharging their role as a Member of the Council. This can range from a request for general information about some aspect of the Council's activities to a request for specific information on behalf of a constituent. Such approaches should normally be made to the Director or another senior Officer of the Directorate concerned.
- 4.2 As regards the legal rights of Members to inspect Council documents, these are covered partly by statute and partly by the common law.
- 4.3 Further and more detailed information regarding Members rights to inspect Council documents is contained in the Access to Information Procedure Rules in Section 15 of this Constitution and Members may obtain advice on their rights from the Council's Monitoring Officer.
- 4.4 Any Council information provided to a Member must only be used by the Member for the purpose for which it was provided i.e. in connection with the proper performance of the Member's duties as a Member of the Council.

## 5. Relationships between Officers and Cabinet Members/Chairpersons

- 5.1 It is clearly important that there should be a close working relationship between Cabinet Members, Chairpersons of Committees etc. and the Officers who report to or interact with the Cabinet and Committees etc. However, such relationships should never be allowed to become so close, or appear to be so close, as to bring into question the Officer's ability to deal impartially with other Members and other party groups.
- 5.2 It must be recognised that in some situations an Officer will be under a professional duty to submit a report. Similarly, a Director or other senior Officer will always be fully responsible for the contents of any report submitted in his/her name. This means that any such report may be amended only where the amendment reflects the professional judgement of the author of the report. This is to be distinguished from a situation where there is a value judgement to be made. Any issues arising between a Cabinet Member or a Chairperson and a Director or other Senior Officer in this area should be referred to the Chief Executive for resolution in conjunction with the Monitoring Officer.

#### 6. Correspondence

6.1 Official letters on behalf of the Council should normally be sent in the name of the appropriate Officer, rather than in the name of a Member. It may be appropriate in certain circumstances (e.g. representations to a Government Minister) for a letter to appear in the name of a Member, but this should be the exception rather than the norm. Letters which, for example, create obligations or give instructions on behalf of the Council should never be sent out in the name of a Member.

# 7. Site Visits / Meetings with Members

- 7.1 The ability of Officers to undertake duties free from political pressure prevails over any provisions which enable Member meetings and site visits to take place. As part of the operational role of the Directorates, Officers will arrange and undertake various meetings or site visits that will assist in the development of recommendations for senior officers or elected Members. It is not always appropriate that these meetings be attended by Elected Members and it is at the discretion of the responsible Officer to determine the merits of inviting either the Local Elected Members or an Elected Member with a particular responsibility or interest in the site or issues relating to that site.
- 7.2 It is recognised that there are occasions when Members need to meet with officers and the electorate or other third parties and attend site visits to carry out their legitimate role as Ward Councillors. Officers will attend site visits as are necessary for the performance of their duties and will, often separate to this function, ensure that Member meetings and site visits are available to Members.
- 7.3 The Development Control Committee has determined that for Planning purposes, only local members can request a site visit. This is usually restricted to sites within the Members own Ward and occasionally sites close to boundaries between wards where the impact of any development may affect an adjacent ward
- 7.4 Elected members will be invited to meetings and site visits when these are requested by individuals such as Assembly Members or Members of Parliament. These requests are likely to have an impact on the ward or areas of responsibility of the Elected Member.

## **SECTION 22**

## **MEMBER ROLE DESCRIPTIONS**

## **Elected Member Role Description**

## **Accountabilities**

- (a) To Full Council.
- (b) To the electorate of their ward.

- (c) Representing and Supporting Communities
  - (i) To represent ward interests.
  - (ii) To be an advocate for the Council in the ward and communities they serve.
  - (iii) To be a channel of communication to the community on council strategies, policies, services and procedures.
  - (iv) To represent individual constituents and local organisations, undertaking casework on their behalf and serving all fairly and equally.
  - (v) To liaise with executive members, other council members, council officers and partner organisations to ensure that the needs of the local communities are identified, understood and supported.
  - (vi) To promote tolerance and cohesion in local communities.
- (d) Making Decisions and Overseeing Council Performance
  - (i) To participate in Full Council meetings, reaching and making informed and balanced decisions, and overseeing performance.
  - (ii) To participate in informed and balanced decision making on committees and panels to which they might be appointed.
  - (iii) To adhere to the principles of democracy and collective responsibility in decision making.
  - (iv) To promote and ensure efficiency and effectiveness in the provision of council and other public services.

- (e) Representing the Council (Subject to Appointment)
  - (i) To represent the Council on local outside bodies as an appointee of the Council.
  - (ii) To represent the Council on local partnership bodies, promoting common interest and co-operation for mutual gain.
  - (iii) To represent and be an advocate for the Council on national bodies and at national events.
- (f) Internal Governance, Ethical Standards and Relationships
  - (i) To promote and support good governance of the Council and its affairs To provide community leadership and promote active citizenship.
  - (ii) To promote and support open and transparent government.
  - (iii) To support, and adhere to respectful, appropriate and effective relationships with employees of the Council.
  - (iv) To adhere to the Member's Code of Conduct, the Member/Officer Protocol and the highest standards of behaviour in public office.
- (g) Personal and Role Development

To participate in opportunities for development provided for members by the authority.

#### Values

To be committed to the values of the Council and the following values in public office:

- (h) openness and transparency;
- (i) honesty and integrity;
- (j) tolerance and respect;
- (k) equality and fairness;
- (I) appreciation of cultural difference;
- (m) sustainability.

# **Leader and Deputy Role Description**

## **Accountabilities**

- (a) To Full Council.
- (b) The Public.

## Role Purpose and Activity

- (c) Providing Political Leadership to the Council
  - (i) To be a political figurehead for the Council; to be the principal political spokesperson for the Council.
  - (ii) To provide leadership in building a political consensus around council policies.
  - (iii) To form a vision for the Council and community.
  - (iv) To provide strong, clear leadership in the co-ordination of policies, strategies and service delivery.
- (d) Appointing the Cabinet
  - (i) To designate the appropriate portfolios.
  - (ii) To appoint appropriate elected members to each portfolio.
  - (iii) To allocate cabinet members to roles with regard to their abilities.
  - (iv) To designate the Deputy Leader.
- (e) Representing and Acting as Ambassador for the Authority
  - (i) To represent the Authority to a high standard. Provide a strong, competent and eloquent figure to represent the Authority both within the County and at external bodies.
  - (ii) To represent the Authority on the WLGA coordinating committee and the WLGA regional partnership board.
  - (iii) To represent the Authority on the Public Services Board.
  - (iv) To be the Council member on the Corporate Joint Committee
  - (v) To provide leadership and support local partnerships and organisations.
  - (vi) To represent the Authority in regional and national bodies as appropriate.
- (f) Providing Leadership within the Portfolio

To fulfil the role of a portfolio holder, having regard to the role purpose and activities, and role specification of an executive member.

- (g) Managing and Leading the Work of the Cabinet and Chairing Meetings
  - (i) To ensure the effective running of the Cabinet by managing the Forward Work Programme and ensuring its continuing development.
  - (ii) To ensure the work of the Cabinet meets national policy objectives.
  - (iii) To advise and mentor other cabinet members in their work.
  - (iv) To chair meetings of the Cabinet in line with the Constitution.
  - (v) In the Leader's absence the Deputy Leader should fulfil this role.
- (h) Participating in the Collective Decision Making of the Cabinet
  - (i) To work closely with other cabinet members to ensure the development of effective council policies and the budgetary framework for the Council, and the delivery of high quality services to local people.
  - (ii) To accept collective responsibility and support decisions made by the Cabinet once they have been made.
- (i) Working with Officers to Lead the Organisation
  - (i) To liaise with the Chief Executive, and other appropriate officers, on a regular basis.
  - (ii) To work with employees of the Council in relation to the strategic vision and direction of the Council, the management roles of officers and the development of policy issues.
- (j) Leading Partnerships and Community Leadership
  - (i) To give leadership to local strategic partnerships and local partners in the pursuit of common aims and priorities.
  - (ii) To negotiate and broker in cases of differing priorities and disagreement.
  - (iii) To act as a leader of the local community by showing vision and foresight.
- (k) Internal Governance, Ethical Standards and Relationships
  - (i) To promote and support good governance of the Council and its affairs.
  - (ii) To provide community leadership and promote active citizenship.
  - (iii) To promote and support open and transparent government.
  - (iv) To support, and adhere to respectful, appropriate and effective relationships with employees of the Council.

(v) To adhere to the Member's Code of Conduct, Member/Officer Protocol and the highest standards of behaviour in public office.

## <u>Values</u>

To be committed to and demonstrate the following values in public office:

- (I) openness and transparency;
- (m) honesty and integrity;
- (n) tolerance and respect;
- (o) equality and fairness;
- (p) appreciation of cultural difference;
- (q) sustainability;
- (r) inclusive leadership;
- (s) consensus building.

## **Deputy Leader**

To fulfil the duties of the Leader in his or her absence.

To assist the Leader in specific duties as required.

## **Cabinet Member Role Description**

## **Accountabilities**

- (a) To the Leader.
- (b) To the Cabinet (through collective responsibility).
- (c) To Full Council.

- (d) Providing Portfolio Leadership
  - (i) To give political direction to officers working within the portfolio.
  - (ii) To gain the respect of officers within the portfolio; provide support to officers in the implementation of portfolio programmes.
  - (iii) To provide leadership in the portfolio.
  - (iv) To liaise with the appropriate scrutiny chair and receive scrutiny reports as required.
  - (v) To be accountable for choices and performance in the portfolio.

- (vi) To have an overview of the performance management, efficiency and effectiveness of the portfolio.
- (vii) To make Executive Decisions within the Portfolio.
- (e) Contributing to the Setting of the Strategic Agenda and Work Programme for the Portfolio
  - (i) To work with officers to formulate policy documents both strategic and statutory. Ensure that the political will of the majority is carried to and through the Cabinet.
  - (ii) To provide assistance in working up and carrying through a strategic work programme both political and statutory. Carry out consultations with stakeholders as required. Make sure that the portfolio's Forward Work Programme is kept up to date and accurate.
- (f) Providing Representation for the Portfolio

To provide a strong, competent and persuasive figure to represent the portfolio. Be a figurehead in meetings with stakeholders.

- (g) Reporting and Accounting
  - (i) To report as appropriate to the Leader, Full Council, Cabinet, appropriate chair of scrutiny, regulatory bodies and the media.
  - (ii) To be the principal political spokesperson for the portfolio.
  - (iii) To appear before scrutiny committees in respect of matters within the portfolio.
- (h) Taking an Active Part in Cabinet Meetings and Decision Making
  - (i) To show an interest in and support for the portfolios of others.
  - (ii) To recognise and contribute to issues which cut across portfolios or are issues of collective responsibility.
- (i) Leading Partnerships and Community Leadership
  - (i) To give leadership to local strategic partnerships and local partners in the pursuit of common aims and priorities.
  - (ii) To negotiate and broker in cases of differing priorities and disagreement.
  - (iii) To act as a leader of the local community by showing vision and foresight.
- (i) Internal Governance, Ethical Standards and Relationships
  - (i) To promote and support good governance of the Council and its affairs.
  - (ii) To provide community leadership and promote active citizenship.

- (iii) To promote and support open and transparent government.
- (iv) To support, and adhere to respectful, appropriate and effective relationships with employees of the Council.
- (v) To adhere to the Members' Code of Conduct, Member/Officer Protocol and the highest standards of behaviour in public office.

#### Values

To be committed to the values of the Council and the following values in public office:

- (k) openness and transparency;
- (I) honesty and integrity;
- (m) tolerance and respect;
- (n) equality and fairness;
- (o) appreciation of cultural difference;
- (p) sustainability;
- (q) inclusive leadership.

## **Mayor Role Description**

## **Accountabilities**

Full Council.

## Role Purpose and Activity

- (n) Acting as a symbol of the Council's democratic authority
  - (i) As the ceremonial head of the Council, to be non-political and uphold the democratic values of the Council.
  - (ii) To represent the Council at civic and ceremonial functions.
- (o) Chairing Council Meetings
  - To preside over meetings of the Council, so that its business can be carried out efficiently.
  - (ii) To ensure the Council conducts its meetings in line with the Council's Standing Orders.
- (p) Upholding and Promoting the Council's Constitution

To ensure the Constitution is adhered to and, if necessary, to rule on the interpretation of the Constitution.

- (q) Internal Governance, Ethical Standards and Relationships
  - (i) To promote and support good governance of the Council and its affairs.
  - (ii) To provide community leadership and promote active citizenship.
  - (iii) To promote and support open and transparent government.
  - (iv) To support, and adhere to respectful, appropriate and effective relationships with employees of the Council.
  - (v) To adhere to the Members' Code of Conduct, Member/Officer Protocol and the highest standards of behaviour in public office.

## (r) Work Programming

To prepare and manage an annual work programme for the Council to meet its legal obligations (eg setting the budget and the council tax and making appointments).

#### Values

To be committed to the values of the Council and the following values in public office:

- (s) openness and transparency;
- (t) honesty and integrity;
- (u) tolerance and respect;
- (v) equality and fairness;
- (w) appreciation of cultural difference;
- (x) sustainability.

## Vice Chair

To fulfil the duties of the Chair in his or her absence.

To assist the Chair in specific duties as required.

#### **Chair of Democratic Services Committee Role Description**

#### <u>Accountabilities</u>

To Full Council.

- (a) Providing Leadership and Direction
  - (i) To provide confident and effective management of meetings to facilitate inclusivity, participation and clear decision making.

- (ii) To lead the committee in its role in:
  - (A) designating the Head of Democratic Services;
  - (B) keeping under review the provision of staff, accommodation and other resources made available to the Head of Democratic Services, ensuring that these are adequate;
  - (C) make annual reports to the full council in relation to the above;
  - (D) appointing sub committees and chairs of sub-committees to undertake functions delegated by the committee;
  - (E) considering reports prepared by the Head of Democratic Services;
  - (F) developing the Authority's member support and development strategy;
  - (G) ensuring that members have access to a reasonable level of training and development as described in the Authority's member development strategy and the Wales Charter for Member Support and Development;
  - (H) ensuring that the budget for member development is sufficient;
  - (I) ensuring that members have access to personal development planning and annual personal development reviews;
  - (J) work with the member support and development champion where relevant to promote the role of members and necessary support and development.
- (iii) To demonstrate integrity and impartiality in decision making which accord with legal, constitutional and policy requirements.
- (b) Promoting the Role of the Democratic Services Committee
  - (i) To act as an ambassador for the DS committee, facilitating understanding of the role.
  - (ii) To act within technical, legal and procedural requirements to oversee the functions of the committee fairly and correctly.
  - (iii) To ensure thoroughness and objectivity in the committee, receiving and responding to professional advice in the conduct of meetings.
- (c) Internal Governance, Ethical Standards and Relationships
  - To develop the standing and integrity of the committee and its decision making.
  - (ii) To understand the respective roles of members, officers and external parties operating within the Democratic Services Committee's area of responsibility.

(iii) To promote and support good governance by the Council.

## <u>Values</u>

To be committed to the values of the council and the following values in public office:

- (d) openness and transparency;
- (e) honesty and integrity;
- (f) tolerance and respect;
- (g) equality and fairness;
- (h) appreciation of cultural differences;
- (i) sustainability.

## **Member of a Democratic Services Committee Role Description**

#### Accountabilities

- (a) To Full Council.
- (b) To the Chair of the Democratic Services Committee.

- (c) Understanding the Nature of the Governance and Audit Committee:
  - (i) To be aware of and effectively undertake the role of the committee in:
    - (A) designating the Head of Democratic Services;
    - (B) keeping under review the provision of staff, accommodation and other resources made available to the Head of Democratic Services, ensuring that these are adequate;
    - (C) make annual reports to the full council in relation to the above;
    - (D) appointing sub committees and chairs of sub-committees to undertake functions delegated by the committee;
    - (E) considering reports prepared by the Head of Democratic Services;
    - (F) developing the Authority's member support and development strategy;
    - (G) ensuring that members have access to a reasonable level of training and development as described in the member development strategy and the Wales Charter for Member Support and Development;
    - (H) ensuring that the budget for member development is sufficient;

- (I) ensuring that members have access to personal development planning and annual personal development reviews.
- (d) To have sufficient technical, legal and procedural knowledge to contribute fairly and correctly to the function of the committee.
- (e) To be thorough and objective in receiving and responding to professional advice in the conduct of meetings and issues before the committee.
- (f) Participating in Meetings and Making Decisions
  - (i) To participate effectively in meetings of the Democratic Services Committee.
  - (ii) To make informed and balanced decisions, within the terms of reference of the committee, which accord with legal, constitutional and policy requirements.
- (g) Internal Governance, Ethical Standards and Relationships
  - (i) To ensure the integrity of the committee's decision making and of his/ her own role by adhering to the Code of Conduct(s) and other constitutional and legal requirements.
  - (ii) To promote and support good governance by the Council.
  - (iii) To understand the respective roles of members, officers and external parties operating within the Democratic Services Committee's area of responsibility.

#### Values

To be committed to the values of the Council and the following values in public office:

- (h) openness and transparency;
- (i) honesty and integrity;
- (j) tolerance and respect;
- (k) equality and fairness;
- (I) appreciation of cultural difference;
- (m) sustainability.

## **Chair of a Regulatory Committee Role Description**

#### Accountabilities

- (a) To Full Council.
- (b) To the members of the regulatory committee.

## Role Purpose and Activity

- (c) Providing Leadership and Direction
  - (i) To provide confident and effective management of meetings to facilitate inclusivity, participation and clear decision making.
  - (ii) To ensure that applicants and other interested parties are satisfied as to the transparency of the regulatory process.
  - (iii) To demonstrate integrity and impartiality in decision making which accord with legal, constitutional and policy requirements.
  - (iv) To delegate actions to sub committees as appropriate
- (d) Promoting the Role of the Regulatory Committee and Quasi-Judicial Decision Making
  - (i) To act as an ambassador for the regulatory committee, facilitating understanding of the role.
  - (ii) To act within technical, legal and procedural requirements to oversee the functions of the committee fairly and correctly.
  - (iii) To ensure thoroughness and objectivity in the committee, receiving and responding to professional advice in the conduct of meetings and in individual cases/applications before formal committee meetings.
- (e) Internal Governance, Ethical Standards and Relationships
  - (i) To develop the standing and integrity of the committee and its decision making.
  - (ii) To understand the respective roles of members, officers and external parties operating within the regulatory committee's area of responsibility.
  - (iii) To promote and support good governance by the Council.

#### Values

To be committed to the values of the council and the following values in public office:

- (f) openness and transparency;
- (g) honesty and integrity;

- (h) tolerance and respect;
- (i) equality and fairness;
- (j) appreciation of cultural differences;
- (k) sustainability.

## Member of a Regulatory Committee Role Description

#### Accountabilities

- (a) To Full Council.
- (b) To the Chair of the regulatory committee.

- (c) Understanding the Nature of the Regulatory Committee and Quasi-Judicial Decision Making
  - (i) To be aware of the quasi-judicial nature of regulatory committee decision making.
  - (ii) To have sufficient technical, legal and procedural knowledge to contribute fairly and correctly to the function of the committee.
  - (iii) To be thorough and objective in receiving and responding to professional advice in the conduct of meetings and individual cases/applications before the committee.
- (d) Participating in Meetings and Making Decisions
  - (i) To participate effectively in meetings of the regulatory committee, ensuring that both local considerations and policy recommendations are balanced to contribute to effective decision making.
  - (ii) To make informed and balanced decisions, within the terms of reference of the committee, which accord with legal, constitutional and policy requirements.
- (e) Internal Governance, Ethical Standards and Relationships
  - (i) To ensure the integrity of the committee's decision making and of his/her own role by adhering to the Code of Conduct(s) and other constitutional and legal requirements.
  - (ii) To promote and support good governance by the Council.
  - (iii) To understand the respective roles of members, officers and external parties operating within the regulatory committee's area of responsibility.

## Values

To be committed to the values of the Council and the following values in public office:

- (f) openness and transparency;
- (g) honesty and integrity;
- (h) tolerance and respect;
- (i) equality and fairness;
- (j) appreciation of cultural difference;
- (k) sustainability.

## **Chair of Standards Committee Role Description**

#### Accountabilities

To Full Council.

- (a) Providing Leadership and Direction
  - (i) To act within technical, legal and procedural requirements to oversee the functions of the committee fairly and correctly.
  - (ii) To ensure thoroughness and objectivity in the committee, receiving and responding to professional advice on the Code of Conduct.
  - (iii) To demonstrate independence, integrity and impartiality in decision making which accord with legal, constitutional and policy requirements.
  - (iv) To provide confident and effective management of meetings to facilitate inclusivity, participation and clear decision making.
  - (v) To lead the committee in its role in:
    - (A) promoting and maintaining high standards of conduct by Councillors and co-opted members;
    - (B) assisting the Councillors and co-opted members to observe the Members' Code of Conduct:
    - advising the Council on the adoption or revision of the Members' Code of Conduct;
    - (D) monitoring the operation of the Members' Code of Conduct;
    - (E) advising, training or arranging to train Councillors, co-opted members on matters relating to the Members' Code of Conduct;

- (F) granting dispensations to Councillors and co-opted members on dealing with any reports from a case tribunal or interim case tribunal, and any report from the Monitoring Officer on any matter referred to that officer by the Public Services Ombudsman for Wales;
- (G) the exercise of these functions in relation to community councils and the members of those community councils.

## <u>Values</u>

To be committed to the values of the council and the following values in public office:

- (b) openness and transparency;
- (c) honesty and integrity;
- (d) tolerance and respect;
- (e) equality and fairness;
- (f) appreciation of cultural differences;
- (g) sustainability.

## Member of a Standards Committee Role Description

## <u>Accountabilities</u>

- (a) To Full Council.
- (b) To the Chair of the Standards Committee.

- (c) Understanding the nature of the Standards committee and effectively fulfilling its functions by:
  - promoting and maintaining high standards of conduct by Councillors and co-opted members;
  - (ii) assisting the Councillors and co-opted members to observe the Members' Code of Conduct;
  - (iii) advising the Council on the adoption or revision of the Members' Code of Conduct:
  - (iv) monitoring the operation of the Members' Code of Conduct;
  - advising, training or arranging to train Councillors, co-opted members on matters relating to the Members' Code of Conduct;
  - (vi) granting dispensations to Councillors and co-opted members on dealing with any reports from a case tribunal or interim case tribunal,

and any report from the Monitoring Officer on any matter referred to that officer by the Public Services Ombudsman for Wales.

- (d) To have sufficient technical, legal and procedural knowledge to contribute fairly and correctly to the function of the committee.
- (e) To be thorough and objective in receiving and responding to professional advice in the conduct of meetings and issues before the committee.
- (f) Participating in Meetings and Making Decisions
  - (i) To participate effectively in meetings of the Standards committee.
  - (ii) To make informed and balanced decisions, within the terms of reference of the committee, which accord with legal, constitutional and policy requirements.
- (g) Internal Governance, Ethical Standards and Relationships
  - (i) To ensure the integrity of the committee's decision making and of his/her own role by adhering to the Code of Conduct(s) and other constitutional and legal requirements.
  - (ii) To promote and support good governance by the Council.
  - (iii) To understand the respective roles of members, officers and external parties operating within the Standards committee's area of responsibility.

## Values

To be committed to the values of the Council and the following values in public office:

- (h) openness and transparency;
- (i) honesty and integrity;
- (j) tolerance and respect;
- (k) equality and fairness;
- (I) appreciation of cultural difference;
- (m) sustainability.

## **Chair of Governance and Audit Committee Role Description**

## Accountabilities

(a) To Full Council.

#### Role Purpose and Activity

(b) Providing Leadership and Direction

- (i) To demonstrate independence, integrity and impartiality in decision making which accord with legal, constitutional and policy requirements.
- (ii) To provide confident and effective management of meetings to facilitate inclusivity, participation and clear decision making.
- (iii) To agree the agendas for Governance and Audit Committee meetings.
- (iv) To lead the committee in its role in:
  - (A) reviewing and scrutinising the authority's financial affairs;
  - (B) making reports and recommendations in relation to the authority's financial affairs;
  - (C) reviewing and assessing the risk management, internal control and corporate governance arrangements of the authority;
  - (D) making reports and recommendations to the authority on the adequacy and effectiveness of those arrangements;
  - (E) overseeing the authority's internal and external audit arrangements;
  - (F) reviewing the financial statements prepared by the authority and approving them when powers are delegated;
  - (G) developing relationships with internal and external auditors and the Authority's Monitoring Officer;
  - (H) developing a Forward Work Programme designed to deliver the governance and audit committees functions;
  - (I) reviewing and self-assessing the performance of the Committee and its members; and
  - (J) reviewing and making recommendations in relation to the self-assessment report pursuant to section 91 of the Local Government and Elections (Wales) Act 2021.
- (c) Promoting the Role of the Governance and Audit Committee
  - (i) To act as an ambassador for the governance and audit committee, facilitating understanding of the role.
  - (ii) To act within technical, legal and procedural requirements to oversee the functions of the committee fairly and correctly.
  - (iii) To ensure thoroughness and objectivity in the committee, receiving and responding to professional advice in the conduct of meetings and in the audit process.
- (d) Internal Governance, Ethical Standards and Relationships

- (i) Understanding the financial risks associated with corporate governance; being satisfied that the authorities assurance statements including the annual governance statement reflects the risk environment and any activities required to improve it.
- ii) To develop the standing and integrity of the committee and its decision making.
- (iii)To understand the respective roles of members, officers and external parties operating within the governance and audit committee's area of responsibility.
- (iv) To promote and support good governance by the Council.

## <u>Values</u>

To be committed to the values of the council and the following values in public office:

- (e) openness and transparency:
- (f) honesty and integrity;
- (g) tolerance and respect;
- (h) equality and fairness;
  - (i) appreciation of cultural differences;
- (i) sustainability.

## Member of a Governance and Audit Committee Role Description

#### Accountabilities

- (a) To Full Council.
- (b) To the Chair of the Governance and Audit Committee.

- (c) Understanding the role of the Governance and Audit Committee and undertaking its functions:
  - (i) reviewing and scrutinising the authority's financial affairs;
  - (ii) making reports and recommendations in relation to the authority's financial affairs;
  - (iii) reviewing and assessing the risk management, internal control and corporate governance arrangements of the authority;
  - (iv) making reports and recommendations to the authority on the adequacy and effectiveness of those arrangements;
  - (v) overseeing the authority's internal and external audit arrangements;

- (vi) reviewing the financial statements prepared by the authority.
- (d) To have sufficient technical, legal and procedural knowledge to contribute fairly and correctly to the function of the committee.
- (e) To be thorough and objective in receiving and responding to professional advice in the conduct of meetings and issues before the committee.
- (f) Participating in Meetings and Making Decisions
  - (i) To participate effectively in meetings of the governance and audit committee; questioning and seeking clarification on matters falling within the committee's remit.
  - (ii) To make informed and balanced decisions, within the terms of reference of the committee, which accord with legal, constitutional and policy requirements.
- (g) Internal Governance, Ethical Standards and Relationships
  - (i) Understanding the financial risks associated with corporate governance; being satisfied that the authorities assurance statements including the annual governance statement reflects the risk environment and any activities required to improve it.
  - (ii) To ensure the integrity of the committee's decision making and of his/her own role by adhering to the Code of Conduct(s) and other constitutional and legal requirements.
  - (iii) To promote and support good governance by the Council.
  - (iv) To understand the respective roles of members, officers and external parties operating within the governance and audit committee's area of responsibility.

#### Values

To be committed to the values of the Council and the following values in public office:

- (h) openness and transparency;
- (i) honesty and integrity;
- (j) tolerance and respect;
- (k) equality and fairness;
- (I) appreciation of cultural difference;
- (m) sustainability.

## Overview and Scrutiny Chair Role Description

#### Accountabilities

- (a) Full Council.
- (b) The Public.

## Role Purpose and Activity

- (c) Providing Leadership and Direction
  - (i) To provide confident and effective management of the member team.
  - (ii) To promote the role of overview and scrutiny within and outside the council, liaising effectively both internally within the council and externally with the Council's partners.
  - (iii) To demonstrate an objective and evidence based approach to overview and scrutiny.
  - (iv) To evaluate the impact and added value of overview and scrutiny activity and identify areas for improvement.

## (d) Managing the Work Programme

- (i) To develop a balanced work programme of the committee which includes pre decision scrutiny, policy development and review, investigative scrutiny, and holding the executive to account including performance monitoring.
- (ii) To ensure the programme takes account of relevant factors such as: the work programmes of the executive and other committees, strategic priorities and risks, and relevant community issues.
- (iii) To ensure that the work programme is delivered.
- (iv) To report on progress against the work programme to Council, and others as appropriate.
- (v) To liaise with officers, other members and community representatives to resource and deliver the work programme.

# (e) Effective Meeting Management

- To set agendas containing clear objectives and outcomes for the meeting.
- (ii) To manage the progress of business at meetings, ensuring that meeting objectives are met, and the code of conduct, standing orders and other constitutional requirements are adhered to.
- (iii) To ensure that the necessary preparation is done beforehand. To ensure that all participants have an opportunity to make an appropriate contribution.

- (f) Community Leadership
  - (i) To act as a focus for liaison between the council, community and external bodies in relation to the scrutiny function.
  - (ii) To build understanding and ownership of the overview and scrutiny function within the community.
  - (iii) To identify relevant community based issues for scrutiny.
  - (iv) To promote the full involvement of external stakeholders for example, service users expert witnesses and partners in scrutiny activity.
- (g) Involvement and Development of Committee Members
  - (i) To encourage effective contributions from all committee members in both committee and task and finish groups.
  - (ii) To assess individual and collective performance within the committee and facilitate appropriate development.
  - (iii) To champion the importance of learning and development.

#### Values

To be committed to the values of the Council and the following values in public office:

- (h) openness and transparency;
- (i) honesty and integrity;
- (j) tolerance and respect;
- (k) equality and fairness;
- (I) appreciation of cultural difference;
- (m) sustainability.

## **Overview and Scrutiny Member Role Description**

## Accountabilities

- (a) Chair of the appropriate Overview and Scrutiny Committee.
- (b) Full Council.
- (c) The public.

#### Role Purpose and Activity

To participate fully in the activities of the Overview and Scrutiny Committee, the development and delivery of its work programme and any associated task and finish groups.

- (d) Reviewing and Developing Policy
  - (i) To assist in the creation, development, improvement and refinement of council policy.
  - (ii) To challenge policies on a sound basis of evidence for example against legislation or local political priority.
  - (iii) To assess impact of existing policy.
- (e) Holding the Executive to Account, Monitoring Performance and Service Delivery
  - (i) To monitor the performance of internal and external providers against standards and targets including questioning of executive and senior officers over time.
  - (ii) To contribute to the identification and mitigation of risk.
  - (iii) To investigate and address the causes of poor performance.
  - (iv) To evaluate the validity of Executive Decisions and challenging decisions through call in where appropriate.
- (f) Promoting the Work of Overview and Scrutiny
  - (i) To promote the role of overview and scrutiny within and outside the council, developing effective internal and external relationships.
  - (ii) To demonstrate an objective and evidence based approach to overview and scrutiny.
  - (iii) To add value to the decision making and service provision of the authority through effective scrutiny.

#### (g) Community Leadership

- (i) To use scrutiny as a means to address community issues and engage the public.
- (ii) To encourage stakeholders to participate in the work of the authority To develop locally viable and acceptable policy solutions.
- (iii) To build a dialogue around priorities, objectives and performance, among communities and stakeholders.

# (h) Meeting Participation

- (i) To make adequate and appropriate preparation for meetings through research and briefings.
- (ii) To participate in a proactive, informed and effective manner taking account of the Code of Conduct, Standing Orders and other constitutional requirements.

## Values

To be committed to the values of the Council and the following values in public office:

- (i) openness and transparency;
- (j) honesty and integrity;
- (k) tolerance and respect;
- (I) equality and fairness;
- (m) appreciation of cultural difference;
- (n) sustainability.

## **Leader of the Opposition Role Description**

## Accountabilities

To the nominating group within the constitution.

- (a) Providing Political Leadership for an Opposition Group
  - (i) To be a political figurehead for the opposition group; to be the principal political spokesperson for the Council's opposition.
  - (ii) To provide leadership in the constructive challenge of the Council's policies.
  - (iii) To constructively challenge the vision for the Council and community where appropriate.
  - (iv) To provide strong, clear leadership in the co-ordination of alternative policies, strategies and service delivery.
- (b) Representing the Authority's Opposition
  - (i) To represent the Opposition Group to a high standard; providing a strong, competent and eloquent figure to represent the opposition within the Authority.
  - (ii) To represent the Council on external bodies.
- (c) Internal Governance, Ethical Standards and Relationships
  - (i) To promote and support good governance of the Council and its affairs.
  - (ii) To provide community leadership and promote active citizenship.
  - (iii) To promote and support open and transparent government.

- (iv) To promote, support, and adhere to respectful, appropriate and effective relationships with employees of the Council.
- (v) To promote, support and adhere to the Member's Code of Conduct, Member/Officer Protocol and the highest standards of behaviour in public office.

#### <u>Values</u>

To be committed to and demonstrate the following values in public office:

- (d) openness and transparency;
- (e) honesty and integrity;
- (f) tolerance and respect;
- (g) equality and fairness;
- (h) appreciation of cultural difference:
- (i) sustainability;
- (j) inclusive leadership;
- (k) consensus building.

## Role of the Deputy Leader of the Opposition

To fulfil the duties of the Leader in his or her absence.

To assist the Leader in specific duties as required.

# **Member Champion Purpose and Role**

#### What are Member Champions?

- (a) Member Champions exist to provide a voice for traditionally underrepresented groups, or issues which need to be kept at the forefront of council business although they may not be the responsibility of any individual or committee.
- (b) Member Champions, (sometimes called lead members) are elected members who in addition to their other council responsibilities make sure that the issue or group that they are championing are taken into account when council policy is being developed and decisions are made. Members act as champions in areas such as children, homelessness, equalities, older people, young people, scrutiny, member support and development, health improvement and anti-poverty. There is a statutory role for a lead member of children's and young people's services with a responsibility for over-seeing the arrangements made under sections 25 and 26 of the 2004 Children Act.
- (c) Guidance on undertaking the role with regard to the subject knowledge that members need is sometimes available from the outside bodies associated with the issue being championed for example the toolkit for older peoples' champions from the LGA. Otherwise they will be reliant on their authority for

guidance in the subject they lead on and also their role as lead member in this area.

## What do they do?

Typically, the lead member will:

- (d) make sure that their area of interest is taken into account when developing policy or making decisions;
- (e) ask questions about performance and resourcing for the area;
- (f) raise the profile of the area and make the authority aware of good practice;
- (g) engage with external bodies who work in the area;
- (h) engage with other officers and members in relation to the role;
- (i) engage with community groups with an interest/stake in the area;
- (j) report action to the council.

## How does their role fit within the Corporate Structure?

- (k) This will vary according to the area/issue that is being championed and how the authority functions. There is potential for confusion and overlap between the role of the member champion and those of the relevant executive member or overview and scrutiny members. The champion role itself could be undertaken by either the relevant executive member or a non-executive member.
- (I) It is therefore important that members and officers work together to agree roles and action for the area being championed and that there are mechanisms for lead members to report on their activities. It is helpful for the authority to draft a protocol which sets out what powers champions have and do not have, such as whether or not they are able to make decisions on behalf of the authority. Similarly, the appointment of champions varies between authorities, and includes appointments being made by full council meetings or by the leader.

#### The Role Description

It is difficult to create a role description that fits with the different roles expected of champions and how they operate in the different authorities' structures. The following is a generic model which will require local adaptation, particularly to reflect the difference that may exist between a statutory role, one undertaken by an executive member and a non-executive lead.

## Member Champion Role Description

#### Accountabilities

To Full Council.

## Role Purpose and Activities

- (a) Within the Council
  - (i) To promote the interest being championed within the Council's corporate and service priorities.
  - (ii) To promote the needs of the client group represented in the interest to the decision makers within the council.
  - (iii) To work with the decision makers in the Council to establish strategies/ policies/work plans connected with the interest.
  - (iv) To maintain an awareness of all matters connected with the interest.
  - (v) To contribute to good practice and the continuous improvement of services and functions related to the interest.
  - (vi) To engage with members in matters related to the interest such as attending Overview and Scrutiny/Cabinet /Full Council meetings etc.
  - (vii) Raising awareness of and taking a lead role in the development of all members and officers in relation to the interest.
- (b) In the Community
  - (i) To raise the profile of the interest in the community.
  - (ii) To engage with citizens and community groups in matters related to the interest.
  - (iii) To lead and support local initiatives related to the interest.

#### Values

To be committed to the values of the Council and the following values in public office:

- (c) openness and transparency;
- (d) honesty and integrity;
- (e) tolerance and respect;
- (f) equality and fairness;
- (g) appreciation of cultural difference;
- (h) sustainability;
- (i) inclusive leadership.

## **Elected Member Person Specification**

To fulfil his or her role as laid out in the role description, an effective member requires the following:

## Representing and Supporting Communities

- (a) Good advocacy skills.
- (b) Interpersonal skills.
- (c) Integrity and the ability to set aside own views and act impartially.
- (d) The ability to present relevant and well-reasoned arguments.
- (e) Good communication skills.

## Making Decisions and Overseeing Council Performance

- (f) Knowledge and understanding of meetings law, rules and conventions.
- (g) An understanding of strategic, policy and service contexts for decisions.
- (h) The ability to challenge ideas and contribute positively to policy development.

## Representing the Council (Subject to Appointment)

- (i) Good public speaking skills.
- (j) Good presentation skills.
- (k) The ability to persuade others and act with integrity.

## Internal Governance, Ethical Standards and Relationships

- (I) An understanding of the roles of officers, members and different agencies.
- (m) Respect for, and desire to work with, different groups and individuals.
- (n) Have knowledge and understanding of the Code of Conduct and Member/Officer Protocol.
- (o) A knowledge and commitment to the values of the Council.

#### Personal and Role Development

- (p) An ability to asses personal and role development needs.
- (q) Desire and skills to participate in development.

#### **Leader Person Specification**

To fulfil his or her role as laid out in the role description, an effective leader requires:

## Providing Political Leadership to the Council

- (a) Knowledge of community strengths, areas of improvement and key issues.
- (b) An understanding of the relationship between national and local politics.
- (c) Have a good strategic awareness of issues facing the Council.
- (d) An understanding of the Council's strategy, policies and operations.

## Appointing the Cabinet\*

- (e) An understanding of the rules for the appointment of Cabinet Members.
- (f) An ability to recognise talent amongst members.
- (g) An ability to negotiate the most advantageous appointments within and across political groups.
- (h) To appraise, guide and mentor senior members.

# Representing and Acting as Ambassador for the Authority

- (i) High level communication skills to communicate to the media, local community and wider audience.
- (j) Good public speaking skills.

## Providing Leadership within the Portfolio

The skills necessary for a cabinet member to fulfil their role.

# Managing and Leading the work of the Cabinet and Chairing Meetings of the Cabinet

- (k) An understanding of the Cabinet procedure rules.
- (I) Skills to Chair meetings, including encouraging participation from all members.
- (m) A knowledge and understanding of national policy objectives.
- (n) An overview of the work being carried out by cabinet members.

#### Participating in the Collective Decision Making of the Cabinet

The ability to constructively challenge decisions and suggest alternatives.

## Working with Officers to Lead the Organisation

An understanding of the roles and responsibilities of the Chief Executive and other officers.

## Leading Partnerships and Community Leadership

(o) Adaptive leadership skills.

- (p) Negotiation and brokerage skills.
- (q) Creative and lateral thinking skills; the ability to see ahead and be predictive.

## Internal Governance, Ethical Standards and Relationships

- (r) An understanding of the roles of officers, members and different agencies.
- (s) Respect for, and desire to work with, different groups and individuals.
- (t) Have knowledge and understanding of the Code of Conduct and Member/Officer Protocol.
- (u) A knowledge and commitment to the values of the Council.

## **Cabinet Member Person Specification**

To fulfil his or her role as laid out in the role description, an effective Cabinet Member requires:

## Portfolio Leadership

- (a) An understanding of the Council's strategy, policies and operations.
- (b) Leadership skills.

# Contributing to the setting of the Strategic Agenda and Work Programme for the Portfolio

- (c) The ability to present to others.
- (d) The ability to exercise strategic awareness and judgement.
- (e) Knowledge of relevant issues and who to involve in decision making.
- (f) The ability to persuade others.
- (g) Knowledge of Council and national objectives.

## Providing Representation for the Portfolio

- (h) Public speaking skills.
- (i) Good presentation skills.

#### Reporting as Appropriate

High level communication skills.

## Taking an Active Part in Cabinet Meetings and Decisions

- (j) The ability to constructively challenge decisions and suggest alternatives.
- (k) The knowledge, confidence and ability to contribute to discussion and resolution of cross cutting and collective issues.

## Leading Partnerships and Community Leadership

- (I) Adaptive leadership skills.
- (m) Negotiation and brokerage skills.
- (n) Creative and lateral thinking skills; the ability to see ahead and be predictive.

## Internal Governance, Ethical Standards and Relationships

- (o) An understanding of the roles of officers, members and different agencies.
- (p) Respect for, and desire to work with, different groups and individuals.
- (q) Have knowledge and understanding of the Code of Conduct and Member/Officer Protocol.
- (r) A knowledge and commitment to the values of the Council.

# **Mayor Person Specification**

To fulfil his or her role as laid out in the role description, an effective member requires the following:

## Acting as a Symbol of the Council's Democratic Authority

- (a) Good public speaking skills.
- (b) An in-depth understanding of role of Chair.

## **Chairing Council Meetings**

- (c) Skills to chair meetings, to ensure business is carried out effectively and all those attending participate.
- (d) An understanding of the Council's Standing Orders.

#### Upholding and Promoting the Council's Constitution

- (e) An understanding of the Council's Constitution.
- (f) An understanding of when to seek the advice of the Monitoring Officer on issues relating to the Constitution.

#### Internal Governance, Ethical Standards and Relationships

- (g) An understanding of the roles of officers, members and different agencies.
- (h) Respect for, and desire to work with, different groups and individuals.
- (i) Have knowledge and understanding of the Code of Conduct and Member/Officer Protocol.
- (j) A knowledge and commitment to the values of the Council.

## Work Programming

The ability and discipline to plan and manage work programmes.

# **Chair of Regulatory Committee Person Specification**

To fulfil his or her role as set out in the role description, an effective regulatory committee chair requires:

## Providing Leadership and Direction

- (a) Ability to conduct meetings to ensure that applicants feel that they have been dealt with fairly and fully even if their application is refused.
- (b) Understanding of the Council's role and ability to ensuring that stake holders are aware of that role.
- (c) Communication skills Knowledge of local issues.
- (d) Ability to manage the work of the committee.
- (e) Ability to support and develop necessary skills in fellow members of the committee.

# <u>Promoting the Role of the Regulatory Committee and Quasi-Judicial Decision</u> Making

- (f) Understanding and appreciation of the regulatory framework.
- (g) Ability to inspire and enthuse committee members for the work of the committee.
- (h) Integrity and the ability to set aside own views and act impartially.
- (i) Knowledge and understanding of the relevant code(s) of conduct and protocols and the ability to champion them.

#### Internal Governance, Ethical Standards and Relationships

- (j) Knowledge and understanding of the Code of Conduct(s) and protocols.
- (k) Knowledge of and commitment to the values of the Council.

## **Regulatory Committee Member Person Specification**

To fulfil his or her role as laid out in the role/job description, an effective member of a regulatory committee requires the following:

# <u>Understanding the Nature of the Regulatory Committee and Quasi-Judicial Decision Making</u>

- (a) Integrity and the ability to set aside own views and act impartially.
- (b) Knowledge of law, policy and procedures for that regulatory/quasi-judicial area.

- (c) Maintenance of knowledge.
- (d) Objectivity and judgement.

## Participating in Meetings and Making Decisions

- (e) Ability to listen and to consider and respect the views of other contributors.
- (f) Good public speaking skills.
- (g) Good advocacy skills.

## Internal Governance, Ethical Standards and Relationships

- (h) Knowledge and understanding of the Code of Conduct(s) and protocols.
- (i) Knowledge of and a commitment to the values of the Council.

# **Scrutiny Chair Person Specification**

To fulfil his or her role laid out in the role description an effective Scrutiny Chair requires:

## Providing Leadership and Direction

- (a) Understanding of council role and functions.
- (b) Understanding of role of scrutiny, terms of reference for the committee, role of chair, and other aspects of the democratic arrangements.
- (c) Understanding of member support functions. Understanding of council priorities and risks Ability to develop work programmes.
- (d) Understanding of community issues.
- (e) Objectivity.
- (f) Negotiation and consensus building.
- (g) Ability to build constructive and 'critical friend' relationships with the Executive.

#### Managing the Work Programme

- (h) Ability to manage projects and resources.
- (i) Ability to manage people.
- (j) Ability to prioritise.
- (k) Ability to report progress to different groups in different styles.

# **Effective Meeting Management**

(I) Understanding and application of meeting protocols, code of conduct, standing orders and other constitutional requirements.

- (m) Ability to chair meetings effectively, managing the agenda and progressing business.
- (n) Ability to facilitate effective discussions.
- (o) Ability to listen and question effectively.

## Community Leadership

- (p) Understanding of the community leadership role.
- (g) Knowledge of local issues and expectations.
- (r) Ability to work effectively with all members of the community and build understanding and ownership of scrutiny.
- (s) Knowledge of the individuals and organisations in the community especially those traditionally excluded.

## Involving and Developing of Committee Members

- (t) Understanding of the role and skills of the scrutiny committee and its individuals.
- (u) Ability to support members and the committee in assessing their performance.
- (v) Ability to identify any training and development needs and Champion and participate in appropriate learning and development.

#### **Scrutiny Member Person Specification**

To fulfil his or her role as laid out in the role description, an effective scrutiny member requires:

## Participating Fully in the Activities of the Scrutiny Function

- (a) Full understanding of the scrutiny remit and role, and terms of reference for their own committee and others.
- (b) Understanding of member support functions.
- (c) Willingness to work within the guidance of the chair.
- (d) Willingness to undertake training as necessary.

#### Reviewing and Developing Policy

- (e) Knowledge of and ability to evaluate existing policy.
- (f) Understanding of best practice.
- (g) Understanding of national and local legislative and policy context.

# Monitoring Performance and Service Delivery and Holding the Executive to Account

- (h) Understanding of the Wales Programme for Improvement (WPI) and associated risk assessment arrangements.
- (i) Understanding of the principles and practice of performance management.
- (j) Understanding of council's performance management arrangements.
- (k) Ability to analyse data and challenge performance.
- (I) Understanding of arrangements for call in.

## Promoting the Work of Scrutiny

- (m) Ability to negotiate and build consensus.
- (n) Ability to act objectively and on the basis of evidence.

## Community Leadership

- (o) An understanding of the community leadership role.
- (p) Knowledge of local issues and expectations.
- (q) Ability to work effectively with all members of the community and build understanding and ownership of scrutiny.
- (r) Knowledge of the individuals and organisations in the community especially those traditionally excluded.

## **Meeting Participation**

- (s) Ability to interpret information and data from a range of sources.
- (t) Understanding and application of meeting protocols, code of conduct, standing orders and other constitutional requirements.
- (u) Ability to participate in meetings including effective listening, questioning and speaking.

## **Leader of the Opposition Person Specification**

To fulfil his or her role as laid out in the role description, an effective leader of the opposition requires:

#### In Providing Political Leadership for the Opposition Group

- (a) Knowledge of community strengths, areas of improvement and key issues.
- (b) An understanding of the relationship between national and local politics.
- (c) Have a good strategic awareness of issues facing the Council.
- (d) An understanding of the Council's strategy, policies and operations.

(e) An ability to challenge different strategies, policies and operations.

## In Representing the Authority's Opposition

- (f) High level communication skills to communicate to the media, local community and wider audience.
- (g) Good public speaking skills.
- (h) Creative and lateral thinking skills; the ability to see ahead and be foresighted.
- (i) Leadership skills.

## Internal Governance, Ethical Standards and Relationships

- (j) An understanding of the roles of officers, members and different agencies Respect for, and willingness to work with, different groups and individuals.
- (k) A thorough knowledge and understanding of the Code of Conduct and Member/Officer Protocol.
- (I) A knowledge of and commitment to the values of the Council.

# **Member Champion Person Specification**

#### Within the Council

- (a) Understanding of the area of interest being championed in terms of council strategies and policy, good practice, improvement and national agendas and the needs of the client group.
- (b) Ability to engage with a range of members and officers around the area of interest and listening to requirements.
- (c) Ability to advocate on behalf of the area of interest within the council.

## In the Community

- (d) Understanding of the needs of the community in relation to the interest
- (e) Ability to engage with citizens and community groups in matters related to the interest.
- (f) Ability to lead and support local initiatives related to the interest.
- (g) Ability to represent the position of the council to the community in relation to the interest.



# Guide to the Constitution of Bridgend County Borough Council

#### Introduction

#### The constitution

The Council is made up of different individuals and bodies, who work together to deliver the Council's functions. The Council's constitution describes the different people and bodies that make up the Council, their functions, and the procedure rules that govern how those bodies work together to deliver services.

It is intended to: -

- enable the Council to make decisions efficiently and effectively;
- support you to participate in decisions that affect you;
- ➤ help Councillors to represent their constituents more effectively; and
- enable you to hold the Council to account.

One of the main aims of the constitution is to set out clearly what you can expect from the Council, and what you can do if your expectations are not met.

#### This guide

This guide is designed to help you to understand how the Council makes decisions and works to deliver services in your area and how you can get involved. It provides an overview of the Council's constitution and explains key sections of the constitution in clear and simple language.

It may also be of use to those organisations that work with the Council to deliver services in the Council's area.

The first part of this guide explains why the constitution is important and how it is reviewed and updated.

It aims to answer questions such as: -

- Why does the Council have a constitution?
- ➤ How can I get a copy of the constitution?
- > Who is responsible for keeping the constitution up to date?

The second part of this guide explains how the Council is structured. It describes the Council's democratic bodies as well as how decisions are made and by whom.

It explains the functions of the Council, its Cabinet and its committees, and which body of the Council is responsible for particular policies and decisions.

It aims to answer questions such as: -

- ➤ What does my local Councillor do?
- What does the Council do?
- What happens at Council meetings?
- How does the Council make decisions?
- ➤ How can I find out what the Council decided about an issue that affects me?
- What does the Cabinet do?
- ➤ How does the Cabinet make decisions?
- How are decisions scrutinised?
- Who is responsible for upholding standards and holding Councillors to account for their conduct?
- What do the Council's committees do?
- What does the Development Control committee do?
- ➤ Who is responsible for governance, audit and risk management?
- Who is responsible for upholding local democracy?
- What does the licensing committee do?

The third part of this guide explains the roles and responsibilities of elected members and paid officers of the Council. It describes the jobs they do, and how they work together to deliver the Council's functions and priorities. It also explains the codes of conduct that govern councillors' and officers' conduct and the standards of behaviour that you can expect from them.

It aims to answer questions such as: -

- What does the Mayor do?
- ➤ What does the Leader do?

- What happens if the Leader is unable to perform their duties?
- ➤ My Councillor is a member of the Cabinet, what does this mean?
- How are members of the Cabinet supported?
- What does the Chief Executive do?
- ➤ What responsibilities does the Monitoring Officer have?
- Who is responsible for ensuring that the Council's democratic policies and procedures are complied with?
- How should Councillors behave?
- How should Officers behave?
- Are there any special requirements when Councillors of Officers decide planning applications?
- ➤ How can I complain about the behaviour of a Councillor?
- ➤ How do I report concerns about the conduct of a paid Officer of the Council?

The fourth part of this guide provides a summary of the key policies and procedures that govern how the Council carries out certain functions.

It aims to answer questions such as: -

- ➤ Where can I find the Council's key plans and strategies?
- ➤ How is the Council's policy framework decided?
- How does the Council set its budget?
- Can decisions be taken that do not comply with the budget or policy framework?
- Can money be moved from one budget head to another once the budget has been adopted?
- What other policies and procedures govern how the Council manages its budget?
- Are there any restrictions on how the Council buys in goods and services?

➤ How does the Council enter into contracts and agreements?

The final part of this guide explains how you can engage with, and get involved in, your Council and local democracy. It explains how your councillors are elected, how you can contact your councillor, raise questions and speak at meetings, and how you can use petitions to raise issues that are important to you with the Council.

It aims to answer questions such as: -

- What is the Council doing to engage with local people?
- Can I send a petition to the Council?
- How can I find out when meetings of the Council and its committees and bodies are taking place?
- ➤ How can I find out what will be discussed at a particular meeting?
- ➤ How can I find out when an issue I am concerned about will be decided?
- Can I ask a committee or body of the Council to look into a particular issue?
- Can I attend meetings of the Full Council?
- Can I attend other meetings?
- Can I speak at a meeting?
- Can I ask a question at a meeting?
- How can I find out what the Council decided?
- ➤ Can I view the Council's accounts to understand how my council tax is spent?

Signposts to the most relevant sections of the constitution and to additional resources available on the Council's website are provided throughout this guide to enable you to read more on a particular topic if you wish.

If there is anything in this guide which is unclear, or anything is missing, please let us know so we can improve it.

#### Part 1 The constitution

The constitution governs the way in which the Council, Councillors and officers working at the Council work together to deliver the Council's functions. It ensures that everyone at the Council acts lawfully, fairly and appropriately and that the Council's functions are performed properly and effectively.



Section 2.1 of the constitution explains the purpose of the constitution.

You can obtain a copy of the constitution from the Council's offices and view it on the Council's website. A copy of the constitution must also be provided to each Councillor when they are elected to the Council.



You can access the Council's constitution on the Council's website.



Section 2.8 of the constitution explains where, when and how the constitution must be published.

The Monitoring Officer is responsible for maintaining and reviewing the constitution. The Monitoring Officer is also responsible for deciding how the constitution should be understood and applied. You can read more about the Monitoring Officer's role in Part 4 of this guide.

The Full Council is responsible for agreeing the constitution. Once the constitution has been agreed, it can only be changed by the Full Council. Usually, changes to the constitution are recommended to the Full Council by the Monitoring Officer.

In some circumstances, the Monitoring Officer may make changes to the constitution. For example, where there is a change in the law that affects the constitution, or where a minor change is needed to clarify a provision in the constitution that is unclear.



Sections 2.4 to 2.6 of the constitution explain how the constitution is agreed and how it can be changed.

The constitution also governs how meetings of the Council and its committees should be conducted. The person chairing a meeting will be responsible for ensuring that the constitution is followed during that meeting.



Key words and phrases are defined and explained in section 2.2 of the constitution.

#### Part 2 The Council's democratic structures

The Council is made up of Councillors who are elected every five years to represent people living in different parts of the Council's area (referred to in the constitution as 'electoral wards').

Councillors are responsible for everyone living in the Council's area, but they have a special duty to people living in their ward.



Section 1.2 of the constitution provides an overview of how individual Councillors work together as the Council.

All Councillors meet together regularly as the Full Council. The Full Council is responsible for setting the Council's budget, policy priorities and overall policy framework.

At the start of every year, the Full Council will elect one Councillor as its Mayor. The Mayor is responsible for chairing meetings of the Full Council and ensuring that decisions are taken properly and in accordance with the rules.

You can read more about these roles in Part 3 of this guide.

The Council appoints or elects a Leader (who may be the leader of the largest political group or coalition of political groups). Individual councillors may also be appointed to the Cabinet by the Leader. The Leader will generally appoint councillors from their own political group to the Cabinet, although they may appoint councillors from any political group to the Cabinet. Members of the Cabinet are responsible for specific policy areas, commonly referred to as portfolios. Where responsibility for a particular work area or function is given to a member of the Cabinet, it is described in the constitution as being delegated to that person.

The Cabinet meets regularly to take collective decisions on those aspects of the Council's work which the Cabinet is responsible for.

Some of the Council's functions are carried out by committees. Committees are small groups of Councillors that meet together to carry out certain of the Council's regulatory and scrutiny functions. Subject to some exceptions, committees comprise councillors from each political group in the same proportion as they are represented on the Full Council.

Until recently all council meetings (whether of the Full Council, the Cabinet or a committee or sub-committee) had to take place in person and only those councillors how were physically present at the meeting could participate in discussions and vote. Councils are now able to meet using video conferencing software. Meetings may either be "remote" (where all councillors are in different places) or "hybrid" (where some people are present in one place, and others are in different places). Even if all

the people who are participating in a meeting choose to be physically present, the Council must ensure that there is the option to join the meeting remotely.

The Full Council may engage such paid staff (referred to as officers) as it considers necessary. The Cabinet, Full Council and committees may delegate functions to officers. Further information on delegations can be found in the officers' scheme of delegations.

You can read more about these structures in the following sections of this Part of the guide.

#### Councillors

Councillors are elected by the people living in a particular ward to represent them on the Council. However, Councillors are also accountable to the wider community and must act in the best interests of everyone living in the Council's area.

If you want to raise an issue with the Council and are not sure who to contact, you may wish to contact your Councillor to ask for help.



You can find out who your local Councillor is and their contact details on the Council's website.

All Councillors are members of the Full Council, and they may also be members of one or more of the Council's members bodies (such as the Cabinet or a committee). Where Councillors act as a member of a particular Council body, they are referred to in the constitution as 'Members'.



This is explained more fully in the list of definitions in section 2.2.2 of the constitution.

Councillors have certain rights that are set out in the constitution, which enable them to raise issues and make representations on your behalf.

For example, Councillors may: -

- see any information which they need in order to fulfil their role as a member of the Council;
- > attend any meeting of the Council, its committees or the Cabinet;
- speak at any meeting of any Council body which they are a member of;
- with the permission of the chair, speak at any meeting of any Council body (even if they are not a member of that body);
- ➤ talk to the Leader, or members of the Cabinet or Council officers about any aspect of Council business; and
- raise complaints.



Councillors' rights to information are set out in section 3 of the constitution.



Councillors' rights to participate in Council proceedings are set out in section 3.3 of the constitution.

Councillors may also raise questions at meetings of the Full Council and the Council's committees and sub-committees, submit motions to the Full Council and call-in decisions.



The rules surrounding how these rights are exercised are set out in sections 4.19 to 4.21 and 7.29 of the constitution.



You can read more about the role of elected Councillors and the qualities that make a good Councillor in the role description (in the constitution) and the person specification (in the constitution).

#### **Full Council**



Section 4 of the constitution governs the Full Council.

#### What does the Full Council do?

The law requires that certain important decisions are taken by all Councillors meeting together as the Full Council. These are referred to in the constitution as functions of the Full Council. The Council may decide that other non-executive functions should be carried out by the Full Council too. These are referred to in the constitution as local choice functions because the Council has a choice about which person or body should carry them out.



Section 14 of the constitution explains the different types of functions that are carried out by the Council and which part of the Council is responsible for carrying them out.

The Full Council is responsible for, amongst other things: -

- > agreeing the constitution, and any changes to the constitution;
- setting the Council's budget;
- developing the Council's single integrated plan (which explains how the Council will deliver its legal duties in areas such as health and social care, children and young people and community safety);
- agreeing key plans and strategies (referred to in the constitution as the 'policy framework');
- reviewing and reporting on the Council's performance;
- electing the Leader and the Mayor; and
- appointing the Chief Executive and other Chief Officers.



All of the functions of the Full Council are listed in section 4.6 of the constitution.

Meetings of the Full Council are chaired by the Mayor. They are responsible for ensuring that meetings are conducted in accordance with the constitution and that decisions are made properly, fairly and lawfully.

You can read more about the role of the Mayor in Part 3 of this guide.

#### What happens at meetings of the Full Council?

The Full Council will meet early in each financial year to elect Councillors to particular positions on the Council, to appoint Councillors to outside bodies and to establish committees and working groups to carry out the Council's business during the year. This is known as the annual meeting.



The timing of the annual meeting and the issues to be decided at that meeting are described in the council procedure rules in section 4.10 of the constitution.

At the annual meeting, the Council will also decide how often all Councillors should meet together as the Full Council. These regular meetings of the Full Meetings are known as ordinary meetings.



Ordinary meetings are carried out in accordance with the council procedure rules in section 4.11 of the constitution.

In some circumstances, it may be necessary for the Full Council to meet before its next scheduled meeting to discuss an issue that is particularly urgent or important. This is known as an extraordinary meeting of the Council.



The process by which an extraordinary meeting may be called and the issues that may be discussed there are set out in section 4.12 of the constitution.

The Full Council is responsible for deciding when and where meetings of the Full Council will be held. The Head of Democratic Services must notify Councillors of this by issuing them with a summons.

The Head of Democratic Services is also responsible for ensuring that the public are told about meetings of the Full Council, by publishing a notice in advance of each meeting. The notice must include certain information such as the date and time of the meeting, where it will be held and how you can access the meeting remotely using video conferencing.

You can read more about participating in meetings of the Council in part 4 of this guide.



Section 4.13 and 4.14 of the constitution govern meeting arrangements and notice requirements.

#### **How does the Full Council make decisions?**

Meetings of the Full Council are carried out in accordance with an agreed set of rules that are designed to ensure that debates are conducted fairly and efficiently, and that every Councillor has the ability to raise questions and to make comments on the public's behalf.



The rules of debate are set out in section 4.22 of the constitution.

Decisions are made by Councillors casting votes for or against a particular decision (referred to in the constitution as a motion). Councillors may also propose amendments (changes) to a particular motion, which must then be voted on.

Historically, votes were conducted by a show of hands, with Councillors asked to raise their hand to indicate whether they were voting for or against a particular motion, or whether they wished to abstain (i.e. to vote neither for nor against the motion). Councils now use a range of methods including electronic voting pads for meetings where Councillors are physically present and in-built electronic voting functions on video conferencing platforms where meetings are conducted remotely or in a hybrid format.

Decisions will usually require a simple majority of those present voting for a particular motion. In other words, more Councillors must vote for a motion than against it. Where a vote is tied once all Councillors have voted, the Mayor will have a second, casting vote.

Councillors may request a recorded vote on a particular issue. In a recorded vote, the Councillors voting for and against a particular motion, and those abstaining, will be written down and recorded in the minutes of the meeting. Councillors may also request that their individual vote on a particular motion is recorded in the minutes.



The Council's voting arrangements are set out in section 4.25 of the constitution.

#### Does every Councillor need to be present before a decision can be taken?

Not every Councillor needs to vote on every decision. In fact, in some circumstances it may not be appropriate for a Councillor to vote on a particular issue (for example, where they have a personal interest in a decision). You can read more about this in part three of this guide.

However, in order for the Council to make a lawful decision a minimum number of Councillors must be present at a meeting. The minimum number of Councillors that must be present at a meeting is referred to in the constitution as the quorum for that meeting.



Section 4.16 of the constitution sets out the minimum number of Councillors that must be present so the Full Council can make a decision.

#### **Are meetings recorded?**

Meetings of the Full Council are broadcast live on the Council's website.

A record of each meeting will be kept in the form of minutes. The minutes will contain a record of the motions that were put to the meeting, and the decisions that were taken. They will also record who was present at the meeting.

The minutes of a meeting of the Full Council will be agreed by the next meeting of the Council and published on the Council's website in accordance with the Council's access to information procedure rules.

The Council also publishes a record of the decisions taken at every meeting of the Full Council, the Cabinet and at committee and sub-committee meetings within seven days of the meeting taking place.

If you want to find out what the Council decided about an issue that affects you, you can read the minutes or the record of decisions for the meeting where that issue was discussed.



Information on the Council's broadcasting arrangements can be found in section 4.32 of the constitution and on the Council's website.



Information on minutes can be found in section 4.26 of the constitution.



The access to information procedure rules can be found in section 15 of the constitution.



Minutes of meetings and records of decisions are available on the Council's website.

#### The Cabinet

The Cabinet is made up of the Leader and individual Councillors appointed to the Cabinet by the Leader.



Sections 5.2 and 5.5 of the constitution explain the composition of the Cabinet and how Councillors are appointed to the Cabinet.

The role of Leader and any role on the Cabinet may be carried out by two or more Councillors on a job-share basis.



Section 5.13 of the constitution explains the job-sharing arrangements for members of the Cabinet.

#### What does the Cabinet do?

The Leader is responsible for carrying out the Council's executive functions. However, in practice the Leader cannot personally carry out every one of these functions, so the Leader delegates responsibility for certain functions to the Cabinet, to members of the Cabinet, to officers of the Council or to another bodies.

Each year, the Leader decides which functions to delegate to whom. This is known as the Council's scheme of delegation.



Sections 5.6 and 5.10.1 to 5.10.3 of the constitution explain how functions are delegated by the Leader.



The Council's scheme of delegation can be found on the Council's website.

#### **Meetings of the Cabinet**

The Leader will decide when the Cabinet will meet, and for how long. They will also chair meetings of the Cabinet.

The constitution prescribes certain matters that must be considered at every meeting of the Cabinet. Additionally, the Leader, members of the Cabinet and certain Chief Officers may require that an item of business is added to the meeting agenda.

Meetings of the Cabinet are carried out in accordance with the rules of procedure and debate set down in the constitution.



The rules of procedure and debate of the Cabinet are set out in sections 5.8 to 5.10 of the constitution.



Section 5.11 of the constitution governs how meetings of the Cabinet are conducted.

#### **How does the Cabinet make decisions?**

The Cabinet is obliged to consult with individual Councillors and with certain committees before it makes certain decision on matters that are not urgent.

The Cabinet is also required by law to consult with the public in advance of taking certain decisions. Even where legislation does not require public consultation, as a matter of good practice the Cabinet may also consult the public in advance of taking other significant decisions. Where the Cabinet undertakes public consultation, the Cabinet must have regard to the consultation responses received when taking their decision.



These consultation requirements are explained in section 15.14 of the consultation.

The Cabinet must keep a record of every decision it makes, including every decision made by an individual member of the Cabinet and the Cabinet's committees.



Section 15.15 of the constitution sets down the requirements for the Cabinet's record of decisions.



You can read the minutes of meetings of the Cabinet on the Council's website.

#### Why does the Council have committees and what do they do?

The law requires the Council to establish committees for the purpose of carrying out certain legal functions of the Council.

The Council may also decide to establish other committees and working groups for the purpose of assisting the Council to conduct its business efficiently and effectively.

#### What do the Council's scrutiny committees do?

#### Overview and Scrutiny Committee

The Overview and Scrutiny Committee is responsible for holding the Cabinet and other bodies of the Council to account.

The Overview and Scrutiny Committee can: -

- review and scrutinise decisions made by the Cabinet and other parts of the Council;
- ➤ make reports and recommendations to the Council or to the Cabinet; and
- make proposals regarding changes to the Council's policies and procedures.



The functions of the Overview and Scrutiny Committee are set out in sections 7.2, 7.4, 7.10 and 7.19 of the constitution.

Members of the Overview and Scrutiny Committee must be told about meetings of other Council bodies. They have special rights to see Council documents.

The Overview and Scrutiny Committee may also require any member of the Cabinet or any senior paid officer of the Council to attend a meeting of the committee to explain a decision that they have taken or their performance.



The Overview and Scrutiny Committee's rights and powers are described in sections 7.22 to 7.25 of the constitution.



The rules around membership of the Overview and Scrutiny Committee, the conduct of committee meetings and the carrying out of the functions are committee are set out in section 7 of the constitution.

#### Standards Committee

The Standards Committee is responsible for promoting high standards of conduct by Councillors, including by advising and training Councillors on the councillor code of conduct and advising the Council on changes to that code.



The functions of the Standards Committee are set out in sections 8.6 to 8.8 of the constitution.

The Standards Committee is also responsible for investigating alleged breaches of the councillor code of conduct and censuring members who are found to have failed to comply with that code.



The procedure for dealing with allegations made against Councillors is set out in the constitution.

The Standards Committee comprises a mixture of Councillors and independent members (who cannot be members or paid officers of the Council or related to members or paid officers of the Council).

Councillors are appointed to the Committee in accordance with the political balance rules (which ensure that the political make-up of the committee reflects the political make-up of the Council).



You can read more about the Standards Committee in section 8 of the constitution.

#### What do the Council's regulatory and other committees do?

#### **Development Control Committee**

The Council is responsible for operating the planning system in its area. This involves preparing local development plans and local planning policies (which govern development in the Council's area) and managing development (by granting or refusing permission for new development).

The Council will appoint individual Councillors to the Committee and agree terms of reference for, and the delegation of powers to, the Committee to enable it to discharge those functions.



You can read more about the Council's Committee in section 9.4 of the constitution.



You can also read more about the Council's planning functions on the Council's website.

#### Licensing Committee

The Council is responsible for deciding whether to grant licences for a wide range of different businesses, services and activities in its area. The Council's licensing functions are delegated to the Council's Licensing Committee.

The Council will appoint individual Councillors to the Licensing Committee and agree terms of reference for, and the delegation of powers to, the Licensing Committee to enable it to discharge those functions.



You can read more about the Council's Licensing Committee in section 9.4 of the constitution.



You can also read more about the Council's licensing functions on the Council's website.

#### Governance and Audit Committee

The Governance and Audit Committee advises the Full Council, the Cabinet and their respective committees and members on matters relating to good governance, financial oversight, risk management and complaints.

The Governance and Audit Committee monitors the effectiveness of the Council's rules and procedures for ensuring that the Council acts lawfully, responsibly and that it is accountable to the public (referred to in the constitution as the Council's governance systems and internal controls). The committee is also responsible for ensuring that the Council's decisions and finances are audited in accordance with

agreed procedures. Lastly, the Governance and Audit Committee must review and comment upon the Council's own corporate self-assessment reports.

The Governance and Audit Committee comprises a mixture of Councillors and lay members (who are not members of the Council). Councillors are appointed to the Committee in accordance with the political balance rules (which ensure that the political make-up of the Council).



Section 9.2 of the constitution explains the Council's arrangements for the Governance and Audit Committee.



The constitution explains the functions of the Governance and Audit Committee.

#### <u>Democratic Services Committee</u>

The Democratic Services Committee is responsible for appointing the Head of Democratic Services, keeping under review the provision of resources to the Head of Democratic Services and supporting non-executive members of the Council.

The Democratic Services Committee is made up of Councillors, though there are limits to the number of Cabinet members who can sit on the Democratic Services Committee. Councillors are appointed to the Committee in accordance with the political balance rules (which ensure that the political make-up of the committee reflects the political make-up of the Council).



Section 9.3 of the constitution explains the Council's arrangements for the Democratic Services Committee.



The constitution explains the functions of the Democratic Services Committee.

#### What are joint committees?

The Council may carry out some of its functions jointly with one or more other local authorities in Wales, and with other public bodies. The Council may decide to do this where, for example, the Council considers that it could better promote the economic, social or environmental wellbeing of people living in its area by coordinating its activities with other local authorities, bodies or people.

The Council can establish a joint committee with other local authorities, bodies or people and delegate decisions and functions to that joint committee. By allowing decisions to be taken by a joint committee, the Council can ensure that decisions can be coordinated between all authorities and bodies represented on the committee.

The Cabinet may also establish joint arrangements with one or more local authorities in Wales for the purpose of jointly exercising executive functions together with those authorities.



Section 10.3 of the constitution governs the circumstances in which the Council or Cabinet may enter into joint arrangements with other authorities and bodies.

The Council and the Cabinet may also delegate or contract out particular functions to another local authority or to another body or organisation.



Sections 10.5 to 10.6 of the constitution explain the Council's delegations and contracting out arrangements.

#### The South East Wales Corporate Joint Committee

The South East Wales Corporate Joint Committee is made up of the Leaders of the constituent Councils, who may decide to co-opt other executive members or partner representatives as appropriate.

The Council must work together with other members of the Corporate Joint Committee when it performs these functions.

The Corporate Joint Committee was established under regulations made by the Welsh Ministers. Consequently, many of the rules and standards that apply to the Corporate Joint Committee are set down in regulations, rather than in the constitution.



You can read more about the Corporate Joint Committee arrangements in section 10 of the constitution.

#### Part 3 Roles and responsibilities

#### What roles do Councillors perform?

In the constitution you can find information about the roles that Councillors may perform on the Council.

#### These include: -

- ➤ Leader and Deputy Leader;
- Mayor and Deputy Mayor;
- ➤ Member of the Cabinet;
- ➤ Assistant to the Executive;
- ➤ Elected Member (Councillor);
- ➤ Chair and / or member of the Democratic Services Committee:
- ➤ Member of the Governance and Audit Committee;
- ➤ Chair and / or member of the Overview and Scrutiny Committee;
- ➤ Member of the Standards Committee:
- Chair and / or member of a regulatory committee;
- ➤ Leader of the Opposition and Deputy Leader of the Opposition; and
- ➤ Member Champion.

In the constitution you will also find information about the roles played by senior paid officials of the Council, including the Chief Executive, the Monitoring Officer, the Chief Finance Officer and the Head of Democratic Services.

This part of the guide provides more information about these important roles.

#### **Mayor and Deputy Mayor**

The Mayor and Deputy Mayor are elected by the Full Council every year.



The procedures that govern the election of the Mayor and the Deputy Mayor and their resignation or dismissal are set out in section 4.7.3 of the constitution.

The Mayor is responsible for: -

- promoting and upholding the constitution;
- presiding over meetings of the Full Council;
- ➤ ensuring that meetings of the Full Council are quorate and conducted in accordance with the Council's procedure rules; and
- ensuring that decisions are taken in accordance with the constitution.

The Mayor is also the civil leader of the Council. They are responsible for promoting the interests and reputation of the Council and for carrying out civic, community and ceremonial activities.

The Deputy Mayor performs the Mayor's functions in their absence.



You can read more about the role and functions of the Mayor and the Deputy Mayor in section 4.7.4 of the constitution.



The role description for the Mayor can be found in the constitution.



The role description for the Deputy Mayor can be found in the constitution.



The person specification for the Mayor can be found in the constitution.

#### **Leader and Deputy Leader**

The Leader is elected by the Full Council.

The Deputy Leader is appointed by the Leader to exercise the Leader's functions in their absence.



The arrangements governing the election of the Leader and the appointment of the Deputy Leader are set out in sections 6.1 to 6.4 of the constitution.

The Leader is responsible for appointing Councillors to the Cabinet and for allocating specific policy areas and responsibilities (portfolios) to members of the Cabinet.

The Leader is also responsible for preparing a scheme describing which functions are to be carried out by which members of the Cabinet (this is referred to in the constitution as the Cabinet scheme of delegation).



See section 6.5 of the constitution for more information.

The Leader also chairs meetings of the Cabinet.

The Leader will act as the Council member of the South East Wales Corporate Joint Committee and the Council's representative on the Public Services Board.

The Leader may appoint other members of the Council to sit on other outside bodies.



See sections 6.5.6 to 6.5.8 of the constitution for more information.



The role description for the Leader can be found in the constitution.



The role description for the Deputy Leader can be found in the constitution.



The person specification for the Leader can be found in the constitution.

#### **Member of the Cabinet**

Members of the Cabinet are responsible for: -

- ➤ taking decisions regarding issues that fall within their area of responsibility;
- playing an active role in Cabinet meetings and decision making;
- ➤ contributing to the development of the Council's forward work programme and to policies and procedures in their area of responsibility;
- providing political leadership to the Council's paid officers on matters they are responsible for;
- ➤ reporting to the Full Council, the Leader, the Cabinet and others on the performance of services which they are responsible for; and
- ➤ participating in the Council's scrutiny processes and procedures, including by explaining decisions they have made and the performance of functions within their area to the Overview and Scrutiny Committee.

You can read more about members of the Cabinet in section 5.5 of the constitution.



The role description for members of the Cabinet can be found in the constitution.



The person specification for members of the Cabinet can be found in the constitution.

#### Assistant to the Executive

Assistants to the Executive support members of the Cabinet by taking on certain tasks and responsibilities on their behalf. They may attend meetings, prepare reports, review papers and draft comments for the member they support.

Assistants to the Executive are not members of the Cabinet and cannot vote at Cabinet Meetings or Cabinet Committee Meetings. However, they are entitled to attend and to speak at meetings of the Cabinet and its committees.



You can read more about Assistants to the Executive in section 5.7 of the constitution.

#### What roles do paid officers of the Council perform?

The elected members of the Council are supported by paid officers of the Council, who are accountable to the Council and responsible for delivering services to the public in

accordance with the policies and procedures agreed by the Full Council, the Cabinet and their committees and bodies.

The **Chief Executive** has overall corporate and operational responsibility for the work of the Council and for all paid officers of the Council.



The Chief Executive's role and responsibilities are described in sections 12.1.2 and 12.2 of the constitution.

The **Monitoring Officer** is responsible for ensuring lawfulness and fairness in the Council's decision making, maintaining and upholding the constitution, receiving reports of alleged breaches of the Council's duties and obligations and conducting investigations into such allegations.



The Monitoring Officer's role and responsibilities are described in section 12.3 of the constitution.

The **Head of Democratic Services** is responsible for advising the Council, its committees and individual Councillors, on the proper discharge of the Council's democratic functions and responsibilities.



The Head of Democratic Services' role and responsibilities are described in section 12.5 of the constitution.

The **Chief Finance Officer** is responsible for ensuring that the Council makes lawful and financially prudent decisions. They are also responsible for the administration of the Council's financial affairs and providing advice to Councillors on the Council's budgetary and other financial procedures.



The Chief Finance Officer's role and responsibilities are described in sections 12.4 and 17 of the constitution.

The Council also employs a number of other Chief Officers as listed in the constitution.

Chief Officers are each accountable to the Council for the financial management and administration of those services and activities allocated to them in accordance with the Council's policies.



The roles and responsibilities of Chief Officers are described in the constitution.



The processes by which officers are recruited, appointed, disciplined and dismissed are set out in section 12.9 of the constitution.

#### How should Councillors and Officers behave?

#### Councillors should comply with the Members' Code of Conduct

Councillors are expected to uphold the highest standards of personal and professional conduct. Those standards are described in the Code of Conduct for Members.

The Code of Conduct for Members is intended to help and guide Councillors in maintaining appropriate standards of conduct when serving their community. In turn, it provides reassurance to the public and helps build their trust in, and respect for, their local representatives.

The Code of Conduct for Members is based upon the 7 Principles of Public Life, which were first set out in the 1995 Nolan Report on Standards in Public Life. Three additional principles were added in the local government principles in Wales.

The Code of Conduct for Members is consistent with, and provides for the practical application of, these principles.

#### > Selflessness

Members must act solely in the public interest. They must never use their position as members to improperly confer an advantage on themselves or to improperly confer an advantage or disadvantage on others.

#### ➤ Honesty

Members must declare any private interests relevant to their public duties and take steps to resolve any conflict in a way that protects the public interest.

#### > Integrity and propriety

Members must not put themselves in a position where their integrity is called into question by any financial or other obligation to individuals or organisations that might seek to influence them in the performance of their duties. Members must on all occasions avoid the appearance of such behaviour.

#### Duty to uphold the law

Members must act to uphold the law and act on all occasions in accordance with the trust that the public has placed in them.

#### > Stewardship

In discharging their duties and responsibilities members must ensure that their authority's resources are used both lawfully and prudently.

#### > Objectivity in decision-making

In carrying out their responsibilities including making appointments, awarding contracts, or recommending individuals for rewards and benefits, members must make decisions on merit. Whilst members must have regard to the professional advice of officers and may properly take account of the views of others, including their political groups, it is their responsibility to decide what view to take and, if appropriate, how to vote on any issue.

#### > Equality and respect

Members must carry out their duties and responsibilities with due regard to the need to promote equality of opportunity for all people, regardless of their gender, race, disability, sex, sexual orientation, marital status, age or religion, and show respect and consideration for others.

#### > Openness

Members must be as open as possible about all their actions and those of their authority. They must seek to ensure that disclosure of information is restricted only in accordance with the law.

#### > Accountability

Members are accountable to the electorate and the public generally for their actions and for the way they carry out their responsibilities as a member. They must be prepared to submit themselves to such scrutiny as is appropriate to their responsibilities.

#### > Leadership

Members must promote and support these principles by leadership and example so as to promote public confidence in their role and in the authority. They must respect the impartiality and integrity of the authority's statutory officers and its other employees.



You can read the Code of Conduct for Members in full in the constitution.

Councillors are expected to ensure that they understand their obligations under the Code and act in a way which shows that they are committed to meeting the high standards of conduct that are expected of them. However, Councillors are provided with training when they are first appointed, and on a regular basis, to support them to comply with the Code of Conduct for Members.

#### Officers should comply with the Officers' Code of Conduct

Paid officers of the Council are responsible for serving the council by providing advice to Councillors, implementing the Council's policies, and delivering services to the local community. In performing their duties, they must act with integrity, honesty, impartiality and objectivity.

The public is entitled to expect the highest standards of conduct from all those who work for the Council. The Code of Conduct for Officers outlines the rules and conditions of service which apply to the Council's employees. It is designed to provide clear guidance to assist them in their day to day work and to allow the public to understand what they can expect when they interact with a Council employee.



You can read the Code of Conduct for Officers in full in the constitution.

#### **How should Councillors and Officers work together?**

Councillors and Officers have different roles, responsibilities and accountabilities, but it is imperative that they work effectively together to perform the Council's functions and to deliver services to residents living in the Council's area.

The Protocol on Member and Officer Relations is designed to clarify the respective roles and responsibilities of Councillors and Officers and to guide them in their dealings with one another. It explains what Councillors and Officers can reasonably expect from one another and how they should work together to achieve their common purpose.

Councillors are entitled to express political views and to support the policies of the party or group to which they belong. Conversely, officers are expected to carry out the Council's business in a politically impartial way and many are restricted from engaging in political activity. The protocol therefore explains how officers can support the policy deliberations by political groupings, while remaining politically neutral.



You can read the Protocol on Member and Officer Relations in full in the constitution.

#### How can I complain about a Councillor?

Section 3.2.2.(g) of the constitution explains how you can submit comments or complaints about the Council's services.

If you wish to complain about a Councillor, then you should contact either the Council's Monitoring Officer or the Public Services Ombudsman for Wales.



The Council's complaints process provides more information about how to complain. It is available on the Council's website.



You can find the Monitoring Officer's contact details on the Council's website.



You can find out more about how to complain to the Public Services Ombudsman for Wales on their website:

https://www.ombudsman.wales/how-to-complain/

#### How can I complain about a service provided by the Council?

If you have a complaint about the Council, its employees or the services it provides you make a complaint via the Council's complaints process which can be found on the Council's website.

#### Part 4 Policies and procedures

#### The Council's key plans and strategies (the policy framework)

#### Where can I find the Council's key plans and strategies?

The Full Council is responsible for setting the Council's key plans and strategies, which together form the Council's policy framework.



The plans and strategies which form the Council's policy framework are listed in section 4.2 of the constitution.



Individual plans and strategies are published on the Council's website.

Additionally, the Council's single integrated plan brings together the Council's plans and strategies for: communities; children and young people; health social care and wellbeing; and community safety.



You can read more about the single integrated plan in section 4.3 of the constitution.



The Council's single integrated plan is published on the Council's website.

#### How is the Council's policy framework decided?

The Cabinet is responsible for consulting with relevant stakeholders, considering any reports or recommendations made by the Council's Overview and Scrutiny Committees and working with the Council's Chief Officers to develop a draft policy framework.

Once the Cabinet has developed a draft policy framework, it will be considered by a meeting of the Full Council. The Full Council may decide to adopt the policy framework, make changes to it, refer it back to the Cabinet so that further work can be done on it, or substitute its own policy framework for the draft prepared by the Cabinet.

If the Full Council decides to adopt the policy framework then it shall take effect immediately. If the Full Council amends the policy framework then the Leader has an opportunity to object to the amendments and to reconvene the Full Council to reconsider the policy framework.

Once adopted, the Cabinet, their committees, individual Councillors and the Council's paid officers are bound to act in accordance with the policy framework.



The process by which the Full Council adopts a policy framework is described in section 16.2 of the constitution.

#### **Budget setting and financial management**

#### How does the Council set its budget?

The Full Council is responsible for agreeing the Council's budget, and for agreeing changes to the Council's budget once it has been formally adopted.

The Council's budget is in two parts: -

- ➤ the revenue budget provides for day to day operating costs and expenditure like staff salaries, rent, and the ongoing costs of providing services.
- ➤ the capital budget provides for one-off costs like a major road improvement scheme, or the purchase or construction of a new car park.

The Cabinet, in consultation with the Chief Finance Officer, is responsible for developing initial proposals for the Council's budget and consulting over these.

Once this consultation has been completed, the Cabinet is responsible for preparing a final budget proposal for consideration by the Full Council.

The Full Council may adopt the budget without making any changes, amend the budget, or ask the Cabinet to reconsider it before adopting it.



The process by which the Full Council adopts its budget is described in sections 16.3.1 and 17 of the constitution.

#### Decisions outside the budget or policy framework

#### Can decisions be taken that do not comply with the budget or policy framework?

The Full Council may decide to make changes to the policy framework.

Other bodies and individuals may only take decisions that do not comply with the policy framework in a limited number of exceptional circumstances, which are set out in the constitution. For example, where an urgent decision must be taken to safeguard the Council's interests and it is not possible to arrange a meeting of the Full Council in time.

However, the Council's overview and scrutiny committee may decide to refer such decisions to the Full Council for further consideration (this is referred to in the constitution as calling-in the decision).

ı	_
ı	•
ı	1

The Council's arrangements for urgent decisions which do not comply with the policy framework are set out in section 16.5 of the constitution.



Section 16.7 of the constitution describes the limited circumstances in which bodies or individuals other than the Full Council may make changes to the policy framework.



The overview and scrutiny committee's powers are described in section 16.8 of the constitution.

### <u>Can money be moved from one budget head to another once the budget has</u> been adopted?

During the year the Cabinet and Chief Officers may need to transfer budgets from one service area to another to reflect changed service needs or priorities in order to deliver the Council's policy framework within the financial limits set by the Council.



The mechanism by which the Cabinet and Chief Officers may move money between budget heads is set out in the constitution.

### What other policies and procedures govern how the Council manages its budget?

The Council is bound by a number of different financial rules and procedures that govern how the Council's spending is planned, committed, reviewed and audited.



The Council's detailed financial procedure rules are set out in section 17 of the constitution.

#### Are there any restrictions on how the Council buys in goods and services?

The Council can enter into contracts to purchase goods and services in much the same way as any other person or organisation. However, the Council must comply with the contract procedure rules in the constitution when it does so.

The purpose of the contract procedure rules is to ensure that the Council complies with relevant legal requirements and secures the most economically advantageous (best value) goods and services for taxpayers.

If you regularly sell goods or services to the Council then you may wish to familiarise with these rules and procedures.



The Council's contract procedure rules are set out in the constitution.

#### **How does the Council enter into contracts and agreements?**

The rules and procedures that govern how agreements, contracts and deeds are authorised and executed on behalf of the Council are set out in section 13 of the constitution.

#### Part 5 How can I get involved?

## How is the Council engaging with local people to encourage them to participate in local democracy?

The Council will publish a public participation strategy which explains how it will encourage local people to participate in the Council's decision-making processes. The Council's public participation strategy describes how the Council will:

- promote awareness of the Council's functions;
- promote awareness of how people can become a Councillor and what the role entails:
- facilitate access to information about decisions made, or to be made, by the Council:
- promoting arrangements by which people may make representations to the Council about decisions it has made or will make in the future:
- ➤ ensure that the public's views are brought to the attention of its Overview and Scrutiny Committee[s]; and
- promote awareness amongst Councillors of the benefits of using social media to communicate with local people.

The Council also engages with local people through citizen panels and surveys and consultations around specific policies and services. These may be initiated by the Council, service departments within the Council or through the work of the Council's committees or sub-committees.

#### Can I send a petition to the Council?

Yes, you can submit petitions to the Council using the Council's petition scheme. The Council is obliged by law to operate a petition scheme, which sets out: -

- how a petition can be submitted to the Council;
- ➤ how and when the Council will acknowledge receipt of a petition;
- ➤ the steps the Council will take in response to a petition; and
- ➤ how any by when the Council will make available its response to a petition to the person who submitted the petition and to the public.

You can read more about submitting a petition to the Council in section 3.2.2.(c) of the constitution.



The Council publishes details of its petition scheme on its website.

### How can I find out when meetings of the Council and its committees and bodies are taking place?

The Council publishes notices of meetings of the Full Council and its committees.



You can find out more about notice requirements in sections 3.11, 4.14 and 15.4 of the constitution.



The Council publishes notices of meetings, and a programme of upcoming meetings, on its website.

#### How can I find out what will be discussed at a particular meeting?

The Council will publish agendas for meetings, together with any background papers and reports in advance of the meeting taking place.

Hard copies of agendas and background papers and reports will also be available at the meeting for those members of the public who wish to attend in person.



You can find out more the information that is made publicly available in advance of a Council meeting in sections 15.5 and 15.8 of the constitution.



The Council publishes meetings agendas and supporting documents and reports on its website.

#### How can I find out when an issue I am interested in will be decided?

The Council publishes a forward work programme, which sets out what decisions will be taken by the Full Council, the Cabinet and what issues the Overview and Scrutiny Committee will be considering, and when these matters will be discussed.



You can find out more about the Council's forward work programme in sections 3.1.1 and 15.13 of the constitution.



The Council publishes its forward work programme on its website.

#### Can I ask a committee or body of the Council to look into a particular issue?

You can ask the chair of a particular body to add an item to the agenda for a future meeting, or attend a meeting and ask that body to look at an issue when it is considering items of future business.

#### Can I observe meetings of the Full Council?

Yes, members of the public can come to observe meetings of the Full Council so long as they are being held in public.

Meetings of the Full Council are also broadcast live on the Council's website so you can watch them in real time remotely if you wish to do so.

The public will only be excluded from meetings whenever it is likely that confidential information would be disclosed to them if they were able to attend. The Council may also exclude the public from a meeting, or part of a meeting, where exempt information would be disclosed. Exempt information includes information that relates to a particular individual or their financial or business affairs, information that is legally privileged or information relating to the prevention, investigation or prosecution of a crime, or other information specified in the constitution.

If you interrupt a meeting of the Council, then you are likely to be warned by the Mayor about causing a disturbance. If you continue to disturb the meeting, then you may be removed.



The rules around excluding the public from meetings are set out in sections 15.10 of the constitution.



The Council's obligations to broadcast meetings of the Council are set out in section 4.32 of the constitution.



The rules around disturbance by members of the public are set out in section 4.30 of the constitution.

#### Can I observe other meetings?

Yes, members of the public can come to observe any meeting which the Council has resolved should be held in public.



Your right to attend meetings of the Council's Cabinet, the council's committees and other bodies are described in section 15.3 of the constitution.

#### Can I speak at a meeting?

Members of the public can speak at any meeting which the Council has resolved should include participation by members of the public.

#### Can I ask a question at a meeting?

You can ask formal questions of members of the Cabinet at meetings of the Full Council.

However, you may only ask a question if you have given the designated officer (also called the "proper officer"), who is often the Monitoring Officer notice in writing (including by email) that you wish to raise a question in advance of the meeting taking place.

You may only ask one question, but if you cannot attend the meeting then you may nominate someone to attend the meeting and ask the question on your behalf.

Time for questions is limited, and questions are permitted in the order in which they were notified to the proper officer, who is often the Monitoring Officer. If time for questions runs out before you are able to ask your question, then you will be provided with a written answer to your question instead.



You can read more about asking a question in section 4.18 of the constitution.

#### How can I find out what the Council decided?

The Council publishes the agenda, reports and the minutes of meetings once they have been agreed. These papers are available for inspection by the public for a minimum of six years from the date of the meeting. The background papers are available to the public for at least four years. You may also be able to view webcast recordings of many meetings.

The Council also has arrangements in place for publishing written records of decisions taken by the Cabinet and the Council's committees and other bodies.



You can read about the Council's arrangements for publishing minutes of meetings in section 15.7 of the constitution.



You can read about the Council's arrangements for publishing a written record of decisions taken by the Cabinet, the Council's committees and individual members of the Cabinet in section 15.15 of the constitution.



You can read meeting minutes, reports and agendas on the Council's website.

#### Can I view the Council's accounts to understand how my council tax is spent?

Yes. The Council is required to publish its accounts and to make them available for inspection by the public. You may raise questions of concerns about the Council's accounts with the Council or with the Council's extremal auditor.



You can find out more about how to view and comment on the Council's accounts in sections 3.1.1 and 15.13 of the constitution.

### A BEGINNER'S GUIDE TO THE CONSTITUTION OF BRIDGEND COUNTY BOROUGH COUNCIL

CONTENTS	WHAT IT DOES	
SECTION 1 – Introduction	This sets the scene for what the Constitution is, and how it operates. It contains brief notes about the Constitution and how the Council operates.	
SECTION 2 – Purpose, Definition, Interpretation and	Establishes the primacy of the Constitution, and sets	
Amendment of the Constitution	out its purpose.	
SECTION 3 – Getting information and Getting involved	Sets out how citizens of Bridgend can interact with the Council.	
SECTION 4 – Full Council	Defines the roles and decisions which are reserved to Full Council, and the types of meetings Council can have.	
SECTIONS 5 AND 6 – Cabinet and the Leader	Defines the role of the Executive, its decision making powers, and deals with appointments to, and removal from, the Executive.	
SECTION 7 – Overview and Scrutiny Committees	Defines the scope and work of the Overview and Scrutiny committees.	
SECTION 8 – Standards Committee	Defines the scope and work of the Standards Committee, and sets out its composition and terms of reference.	
SECTION 9 – Regulatory Committees	Allows the Council to establish other committees.	
SECTIONS 10 AND 11 – Joint Committees and	Allows the Council to form joint arrangements with	
Corporate Joint Committees	other authorities or other bodies.	

SECTION 12 – Officers	Establishes the primacy of the Chief Executive, and
	defines the roles and responsibilities of the Monitoring
	Officer and Chief Finance Officer.
	Provides the Senior Management Structure.
	Establishes the rules for the recruitment and dismissal
	of officers.
SECTION 13 – Finance, Contracts and Legal Matters	Establishes the basis for the Council's Standing
	Orders and the execution of documents.
SECTION 14 – Responsibility for Functions	Sets the parameters for, and principles of, decision
	making in the Council and provides guidance on the
	types of decisions to be recorded under Regulations.
SECTION 15 – Access to Information Rules	Establishes the rules for the publication or withholding
	of information.
SECTION 16 – Budget and Policy Framework	Establishes the rules for the annual budget setting
Procedure Rules	process, and for in-year changes to the budget.
SECTION 17 – Financial Procedure Rules	Incorporates the Bridgend County Borough Council
	Financial Procedure Rules as part of the Constitution.
SECTION 18 – Contract Procedure Rules	Incorporates the Bridgend County Borough Council
	Contract Procedure Rules as part of the Constitution.
SECTION 19 – Code of Conduct for Members	Sets the standards for Member conduct towards
	Citizens, Officers and Members.
SECTION 20 – Officers' Code of Conduct	Sets the standards for Officer conduct and neutrality,
	and limits on outside working.
SECTION 21 – Protocol on Member / Officer	Provides guidance on the interaction between
Relations	Members and Officers, and their respective roles.
SECTION 22 – Member Role Descriptions	Sets out role descriptions and person specifications.